



Awel y Môr Offshore Wind Farm

Book of Reference (Clean) Deadline 6

Date: 20 February 2023 Revision: G Document Reference: 6.7 Application Reference: 4.3





Copyright ©2023 RWE Renewables UK

REVISION	DATE	STATUS/ REASON FOR ISSUE	AUTHOR:	CHECKED BY:	APPROVED BY:
A	March 2022	ES	Dalcour McClaren	RWE	RWE
В	May 2022	S.56, S.51 and Relevant Reps	Dalcour McClaren	RWE	RWE
С	October 2022	Deadline 1	Dalcour McClaren	RWE	RWE
D	November 2022	Deadline 2	Dalcour McClaren	RWE	RWE
E	January 2023	Deadline 4	Dalcour McClaren	RWE	RWE
F	February 2023	Deadline 5	Dalcour McClaren	RWE	RWE
G	February 2023	Deadline 6	Dalcour McClaren	RWE	RWE



RWE Renewables UK Swindon Limited

Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB T +44 (0)8456 720 090

Registered office: RWE Renewables UK Swindon Limited Windmill Hill Business Park Whitehill Way Swindon



Contents

1	Introduction
2	Rights which may be acquired9
3	Structure of this book of reference
4	Book of reference notes
5	Book of Reference parts 1-5

Tables

Table 1: Land plan colours relating to acquisition 7
Table 2: Rights sought for the authorised development



1 Introduction

- 1 This Book of Reference accompanies the proposed Development Consent Order ("the Order") for the Awel y Môr Offshore Wind Farm ("the authorised development"), as required by Regulation 5(2)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 ("the Regulations").
- 2 The land described in this Book of Reference, which is to be subject to powers of compulsory acquisition and/or temporary possession, is required to deliver the works comprising the authorised development which are described in detail in Part 1 of Schedule 1 of the Order and shown on the works plans. This Book of Reference is part of the application documents for the authorised development it should be read in conjunction with the Land Plans (REP3a-011, Document 6.18 of the Applicant's Deadline 6 submission, Document 6.17 of the Applicant's Deadline 6 submission and Document 6.24 of the Applicant's Deadline 6 submission), the Statement of Reasons (Document 6.6 of the Applicant's Deadline 6 submission) and the draft DCO (Document 6.5 of the Applicant's Deadline 6 submission).
- 3 Every parcel of land that is affected is identified on a plot by plot basis and a unique number has been ascribed to each plot. The plots are shown on the Land Plans which accompany the Order. For each plot it identifies whether the Applicant (Awel y Môr Offshore Wind Farm Limited or the undertaker) is seeking the power to acquire that plot outright, the power to create and/or acquire permanent rights, or the power to create and/or acquire temporary rights of possession and use. It also lists the persons with land outside the order limits who may be entitled to make a relevant claim for compensation due to the effects of construction or when the authorised development is operational.
- 4 The Order seeks powers to compulsorily acquire land and new rights (both temporary and permanent) for the purposes of the construction, operation and maintenance of the authorised development. Some of the plots identified in Part 1 of the Book of Reference will be subject to acquisition of permanent rights (including restrictive covenants) pursuant to Article 20 (Compulsory acquisition of rights) and Schedule 7 (Land in which only new rights etc. may be required) of the Order. These plots are shown coloured blue on the land plans.



- 5 Plots that are subject to powers of temporary possession only, such as for the purpose of access or areas only required during construction are listed in Schedule 6 of the Order and shown coloured yellow on the land plans.
- 6 The colours shown on the land plans indicate the type of acquisition sought as set out in the Table 1 below.



Table 1: Land plan colours relating to acquisition

COLOUR OF THE PLOT ON LAND PLANS	DESCRIPTION OF ACQUISITION SOUGHT IN BOOK OF REFERENCE	ACQUISITION SOUGHT	PRINCIPAL RELEVANT DCO ARTICLE(S)
Pink	"Acquisition of freehold"	Compulsory acquisition of all interests and rights in land (including as required, subsoil, surface land)	Articles 18, 20
Blue	"Acquisition of new rights and imposition of restrictions" and "Acquisition of new rights"	Creation and compulsory acquisition of new rights (including where necessary, a right to impose restrictive covenants)	Articles 20, 22
Yellow	"Temporary possession"	Temporary possession and use of land, primarily during construction	Articles 27, 28



7 The Applicant has taken the cautious approach of seeking powers of compulsory acquisition (or rights of use) in respect of all plots of land required for the scheme even where it already holds an interest or presumes it holds an interest in the land. This approach has been taken to ensure that it has the right to acquire the interests it needs in the whole of the Order limits even where an unidentified owner later asserts an interest in land which the Applicant believes it owns or has rights.



2 Rights which may be acquired

- 8 For plots in which rights are to be acquired or restrictive covenants are to be imposed, the rights sought for the authorised development have been categorised as shown in the Table 2 below. This means that where the entry in this Book of Reference specifies that 'all cable rights' are to be acquired, all of the rights shown in the cable rights line of table will be sought. Where only some of a category of rights is required the specific rights sought will be listed for the relevant plot.
- 9 In some cases more than one category of rights will be sought in a single plot. For example cable rights to install, operate and maintain the cable and restrictive covenants to protect the cable once it has been installed will be required on the cable route plots.



Table 2: Rights sought for the authorised development

(1) NUMBER OF PLOT SHOWN ON LAND PLANS	(2) PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED
1, 28, 29, 30, 31 32, 33, 34, 35, 87, 97, 101, 103	Cable rights and restrictive covenants
105, 118, 139, 140, 157, 162 184, 187, 190	1. Cable rights
194, 242, 247, 250, 257, 258, 259, 261, 269 272, 275, 285, 293, 294, 297 309, 316, 317	Rights for the purposes of the construction, installation, operation, maintenance and decommissionin development to:
318, 319, 335, 347, 349, 353 359, 365, 374, 401, 403, 413 435, 437, 438 453, 462, 465	(a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgr replace the electricity cables (including the removal of materials including spoil) in, under, over and, such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measure and manholes, manhole covers and other equipment which is ancillary to the purposes of transmittir electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trene including (but not limited to) directional drilling beneath sea defences, watercourses, roads and raily
400, 471	(b) enter, be on, and break open and break up the surface of the Land and remain with or without paparatus and equipment which is ancillary to the purposes of transmitting electricity along the cab
	(c) to benefit from continuous vertical and lateral support for the authorised development;
	(d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting, removi
	(e) construct and install and thereafter use the Land for all necessary purposes for the commissioning and maintenance of the cables in, on or under the Land;
	(f) place and use plant, machinery, structures and temporary structures within the Land for the purport construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the signage and provide measures for the benefit of public and personnel safety;
	(g) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working area trenchless installation technique compounds and working areas;
	(h) construct, lay down, use and remove temporary access roads including any necessary hard stan including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary of water courses and drains during any period during which construction, maintenance, repair or rem
	(i) effect access to the highway;



ning of the authorised

grade, inspect, remove and ad/or on the Land, together with tres, cable marker posts, chambers tting electricity along such enchless installation techniques ailways;

it plant, vehicles, machinery, ables;

als for the purposes of laying down, oving and replacing the cables;

ng, construction, repair, testing

poses of the installation, the cables and to erect temporary

eas and compounds including

anding and other surface materials ary bridging, culverting or diversion enewal is being carried out;

(1)	(2)
NUMBER OF PLOT SHOWN ON LAND PLANS	PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED
	(j) make such investigations in or on the Land as required;
	(k) fell, lop or cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be a land which would if not felled, lopped, cut or removed would obstruct or interfere with the operation
	(I) remove and discharge water from the Land, and to install, retain, use, maintain, inspect, alter, re replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into a drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or c Land or reinstate the any existing drainage scheme on the Land;
	(m) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus in electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service cables or conduits or apparatus of statutory undertakers);
	(n) remove fences and structures within the Land during any period during which construction, main being carried out (subject to erection of any temporary stock-proof fencing as is reasonably require suitable replacement of the fences or structures following the exercise of the rights);
	(o) store and stockpile materials (including excavated material);
	(p) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and it trenchless installation technique, to keep in place and monitor the same through construction, main decommissioning and to reinstate the Land;
	(q) to excavate materials below ground level, including soils, and to store and re-use or dispose of t undertake any works, including works of protection or removal of archaeological remains as may b of investigation approved under this Order;
	(r) lay out temporary paths and bridleways for public use;
	(s) to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out m remediation works for environmental or ecological mitigation or enhancement works, including tem measures and the installation of temporary barriers for the protection of fauna;
	(t) carry out such works (together with associated fencing) required by a planning permission and/or over the Land and/or in accordance with any necessary licences relating to protected species and
	(u) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, main cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and together with conduits or pipes for containing the same in and under the Land.
	suitable replacement of the fences or structures following the exercise of the rights); (o) store and stockpile materials (including excavated material); (p) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and trenchless installation technique, to keep in place and monitor the same through construction, mai decommissioning and to reinstate the Land; (q) to excavate materials below ground level, including soils, and to store and re-use or dispose of t undertake any works, including works of protection or removal of archaeological remains as may b of investigation approved under this Order; (r) lay out temporary paths and bridleways for public use; (s) to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out m remediation works for environmental or ecological mitigation or enhancement works, including ten measures and the installation of temporary barriers for the protection of fauna; (t) carry out such works (together with associated fencing) required by a planning permission and/o over the Land and/or in accordance with any necessary licences relating to protected species and (u) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, main cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and



standing on the Land or other on of the cables;

emove, refurbish, reconstruct, and manage waterflows in any cleanse drainage schemes on the

ncluding but not limited to cables and any ancillary ce media (including the pipes,

intenance, repair or renewal is ed and the re-instatement or

monitoring the use of any intenance repair, replacement or

the same, and in so excavating to be required by any written scheme

nitigation, maintenance, nporary works for noise alleviation

or consent now or to be granted d/or wildlife; and

ntain and inspect underground d other conducting media

(1) NUMBER OF PLOT SHOWN ON LAND PLANS	(2) PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED
	2. Restrictive covenants
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
	(a) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection construction erection or works of any kind (including the foundations or footings thereto);
	(b) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or w whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably w proposed surfacing would not cause damage to relevant part of the authorised development nor m expensive to maintain the authorised development);
	(c) to prevent anything to be done by way of excavation of any kind in the Land nor any activities we decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the consense reasonably required for agricultural activities (being ploughing to no deeper than 0.6m for the purpose required to be carried out by National Grid in order to exercise their statutory functions or rights in relativities within the Land;
	(d) to prevent the planting or growing within the Land of any trees, shrubs or underwood without the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed tr not cause damage to the relevant part of the authorised development nor make it materially more of the relevant part of the authorised development);
	(e) to prevent anything being done which may interfere with free flow and passage of electricity or to cables or support for the authorised development;
	(f) to prevent carrying out operations or actions (including but not limited to blasting and piling) whic interfere with the exercise of the rights or damage the authorised development; and
	(g) to prevent any activity which would in the reasonable opinion of the undertaker result in the distu areas or areas of habitat creation including any ploughing or grazing without the prior written conser
100	Cable rights, transition joint bay rights and restrictive covenants
	1. Cable rights



ction of any buildings or

with any other material or surface withheld or delayed if the make it materially more difficult or

which would alter, increase or e undertaker save as are coses of arable farming) or are elation to their apparatus (if any)

e consent in writing of the I trees, shrubs or underwood would e difficult or expensive to access

telecommunications through the

hich may obstruct, interrupt, or

turbance of ecological mitigation ent of the undertaker.

(1)	(2)
NUMBER OF PLOT SHOWN ON LAND PLANS	PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED
	Rights for the purposes of the construction, installation, operation, maintenance and decommissionidevelopment to:
	(a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upg replace the electricity cables (including the removal of materials including spoil) in, under, over and such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measure and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitti electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to tren including (but not limited to) directional drilling beneath sea defences, watercourses, roads and rail
	(b) enter, be on, and break open and break up the surface of the Land and remain with or without apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cal
	(c) to benefit from continuous vertical and lateral support for the authorised development;
	(d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and material installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting, remov
	(e) construct and install and thereafter use the Land for all necessary purposes for the commissionin and maintenance of the cables in, on or under the Land;
	(f) place and use plant, machinery, structures and temporary structures within the Land for the purp construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the signage and provide measures for the benefit of public and personnel safety;
	(g) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working area trenchless installation technique compounds and working areas;
	(h) construct, lay down, use and remove temporary access roads including any necessary hard star including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary of water courses and drains during any period during which construction, maintenance, repair or re
	(i) effect access to the highway;
	(j) make such investigations in or on the Land as required;
	(k) fell, lop or cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be s land which would if not felled, lopped, cut or removed would obstruct or interfere with the operatio
	(I) remove and discharge water from the Land, and to install, retain, use, maintain, inspect, alter, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into an



ning of the authorised

grade, inspect, remove and d/or on the Land, together with res, cable marker posts, chambers ting electricity along such nchless installation techniques ilways;

t plant, vehicles, machinery, bles;

als for the purposes of laying down, wing and replacing the cables;

ng, construction, repair, testing

poses of the installation, the cables and to erect temporary

eas and compounds including

anding and other surface materials ry bridging, culverting or diversion enewal is being carried out;

standing on the Land or other on of the cables;

emove, refurbish, reconstruct, and manage waterflows in any

(1)	(2)
NUMBER OF PLOT SHOWN ON LAND PLANS	PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED
	drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or c Land or reinstate the any existing drainage scheme on the Land;
	(m) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus in electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, servic cables or conduits or apparatus of statutory undertakers);
	(n) remove fences and structures within the Land during any period during which construction, main being carried out (subject to erection of any temporary stock-proof fencing as is reasonably require suitable replacement of the fences or structures following the exercise of the rights);
	(o) store and stockpile materials (including excavated material);
	(p) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and r trenchless installation technique, to keep in place and monitor the same through construction, main decommissioning and to reinstate the Land;
	(q) to excavate materials below ground level, including soils, and to store and re-use or dispose of t undertake any works, including works of protection or removal of archaeological remains as may b of investigation approved under this Order;
	(r) lay out temporary paths and bridleways for public use;
	(s) to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out m remediation works for environmental or ecological mitigation or enhancement works, including tem measures and the installation of temporary barriers for the protection of fauna;
	(t) carry out such works (together with associated fencing) required by a planning permission and/or over the Land and/or in accordance with any necessary licences relating to protected species and
	(u) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, main cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and together with conduits or pipes for containing the same in and under the Land.
	2. Transition joint bay rights
	Rights for the purposes of the construction, installation, operation, maintenance and decommission development to:



cleanse drainage schemes on the

ncluding but not limited to cables and any ancillary ce media (including the pipes,

intenance, repair or renewal is ed and the re-instatement or

monitoring the use of any intenance repair, replacement or

the same, and in so excavating to be required by any written scheme

nitigation, maintenance, nporary works for noise alleviation

or consent now or to be granted d/or wildlife; and

ntain and inspect underground d other conducting media

ning of the authorised

(1) NUMBER OF PLOT SHOWN ON LAND PLANS	(2) Purpose for which rights may be acquired
	(a) lay down, install, adjust, alter, construct, create, use, maintain, repair, renew, upgrade, inspect, re joint bays; and
	(b) to adjust, alter, remove, replace, and create tunnels under sea defences including walls and gro
	3. Restrictive covenants
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
	(a) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erect construction erection or works of any kind (including the foundations or footings thereto);
	(b) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or w whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably w proposed surfacing would not cause damage to relevant part of the authorised development nor m expensive to maintain the authorised development);
	(c) to prevent anything to be done by way of excavation of any kind in the Land nor any activities we decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the reasonably required for agricultural activities (being ploughing to no deeper than 0.6m for the purpor required to be carried out by National Grid in order to exercise their statutory functions or rights in relativitient the Land;
	(d) to prevent the planting or growing within the Land of any trees, shrubs or underwood without the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed tr not cause damage to the relevant part of the authorised development nor make it materially more of the relevant part of the authorised development);
	(e) to prevent anything being done which may interfere with free flow and passage of electricity or t cables or support for the authorised development;



remove and replace transition

roynes.

ction of any buildings or

with any other material or surface withheld or delayed if the make it materially more difficult or

which would alter, increase or e undertaker save as are coses of arabale farming) or are elation to their apparatus (if any)

e consent in writing of the I trees, shrubs or underwood would e difficult or expensive to access

telecommunications through the

(1) NUMBER OF PLOT SHOWN ON LAND PLANS	(2) PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED
	 (f) to prevent carrying out operations or actions (including but not limited to blasting and piling) which interfere with the exercise of the rights or damage the authorised development; and (g) to prevent any activity which would in the reasonable opinion of the undertaker result in the disturates or areas of habitat creation including any ploughing or grazing without the prior written conservation.
89, 90, 91, 99, 131, 147, 216, 217, 260, 291, 320, 321, 322, 342, 381, 382 383, 384, 385 386, 387, 388 389, 390, 391 392, 393, 394 395, 396, 397 398, 399, 400, 402, 425, 444	Cable rights and restrictive covenants under existing infrastructure 1. Cable rights Rights for the purposes of the construction, installation, operation, maintenance and decommissionir development to: (a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgr replace the electricity cables (including the removal of materials including spoil) in, under, over and, such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measure and manholes, manhole covers and other equipment which is ancillary to the purposes of transmittir electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trend including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railv (b) enter, be on, and break open and break up the surface of the Land and remain with or without p apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cab (c) to benefit from continuous vertical and lateral support for the authorised development: (d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting, removi (e) construct and install and thereafter use the Land for all necessary purposes for the commissioning and maintenance of the cables in, on or under the Land: (f) place and use plant, machinery, structures and temporary structures within the Land for the purpo construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of th signage and provide measures for the benefit of public and personnel safety;
	(g) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working area trenchless installation technique compounds and working areas;



nich may obstruct, interrupt, or

turbance of ecological mitigation ent of the undertaker.

ning of the authorised

grade, inspect, remove and ad/or on the Land, together with tres, cable marker posts, chambers tting electricity along such enchless installation techniques ailways;

t plant, vehicles, machinery, bles;

als for the purposes of laying down, oving and replacing the cables;

ng, construction, repair, testing

poses of the installation, the cables and to erect temporary

eas and compounds including

(1)	(2)
NUMBER OF PLOT SHOWN ON LAND PLANS	PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED
	(h) construct, lay down, use and remove temporary access roads including any necessary hard star including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary of water courses and drains during any period during which construction, maintenance, repair or re
	(i) effect access to the highway;
	(j) make such investigations in or on the Land as required;
	(k) fell, lop or cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be s land which would if not felled, lopped, cut or removed would obstruct or interfere with the operatio
	(I) remove and discharge water from the Land, and to install, retain, use, maintain, inspect, alter, represented and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into an drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or cluband or reinstate the any existing drainage scheme on the Land;
	(m) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus in electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications of equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service cables or conduits or apparatus of statutory undertakers);
	(n) remove fences and structures within the Land during any period during which construction, main being carried out (subject to erection of any temporary stock-proof fencing as is reasonably require suitable replacement of the fences or structures following the exercise of the rights);
	(o) store and stockpile materials (including excavated material);
	(p) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and r trenchless installation technique, to keep in place and monitor the same through construction, main decommissioning and to reinstate the Land;
	(q) to excavate materials below ground level, including soils, and to store and re-use or dispose of the undertake any works, including works of protection or removal of archaeological remains as may be of investigation approved under this Order;
	(r) lay out temporary paths and bridleways for public use;
	(s) to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out m remediation works for environmental or ecological mitigation or enhancement works, including tem measures and the installation of temporary barriers for the protection of fauna;
	(t) carry out such works (together with associated fencing) required by a planning permission and/or over the Land and/or in accordance with any necessary licences relating to protected species and



anding and other surface materials ry bridging, culverting or diversion enewal is being carried out;

- standing on the Land or other on of the cables;
- emove, refurbish, reconstruct,
- and manage waterflows in any cleanse drainage schemes on the

ncluding but not limited to cables and any ancillary ce media (including the pipes,

intenance, repair or renewal is ed and the re-instatement or

monitoring the use of any intenance repair, replacement or

the same, and in so excavating to be required by any written scheme

nitigation, maintenance, nporary works for noise alleviation

or consent now or to be granted d/or wildlife; and

(1) NUMBER OF PLOT SHOWN ON LAND PLANS	(2) PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED
	(u) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maint cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and o together with conduits or pipes for containing the same in and under the Land.
	2. Restrictive covenants
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
	(a) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erect construction, erection or works of any kind (including the foundations or footings thereto) other than benefit of existing highway or railway infrastructure;
	(b) to prevent anything to be done by way of excavation of any kind in the Land nor any activities we decrease ground cover or soil levels by greater than 1 metre whatsoever without the consent in writing where such works are reasonably required for the exercise of statutory functions or rights in relation at the Land and will not damage, undermine or interfere with the cables;
	(c) to prevent the planting or growing within the Land of any trees, shrubs or underwood without the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed t not cause damage to the relevant part of the authorised development nor make it materially more the relevant part of the authorised development);
	(d) to prevent anything being done which may interfere with free flow and passage of electricity or t cables or support for the authorised development; and
	(e) to prevent carrying out operations or actions (including but not limited to blasting and piling) whi interfere with the exercise of the rights or damage the authorised development.
6, 8, 11, 12, 14	Access rights
36, 41, 42, 48,	
51, 53, 55, 56, 58, 61, 63, 66, 76, 77, 78, 82, 83, 102, 106,	Rights for the purposes of the construction, installation, operation, maintenance and decommissionin development to:
114, 115, 116	
125, 126, 129	(a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials
134, 135, 136,	the inspection, testing, maintenance, renewal, upgrading, replacement and removal of the cables
142, 152, 153, 154, 155, 169	

 \sim



ntain and inspect underground d other conducting media

ction of any buildings or n those related to works for the

which would alter, increase or iting of the undertaker, save any public highway or railway on

e consent in writing of the I trees, shrubs or underwood would e difficult or expensive to access

r telecommunications through the

hich may obstruct, interrupt, or

ning of the authorised

als for the purposes of the Works, s and connection into any

(1) NUMBER OF PLOT SHOWN ON LAND PLANS	(2) PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED
174, 175, 178, 179, 180, 183,	adjacent cables and associated works, to take plant and equipment on to adjoining land and make
196, 213, 214	Land which is ancillary for the purposes of exercise of the rights;
220, 222, 225	(b) to construct, use, maintain and improve a permanent means of access including visibility splays, a
226, 233, 235	widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the Land, a
236, 238, 239	(c) construct, lay down, use and remove temporary access roads including any necessary hard stand including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary
240, 241, 243	of water courses and drains during any period during which construction, maintenance, repair, renew
244, 245, 246	carried out;
252, 253, 254	(d) erect temporary bridges and supporting or protective structures for the purposes of access to adj
255, 256, 262	(e) place and use plant, machinery, structures and temporary structures within the Land, and to erec
263, 264, 265	provide measures for the benefit of public and personnel safety;
266, 267, 268	(f) fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be pres
278, 280, 281	of enabling the right to pass and re-pass to adjoining land;
286, 287, 288	(g) repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding together with the right to maintain, inspect and replant such trees, shrubs and landscaping;
292, 295, 296	(h) erect and remove temporary fencing, gates, walls, barriers or other means of enclosure; and
298, 299, 300	(i) lay out temporary paths and bridleways for public use as temporary diversions for public rights of w
301, 302, 304	during any period in which construction, maintenance, repair or renewal decommissioning is being c
305, 306, 336	
337, 344, 346	
354, 355, 356	
358, 366, 367	
368, 371, 372	
373, 377, 379	
380, 407, 412	
429, 436, 439	
440, 461, 467	
468, 469, 470	

~~~~~



#### ke such investigations in or on the

, and retain, maintain, straighten, , adjoining land and highway;

anding and other surface materials ry bridging, culverting or diversion ewal or decommissioning is being

djoining land;

ect temporary signage and

esent on the Land for the purpose

ng and other ecological measures

way which are interfered with carried out.

| (1)<br>NUMBER OF PLOT SHOWN ON LAND<br>PLANS | (2)<br>PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED                                                                                                                                                                                                                                                                                                                                                                        |
|----------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| 418, 419, 420<br>421, 422, 423               | Visibility splay and highway verge rights and restrictive covenants                                                                                                                                                                                                                                                                                                                                                    |
| 426, 427, 428<br>430, 431, 432               | 1. Visibility splay and highway verge rights                                                                                                                                                                                                                                                                                                                                                                           |
|                                              | Rights for the purposes of the construction, installation, operation, maintenance and decommissionin development to:                                                                                                                                                                                                                                                                                                   |
|                                              | (a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials<br>and make such investigations in or on the Land which is ancillary for the purposes of exercise of the rig                                                                                                                                                                                                       |
|                                              | (b) to construct, use, maintain and improve visibility splays, and retain, maintain, straighten, widen, re existing access routes and visibility splays for the purposes of accessing the Land, adjoining land and                                                                                                                                                                                                     |
|                                              | (c) fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be pres<br>of maintaining visibility required for accesses.                                                                                                                                                                                                                                                              |
|                                              | 2. Restrictive covenants                                                                                                                                                                                                                                                                                                                                                                                               |
|                                              | A restrictive covenant over the Land for the benefit of the remainder of the Order land to prevent an<br>the Land or any part thereof for the purpose of the erection of any buildings or construction, erection<br>the foundations or footings thereto) which would interfere with use as a visibility splay, other than work<br>authority or which are otherwise for the benefit of existing highway infrastructure. |
| 145, 150, 160<br>164, 168, 192               | Mitigation work areas access rights, mitigation rights and restrictive covenants                                                                                                                                                                                                                                                                                                                                       |
| 409, 459                                     | 1. Permanent access rights                                                                                                                                                                                                                                                                                                                                                                                             |
|                                              | Rights for the purposes of the construction, installation, operation, maintenance and decommissionin development to:                                                                                                                                                                                                                                                                                                   |
|                                              | (a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials<br>the inspection, testing, maintenance, renewal, upgrading, replacement and removal of the cables a                                                                                                                                                                                                               |



ning of the authorised

als for the purposes of the Works, e rights;

repair, alter, upgrade and use Id highway; and

resent on the Land for the purpose

anything to be done in or upon on or works of any kind (including orks undertaken by the highway

ning of the authorised

als for the purposes of the Works, s and connection into any

| (1)                                | (2)                                                                                                                                                                                                                                                                                                                         |
|------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| NUMBER OF PLOT SHOWN ON LAND PLANS | PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED                                                                                                                                                                                                                                                                                    |
|                                    | adjacent cables and associated works, to take plant and equipment on to adjoining land and mak<br>Land which is ancillary for the purposes of exercise of the rights;                                                                                                                                                       |
|                                    | (b) to construct, use, maintain and improve a permanent means of access including visibility splays, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the Land,                                                                                                                   |
|                                    | (c) construct, lay down, use and remove temporary access roads including any necessary hard star<br>including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary<br>of water courses and drains during any period during which construction, maintenance, repair, rene<br>carried out; |
|                                    | (d) erect temporary bridges and supporting or protective structures for the purposes of access to ac                                                                                                                                                                                                                        |
|                                    | (e) place and use plant, machinery, structures and temporary structures within the Land, and to ere provide measures for the benefit of public and personnel safety;                                                                                                                                                        |
|                                    | (f) fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be pre<br>of enabling the right to pass and re-pass to adjoining land;                                                                                                                                                        |
|                                    | (g) repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding together with the right to maintain, inspect and replant such trees, shrubs and landscaping;                                                                                                                             |
|                                    | (h) erect and remove temporary fencing, gates, walls, barriers or other means of enclosure; and                                                                                                                                                                                                                             |
|                                    | (i) lay out temporary paths and bridleways for public use as temporary diversions for public rights of during any period in which construction, maintenance, repair or renewal decommissioning is being                                                                                                                     |
|                                    | 2. Mitigation works rights                                                                                                                                                                                                                                                                                                  |
|                                    | Rights for the purposes of the construction, installation, operation, maintenance and decommissioni development to:                                                                                                                                                                                                         |
|                                    | (a) install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, wo seeding and other ecological measures together with the right to maintain, inspect and replant suc                                                                                                                    |
|                                    | (b) install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitig<br>remediation works for environmental or ecological mitigation or enhancement works, including tem<br>of temporary barriers for the protection of fauna; and                                                       |



#### ke such investigations in or on the

, and retain, maintain, straighten, , adjoining land and highway;

anding and other surface materials ry bridging, culverting or diversion ewal or decommissioning is being

djoining land;

ect temporary signage and

esent on the Land for the purpose

ng and other ecological measures

way which are interfered with carried out.

ning of the authorised

bodland, shrubs, hedgerows, ch trees, shrubs and landscaping; igation, maintenance, nporary works and the installation

| (1)<br>NUMBER OF PLOT SHOWN ON LAND<br>PLANS | (2)<br>PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|----------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                                              | (c) carry out such works (together with associated fencing) required by a planning permission and/or over the Land in accordance with any necessary licences relating to protected species and/or wildling to protect the species and/or wildling to protect the species and s |
|                                              | 3. Restrictive covenants                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                                              | A restrictive covenant over the Land for the benefit of the remainder of the Order land to prevent an reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas any ploughing or grazing, during the period within which the undertaker is bound by any consent to r mitigation areas or areas of habitat creation, without the prior written consent of the undertaker.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 415                                          | Drainage rights and restrictive covenants                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|                                              | 1. Drainage rights                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|                                              | Rights for the purposes of the construction, installation, operation, maintenance and decommissionin development to:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                                              | (a) remove and discharge water from the Land and to install, retain, use, maintain, inspect, alter, remove and context and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and drains, watercourses and culverts, to lay down, install, adjust, alter, construct, create, use, maintain, remove and replace a drainage scheme on the Land (the "drainage works");                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                              | (b) inspect, use mechanical excavation (including directional drilling and/or digging), reinstate, remo<br>parts of any drainage system on the Land for the purposes of the drainage works (including connection<br>land drain as at the date of the drainage works);                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|                                              | (c) enter, be on, and break up the surface of the Land and remain with or without plant, vehicles, ma<br>equipment which is ancillary to the purposes of the drainage works;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                                              | (d) store and stockpile materials (including excavated material);                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                                              | (e) create boreholes and trail excavation pits for the purposes of intrusively surveying the land and m<br>trenchless installation technique, to keep in place and monitor the same through construction, maint<br>decommissioning and to reinstate the Land;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |



/or consent now or to be granted dlife

any activity which would in the eas of habitat creation including o maintain that ecological

ning of the authorised

emove, refurbish, reconstruct, and manage waterflows in any , repair, renew, upgrade, inspect,

move, move or alter such part or cting the drainage works to any

machinery, apparatus and

monitoring the use of any intenance repair, replacement or

| (1)                                | (2)                                                                                                                                                                                                                                                                                                                                                                               |
|------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| NUMBER OF PLOT SHOWN ON LAND PLANS | PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED                                                                                                                                                                                                                                                                                                                                          |
|                                    | (f) to excavate materials below ground level, including soils, and to store and re-use or dispose of the<br>undertake any works, including works of protection or removal of archaeological remains as may be<br>of investigation approved under this Order;                                                                                                                      |
|                                    | (g) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and material works;                                                                                                                                                                                                                                                                         |
|                                    | (h) place and use plant, machinery, structures and temporary structures within the Land, and to ere provide measures for the benefit of public and personnel safety for the purposes of the drainage we                                                                                                                                                                           |
|                                    | (i) erect fencing, gates, walls, barriers or other means of enclosure, and create secure works areas of temporary trenchless installation technique compounds and working areas for the purposes of the o                                                                                                                                                                         |
|                                    | (j) construct, lay down, use and remove temporary access roads including any necessary hard stand<br>including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary<br>of water courses and drains during any period during which construction, maintenance, repair or re-<br>being carried out;                                               |
|                                    | (k) effect access to the highway;                                                                                                                                                                                                                                                                                                                                                 |
|                                    | (I) make such investigations in or on the Land as required for the purposes of the drainage works;                                                                                                                                                                                                                                                                                |
|                                    | (m) use or resort to trenchless installation techniques including (but not limited to) directional drilling works;                                                                                                                                                                                                                                                                |
|                                    | (n) fell, lop or cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be shand which would if not felled, lopped, cut or removed obstruct or interfere with the drainage works                                                                                                                                                                          |
|                                    | (o) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus in electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service cables or conduits or apparatus of statutory undertakers); |
|                                    | (p) to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out m<br>remediation works for environmental or ecological mitigation or enhancement works, including tem<br>measures and the installation of temporary barriers for the protection of fauna; and                                                                                |
|                                    | (q) to carry out such works (together with associated fencing) required by a planning permission an granted over the Land in accordance with any necessary licences relating to protected species an                                                                                                                                                                              |
|                                    | 2. Restrictive covenants                                                                                                                                                                                                                                                                                                                                                          |



he same, and in so excavating, to be required by any written scheme

als for the purposes of the drainage

- ect temporary signage and vorks;
- or compounds including drainage works;
- nding and other surface materials ry bridging, culverting or diversion enewal of any drainage work is

in connection with the drainage

- standing on the Land or other s;
- ncluding but not limited to
- cables and any ancillary
- ce media (including the pipes,

mitigation, maintenance, nporary works for noise alleviation

nd/or consent now or to be nd/or wildlife.

| (1)<br>NUMBER OF PLOT SHOWN ON LAND<br>PLANS | (2)<br>PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED                                                                                                                                                                                                                                                                                                                                                                             |
|----------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                                              | A restrictive covenant over the Land for the benefit of the remainder of the Order land to:                                                                                                                                                                                                                                                                                                                                 |
|                                              |                                                                                                                                                                                                                                                                                                                                                                                                                             |
|                                              | (a) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erect construction erection or works of any kind (including the foundations or footings thereto); and                                                                                                                                                                                                                        |
|                                              | (b) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or w<br>whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably w<br>proposed surfacing would not cause damage to relevant part of the authorised development nor m<br>expensive to maintain the authorised development).                                                           |
| 141, 143, 144                                | Mitigation work rights and restrictive covenants                                                                                                                                                                                                                                                                                                                                                                            |
| 146, 148, 149                                |                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 151, 156, 158                                | 1. Mitigation works rights                                                                                                                                                                                                                                                                                                                                                                                                  |
| 163, 165, 166                                |                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 167, 177, 185                                | Rights for the purposes of the construction, installation, operation, maintenance and decommissionin                                                                                                                                                                                                                                                                                                                        |
| 186, 188, 189                                | development to:                                                                                                                                                                                                                                                                                                                                                                                                             |
| 191, 193 198                                 |                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 199, 200, 201                                | (a) install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, woo                                                                                                                                                                                                                                                                                                                      |
| 202, 203, 204                                | seeding and other ecological measures together with the right to maintain, inspect and replant such                                                                                                                                                                                                                                                                                                                         |
| 206, 207, 208                                | (b) install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitig<br>remediation works for environmental or ecological mitigation or enhancement works, including temp                                                                                                                                                                                                                |
| 209, 360, 361                                | of temporary barriers for the protection of fauna; and                                                                                                                                                                                                                                                                                                                                                                      |
| 362, 363, 364                                | (c) carry out such works (together with associated fencing) required by a planning permission and/o                                                                                                                                                                                                                                                                                                                         |
| 376, 378, 408                                | over the Land in accordance with any necessary licences relating to protected species and/or wildli                                                                                                                                                                                                                                                                                                                         |
| 410, 445,446                                 |                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 447, 448, 456                                | 2. Restrictive covenants                                                                                                                                                                                                                                                                                                                                                                                                    |
| 457, 458, 460                                |                                                                                                                                                                                                                                                                                                                                                                                                                             |
|                                              | A restrictive covenant over the Land for the benefit of the remainder of the Order land to prevent ar<br>reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or area<br>any ploughing or grazing, during the period within which the undertaker is bound by any consent to<br>mitigation areas or areas of habitat creation, without the prior written consent of the undertaker. |



ction of any buildings or

with any other material or surface withheld or delayed if the make it materially more difficult or

ning of the authorised

oodland, shrubs, hedgerows, ch trees, shrubs and landscaping;

igation, maintenance, nporary works and the installation

/or consent now or to be granted dlife

any activity which would in the eas of habitat creation including o maintain that ecological

| (1)<br>NUMBER OF PLOT SHOWN ON LAND<br>PLANS | (2)<br>PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|----------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| 472, 473, 478                                | National Grid substation works area rights                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| 479, 480, 481<br>482, 483, 484               | Rights for the purposes of the construction, installation, operation, maintenance and decommissionin development to:                                                                                                                                                                                                                                                                                                                                                                            |
| 485, 486                                     | (a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrareplace the electricity cables (including the removal of materials including spoil) in, under, over and/such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures and manholes, manhole covers and other equipment which is ancillary to the purposes of transmittin electricity cables (all collectively referred to as the "cables"); |
|                                              | (b) enter, be on, and break open and break up the surface of the Land and remain with or without p<br>apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cabl                                                                                                                                                                                                                                                                                     |
|                                              | (c) to benefit from continuous vertical and lateral support for the authorised development;                                                                                                                                                                                                                                                                                                                                                                                                     |
|                                              | (d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting the authremoving and replacing the cables;                                                                                                                                                                                                                                               |
|                                              | (e) to use, maintain, renew improve and alter existing accesses, roads, streets, tracks or ways over the not exclusive and exercise of this right must not prevent or unreasonably inhibit use by other parties;                                                                                                                                                                                                                                                                                |
|                                              | (f) construct, lay down, use and remove temporary access roads including any necessary hard stand<br>including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary<br>of water courses and drains during any period during which construction, maintenance, repair, renew<br>carried out;                                                                                                                                                                   |
|                                              | (g) place and use plant, machinery, structures and temporary structures within the Land, and to erect provide measures for the benefit of public and personnel safety;                                                                                                                                                                                                                                                                                                                          |
|                                              | (h) fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be pres<br>of enabling the right to pass and re-pass to adjoining land; and                                                                                                                                                                                                                                                                                                                       |
|                                              | (i) remove and discharge water from the Land and to install, retain, use, maintain, inspect, alter, remove replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and drains, watercourses and culverts, to lay down, install, adjust, alter, construct, create, use, maintain, remove and replace a drainage scheme on the Land; and                                                                                                              |
|                                              | (j) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus inclue<br>electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications ca                                                                                                                                                                                                                                                                    |



ning of the authorised

grade, inspect, remove and d/or on the Land, together with res, cable marker posts, chambers ting electricity along such

t plant, vehicles, machinery, bles;

als for the purposes of laying down, uthorised development and for

he Land, providing that such use is ;

nding and other surface materials ry bridging, culverting or diversion ewal or decommissioning is being

ect temporary signage and

resent on the Land for the purpose

move, refurbish, reconstruct, and manage waterflows in any , repair, renew, upgrade, inspect,

cluding but not limited to cables and any ancillary

| (1)<br>NUMBER OF PLOT SHOWN ON LAND<br>PLANS | (2)<br>PURPOSE FOR WHICH RIGHTS MAY BE ACQUIRED                                                                                                                                                                                                                                                                                                                                                 |  |  |
|----------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                                              | equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service cables or conduits or apparatus of statutory undertakers).                                                                                                                                                                                                                                  |  |  |
| 487, 488, 489                                | National Grid substation access area rights                                                                                                                                                                                                                                                                                                                                                     |  |  |
| 490, 491, 492<br>496, 497                    | Rights for the purposes of the construction, installation, operation, maintenance and decommissionin development to:                                                                                                                                                                                                                                                                            |  |  |
|                                              | (a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting the authremoving and replacing the cables;                                                                                                                                               |  |  |
|                                              | (b) to use, maintain, renew improve and alter existing accesses, roads, streets, tracks or ways over the not exclusive and exercise of this right must not prevent or unreasonably inhibit use by other parties;                                                                                                                                                                                |  |  |
|                                              | (c) construct, lay down, use and remove temporary access roads including any necessary hard stand<br>including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary<br>of water courses and drains during any period during which construction, maintenance, repair, renew<br>carried out;                                                                   |  |  |
|                                              | (d) place and use plant, machinery, structures and temporary structures within the Land, and to erect provide measures for the benefit of public and personnel safety;                                                                                                                                                                                                                          |  |  |
|                                              | (e) fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be pre<br>of enabling the right to pass and re-pass to adjoining land; and                                                                                                                                                                                                                        |  |  |
|                                              | (f) remove and discharge water from the Land and to install, retain, use, maintain, inspect, alter, remove replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and drains, watercourses and culverts, to lay down, install, adjust, alter, construct, create, use, maintain, remove and replace a drainage scheme on the Land; and              |  |  |
|                                              | (g) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus incl<br>electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications ca<br>equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service<br>cables or conduits or apparatus of statutory undertakers). |  |  |



ce media (including the pipes,

ning of the authorised

als for the purposes of laying down, uthorised development and for

the Land, providing that such use is ;

anding and other surface materials ry bridging, culverting or diversion ewal or decommissioning is being

ect temporary signage and

resent on the Land for the purpose

move, refurbish, reconstruct, and manage waterflows in any , repair, renew, upgrade, inspect,

ncluding but not limited to cables and any ancillary ce media (including the pipes,

### 3 Structure of this book of reference

- 10 This Book of Reference is comprised of five Parts, in accordance with Regulation 7(1) of the Regulations, as follows:
  - Part 1 contains the names and addresses of those who own, lease, occupy or have another interest in the land that will be affected by the authorised development and the rights contained in the Order. A person is within Category 1 if the Applicant after making diligent inquiry, knows that they are an owner, lessee, tenant, or occupier of the land. A person is within Category 2 if they are interested in the land or have the power to sell and convey or release the land.

Parties recorded in Part 3 of this Book of Reference have also been included in Part 1.

- Part 2 lists persons who may be entitled to make a relevant claim, also called category 3 persons. A person is within Category 3 if the Applicant believes that, if the Order were to be made and fully implemented, they would or might be entitled to make a relevant claim as defined in section 57(6) of the 2008 Act. A relevant claim is a claim under section 10 of the Compulsory Acquisition Act 1965, a claim under part 1 of the Land Compensation Act 1973, or a claim under section 152(3) of the 2008 Act. This part therefore contains the names and addresses of those who have an interest in the Order land and may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the authorised development. This part also contains the names and addresses of those without an interest in the Order land, who may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the authorised development;
- Part 3 contains the names and addresses of those entitled to enjoy easements or other private rights which it is proposed may be extinguished, suspended or interfered with in connection with the authorised development, pursuant to the Order;
- Part 4 identifies plots in which there is a Crown interest that will be affected by the authorised development and the rights contained in the Order; and
- Part 5 identifies plots which constitute "special category land" for the purposes of section 132 of the Planning Act 2008 that will be affected by the authorised development and the rights contained in the Order. This is land –



- the acquisition of which is subject to special parliamentary procedure;
- which is special category land
- which is replacement land

The Applicant believes that if the Order land listed within Part 5, when burdened with the Order right, will be no less advantageous than it was before to the following persons-

- the persons in whom it is vested,
- other persons, if any, entitled to rights of common or other rights, and
- the public.

The Applicant is therefore of the opinion that while it has identified open space within the Order Land, there is no requirement for the provision of replacement or exchange land.



### 4 Book of reference notes

- 11 Part 1 of this Book of Reference provides the area in square metres of all land included in the DCO. Areas have not been repeated in Parts 2 to 5 of this Book of Reference.
- 12 The term 'approximately' is used before all plot area measurements as these measurements are given in square metres and each measurement is rounded up to the nearest whole square metre.



### 5 Book of Reference parts 1-5



|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                                |                                                                                                                                |                                                                                            |                                                                                            |                                                                                                                                                                                                                                                                                               |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                | Qualifying persons under Regulatio                                                                                             | Category 1<br>n 7(1)(a) of the Infrastructure Planning                                     | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                    |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                |                                                                                                                                | Procedures) Regulations 2009                                                               |                                                                                            |                                                                                                                                                                                                                                                                                               |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                                            | Owners or Reputed Owners                                                                                                       | Lessees or Tenants                                                                         | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                         |  |  |
| 1                       | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 217151 square metres<br>of foreshore land<br>seaward of the Mean<br>High Water lying to the<br>north of Prestatyn<br>(excluding all interests of<br>the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                     |  |  |
| 2                       | Temporary Possession                                                                                                                                                                                                                                                                                 | 111434 square metres<br>of foreshore land<br>seaward of the Mean<br>High Water lying to the<br>north of Prestatyn<br>(excluding all interests of<br>the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter) |  |  |
| 3                       | Temporary Possession                                                                                                                                                                                                                                                                                 | 8428 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)           | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | NONE                                                                                                                                                                                                                                                                                          |  |  |
| 4                       | Temporary Possession                                                                                                                                                                                                                                                                                 | 155 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)            | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Unknown                                                                                                                                                                                                                                                                                       |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                       |                                                                                                                                                                                                                                                                               |                                                                                                                              |                                                                                            |                                                                                                                                                                                                                                                                                                                |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                                                            | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and |                                                                                            |                                                                                                                                                                                                                                                                                                                |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                                   | Owners or Reputed Owners                                                                                                                                                                                                                                                      | Procedures) Regulations 2009<br>Lessees or Tenants                                                                           | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                          |  |  |
| 4 cont'd                |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       | Unknown                                                                                                                                                                                                                                                                       |                                                                                                                              |                                                                                            |                                                                                                                                                                                                                                                                                                                |  |  |
| 5                       | Temporary Possession                                                                                                                                                                                                                                                                                 | 33060 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                                                                                                                                                      | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | NONE                                                                                                                                                                                                                                                                                                           |  |  |
| 6                       | Acquisition of new rights                                                                                                                                                                                                                                                                            | 3496 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)  | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                                                                                                                                                      | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | NONE                                                                                                                                                                                                                                                                                                           |  |  |
| 7                       | Temporary Possession                                                                                                                                                                                                                                                                                 | 34 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>at Rhyl (excluding all<br>interests of the Crown)                          | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown<br>(in respect of mines and minerals) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder)       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                  |                                                                                                                                                              |                    |                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------|--------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                  | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                    |                                                                                            | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                              | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
| 8                       | Acquisition of new rights                                                                                                                                                                                                                                                                            | 4615 square metres of<br>footway (north of Rhyl<br>Coast Road, A548) and<br>footpath (DE 207 11) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals)                  | NONE               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Natural Resources Body For<br>Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883) |  |  |
| 9                       | Temporary Possession                                                                                                                                                                                                                                                                                 | 2 square metres of<br>footway (north of Rhyl<br>Coast Road, A548)                                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                   | NONE               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009)                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                 |                                                                                                                                                                                                                                                               |                                                                                                                        |                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                 |                                                                                                                                                                                                                                                               |                                                                                                                        |                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                 | Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                |                                                                                                                        |                                                                                                                                                                                                                                    | Category 2                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                             | Owners or Reputed Owners                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                     | Occupiers                                                                                                                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 9 cont'd                |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                 | Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                                                 |                                                                                                                        |                                                                                                                                                                                                                                    | Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883)                                 |  |  |
| 10                      | Temporary Possession                                                                                                                                                                                                                                                                                 | 113 square metres of<br>foreshore landward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder)                                                                                                             | Unknown                                                                                                                                               |  |  |
| 11                      | Acquisition of new rights                                                                                                                                                                                                                                                                            | 12 square metres of<br>slipway seaward of the<br>Mean High Water lying<br>to the north of Rhyl<br>(excluding all interests of<br>the Crown)     | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown                                                                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder)<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH | Unknown                                                                                                                                               |  |  |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                              |                                                                                                                          |                                                                                                                                                              |                    |                                                                                                          |                                                                                                                                                                                                                                                                                         |  |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------|----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                                                                                                                                                                                                                                                                                                      |                              |                                                                                                                          | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                    |                                                                                                          | Category 2                                                                                                                                                                                                                                                                              |  |
| Number on<br>Land Plans                                                                                                                                                                                                                                                                              | Extent of acquisition or use | Description of land                                                                                                      | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants | Occupiers                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                   |  |
| 12                                                                                                                                                                                                                                                                                                   | Acquisition of new rights    | 86 square metres of<br>slipway landward of<br>Mean High Water north<br>of Rhyl (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as assumed owner)<br>Unknown                    | NONE               | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH | Unknown                                                                                                                                                                                                                                                                                 |  |
| 13                                                                                                                                                                                                                                                                                                   | Temporary Possession         | 345 square metres of<br>footway and grassed<br>area (north of Garford<br>Road)                                           | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                   | NONE               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN               | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>The Natural Resources Body For<br>Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access) |  |
| 14                                                                                                                                                                                                                                                                                                   | Acquisition of new rights    | 347 square metres of<br>footway, hardstanding<br>and grassed area (north<br>of Garford Road)                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                   | NONE               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN               | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                    |                                                                                                                                             |                    |                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|--------------------|--------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                    |                                                                                                                                             |                    |                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                    | Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |                    |                                                                                            | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                           |  |  |
| 14 cont'd               |                                                                                                                                                                                                                                                                                                      |                                                                                                                    |                                                                                                                                             |                    |                                                                                            | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Natural Resources Body For<br>Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 OTP<br>(in respect of right of access)                                                                                                                                                                          |  |  |
| 15                      | Temporary Possession                                                                                                                                                                                                                                                                                 | 3125 square metres of<br>hardstanding and<br>grassed area (north of<br>Brynhedydd Bay) and<br>footpath (DE 207 11) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883) |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                  | BOOK OF RI<br>powers of compulsory acquisition, (ii)                   | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire                                            | to carry out protective works (Regulatio                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------|------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                              |                                                                                                 | Qualifying persons under Regulation                                    | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                             | Owners or Reputed Owners                                               | Lessees or Tenants                                                                                                                                           | Occupiers                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 16                      | Temporary Possession         | 2728 square metres of<br>private road, buildings<br>and hardstanding at Rhyl<br>Golf Club, Rhyl | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE | NONE                                                                                                                                                         | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE | AP Wireless II (UK) Limited<br>A P Wireless Ltd<br>16-18 Conduit Street<br>LICHFIELD<br>WS13 6JR<br>(in respect of rights stated in Lease<br>dated 28th September 2012)<br>Cornerstone Telecommunications<br>Infrastructure Limited<br>Hive 2<br>1530 Arlington Business Park<br>Theale<br>READING<br>Berkshire<br>RG7 4SA<br>(in respect of rights granted by Lease<br>dated 17th November 2017)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>EE Limited<br>Trident Place<br>Mosquito Way<br>HATFIELD<br>Hertfordshire<br>AL10 9BW<br>(in respect of rights stated in Lease<br>dated 31st March 2004) |  |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|-------------------------|------------------------------|--------------------------------|---------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                         | Category 1                                                                                                     |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|                         |                              |                                | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planning (<br>Procedures) Regulations 2009                                     | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                | Lessees or Tenants                                                                                             | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| 16 cont'd               |                              |                                |                                                         |                                                                                                                |                                         | Hutchison 3G UK Limited<br>450 Longwater Avenue<br>READING<br>RG2 6GF<br>(in respect of rights stated in Lease<br>dated 31st March 2004)<br>Lloyds Bank PLC<br>25 Gresham Street<br>LONDON<br>EC2V 7HN<br>(as mortgagee for Rhyl Golf Club<br>Limited)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications<br>apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                 | BOOK OF R<br>powers of compulsory acquisition, (ii)                    | ITM DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                           | o carry out protective works (Regulatio                                                                                                         | n 7(1)(a))                                                                                                                                              |
|-------------------------|------------------------------|----------------------------------------------------------------|------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                | Qualifying persons under Regulati                                      | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                         | (Applications: Prescribed Forms and                                                                                                             | Category 2                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                            | Owners or Reputed Owners                                               | Lessees or Tenants                                                                                                                              | Occupiers                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009   |
| 17                      | Temporary Possession         | 12 square metres of<br>hardstanding at Rhyl<br>Golf Club, Rhyl | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE | AP Wireless II (UK) Limited<br>A P Wireless Ltd<br>16-18 Conduit Street<br>LICHFIELD<br>WS13 6JR                                                | AP Wireless II (UK) Limited<br>A P Wireless Ltd<br>16-18 Conduit Street<br>LICHFIELD<br>WS13 6JR                                                | EE Limited<br>Trident Place<br>Mosquito Way<br>HATFIELD<br>Hertfordshire<br>AL10 9BW<br>(in respect of rights stated in Lease<br>dated 31st March 2004) |
|                         |                              |                                                                |                                                                        | Cornerstone Telecommunications<br>Infrastructure Limited<br>Hive 2<br>1530 Arlington Business Park<br>Theale<br>READING<br>Berkshire<br>RG7 4SA | Cornerstone Telecommunications<br>Infrastructure Limited<br>Hive 2<br>1530 Arlington Business Park<br>Theale<br>READING<br>Berkshire<br>RG7 4SA | Hutchison 3G UK Limited<br>450 Longwater Avenue<br>READING<br>RG2 6GF<br>(in respect of rights stated in Lease<br>dated 31st March 2004)                |
|                         |                              |                                                                |                                                                        |                                                                                                                                                 |                                                                                                                                                 | Lloyds Bank PLC<br>25 Gresham Street<br>LONDON<br>EC2V 7HN<br>(as mortgagee for Rhyl Golf Club<br>Limited)                                              |
|                         |                              |                                                                |                                                                        |                                                                                                                                                 |                                                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)           |

|                         | Land which is p              | proposed to be subject to: (i)                                  | BOOK OF RE<br>powers of compulsory acquisition, (ii)                   | rm DEVELOPMENT CONSENT ORDER<br>:FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire                                           | to carry out protective works (Regulatio                               | n 7(1)(a))                                                                                                                                                                                                                                                                     |
|-------------------------|------------------------------|-----------------------------------------------------------------|------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                 | Qualifying persons under Regulation                                    | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                        |                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                             | Owners or Reputed Owners                                               | Lessees or Tenants                                                                                                                                           | Occupiers                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                          |
| 18                      | Temporary Possession         | 206 square metres of<br>grassed area at Rhyl Golf<br>Club, Rhyl | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE | NONE                                                                                                                                                         | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of right of access)                                                                                                                                                  |
| 19                      | Temporary Possession         | 15 square metres of<br>grassed area at Rhyl Golf<br>Club, Rhyl  | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE | NONE                                                                                                                                                         | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of right of access)                                                                                                                                                  |
| 20                      | Temporary Possession         | 123 square metres of<br>private road at Rhyl Golf<br>Club, Rhyl | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE | NONE                                                                                                                                                         | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                        |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                   | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                     | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                         |  |  |  |
| 21                      | Temporary Possession                                                                                                                                                                                                                                                                                 | 483 square metres of<br>footway of public road<br>(north of Rhyl Coast<br>Road, A548) | Carol Susan Hough<br>186 Rhyl Coast Road<br>RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>David Alan Hammans<br>192 Rhyl Coast Road<br>Y RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Eileen Mary Harpin<br>184 Rhyl Coast Road<br>Y RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Category 1                                                                                                        |                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and     | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                                | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 21 cont'd               |                              |                               | Gary Michael Hughes<br>188 Rhyl Coast Road<br>RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Gillian Wendy Callaghan<br>180 Rhyl Coast Road<br>Y RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Graham Turner<br>178 Rhyl Coast Road<br>RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Jean Dorothy Turner<br>178 Rhyl Coast Road<br>RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Jean Dorothy Turner<br>178 Rhyl Coast Road<br>RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Joseph Callaghan<br>180 Rhyl Coast Road<br>Y RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                                   |                                         | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)       |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulation | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Category 1                                                                                                     |                                          |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | n 7(1)(a) of the Infrastructure Planning (<br>Procedures) Regulations 2009                                     | Applications: Prescribed Forms and       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | Lessees or Tenants                                                                                             | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 21 cont'd               |                              |                               | Lynda May Hughes<br>188 Rhyl Coast Road<br>RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Roberta Walker Belmont<br>176 Rhyl Coast Road<br>RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Thomas David Belmont<br>176 Rhyl Coast Road<br>RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Thomas David Belmont<br>176 Rhyl Coast Road<br>RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown<br>Wendy Susan Morrall<br>182 Rhyl Coast Road<br>Y RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                |                                          |                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                 | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                | Qualifying persons under Regulatio                                                                                                                                                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                                                                                                        | Category 2                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                            | Owners or Reputed Owners                                                                                                                                                                                                                                                    | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                    |
| 21 cont'd               |                              |                                                                                                                                                | William Derek Lloyd<br>190 Rhyl Coast Road<br>RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway)<br>William Thomas Morrall<br>182 Rhyl Coast Road<br>Y RHYL<br>LL18 3RE<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                                   |                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                          |
| 22                      | Temporary Possession         | 152117 square metres<br>of grassed area and<br>hardstanding (Rhyl Golf<br>Club, north of Rhyl Coast<br>Road, A548) and<br>footpath (DE 207 11) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals)                                                                                                                                 | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 207 11)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                  | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                 |                                                                                                                                                                                                                                                               | Category 1                                                                                                        |                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                 |
|                         |                              |                                                                                                                                                 | Qualifying persons under Regulatio                                                                                                                                                                                                                            | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                             | Owners or Reputed Owners                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                           |
| 22 cont'd               |                              |                                                                                                                                                 |                                                                                                                                                                                                                                                               |                                                                                                                   |                                                                                            | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |
| 23                      | Temporary Possession         | 34 square metres of<br>foreshore landward of<br>the Mean High Water<br>lying to the north of<br>Rhyl, (excluding all<br>interests of the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                        | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Unknown                                                                                                                                                                                                                                                                                                                                                                                         |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                           | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                      | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire                                             | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                              |                                                                                                                          | Qualifying persons under Regulatio                                                                                                          | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                      | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                   |  |
| 24                      | Temporary Possession         | 4159 square metres of<br>foreshore landward of<br>the Mean High Water<br>and footway (north of<br>Rhyl Coast Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883)                                                                                                                                          |  |
| 25                      | Temporary Possession         | 15 square metres of<br>hardstanding (north of<br>Rhyl Coast Road, A548)                                                  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009)<br>The Natural Resources Body For<br>Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883) |  |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                                     | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire      | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                     |
|-------------------------|----------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                                    |                                                                                                                                                                                                                                                                               | Category 1                                                                                                             |                                                                                            |                                                                                                                                                                                                                                                                                                                |
|                         |                                                          |                                                                                                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                                                            | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                               | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                                | Owners or Reputed Owners                                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                     | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                          |
| 26                      | Plot No Longer Included In<br>DCO                        | Plot No Longer Included<br>In DCO                                                                                                                  | Plot No Longer Included In DCO                                                                                                                                                                                                                                                | Plot No Longer Included In DCO                                                                                         | Plot No Longer Included In DCO                                                             | Plot No Longer Included In DCO                                                                                                                                                                                                                                                                                 |
| 27                      | Temporary Possession                                     | 1209 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>at Rhyl (excluding all<br>interests of the Crown)                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown<br>(in respect of mines and minerals) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883) |
| 28                      | Acquisition of new rights and imposition of restrictions | 16 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                                    | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009)                                                                                                                          |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                                       | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire       | o carry out protective works (Regulatio                                                                                | n 7(1)(a))                                                                                                                                                                            |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                                      |                                                                                                                                                                                                                                                               | Category 1                                                                                                             |                                                                                                                        |                                                                                                                                                                                       |
|                         |                                                          |                                                                                                                                                      | Qualifying persons under Regulatio                                                                                                                                                                                                                            | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                               | (Applications: Prescribed Forms and                                                                                    | Category 2                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                                  | Owners or Reputed Owners                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                     | Occupiers                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                 |
| 28 cont'd               |                                                          |                                                                                                                                                      | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown<br>(in respect of mines and minerals)                                                                               |                                                                                                                        |                                                                                                                        | Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883)                                                                 |
| 29                      | Acquisition of new rights and imposition of restrictions | 466 square metres of<br>foreshore landward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Unknown                                                                                                                                                                               |
| 30                      | Acquisition of new rights and imposition of restrictions | 8349 square metres of<br>foreshore landward of<br>the Mean High Water<br>and footway lying to the<br>north of Rhyl                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                    | NONE                                                                                                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                             | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                       | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire       | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                     |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                      | Qualifying persons under Regulatio                                                                                                                                                                                                                                            | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                 | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                  | Owners or Reputed Owners                                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                     | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                          |
| 30 cont'd               |                                                             |                                                                                                                                                      | Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                                                                 |                                                                                                                        |                                                                                            | Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883)                                                                                                                                                                                          |
| 31                      | Acquisition of new rights and<br>imposition of restrictions | 218 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown<br>(in respect of mines and minerals) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883) |
| 32                      | Acquisition of new rights and imposition of restrictions    | 6074 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                                    | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter as contained in a<br>Lease dated 8th July 2009)                                                                                                                          |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                           | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                  |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                          |                                                                                                                                                                                 | Category 1                                                                                                        |                                                                                            |                                                                                                                                                             |
|                         |                                                             |                                                                                                                          | Qualifying persons under Regulatio                                                                                                                                              | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                  |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                      | Owners or Reputed Owners                                                                                                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009       |
| 32 cont'd               |                                                             |                                                                                                                          | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                   |                                                                                            | Unknown<br>(in respect of right of access and<br>maintenance as contained in an<br>Indenture dated 20th October 1883)                                       |
| 33                      | Acquisition of new rights and<br>imposition of restrictions | 63 square metres of<br>footway (north of Rhyl<br>Coast Road, A548)                                                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown                                                     | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Unknown                                                                                                                                                     |
| 34                      | Acquisition of new rights and imposition of restrictions    | 3534 square metres of<br>foreshore landward of<br>the Mean High Water<br>and footway (north of<br>Rhyl Coast Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown                                                     | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Unknown<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                     | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights 1<br>f Denbighshire | to carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                  |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                                            | Category 1<br>n 7(1)(a) of the Infrastructure Planning                                                           | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                  |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                | Owners or Reputed Owners                                                                                                                                                                                                                                      | Procedures) Regulations 2009<br>Lessees or Tenants                                                               | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009       |
| 35                      | Acquisition of new rights and<br>imposition of restrictions | 9198 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>and footway at<br>Prestatyn (excluding all<br>interests of the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Unknown                                                                                                                                                     |
| 36                      | Acquisition of new rights                                   | 160 square metres of<br>footway (north of Rhyl<br>Coast Road, A548)                                                                                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown                                                                                                                                   | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Unknown |
| 37                      | Temporary Possession                                        | 275 square metres of<br>foreshore landward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown                                                                                                                                   | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Unknown |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                            | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire       | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                           | Qualifying persons under Regulatio                                                                                                                                                                                                                            | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                 | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                       | Owners or Reputed Owners                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                     | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                         |
| 38                      | Temporary Possession         | 379 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown)  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown | NONE                                                                                                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Unknown<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)                                                                                                                                   |
| 39                      | Temporary Possession         | 3649 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                                  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                                            | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                           |                                                                                            | Category 1                                                                                                       |                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                         |                              |                                                                                                                                                                           | Qualifying persons under Regulatio                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                       | Owners or Reputed Owners                                                                   | Lessees or Tenants                                                                                               | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                 |
| 39 cont'd               |                              |                                                                                                                                                                           |                                                                                            |                                                                                                                  |                                                                                            | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                  |
| 40                      | Temporary Possession         | 1228 square metres of<br>foreshore landward of<br>the Mean High Water<br>and hardstanding lying<br>to the north of Prestatyn<br>and footpath<br>(DE 205 48,<br>DE 205 49) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of access) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                   | BOOK OF RE powers of compulsory acquisition, (ii)                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                  | Qualifying persons under Regulatic                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                              | Owners or Reputed Owners                                                                                                    | Lessees or Tenants                                                                                                | Occupiers                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                 |
| 41                      | Acquisition of new rights    | 1013 square metres of<br>footway (north of Rhyl<br>Coast Road, A548) and<br>footpath (DE 205 48) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                  | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN            | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of access) |
| 42                      | Acquisition of new rights    | 35 square metres of<br>footway (north of Rhyl<br>Coast Road, A548)                               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                                                  | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire       | o carry out protective works (Regulation                                                                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                                 | Qualifying persons under Regulatio                                                                                                                                                                                                                            | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                 | (Applications: Prescribed Forms and                                                                                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                             | Owners or Reputed Owners                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                     | Occupiers                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 43                      | Temporary Possession         | 67 square metres of<br>footway and foreshore<br>landward of the Mean<br>High Water (north of<br>Rhyl Coast Road, A548)<br>and footpath<br>(DE   205   49)                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown                                                                                                                                   | NONE                                                                                                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 205 49)<br>Unknown | Unknown                                                                                                                                               |
| 44                      | Temporary Possession         | 47 square metres of<br>hardstanding and<br>foreshore land seaward<br>of the Mean High Water<br>(north of Rhyl Coast<br>Road, A548) (excluding<br>all interests of the<br>Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown | NONE                                                                                                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown                                       | Unknown                                                                                                                                               |
| 45                      | Temporary Possession         | 183 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown)                        | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                    | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                  | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)               |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                              | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire      | o carry out protective works (Regulatio                                                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                             |                                                                                                                                | Category 1                                                                                                             |                                                                                                                        |                                                                                                                                                                                                                                                                                               |
|                         |                              |                                                                                                                                                             | Qualifying persons under Regulatio                                                                                             | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                               | (Applications: Prescribed Forms and                                                                                    | Category 2                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                         | Owners or Reputed Owners                                                                                                       | Lessees or Tenants                                                                                                     | Occupiers                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                         |
| 45 cont'd               |                              |                                                                                                                                                             | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner) |                                                                                                                        |                                                                                                                        | Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)                                                                                                                                                                                                      |
|                         |                              |                                                                                                                                                             | Unknown<br>(in respect of mines and minerals)                                                                                  |                                                                                                                        |                                                                                                                        | Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951)                                                                                                                                                                                      |
| 46                      | Temporary Possession         | 28705 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of<br>Prestatyn, (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                             | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter) |
| 47                      | Temporary Possession         | 73 square metres of<br>foreshore landward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown)         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                     |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                           |                                                                                                                                           |                                                                                                                        |                                                                                            |                                                                                                                                                                                                                                                                                               |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                           |                                                                                                                                           | Category 1                                                                                                             |                                                                                            |                                                                                                                                                                                                                                                                                               |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                           | Qualifying persons under Regulatio                                                                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                               | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                    |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                                       | Owners or Reputed Owners                                                                                                                  | Lessees or Tenants                                                                                                     | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                         |  |  |  |
| 47 cont'd               |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                           | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown |                                                                                                                        |                                                                                            | Unknown                                                                                                                                                                                                                                                                                       |  |  |  |
| 48                      | Acquisition of new rights                                                                                                                                                                                                                                                                            | 4312 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter) |  |  |  |
| 49                      | Temporary Possession                                                                                                                                                                                                                                                                                 | 7 square metres of<br>footway seaward of the<br>Mean High Water (north<br>of Victoria Road, A548)<br>(excluding all interests of<br>the Crown)            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                       |                                                                                                                                             |                                                                          |                                                                                            |                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                       |                                                                                                                                             | Category 1                                                               |                                                                                            |                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                       | Qualifying persons under Regulatio                                                                                                          | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                      |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                   | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                       | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                           |  |  |  |
| 49 cont'd               |                                                                                                                                                                                                                                                                                                      |                                                                                                                                       | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                    |                                                                          |                                                                                            | Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)                                                                                                                                                                                                                                                        |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                       | Unknown<br>(in respect of mines and minerals)                                                                                               |                                                                          |                                                                                            | Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951)                                                                                                                                                                                                                                        |  |  |  |
| 50                      | Temporary Possession                                                                                                                                                                                                                                                                                 | 17734 square metres of<br>footway landward of the<br>Mean High Water (north<br>of Victoria Road, A548)<br>and footpath<br>(DE 205 49) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |  |  |  |
| 51                      | Acquisition of new rights                                                                                                                                                                                                                                                                            | 5853 square metres of<br>footway (north of<br>Victoria Road, A548) and<br>footpath (DE 205 49)                                        | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                  | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                                                                         |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                       |                                                                                                                                                                                                                                                               |                                                                                                                        |                                                                                                                        |                                                                                                                                                                                                      |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                                            | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                 | (Applications: Prescribed Forms and                                                                                    | Category 2                                                                                                                                                                                           |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                                   | Owners or Reputed Owners                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                     | Occupiers                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                |  |  |  |
| 51 cont'd               |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       | Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                                                 |                                                                                                                        |                                                                                                                        | Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |  |  |  |
| 52                      | Temporary Possession                                                                                                                                                                                                                                                                                 | 3329 square metres of<br>foreshore landward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Unknown                                                                                                                                                                                              |  |  |  |
| 53                      | Acquisition of new rights                                                                                                                                                                                                                                                                            | 56 square metres of<br>foreshore landward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown)   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)                                                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                            |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                            | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire       | o carry out protective works (Regulatio                                                                                | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                           | Qualifying persons under Regulatio                                                                                                        | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                 | (Applications: Prescribed Forms and                                                                                    | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                       | Owners or Reputed Owners                                                                                                                  | Lessees or Tenants                                                                                                     | Occupiers                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 53 cont'd               |                              |                                                                                                                                                           | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown |                                                                                                                        |                                                                                                                        | Unknown                                                                                                                                               |
| 54                      | Temporary Possession         | 1968 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                             | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)             |
| 55                      | Acquisition of new rights    | 1175 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                             | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)             |
| 56                      | Acquisition of new rights    | 167 square metres of<br>foreshore landward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown)      | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)                          | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Unknown                                                                                                                                               |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                            | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                           | Qualifying persons under Regulatio                                                                                                        | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                       | Owners or Reputed Owners                                                                                                                  | Lessees or Tenants                                                                                               | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 56 cont'd               |                              |                                                                                                                                                           | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown |                                                                                                                  |                                                                                            |                                                                                                                                                       |
| 57                      | Temporary Possession         | 2173 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)             |
| 58                      | Acquisition of new rights    | 1839 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)             |
| 59                      | Temporary Possession         | 174 square metres of<br>foreshore land seaward<br>of the Mean High Water<br>lying north of Prestatyn<br>(excluding all interests of<br>the Crown)         | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)             |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                      | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                                                | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                                                            | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                        |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 60                      | Temporary Possession         | 26 square metres of<br>foreshore landward of<br>the Mean High Water<br>lying north of Prestatyn<br>(excluding all interests of<br>the Crown)        | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder)                                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Unknown                                                                                                                                               |
| 61                      | Acquisition of new rights    | 15 square metres of<br>foreshore landward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder)                                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed leaseholder) | Unknown                                                                                                                                               |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                        | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                       | Qualifying persons under Regulatio                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                   | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                               | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |
| 62                      | Temporary Possession         | 795 square metres of<br>footway and foreshore<br>landward of the Mean<br>High Water (north of<br>Victoria Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed<br>of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of easement dated 7th March 2005) |
| 63                      | Acquisition of new rights    | 155 square metres of<br>slipway at Ffrith Beach<br>landward of Mean High<br>Water (north of Victoria<br>Road, A548)   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed<br>of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of easement dated 7th March 2005) |

|                         | Land which is j              | proposed to be subject to: (i)                                                                                               | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                             |  |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                              |                                                                                                                              | Qualifying persons under Regulatio                                                                                                          | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                          | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |  |
| 64                      | Temporary Possession         | 94 square metres of<br>hardstanding and<br>foreshore landward of<br>the Mean High Water<br>(north of Victoria Road,<br>A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN            | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed<br>of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of easement dated 7th March 2005) |  |
| 65                      | Temporary Possession         | 25 square metres of<br>hardstanding and<br>foreshore landward of<br>the Mean High Water<br>(north of Victoria Road,<br>A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown                 | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown | Unknown                                                                                                                                                                                                                                                                                                                                |  |
| 66                      | Acquisition of new rights    | 17 square metres of<br>footway (north of<br>Victoria Road, A548)                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown                 | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown | Unknown                                                                                                                                                                                                                                                                                                                                |  |

|                         | Land which is p              | proposed to be subject to: (i)                                          | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                      | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|-------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                         | Qualifying persons under Regulatio                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                   | Category 2                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                     | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                               | Occupiers                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |
| 67                      | Temporary Possession         | 53 square metres of<br>grassed area (north of<br>Victoria Road, A548)   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown                 | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown | Unknown                                                                                                                                                                                                                                                                                                                                |
| 68                      | Temporary Possession         | 3178 square metres of<br>grassed area (north of<br>Victoria Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN            | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed<br>of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of easement dated 7th March 2005) |
| 69                      | Temporary Possession         | 3442 square metres of<br>grassed area (north of<br>Victoria Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                  | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN            | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)                                                                                                                                                                                    |

|                         | Land which is p                   | proposed to be subject to: (i)    | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|-----------------------------------|-----------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                   |                                   | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                   |                                         | Category 2                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use      | Description of land               | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants                                                                                                | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                           |
| 69 cont'd               |                                   |                                   | Unknown<br>(in respect of mines and minerals)                                                                                                                |                                                                                                                   |                                         | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |
| 69a                     | Plot No Longer Included In<br>DCO | Plot No Longer Included<br>In DCO | Plot No Longer Included In DCO                                                                                                                               | Plot No Longer Included In DCO                                                                                    | Plot No Longer Included In DCO          | Plot No Longer Included In DCO                                                                                                                                                                                                                                                                                                                  |

|                         | Land which is p              | proposed to be subject to: (i)                                        | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|-----------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                       | Qualifying persons under Regulatio                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                   | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                                | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                           |
| 70                      | Temporary Possession         | 45 square metres of<br>footway (north of<br>Victoria Road, A548)      | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |
| 71                      | Temporary Possession         | 55 square metres of<br>grassed area (north of<br>Victoria Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                               | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                              | Qualifying persons under Regulatio                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                          | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                                | Occupiers                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                  |
| 72                      | Temporary Possession         | 891 square metres of<br>grassed area and track<br>(north of Victoria Road,<br>A548)                          | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL                                          | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |
| 73                      | Temporary Possession         | 57 square metres of<br>grassed area at Ffrith<br>Beach Festival Gardens<br>(north of Victoria Road,<br>A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL                                          | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)                                                                                                                                                                                                                                                                    |

|                         | Land which is p              | proposed to be subject to: (i)                                                                       | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                      |                                                                                                                                             | Category 1                                                                                                        |                                                                          |                                                                                                                                                                                                                                                                                                                                                 |
|                         |                              |                                                                                                      | Qualifying persons under Regulatio                                                                                                          | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                      | Category 2                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                  | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                                | Occupiers                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                           |
| 73 cont'd               |                              |                                                                                                      |                                                                                                                                             |                                                                                                                   |                                                                          | Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951)                                                                                                                                                                                                                                        |
| 74                      | Temporary Possession         | 66 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north<br>of Victoria Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL                                          | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |
| 75                      | Temporary Possession         | 2624 square metres of<br>grassed area (north of<br>Victoria Road, A548)                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL                                          | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                  |

|                         | Land which is p              | proposed to be subject to: (i)                                                         | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                        | Qualifying persons under Regulatio                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                       | Category 2                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                    | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                                | Occupiers                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                   |
| 75 cont'd               |                              |                                                                                        |                                                                                                                                             |                                                                                                                   |                                                                                                           | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |
| 76                      | Acquisition of new rights    | 469 square metres of<br>private road and<br>hardstanding (north of<br>Ferguson Avenue) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX         | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                                                                                 |

|                         | Land which is p              | proposed to be subject to: (i)                                                 | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|--------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                |                                                                                                                                                            | Category 1                                                                                                       |                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|                         |                              |                                                                                | Qualifying persons under Regulatio                                                                                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                            | Owners or Reputed Owners                                                                                                                                   | Lessees or Tenants                                                                                               | Occupiers                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                    |
| 76 cont'd               |                              |                                                                                |                                                                                                                                                            |                                                                                                                  |                                                                                                           | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |
| 77                      | Acquisition of new rights    | 22 square metres of<br>private road and verge<br>(north of Ferguson<br>Avenue) | NS Shelfco 1 Limited<br>Centrix House<br>26 Crow Lane East<br>NEWTON-LE-WILLOWS<br>Merseyside<br>WA12 9UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | NS Shelfco 1 Limited<br>Centrix House<br>26 Crow Lane East<br>NEWTON-LE-WILLOWS<br>Merseyside<br>WA12 9UY | Scottish & Newcastle Limited<br>3-4 Broadway Park<br>South Gyle Broadway<br>EDINBURGH<br>EH12 9JZ<br>(as mortgagee for NS Shelfco 1<br>Limited)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                      |

|                         | Land which is p              | proposed to be subject to: (i)                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | to carry out protective works (Regulatio                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                   | Qualifying persons under Regulatio                                                                                                                         | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                               | Owners or Reputed Owners                                                                                                                                   | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                 |
| 78                      | Acquisition of new rights    | 10 square metres of<br>public road and verge<br>(Ferguson Avenue) | NS Shelfco 1 Limited<br>Centrix House<br>26 Crow Lane East<br>NEWTON-LE-WILLOWS<br>Merseyside<br>WA12 9UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Scottish & Newcastle Limited<br>3-4 Broadway Park<br>South Gyle Broadway<br>EDINBURGH<br>EH12 9JZ<br>(as mortgagee for NS Shelfco 1<br>Limited)<br>Unknown<br>(in respect of restrictive covenants) |
| 79                      | Temporary Possession         | 52 square metres of<br>public road (Ferguson<br>Avenue)           | NS Shelfco 1 Limited<br>Centrix House<br>26 Crow Lane East<br>NEWTON-LE-WILLOWS<br>Merseyside<br>WA12 9UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                                     |

|                         | Land which is p              | proposed to be subject to: (i)                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|------------------------------|-------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                   |                                                                                                                                                            |                                                                                                                   |                                                                                                           |                                                                                                                                                                                                                                                                                                                                                      |
|                         |                              |                                                                   | Qualifying persons under Regulatio                                                                                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                       | Category 2                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                               | Owners or Reputed Owners                                                                                                                                   | Lessees or Tenants                                                                                                | Occupiers                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                |
| 79 cont'd               |                              |                                                                   |                                                                                                                                                            |                                                                                                                   |                                                                                                           | Scottish & Newcastle Limited<br>3-4 Broadway Park<br>South Gyle Broadway<br>EDINBURGH<br>EH12 9JZ<br>(as mortgagee for NS Shelfco 1<br>Limited)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants) |
| 80                      | Temporary Possession         | 24 square metres of<br>private road (north of<br>Ferguson Avenue) | NS Shelfco 1 Limited<br>Centrix House<br>26 Crow Lane East<br>NEWTON-LE-WILLOWS<br>Merseyside<br>WA12 9UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                              | NS Shelfco 1 Limited<br>Centrix House<br>26 Crow Lane East<br>NEWTON-LE-WILLOWS<br>Merseyside<br>WA12 9UY | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                                    |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                           | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                      | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                          |                                                                                                                                             | Category 1                                                                                                       |                                                                                            |                                                                                                                                                                                                                                                                                                                                                      |
|                         |                              |                                                                                                                          | Qualifying persons under Regulatio                                                                                                          | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                      | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                               | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                |
| 80 conť d               |                              |                                                                                                                          |                                                                                                                                             |                                                                                                                  |                                                                                            | Scottish & Newcastle Limited<br>3-4 Broadway Park<br>South Gyle Broadway<br>EDINBURGH<br>EH12 9JZ<br>(as mortgagee for NS Shelfco 1<br>Limited)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants) |
| 81                      | Temporary Possession         | 1729 square metres of<br>grassed area and<br>footway (north of<br>Victoria Road, A548 and<br>west of Ferguson<br>Avenue) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights t<br>Denbighshire | to carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulation                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                       | (Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                           | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| 81 cont'd               |                              |                               |                                                         |                                                                                                              |                                          | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                      | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                       | Qualifying persons under Regulatio                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                   | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                               | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 82                      | Acquisition of new rights    | 225 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north<br>of Victoria Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                         | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                        | Qualifying persons under Regulatio                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                    | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                                | Occupiers                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 83                      | Acquisition of new rights    | 1746 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north<br>of Victoria Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL                                          | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                       | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                      | Qualifying persons under Regulatio                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                      | Category 2                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                  | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                                | Occupiers                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                           |
| 84                      | Temporary Possession         | 27 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north<br>of Victoria Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL                                          | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |
| 85                      | Temporary Possession         | 19 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north<br>of Victoria Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL                                          | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                               | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                      | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                 | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                              |                                                                                                                                             | Category 1                                                                                                     |                                                                          |                                                                                                                                                                                                                                                                                                                                                 |
|                         |                                                          |                                                                                                              | Qualifying persons under Regulatio                                                                                                          | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and                                      | Category 2                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                          | Owners or Reputed Owners                                                                                                                    | Lessees or Tenants                                                                                             | Occupiers                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                           |
| 86                      | Temporary Possession                                     | 16 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north<br>of Victoria Road, A548)         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL                                       | Lakeside (Prestatyn) Limited<br>44 Rowney Croft<br>BIRMINGHAM<br>B28 OPL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in<br>Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as<br>contained in a Transfer dated 15th<br>August 1951) |
| 87                      | Acquisition of new rights and imposition of restrictions | 30836 square metres of<br>grassed area (Rhyl Golf<br>Club, north of Rhyl Coast<br>Road, A548)                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals) | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE                                         | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE   | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)                                                                                                                                                                                                      |
| 88                      | Temporary Possession                                     | 14607 square metres of<br>grassed area and<br>footway (Rhyl Golf Club,<br>north of Rhyl Coast<br>Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                  | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE                                         | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE   | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)                                                                                                                                                                                                      |

|                         | Land which is p                                             | proposed to be subject to: (i)                                 | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                 | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                |                                                                                                                                                                                                                                                                                                                                                                                                 | Category 1                                                                                                       |                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                                                             |                                                                | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                              | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                            | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                        | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                         |
| 88 cont'd               |                                                             |                                                                | Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                  |                                                                                                                                                                                         | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                                                                                                                                                                                                                                                                                               |
| 89                      | Acquisition of new rights and<br>imposition of restrictions | 361 square metres of<br>public road (Rhyl Coast<br>Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>(in respect of subsoil beneath half<br>width of public highway)<br>Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>(in respect of subsoil beneath half<br>width of public highway) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown |

|                         | Land which is p                                          | proposed to be subject to: (i)                                         | BOOK OF RE powers of compulsory acquisition, (ii)                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                  |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                        |                                                                                                                             | Category 1                                                                                                        |                                                                                                       |                                                                                                                                                                                                                                                                                             |
|                         |                                                          |                                                                        | Qualifying persons under Regulatio                                                                                          | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                   | Category 2                                                                                                                                                                                                                                                                                  |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                    | Owners or Reputed Owners                                                                                                    | Lessees or Tenants                                                                                                | Occupiers                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                       |
| 89 cont'd               |                                                          |                                                                        |                                                                                                                             |                                                                                                                   |                                                                                                       | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                             |
| 90                      | Acquisition of new rights and imposition of restrictions | 5 square metres of<br>hardstanding (north of<br>Rhyl Coast Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown | Unknown                                                                                                                                                                                                                                                                                     |
| 91                      | Acquisition of new rights and imposition of restrictions | 1363 square metres of<br>public road (Rhyl Coast<br>Road, A548)        | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                  | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN            | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                        | BOOK OF RE powers of compulsory acquisition, (ii)                                          | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                       | Qualifying persons under Regulation                                                        | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                             | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                   | Owners or Reputed Owners                                                                   | Lessees or Tenants                                                                                                  | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                               |
| 91 cont'd               |                              |                                                                                       |                                                                                            |                                                                                                                     |                                                                                            | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |
| 92                      | Temporary Possession         | 232 square metres of<br>footway of public road<br>(south of Rhyl Coast<br>Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | NONE                                                                                                                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                                                                                                                                                                           |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF RE<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|--------------------------------|--------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                | Qualifying persons under Regulatio                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                               | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 92 cont'd               |                              |                                |                                                        |                                                                                                                  |                                         | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                                  |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                       | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                           | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                         | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                   | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                              | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                       |
| 92 cont'd               |                              |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                 |                                                                                                                      | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                             |
| 93                      | Temporary Possession         | 116 square metres of<br>footway of public road<br>(south of Rhyl Coast<br>Road, A548) | Anthony William Higgs<br>22 Maxstoke Close<br>REDDITCH<br>Worcestershire<br>B98 0EJ<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Diane Elizabeth Inglis<br>24 Pennine Road<br>Woodley<br>STOCKPORT<br>Cheshire<br>SK6 1JS<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                      | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                              | n 7(1)(a))                                                                                                                                                    |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Category 1                                                                                                       |                                                                                                      |                                                                                                                                                               |
|                         |                              |                                                                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                  | Category 2                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Lessees or Tenants                                                                                               | Occupiers                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009         |
| 93 cont'd               |                              |                                                                                                                     | Joyce Anne Signol<br>Terfyn Pella Camp<br>Rhyl Coast Road<br>RHYL<br>Denbighshire<br>LL18 3YH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Stewart Nicholas Signol<br>244 Colne Road<br>Sough<br>BARNOLDSWICK<br>Lancashire<br>BB18 6TD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                  |                                                                                                      | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)               |
| 94                      | Temporary Possession         | 11 square metres of<br>electricity substation<br>(west of Sherwood<br>Avenue and south of<br>Rhyl Coast Road, A548) | Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE                                                                                                                                                                                                                                                                                                                                                                                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET             | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for<br>maintenance) |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|--------------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                          | Category 1                                                                                                    |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|                         |                              |                                | Qualifying persons under Regulatior                      | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 94 cont'd               |                              |                                |                                                          |                                                                                                               |                                         | lain Alexander Douglas Blair<br>Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as<br>contained in a Conveyance dated<br>30th November 1906)<br>Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(in respect of restrictive covenants as<br>contained in a Conveyance dated<br>16th October 1933)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p              | proposed to be subject to: (i                                                         | BOOK OF RE<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                       |                                                                                                                                                                                                                                                                                                                              | Category 1                                                                                                      |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|                         |                              |                                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                   | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                     | Lessees or Tenants                                                                                              | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 95                      | Temporary Possession         | 808 square metres of<br>footway of public road<br>(south of Rhyl Coast<br>Road, A548) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |

|                         | Land which is j              | proposed to be subject to: (i                                                                                            | BOOK OF RE<br>) powers of compulsory acquisition, (ii)                                                                                                                  | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                          |                                                                                                                                                                         | Category 1                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                         |                              |                                                                                                                          | Qualifying persons under Regulation                                                                                                                                     | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                      | Owners or Reputed Owners                                                                                                                                                | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 96                      | Temporary Possession         | 18635 square metres of<br>buildings, caravans,<br>hardstanding and<br>grassed area (Robin<br>Hood Holiday Camp,<br>Rhyl) | Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                              | Joanne Miller<br>Caravan 20 Silver Birch Way<br>Lyons Robin Hood Holiday Camp<br>Rhyl Coast Road<br>RHYL<br>Denbighshire<br>LL18 3UU<br>(as occupier of Caravan 20 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE<br>Robert Miller<br>Caravan 20 Silver Birch Way<br>Lyons Robin Hood Holiday Camp<br>Rhyl Coast Road<br>RHYL<br>Denbighshire<br>LL18 3UU<br>(as occupier of Caravan 20 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 96 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for<br>maintenance)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Iain Alexander Douglas Blair<br>Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as<br>contained in a Conveyance dated<br>30th November 1906)<br>Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(in respect of restrictive covenants as<br>contained in a Conveyance dated<br>16th October 1933) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFE powers of compulsory acquisition, (ii) rig | DEVELOPMENT CONSENT ORDER<br>RENCE - PART 1<br>ht to use the land, and/or (iii) righ<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                         | n 7(1)(a))                                                                                                                                                          |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                         | Category 1                                                                                           |                                                                                                                                                                                     |                                                                                                                                                                     |
|                         |                              |                               | Qualifying persons under Regulation                     | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                 | ng (Applications: Prescribed Forms and                                                                                                                                              | Category 2                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                   | Occupiers                                                                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009               |
| 96 cont'd               |                              |                               |                                                         |                                                                                                      | Unknown<br>(as occupier of Caravan 78 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                             | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                     |
|                         |                              |                               |                                                         |                                                                                                      | Unknown<br>(as occupier of Caravan 97 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                             | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights stated in Lease<br>dated 14 May 1959) |
|                         |                              |                               |                                                         |                                                                                                      | Unknown<br>(as occupier of Caravan 73 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                             | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                       |
|                         |                              |                               |                                                         |                                                                                                      | Unknown<br>(as occupier of Caravan 98 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 110 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) | Unknown<br>(in respect of right of way as<br>contained in a Conveyance dated<br>10th December 1935)                                                                 |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                          |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                          |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 96 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 86 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 92 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 90 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 77 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 95 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 76 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 104 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 82 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 74 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 105 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 72 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 72 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 70 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 70 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 68 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 91 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 88 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br> |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | (as occupier of Caravan 84 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                          |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                          |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 96 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 85 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 111 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 63 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 71 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 99 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 103 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 83 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 80 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                   | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                               |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 89 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 79 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 87 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 81 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 93 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 42<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 50 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 48 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                         | Category 1                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                         | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                         |                                                                                                            | Unknown<br>(as occupier of Caravan 59 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 75 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 60 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 67 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 28<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 48<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 48<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 76 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 55 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Plannii<br>Procedures) Regulations 2009                                    | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                         | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 39<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 70 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 68 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 47 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 65 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 65 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 37<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 56 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 69 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>f Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                       | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                           | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                              | Unknown<br>(as occupier of Caravan 45<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 38<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 61 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 83 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 73 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 73 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 52 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 66 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 74 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 47<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 36<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 45 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 45 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 40<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 33<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 33<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 41 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 78 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 49<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) righ<br>Denbighshire | ts to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | <ul> <li>Unknown <ul> <li>(as occupier of Caravan 53 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 54 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 57 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 57 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 43</li> <li>Hawthorn Crescent, Lyons Robin</li> <li>Hood Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 62 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 58 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 72 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 72 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> </ul> |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) righ<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                       | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                          | <ul> <li>Unknown <ul> <li>(as occupier of Caravan 51 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 44</li> <li>Hawthorn Crescent, Lyons Robin</li> <li>Hood Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 63 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 63 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 79 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 64 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 34</li> <li>Hawthorn Crescent, Lyons Robin</li> <li>Hood Holiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 42 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Hoiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 42 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Hoiday Camp)</li> </ul> </li> <li>Unknown <ul> <li>(as occupier of Caravan 42 Silver</li> <li>Birch Way, Lyons Robin Hood</li> <li>Hoiday Camp)</li> </ul> </li> </ul> |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 96 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 73A Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 46<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 50<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 71 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 32<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 41<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 41<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 49 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 52<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                   | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                               |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                        | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 46 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 48 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 54 The<br>Poplars, Lyons Robin Hood Holiday     |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Camp)<br>Unknown<br>(as occupier of Caravan 47 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 50 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                                   |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 49 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 41 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 52 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                               |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                               |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 96 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 51 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 46 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 43 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 53 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 44 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 45 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 27 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 13<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatior                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | a 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 21 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 23 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 43 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 9 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 8 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 8 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 20<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 27<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 7 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                         | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 34 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 6 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 25 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 38 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 14<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 44 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 44 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 31<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 30 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                            |  |  |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                               |                                                                                                                                                       |  |  |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 21<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |  |  |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 19 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 31 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 37 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 36 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 29 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 15 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 16<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | n 7(1)(a))                                                                                                                                            |  |  |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                       |  |  |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Category 2                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 17<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 30<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 39 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 26 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 26 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 26 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 15<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 23<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 11 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br> |                                                                                                                                                       |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 96 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 40 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 22<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 29<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 12 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 12<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 22<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 22 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 25<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 19<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 96 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 18 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 13 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 10<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 28 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 10 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 35 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 35 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 17 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 24 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 18<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 24<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 14 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 9 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 6 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 6 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 5 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 4 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 4 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 26<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 32 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 11<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 33 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 61 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 68 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 68 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 66 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 59 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 65 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                   | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 56 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 63 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 55 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 55 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 58 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 58 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 57 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 62 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 62 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br> |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | (as occupier of Caravan 60 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                          | Category 1                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |
|                         |                              |                                | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                         | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                                |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 87 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 91 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 85 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 82 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 84 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 84 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 90 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 89 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 89 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 93 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | a 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 88 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 92 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 80 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 81 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 77 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 77 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 86 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 46 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 42 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 32 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 37 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 29 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 34 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 35 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 33 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 38 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 40 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 36 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatior                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 30 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 31 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 39 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 12 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 20 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 8 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 13 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 33 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulatior                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                         | Category 1                                                                                                  |                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                    | (Applications: Prescribed Forms and                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                          | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                         |                                                                                                             | Unknown<br>(as occupier of Caravan 3 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                             | Unknown<br>(as occupier of Caravan 10 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                             | Unknown<br>(as occupier of Caravan 1 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                             | Unknown<br>(as occupier of Caravan 16 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                             | Unknown<br>(as occupier of Caravan 1 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                             | Unknown<br>(as occupier of Silver Birch Way,<br>Lyons Robin Hood Holiday Camp)               |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                             | Unknown<br>(as occupier of Caravan 2 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                             | Unknown<br>(as occupier of Caravan 5 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>f Denbighshire | s to carry out protective works (Regulation                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                         | Category 1                                                                                                   |                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                       | ng (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                           | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 2 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 1 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 6 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 21 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                              | Unknown<br>(as occupier of Holly Grove, Lyons<br>Robin Hood Holiday Camp)                    |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 26 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 14 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 25 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>f Denbighshire | to carry out protective works (Regulation                                                   | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|---------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                         | Category 1                                                                                                    |                                                                                             |                                                                                                                                                       |
|                         |                              |                                | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planniı<br>Procedures) Regulations 2009                                       | g (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                | Lessees or Tenants                                                                                            | Occupiers                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                                |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 4 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 27 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)     |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                               | Unknown<br>(as occupier of Hawthorn Crescent,<br>Lyons Robin Hood Holiday Camp)             |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 2 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 9 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 7 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 8 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 3 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 18 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 3 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 28 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 28 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 11 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 11 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 15 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 24 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 24 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br> |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                      | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                  |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                           | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 19 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 17 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown                                    |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | (as occupier of Caravan 4 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 7 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                      |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 5 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 71 The<br>Dealar, Lyons Robin Hood Holiday                                                          |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 80 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 72 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                         |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                         |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 96 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 76 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 75 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 77 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 69 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 73 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 67 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 78 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 79 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                       | Category 1                                                                                                |                                                                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                   | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                              | Lessees or Tenants                                                                                        | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 74 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 70 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 23 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 48 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 40 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 47 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 54 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 53 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Plannii<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 96 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 34 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 44 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 59 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | <ul> <li>(as occupier of Caravan 29 Holly<br/>Grove, Lyons Robin Hood Holiday<br/>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 49 Holly<br/>Grove, Lyons Robin Hood Holiday<br/>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 65 Holly<br/>Grove, Lyons Robin Hood Holiday<br/>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 36 Holly<br/>Grove, Lyons Robin Hood Holiday<br/>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 36 Holly<br/>Grove, Lyons Robin Hood Holiday<br/>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 67 Holly<br/>Grove, Lyons Robin Hood Holiday<br/>Camp)</li> </ul> |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                    |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                    |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                             | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 96 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 41 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | (as occupier of Caravan 38 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | (as occupier of Caravan 30 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 31 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 42 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 37 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 57 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 32 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 75 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 66 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 43 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 58 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 55 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 55 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 64 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 46 Holly |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 60 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) righ<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|----------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                       | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                          | <ul> <li>Unknown         <ul> <li>(as occupier of Caravan 62 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> </ul> </li> <li>Unknown         <ul> <li>(as occupier of Caravan 52 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> </ul> </li> <li>Unknown         <ul> <li>(as occupier of Caravan 52 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> </ul> </li> <li>Unknown         <ul> <li>(as occupier of Caravan 45 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> </ul> </li> <li>Unknown         <ul> <li>(as occupier of Caravan 50 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown             <ul> <li>(as occupier of Caravan 50 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown             <ul> <li>(as occupier of Caravan 56 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown             <ul> <li>(as occupier of Caravan 35 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown             <ul> <li>(as occupier of Caravan 51 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown             <ul> <li>(as occupier of Caravan 51 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 51 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 61 Holly</li></ul></li></ul></li></ul></li></ul></li></ul></li></ul></li></ul> |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) righ <sup>.</sup><br>Denbighshire | s to carry out protective works (Regulation                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                            |                                                                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                                | ng (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                                    | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 81 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 83 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 88 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 87 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 82 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 84 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 85 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 86 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                       | Category 1                                                                                                |                                                                                          |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                   | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                              | Lessees or Tenants                                                                                        | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 89 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 90 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 100 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 112 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 119 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 108 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 106 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 94 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                          |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 96 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 109 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 118 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 120 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 102 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 107 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 117 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 114 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 116 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                         |                                                                                          |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    | Category 1                                                              |                                                                                          |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                    | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                      | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 96 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 101 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 121 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 113 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 115 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 122 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 123 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                           | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                  | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                          | Qualifying persons under Regulatio                                                                                                                                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                      | Owners or Reputed Owners                                                                                                                                                | Lessees or Tenants                                                                                             | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                  |
| 97                      | Acquisition of new rights and<br>imposition of restrictions | 22566 square metres of<br>buildings, caravans,<br>hardstanding and<br>grassed area (Robin<br>Hood Holiday Camp,<br>Rhyl) | Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                           | Joanne Miller<br>Caravan 20 Silver Birch Way<br>Lyons Robin Hood Holiday Camp<br>Rhyl Coast Road<br>RHYL<br>Denbighshire<br>LL18 3UU<br>(as occupier of Caravan 20 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE<br>Robert Miller<br>Caravan 20 Silver Birch Way<br>Lyons Robin Hood Holiday Camp | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for<br>maintenance)<br>lain Alexander Douglas Blair<br>Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as<br>contained in a Conveyance dated<br>30th November 1906)<br>Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON |
|                         |                                                             |                                                                                                                          |                                                                                                                                                                         |                                                                                                                | Rhyl Coast Road<br>RHYL<br>Denbighshire<br>LL18 3UU<br>(as occupier of Caravan 20 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)                                                                                                                                                                                                                                                                                             | NW1 2DN<br>(in respect of restrictive covenants as<br>contained in a Conveyance dated<br>16th October 1933)                                                                                                                                                                                                                                                                                                                            |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>f Denbighshire | s to carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                         | Category 1                                                                                                    |                                                                                          |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                       | g (Applications: Prescribed Forms and                                                    | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                            | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 96 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)       |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 78 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)         |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 97 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  | Unknown<br>(in respect of right of way as<br>contained in a Conveyance dated<br>10th December 1935)                                                   |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 73 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  | Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                 |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 98 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 110 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 2                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 86 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 92 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 90 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 77 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 95 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 95 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 76 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 104 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 104 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown |                                                                                                                                                       |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | (as occupier of Caravan 82 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFI<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) righ<br>Denbighshire | ts to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                       | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                            |                                                                                                          | <ul> <li>Unknown         <ul> <li>(as occupier of Caravan 74 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> </ul> </li> <li>Unknown         <ul> <li>(as occupier of Caravan 105 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> </ul> </li> <li>Unknown         <ul> <li>(as occupier of Caravan 105 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> </ul> </li> <li>Unknown         <ul> <li>(as occupier of Caravan 72 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> </ul> </li> <li>Unknown         <ul> <li>(as occupier of Caravan 70 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> </ul> </li> <li>Unknown         <ul> <li>(as occupier of Caravan 70 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown             <ul> <li>(as occupier of Caravan 68 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown             <ul> <li>(as occupier of Caravan 91 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown             <ul> <li>(as occupier of Caravan 88 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown             <ul> <li>(as occupier of Caravan 88 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown             <ul> <li>(as occupier of Caravan 88 Holly</li> <li>Grove, Lyons Robin Hood Holiday</li> <li>Camp)</li> <li>Unknown</li> <li>(as occupier of Carav</li></ul></li></ul></li></ul></li></ul></li></ul></li></ul></li></ul> |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                       |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                       |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Category 2                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 85 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 111 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 63 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 71 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 99 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 99 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 103 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 83 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 80 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 89 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 79 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 87 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 81 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 93 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 93 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 42<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 50 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 48 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFI<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 59 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 75 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 60 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 67 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 28<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 48<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 48<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 76 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 55 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                   | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 39<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 70 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 68 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 47 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 65 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 37<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 56 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 56 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 69 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                          | Category 1                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |
|                         |                              |                                | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                         | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                                |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 45<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 38<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 61 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 83 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 73 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 73 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 52 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 66 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 74 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                         | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 47<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 36<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 45 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 40<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 33<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 33<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 41 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 78 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 49<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                       |

|                              |                     |                                     | Category 1                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|------------------------------|---------------------|-------------------------------------|-----------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                              |                     | Qualifying persons under Regulation | 7(1)(a) of the Infrastructure Plannii<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| ixtent of acquisition or use | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                    | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|                              |                     |                                     |                                                                       | Unknown<br>(as occupier of Caravan 53 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 54 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 57 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 43<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 62 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 58 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 58 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 72 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 51 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                              |                     |                                     |                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | (as occupier of Caravan 53 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)       Unknown<br>(as occupier of Caravan 54 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)         Unknown<br>(as occupier of Caravan 54 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)       Unknown<br>(as occupier of Caravan 53 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)         Unknown<br>(as occupier of Caravan 43<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)       Unknown<br>(as occupier of Caravan 43<br>Hawthorn Crescent, Lyons Robin<br>Hood<br>Holiday Camp)         Unknown<br>(as occupier of Caravan 58 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)       Unknown<br>(as occupier of Caravan 58 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)         Unknown<br>(as occupier of Caravan 58 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)       Unknown<br>(as occupier of Caravan 72 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFI<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 51 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 44<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 63 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 79 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 64 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 64 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 34<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 42 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 35<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFI<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 73A Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 46<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 50<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 71 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 32<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 41<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 41<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 49 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 52<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 46 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 48 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 54 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 54 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 54 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 47 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 50 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 49 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 49 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br> |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Camp)<br>Unknown<br>(as occupier of Caravan 52 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulation                                                          | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                 |                                                                                                    |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                    | g (Applications: Prescribed Forms and                                                              | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                         | Occupiers                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 51 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | (as occupier of Caravan 46 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown            |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | (as occupier of Caravan 43 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                       |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 53 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 44 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 45 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 27 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 13<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)      |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 21 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 23 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 43 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 9 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 8 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 8 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 20<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 27<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 7 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 34 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 6 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 25 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 38 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 14<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 44 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 44 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 31<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 30 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 21<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 19 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 31 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 37 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 36 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 36 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 29 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 15 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 15 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 16<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 17<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 30<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 39 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 26 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 15<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 23<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 23<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 11 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 16 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 40 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 22<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 29<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 12 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 12<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 22<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 22 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 25<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 19<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 18 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 13 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 10<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 28 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 10 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 35 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 35 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 17 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 24 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>f Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                         | Category 1                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                        | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                            | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 18<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 24<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 14 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 9 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 6 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 6 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 5 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 5 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 4 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 26<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 32 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 11<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 33 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 61 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 68 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 68 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 66 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 59 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 65 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 56 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 63 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 55 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 58 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 57 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 57 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 62 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 64 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 60 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 87 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 91 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 85 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 82 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 84 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 84 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 90 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 89 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 93 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 88 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 92 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 80 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 81 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 77 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 77 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 86 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 46 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 42 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 32 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFE<br>powers of compulsory acquisition, (ii) rig | DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|---------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                              |                                                                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                      | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                            |                                                                                                         | Unknown<br>(as occupier of Caravan 37 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                         | Unknown<br>(as occupier of Caravan 29 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                         | Unknown<br>(as occupier of Caravan 34 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                         | Unknown<br>(as occupier of Caravan 35 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                         | Unknown<br>(as occupier of Caravan 33 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                         | Unknown<br>(as occupier of Caravan 38 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                         | Unknown<br>(as occupier of Caravan 40 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                         | Unknown<br>(as occupier of Caravan 36 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatior                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 30 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 31 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 39 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 12 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 20 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 8 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 13 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 33 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatior                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                         | Category 1                                                                                                    |                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                       | g (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                            | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 3 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 10 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 1 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 16 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 1 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Silver Birch Way,<br>Lyons Robin Hood Holiday Camp)               |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 2 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                         |                                                                                                               | Unknown<br>(as occupier of Caravan 5 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

| Land which is p              | roposed to be subject to: (i) | BOOK OF REFE<br>powers of compulsory acquisition, (ii) rig | ERENCE - PART 1<br>ght to use the land, and/or (iii) righ                                                                                                   | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|------------------------------|-------------------------------|------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                              |                               | Qualifying persons under Regulation                        |                                                                                                                                                             | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Procedures) Regulations 2009<br>Lessees or Tenants                                                                                                          | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|                              |                               |                                                            |                                                                                                                                                             | Unknown<br>(as occupier of Caravan 2 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 1 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 6 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 6 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 21 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 21 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br> | Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                              |                               |                                                            | BOOK OF REFI<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) rig<br>County of<br>Qualifying persons under Regulation | County of Denbighshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planni Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                       | BOOK OF REFERENCE - PART I         Land which is proposed to be subject to: (I) powers of compulsoy acquisition, (II) right to use the land, and/or (iii) rights to carry out protective works (Regulation Caunty of Denbighshire         Category 1          Category 1 |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>f Denbighshire | s to carry out protective works (Regulation                                                 | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|---------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                         | Category 1                                                                                                   |                                                                                             |                                                                                                                                                       |
|                         |                              |                                | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planniı<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                      | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                | Lessees or Tenants                                                                                           | Occupiers                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                                |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 4 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 27 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)     |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                              | Unknown<br>(as occupier of Hawthorn Crescent,<br>Lyons Robin Hood Holiday Camp)             |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 2 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 9 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 7 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 8 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                                |                                                         |                                                                                                              | Unknown<br>(as occupier of Caravan 3 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |
|                         |                              |                                | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                                |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 18 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 3 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 28 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 28 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 11 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 11 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 15 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 24 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 24 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br> |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                        | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                             | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                          |                                                                                                                | Unknown<br>(as occupier of Caravan 19 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 17 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 4 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 4 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 7 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 7 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 5 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 71 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 80 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                | Unknown<br>(as occupier of Caravan 72 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 76 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 75 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 77 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 69 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 73 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 67 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 78 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 79 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                   | ng (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 74 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 70 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 23 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 48 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 40 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 47 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 54 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 53 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                            | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                        |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                 | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 34 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 44 Holly |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown                                                                                    |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | (as occupier of Caravan 59 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                           |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 29 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 49 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 65 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 36 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 67 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                        | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                    |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                             | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 41 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | (as occupier of Caravan 38 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                       |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 30 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 31 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 42 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 37 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 57 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 32 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 75 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 66 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 43 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 58 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 55 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 55 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 64 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 46 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 60 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 62 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 52 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 45 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 45 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 50 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 50 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 56 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 35 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 35 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br> |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) righ <sup>.</sup><br>Denbighshire | s to carry out protective works (Regulation                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                            |                                                                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                                | ng (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                                    | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 81 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 83 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 88 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 87 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 82 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 84 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 85 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                                       | Unknown<br>(as occupier of Caravan 86 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                       | Category 1                                                                                                |                                                                                          |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                   | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                              | Lessees or Tenants                                                                                        | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 89 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 90 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 100 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 112 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 119 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 108 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 106 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 94 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                         | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                     |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                              | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 97 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 109 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | (as occupier of Caravan 118 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown            |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | (as occupier of Caravan 120 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                       |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 102 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 107 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 117 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 114 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 116 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                         |                                                                                          |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    | Category 1                                                              |                                                                                          |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                    | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                      | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 97 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 101 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 121 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 113 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 115 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 122 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                         | Unknown<br>(as occupier of Caravan 123 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights 1<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                   | n 7(1)(a))                                                                                                                                                    |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                       | Qualifying persons under Regulatio                                                                                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                                                                       | Category 2                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                   | Owners or Reputed Owners                                                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009         |
| 98                      | Temporary Possession         | 7994 square metres of<br>private road, buildings,<br>caravans, hardstanding<br>and grassed area (Robin<br>Hood Holiday Camp,<br>Rhyl) | Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE | NONE                                                                                                             | Joanne Miller<br>Caravan 20 Silver Birch Way<br>Lyons Robin Hood Holiday Camp<br>Rhyl Coast Road<br>RHYL<br>Denbighshire<br>LL18 3UU<br>(as occupier of Caravan 20 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)              |
|                         |                              |                                                                                                                                       | Unknown<br>(in respect of mines and minerals)                                                                          |                                                                                                                  | Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE                                                                                                    | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for<br>maintenance) |
|                         |                              |                                                                                                                                       |                                                                                                                        |                                                                                                                  | Robert Miller<br>Caravan 20 Silver Birch Way<br>Lyons Robin Hood Holiday Camp<br>Rhyl Coast Road<br>RHYL<br>Denbighshire<br>LL18 3UU<br>(as occupier of Caravan 20 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulatio                                                | n 7(1)(a))                                                                                                                                                                                     |
|-------------------------|------------------------------|--------------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                          | Category 1                                                                                                 |                                                                                         |                                                                                                                                                                                                |
|                         |                              |                                | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                    | g (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                         | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                          |
| 98 cont'd               |                              |                                |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 96 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) | lain Alexander Douglas Blair<br>Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as<br>contained in a Conveyance dated<br>30th November 1906) |
|                         |                              |                                |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 78 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) | Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(in respect of restrictive covenants as<br>contained in a Conveyance dated<br>16th October 1933)             |
|                         |                              |                                |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 97 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                                |
|                         |                              |                                |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 73 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights stated in Lease<br>dated 14 May 1959)                            |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFI<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                                |                                                                                          |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                        | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 98 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)         |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 110 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) | Unknown<br>(in respect of right of way as<br>contained in a Conveyance dated<br>10th December 1935)                                                   |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 86 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 92 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 90 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 77 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)  |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) righ<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                               | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|----------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                               |                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                   | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                    | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                       | Occupiers                                                                                                                                                                                                                                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                          | Unknown<br>(as occupier of Caravan 95 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 76 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 104 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 82 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                          | Unknown<br>(as occupier of Caravan 74 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 105 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 72 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 70 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 68 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 91 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 88 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 84 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 85 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 85 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 111 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 63 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 63 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | (as occupier of Caravan 71 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 99 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 103 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 83 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 80 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 89 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 89 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 79 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 87 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 87 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | (as occupier of Caravan 81 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 93 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 42<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 50 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 48 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 59 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 59 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 75 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 60 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 60 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 67 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 28<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 48<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 76 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 55 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 39<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 70 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 70 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 68 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 47 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 65 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 37<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 56 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 69 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 45<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 38<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 38<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 61 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 83 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 73 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 52 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 66 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 74 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 47<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 36<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 45 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 45 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 40<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 33<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 41 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 78 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 78 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 49<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 53 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 53 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 54 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 54 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br> |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 62 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 58 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 72 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 51<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 51 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 51 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 44<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 63 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 79 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 64 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 34<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 42 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 35<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 73A Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 46<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 46<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 50<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 71 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 32<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 41<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 49 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 52<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 46 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 46 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 48 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 54 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 47 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 50 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 49 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 41 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 52 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 51 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 46 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 43 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 53 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                               |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                               |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                         | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 44 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 45 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 27 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 13<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 21 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 23 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 43 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 9 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planniı<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 8 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 20<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 27<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 7 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 34 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 6 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 6 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 25 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 38 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 14<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 44 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 31<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 31<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 30 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 30 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 21<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 19 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 19 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br> |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFI<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) righ<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                       | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                            |                                                                                                          | Unknown<br>(as occupier of Caravan 36 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 29 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 15 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 15 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 16<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 16<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 17<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 30<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 30<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br> |                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|---------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                         | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                       |
|                         |                              |                                | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                                |                                                         |                                                                                                           | Unknown<br>(as occupier of Caravan 15<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 23<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 11 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 16 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 40 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 40 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 22<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 29<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 12 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                         | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 12<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 22 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 25<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 25<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 19<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 19<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)Unknown<br>(as occupier of Caravan 18 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 13 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br>(as occupier of Caravan 13 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)Unknown<br> |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 10 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 35 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 17 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 24 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 18<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 24<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 24<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 14 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 9 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 6 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 5 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 4 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 26<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 32 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 32 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 11<br>Hawthorn Crescent, Lyons Robin<br>Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 33 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 61 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 68 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 66 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 59 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 65 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 56 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 56 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 63 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 55 The |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 58 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | to carry out protective works (Regulation                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                   | g (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 57 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 62 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 64 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 60 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 87 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 91 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 85 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 82 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |
|                         |                              |                                | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                                |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 84 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 90 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 89 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 93 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 88 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 88 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 92 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 80 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 80 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 81 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFE<br>powers of compulsory acquisition, (ii) rig | DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatior                                                                                                                                                                                                                                                                                                                                        | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|---------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                            |                                                                                                         | Unknown<br>(as occupier of Caravan 77 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 86 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 42 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 32 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                         | Unknown<br>(as occupier of Caravan 37 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 29 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 34 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 35 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)           |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulation                                               | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                 |                                                                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                      | g (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                         | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 33 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 38 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 40 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 36 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 30 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 31 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 39 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                            | Unknown<br>(as occupier of Caravan 12 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 20 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 8 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 13 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 33 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 3 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 10 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 1 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 16 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                 |                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                    | g (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                         | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 1 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)  |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Silver Birch Way,<br>Lyons Robin Hood Holiday Camp)               |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 2 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 5 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 2 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 1 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 6 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |
|                         |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Caravan 21 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                          | Category 1                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                    | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                         | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                          |                                                                                                            | Unknown<br>(as occupier of Holly Grove, Lyons<br>Robin Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 26 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 14 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 25 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 4 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 4 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 27 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Hawthorn Crescent,<br>Lyons Robin Hood Holiday Camp)<br>Unknown<br>(as occupier of Caravan 2 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                           | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                       | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                           |                                                                                                           | Unknown<br>(as occupier of Caravan 9 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 7 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 8 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 3 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 18 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 18 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 3 Silver<br>Birch Way, Lyons Robin Hood<br>Holiday Camp)<br>Unknown<br>(as occupier of Caravan 28 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 28 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFI<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | to carry out protective works (Regulation                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                                |                                                                                              |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Plannii<br>Procedures) Regulations 2009                                     | g (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                        | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 15 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 24 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 22 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 39 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 19 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 17 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)      |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 4 Hawthorn<br>Crescent, Lyons Robin Hood<br>Holiday Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 7 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)       |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 5 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 71 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 80 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 80 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 72 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 72 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 76 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 75 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)Unknown<br> |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Camp)<br>Unknown<br>(as occupier of Caravan 69 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFE<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                                |                                                                                         |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                        | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 73 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 67 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 78 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 79 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 74 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 70 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 23 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |
|                         |                              |                               |                                                            |                                                                                                           | Unknown<br>(as occupier of Caravan 48 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 40 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 47 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 54 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 53 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 34 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 34 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 44 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 59 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 29 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                            |  |  |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                       |  |  |
|                         |                              |                               | Qualifying persons under Regulation                      | 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                      | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | Category 2                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 98 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 49 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 65 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 36 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 36 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 67 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 67 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 41 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 38 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br>(as occupier of Caravan 38 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)Unknown<br> |                                                                                                                                                       |  |  |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 31 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | n 7(1)(a))                                                                                                                                            |  |  |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                              |                               |                                                          | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       |  |  |
|                         |                              |                               | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Category 2                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 98 cont'd               |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 42 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 37 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 57 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 32 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 75 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 75 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 66 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 43 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                       |  |  |
|                         |                              |                               |                                                          |                                                                                                           | Unknown<br>(as occupier of Caravan 58 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                       |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 55 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 64 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 46 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 60 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 62 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 62 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 52 Holly<br>Grove, Lyons Robin Hood Holiday |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Camp)<br>Unknown<br>(as occupier of Caravan 45 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 50 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|-------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                       | Category 1                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulatior                   | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                    | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                              | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 98 cont'd               |                              |                               |                                                       |                                                                                                           | Unknown<br>(as occupier of Caravan 56 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 35 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 51 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                       |
|                         |                              |                               |                                                       |                                                                                                           | <ul> <li>(as occupier of Caravan 61 Holly<br/>Grove, Lyons Robin Hood Holiday<br/>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 81 The<br/>Poplars, Lyons Robin Hood Holiday<br/>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 83 The<br/>Poplars, Lyons Robin Hood Holiday<br/>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 88 The<br/>Poplars, Lyons Robin Hood Holiday<br/>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 88 The<br/>Poplars, Lyons Robin Hood Holiday<br/>Camp)</li> <li>Unknown</li> <li>(as occupier of Caravan 87 The<br/>Poplars, Lyons Robin Hood Holiday<br/>Camp)</li> </ul> |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                         |                                                                                                                                                                           |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|-------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                              |                                                                                                                                                                           |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                                                                                                    | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                      | Occupiers                                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 82 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 84 The<br>Poplars, Lyons Robin Hood Holiday |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Camp)<br>Unknown<br>(as occupier of Caravan 85 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                          |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 86 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                                   |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 89 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                                   |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 90 The<br>Poplars, Lyons Robin Hood Holiday<br>Camp)                                                                                   |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 100 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                  |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                         | Unknown<br>(as occupier of Caravan 112 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                                                                                  |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                     |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                     |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                              | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 119 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | (as occupier of Caravan 108 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 106 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 94 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)             |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 109 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 118 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 120 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 102 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                                                                                     |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                                                                                     |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                                                              | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 98 cont'd               |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 107 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | (as occupier of Caravan 117 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)                       |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 114 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 116 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 101 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 121 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 113 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        | Unknown<br>(as occupier of Caravan 115 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)            |                                                                                                                                                       |  |  |  |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                          | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                     |
|-------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                         | Qualifying persons under Regulatio                                                                              | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                                  | Category 2                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                     | Owners or Reputed Owners                                                                                        | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                          |
| 98 cont'd               |                                                          |                                                                                                         |                                                                                                                 |                                                                                                                  | Unknown<br>(as occupier of Caravan 122 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp)<br>Unknown<br>(as occupier of Caravan 123 Holly<br>Grove, Lyons Robin Hood Holiday<br>Camp) |                                                                                                                                                                                                                                                                |
| 99                      | Acquisition of new rights and imposition of restrictions | 2590 square metres of<br>railway (Prestatyn and<br>Rhyl) and hedgerow                                   | Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(as reputed owner)<br>Unknown | NONE                                                                                                             | NONE                                                                                                                                                                                 | Unknown                                                                                                                                                                                                                                                        |
| 100                     | Acquisition of new rights and imposition of restrictions | 27103 square metres of<br>agricultural land (north<br>of The Cut and south of<br>Rhyl Coast Road, A548) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                       | NONE                                                                                                             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                                                                            | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                        |                                                                                           |                                                                          |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                        |                                                                                           | Category 1                                                               |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                        | Qualifying persons under Regulatio                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                    | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                       | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                     |  |  |  |
| 100 conť d              |                                                                                                                                                                                                                                                                                                      |                                                                                                        |                                                                                           |                                                                          |                                                                                           | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for<br>maintenance as contained in an<br>Agreement dated 14th February<br>1964)                                                                                                                                                                                                                                                                                          |  |  |  |
| 101                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 2759 square metres of<br>agricultural land (north<br>of The Cut and south of<br>Rhyl Coast Road, A548) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                     | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for<br>maintenance as contained in an<br>Agreement dated 14th February<br>1964) |  |  |  |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                        | BOOK OF RI<br>powers of compulsory acquisition, (ii)                                      | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) right<br>of Denbighshire | s to carry out protective works (Regulatio                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|----------------------------------------------------------|-------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                          |                                                                                                       |                                                                                           | Category 1                                                                                                       |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
|                         |                                                          |                                                                                                       | Qualifying persons under Regulation                                                       | on 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                         | ng (Applications: Prescribed Forms and                                                    | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                   | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                                               | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                     |  |
| 102                     | Acquisition of new rights                                | 15 square metres of<br>agricultural land (north<br>of The Cut and south of<br>Rhyl Coast Road, A548)  | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for<br>maintenance as contained in an<br>Agreement dated 14th February<br>1964) |  |
| 103                     | Acquisition of new rights and imposition of restrictions | 323 square metres of<br>agricultural land (north<br>of The Cut and south of<br>Rhyl Coast Road, A548) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted)                                                                                                                                                                                                                                                                                                                                                                               |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                 |                                                                                           |                                                                           |                                                                                           |                                                                                                                                                                                                                                                                                                                                                     |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                 |                                                                                           | Category 1                                                                |                                                                                           |                                                                                                                                                                                                                                                                                                                                                     |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                 | Qualifying persons under Regulatic                                                        | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                          |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                             | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                        | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                               |  |  |
| 103 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                 |                                                                                           |                                                                           |                                                                                           | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for<br>maintenance as contained in an<br>Agreement dated 14th February<br>1964) |  |  |
| 104                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 417 square metres of<br>hedgerow, track and<br>watercourse (The Cut)<br>(north of Dyserth Road,<br>B5119 and south of Rhyl<br>Coast Road, A548) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                      | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)                                                               |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                     | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                    |                                                                                                                                            | Category 1                                                                                                       |                                                                                           |                                                                                                                                                                                                                                                                                                                                                   |
|                         |                                                             |                                                                                                                                                    | Qualifying persons under Regulatio                                                                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                | Owners or Reputed Owners                                                                                                                   | Lessees or Tenants                                                                                               | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                             |
| 104 cont'd              |                                                             |                                                                                                                                                    |                                                                                                                                            |                                                                                                                  |                                                                                           | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for<br>maintenance as contained in an<br>Agreement dated 14th February<br>1964)                                                                                                                  |
| 105                     | Acquisition of new rights and<br>imposition of restrictions | 8131 square metres of<br>agricultural land and<br>watercourse (The Cut)<br>(north of Dyserth Road,<br>B5119 and south of Rhyl<br>Coast Road, A548) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Conveyance dated 1st May 1959) |
| 106                     | Acquisition of new rights                                   | 259 square metres of<br>agricultural land (north<br>of Dyserth Road, B5119<br>and south of The Cut)                                                | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                                  | NONE                                                                                                             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)                                                                                                                                                                                                                                   |

|                         | Land which is p              | proposed to be subject to: (i)                                                                   | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                  | Qualifying persons under Regulatio                                                                                                         | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                              | Owners or Reputed Owners                                                                                                                   | Lessees or Tenants                                                                                               | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                             |
| 106 cont'd              |                              |                                                                                                  | Unknown<br>(in respect of mines and minerals)                                                                                              |                                                                                                                  |                                                                                           | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Conveyance dated 1st May 1959)                                                                                                                    |
| 107                     | Temporary Possession         | 17 square metres of<br>watercourse (The Cut)<br>and banks (north of<br>Dyserth Road, B5119)      | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Conveyance dated 1st May 1959) |
| 108                     | Temporary Possession         | 1 square metre of<br>agricultural land (north<br>of Dyserth Road, B5119<br>and south of The Cut) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                                  | NONE                                                                                                             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)                                                                                                                                                                                                                                   |

|                         | Land which is p              | proposed to be subject to: (i)                                                                       | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                      | Qualifying persons under Regulatio                                                                                                         | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                  | Owners or Reputed Owners                                                                                                                   | Lessees or Tenants                                                                                               | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                             |
| 108 cont'd              |                              |                                                                                                      | Unknown<br>(in respect of mines and minerals)                                                                                              |                                                                                                                  |                                                                                           | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Conveyance dated 1st May 1959)                                                                                                                    |
| 109                     | Temporary Possession         | 1489 square metres of<br>agricultural land (north<br>of Dyserth Road, B5119<br>and south of The Cut) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Conveyance dated 1st May 1959) |
| 110                     | Temporary Possession         | 470 square metres of<br>hedgerow (north of<br>Dyserth Road, B5119<br>and south of The Cut)           | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                                  | NONE                                                                                                             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)                                                                                                                                                                                                                                   |

|                         | Land which is p                                                                                                                                              | proposed to be subject to: (i)                                   | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                      | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                  |                                                                                           | (Applications: Prescribed Forms and                                                                                 | Category 2                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                 | Description of land                                              | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                                                  | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                  |
| 110 cont'd              |                                                                                                                                                              |                                                                  | Unknown<br>(in respect of mines and minerals)                                             |                                                                                                                     |                                                                                           | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Conveyance dated 1st May 1959)                                                                                                                                                                                                         |
| 111                     | Temporary Possession                                                                                                                                         | 2039 square metres of<br>track (north of Dyserth<br>Road, B5119) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                                | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is j              | proposed to be subject to: (i)                                                                                                 | BOOK OF R<br>powers of compulsory acquisition, (ii)                                       | ITM DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) righ<br>of Denbighshire | ts to carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                |                                                                                           |                                                                                                                   |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|                         |                              |                                                                                                                                | Qualifying persons under Regulati                                                         | on 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                                            | ing (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                            | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                                                | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| 112                     | Temporary Possession         | 6479 square metres of<br>agricultural land and<br>track (north of Dyserth<br>Road, B511 and south of<br>Rhyl Coast Road, A548) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                              | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for<br>maintenance as contained in an<br>Agreement dated 14th February<br>1964) |

|                         | Land which is p              | proposed to be subject to: (i)                                              | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights t<br>f Denbighshire | to carry out protective works (Regulatio                                                                                            | n 7(1)(a))                                                                                                                                                                                         |
|-------------------------|------------------------------|-----------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                             |                                                                                                                                     | Category 1                                                                                                        |                                                                                                                                     |                                                                                                                                                                                                    |
|                         |                              |                                                                             | Qualifying persons under Regulatio                                                                                                  | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                 | Category 2                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                         | Owners or Reputed Owners                                                                                                            | Lessees or Tenants                                                                                                | Occupiers                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                              |
| 113                     | Temporary Possession         | 2 square metres of<br>agricultural land (north<br>of Dyserth Road, B5119)   | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY                                    | NONE                                                                                                              | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY                                    | Barclays Security Trustee Limited<br>1 Churchill Place<br>LONDON<br>E14 5HP<br>(as mortgagee for The Executor of the<br>Estate of the Late Elwyn Lloyd<br>Thomas and Frances Mary Wilma<br>Thomas) |
|                         |                              |                                                                             | The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY |                                                                                                                   | The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and<br>maintain cable)                                      |
|                         |                              |                                                                             | Unknown<br>(in respect of mines and minerals)                                                                                       |                                                                                                                   |                                                                                                                                     | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                    |
| 114                     | Acquisition of new rights    | 681 square metres of<br>agricultural land (north<br>of Dyserth Road, B5119) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                           | NONE                                                                                                              | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                           | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)                                 |

|                         | Land which is p              | proposed to be subject to: (i)                                              | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                             | Qualifying persons under Regulatio                                                        | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                         | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                                                | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                 |
| 114 cont'd              |                              |                                                                             |                                                                                           |                                                                                                                   |                                                                                           | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                      |
| 115                     | Acquisition of new rights    | 449 square metres of<br>agricultural land (north<br>of Dyserth Road, B5119) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                              | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                  |                                                                                           |                                                                                         |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                  | Qualifying persons under Regulation                                                       | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                              | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                      | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |  |
| 116                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 3362 square metres of<br>track (north of Dyserth<br>Road, B5119) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                    | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                      | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                             | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                             | n 7(1)(a))                                                                                                                                                                                                                       |
|-------------------------|------------------------------|---------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                     |                                                                                                                                     | Category 1                                                                                                       |                                                                                                                                     |                                                                                                                                                                                                                                  |
|                         |                              |                                                                     | Qualifying persons under Regulatio                                                                                                  | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                 | Category 2                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                 | Owners or Reputed Owners                                                                                                            | Lessees or Tenants                                                                                               | Occupiers                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                            |
| 116 cont'd              |                              |                                                                     |                                                                                                                                     |                                                                                                                  |                                                                                                                                     | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for<br>maintenance as contained in an<br>Agreement dated 14th February<br>1964) |
| 117                     | Temporary Possession         | 2 square metres of<br>wooded area (north of<br>Dyserth Road, B5119) | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY                                    | NONE                                                                                                             | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY                                    | Barclays Security Trustee Limited<br>1 Churchill Place<br>LONDON<br>E14 5HP<br>(as mortgagee for The Executor of the<br>Estate of the Late Elwyn Lloyd<br>Thomas and Frances Mary Wilma<br>Thomas)                               |
|                         |                              |                                                                     | The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY |                                                                                                                  | The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and<br>maintain cable)                                                                    |
|                         |                              |                                                                     | Unknown<br>(in respect of mines and minerals)                                                                                       |                                                                                                                  |                                                                                                                                     | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                                                                  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                | BOOK OF RE powers of compulsory acquisition, (ii) r                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                               | Qualifying persons under Regulatio                                                        | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                           | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                                                | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                  |
| 117 cont'd              |                                                             |                                                                                               |                                                                                           |                                                                                                                   |                                                                                           | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                                                                                        |
| 118                     | Acquisition of new rights and<br>imposition of restrictions | 21327 square metres of<br>agricultural land and<br>hedgerow (north of<br>Dyserth Road, B5119) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                              | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                               | BOOK OF RE powers of compulsory acquisition, (ii)                                         | TT DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights to<br>of Denbighshire | to carry out protective works (Regulatio                                                  | n 7(1)(a))                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                              | Qualifying persons under Regulatic                                                        | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                              | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                          | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                                                   | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                 |
| 119                     | Temporary Possession         | 974 square metres of<br>agricultural land (north<br>of Dyserth Road, B5119)  | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                                 | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies) |
| 120                     | Temporary Possession         | 9265 square metres of<br>agricultural land (north<br>of Dyserth Road, B5119) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                                 | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                               | BOOK OF RI<br>powers of compulsory acquisition, (ii)                                             | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                  |                                                                                                                   |                                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                          | Owners or Reputed Owners                                                                         | Lessees or Tenants                                                                                                | Occupiers                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                    |
| 121                     | Temporary Possession         | 2650 square metres of<br>agricultural land,<br>hedgerow and track<br>(north of Dyserth Road,<br>B5119)                                                       | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB        | NONE                                                                                                              | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB        | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |
| 122                     | Temporary Possession         | 2 square metres of<br>wooded area (north of<br>Dyserth Road, B5119)                                                                                          | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY | NONE                                                                                                              | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY | Barclays Security Trustee Limited<br>1 Churchill Place<br>LONDON<br>E14 5HP<br>(as mortgagee for The Executor of the<br>Estate of the Late Elwyn Lloyd<br>Thomas and Frances Mary Wilma<br>Thomas)                                                                                                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i)                                              | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|-----------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                             |                                                                                                                                                                                                                                         | Category 1                                                                                                        |                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                         |
|                         |                              |                                                                             | Qualifying persons under Regulatio                                                                                                                                                                                                      | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                 | Category 2                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                         | Owners or Reputed Owners                                                                                                                                                                                                                | Lessees or Tenants                                                                                                | Occupiers                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                   |
| 122 cont'd              |                              |                                                                             | The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY<br>Unknown<br>(in respect of mines and minerals)                                                    |                                                                                                                   | The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and<br>maintain cable)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                        |
| 123                     | Temporary Possession         | 843 square metres of<br>agricultural land (south<br>of Dyserth Road, B5119) | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY<br>The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY | NONE                                                                                                              | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY                                    | Barclays Security Trustee Limited<br>1 Churchill Place<br>LONDON<br>E14 5HP<br>(as mortgagee for The Executor of the<br>Estate of the Late Elwyn Lloyd<br>Thomas and Frances Mary Wilma<br>Thomas)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights to lay and<br>maintain services) |

|                         | Land which is p              | proposed to be subject to: (i)                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                             | n 7(1)(a))                                                                                                                                                                                         |
|-------------------------|------------------------------|-----------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                       | Qualifying persons under Regulatio                                                                                                  | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                 | Category 2                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                   | Owners or Reputed Owners                                                                                                            | Lessees or Tenants                                                                                               | Occupiers                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                              |
| 123 cont'd              |                              |                                                                       |                                                                                                                                     |                                                                                                                  |                                                                                                                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                      |
| 124                     | Temporary Possession         | 113 square metres of<br>wooded area (north of<br>Dyserth Road, B5119) | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY                                    | NONE                                                                                                             | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY                                    | Barclays Security Trustee Limited<br>1 Churchill Place<br>LONDON<br>E14 5HP<br>(as mortgagee for The Executor of the<br>Estate of the Late Elwyn Lloyd<br>Thomas and Frances Mary Wilma<br>Thomas) |
|                         |                              |                                                                       | The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY |                                                                                                                  | The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and<br>maintain cable)                                      |
|                         |                              |                                                                       | Unknown<br>(in respect of mines and minerals)                                                                                       |                                                                                                                  |                                                                                                                                     | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                                    |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                          |                                                                                                                      |                                                                                                                                                            |  |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Category 1                                                               |                                                                                                                      |                                                                                                                                                            |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                  | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                          | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                 |  |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                              | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                    | Lessees or Tenants                                                       | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009      |  |  |  |  |
| 124 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                          |                                                                                                                      | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)            |  |  |  |  |
| 125                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 56 square metres of<br>verge of public road<br>(north of Dyserth Road,<br>B5119) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY<br>(in respect of subsoil beneath half<br>width of public highway)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown |  |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                 | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|----------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                |                                                                                                                                                                                                                   | Category 1                                                                                                        |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|                         |                              |                                                                | Qualifying persons under Regulatio                                                                                                                                                                                | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                            | Owners or Reputed Owners                                                                                                                                                                                          | Lessees or Tenants                                                                                                | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                    |
| 125 cont'd              |                              |                                                                | The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                   |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 126                     | Acquisition of new rights    | 21 square metres of<br>track (north of Dyserth<br>Road, B5119) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                                                                                                         | NONE                                                                                                              | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                              | BOOK OF R powers of compulsory acquisition, (ii)                                          | ITM DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) righ<br>of Denbighshire                                            | ts to carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|-----------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                             | Qualifying persons under Regulati                                                         | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                         | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                    |
| 127                     | Temporary Possession         | 106 square metres of<br>hedgerow (north of<br>Dyserth Road, B5119)          | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                                                                         | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |
| 128                     | Temporary Possession         | 305 square metres of<br>agricultural land (north<br>of Dyserth Road, B5119) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                                                                         | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)                                                                                                                                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i)                                              | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                      | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                             | Qualifying persons under Regulatic                                                        | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                         | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                                                | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                 |
| 128 cont'd              |                              |                                                                             |                                                                                           |                                                                                                                   |                                                                                           | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                    |
| 129                     | Acquisition of new rights    | 593 square metres of<br>agricultural land (north<br>of Dyserth Road, B5119) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                              | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted<br>dated 29 March 2017)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Hugh Henry Wynne<br>Davies) |

|                         | Land which is p              | proposed to be subject to: (i)                                                              | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                          | n 7(1)(a))                                                                                                                                                                                         |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                             |                                                                                                                                     | Category 1                                                                                                       |                                                                                                  |                                                                                                                                                                                                    |
|                         |                              |                                                                                             | Qualifying persons under Regulatio                                                                                                  | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                              | Category 2                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                         | Owners or Reputed Owners                                                                                                            | Lessees or Tenants                                                                                               | Occupiers                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                              |
| 129 cont'd              |                              |                                                                                             |                                                                                                                                     |                                                                                                                  |                                                                                                  | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                                    |
| 130                     | Temporary Possession         | 150 square metres of<br>agricultural land and<br>hedgerow (south of<br>Dyserth Road, B5119) | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY                                    | NONE                                                                                                             | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY | Barclays Security Trustee Limited<br>1 Churchill Place<br>LONDON<br>E14 5HP<br>(as mortgagee for The Executor of the<br>Estate of the Late Elwyn Lloyd<br>Thomas and Frances Mary Wilma<br>Thomas) |
|                         |                              |                                                                                             | The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY |                                                                                                                  |                                                                                                  | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights to lay and<br>maintain apparatus)                                 |
|                         |                              |                                                                                             |                                                                                                                                     |                                                                                                                  |                                                                                                  | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                                    |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                            | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                 |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                           | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Category 2                                                                                                       |                                          |                                                                                                                                                            |
|                         |                                                             |                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Procedures) Regulations 2009                                                                                     | , , , , , , , , , , , , , , , , , , ,    |                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                               | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009      |
| 131                     | Acquisition of new rights and<br>imposition of restrictions | 444 square metres of<br>public road (Dyserth<br>Road, B5119) (excluding<br>all interests of the<br>Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY<br>(in respect of subsoil beneath half<br>width of public highway)<br>George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of subsoil beneath half<br>width of public highway)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                             | NONE                                     | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire                                             | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                         |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Lessees or Tenants                                                                                                                                           | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 131 cont'd              |                              |                              | Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of subsoil beneath half<br>width of public highway)<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN<br>(in respect of subsoil beneath half<br>width of public highway)<br>The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                                                              |                                         |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                        |                                                                         |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                               | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                     | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                     | Occupiers                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 132                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 48 square metres of<br>hedgerow (north of<br>Dyserth Road, B5119) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                   | Tudor Williams<br>Pydew Farm<br>Ffordd Ffynnon<br>PRESTATYN<br>LL19 8BE | NONE                                                                                                                                                  |  |  |  |

|                         | Land which is j              | proposed to be subject to: (i                                                               | BOOK OF R<br>) powers of compulsory acquisition, (ii)                                                                        | rm DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                             | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and |                                                                                                                    |                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|                         |                              |                                                                                             |                                                                                                                              | Procedures) Regulations 2009                                                                                       |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                         | Owners or Reputed Owners                                                                                                     | Lessees or Tenants                                                                                                 | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 133                     | Temporary Possession         | 677 square metres of<br>agricultural land and<br>hedgerow (south of<br>Dyserth Road, B5119) | Richard Gratton<br>8 Eldon Drive<br>ABERGELE<br>LL22 7BY<br>(as reputed owner)<br>Unknown                                    | NONE                                                                                                               | Richard Gratton<br>8 Eldon Drive<br>ABERGELE<br>LL22 7BY<br>(as reputed owner)<br>Unknown | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of right to take water and<br>use water pipe)<br>lola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of right to take water and<br>use water pipe)<br>Jason Wayne Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to<br>septic tank)<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN<br>(in respect of right to take water and<br>use water pipe) |

|                         | Land which is p              | proposed to be subject to: (i)                                   | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                      | rm DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                         |
|-------------------------|------------------------------|------------------------------------------------------------------|-------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                  | Qualifying persons under Regulation                                                       | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                              | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                              | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                                                   | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                              |
| 133 cont'd              |                              |                                                                  |                                                                                           |                                                                                                                      |                                                                                           | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Pamela Elaine Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to<br>septic tank)<br>Unknown                         |
| 134                     | Acquisition of new rights    | 5 square metres of<br>hedgerow (south of<br>Dyserth Road, B5119) | Richard Gratton<br>8 Eldon Drive<br>ABERGELE<br>LL22 7BY<br>(as reputed owner)<br>Unknown | NONE                                                                                                                 | Richard Gratton<br>8 Eldon Drive<br>ABERGELE<br>LL22 7BY<br>(as reputed owner)<br>Unknown | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of right to take water and<br>use water pipe)<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of right to take water and<br>use water pipe) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulation                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                        | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 134 cont'd              |                              |                               |                                                          |                                                                                                               |                                         | Jason Wayne Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to<br>septic tank)<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN<br>(in respect of right to take water and<br>use water pipe)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Pamela Elaine Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to<br>septic tank)<br>Unknown |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                          |                                                                                                                                                           |                                                                                                                                                            |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                              | Category 1                                                               |                                                                                                                                                           |                                                                                                                                                            |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                             | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                                                       | Category 2                                                                                                                                                 |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                         | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                     | Lessees or Tenants                                                       | Occupiers                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009      |  |  |  |
| 135                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 19 square metres of<br>verge of public road<br>(south of Dyserth Road,<br>B5119) and footpath<br>(DE 206 5) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of subsoil beneath half<br>width of public highway)<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>(in respect of footpath DE 206 5) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                              | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                                        |                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                 |
|                         |                              |                                                                                                             | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                               | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                                                 | Category 2                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                         | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                           |
| 135 cont'd              |                              |                                                                                                             | Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN<br>(in respect of subsoil beneath half<br>width of public highway)<br>Richard Gratton<br>8 Eldon Drive<br>ABERGELE<br>LL22 7BY<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                   |                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                 |
| 136                     | Acquisition of new rights    | 16 square metres of<br>verge of public road<br>(south of Dyserth Road,<br>B5119) and footpath<br>(DE 206 5) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                                                                                                                                         | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>(in respect of public footpath<br>DE 206 5) | Richard Gratton<br>8 Eldon Drive<br>ABERGELE<br>LL22 7BY<br>(in respect of right to passage of<br>service media)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline) |

|                         | Land which is p              | proposed to be subject to: (i)                                                             | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                                                                                                                                        | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                            |                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                       |
|                         |                              |                                                                                            | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                              | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                             | outogo, y _                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                        | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                        | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 136 cont'd              |                              |                                                                                            | Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN<br>Unknown<br>(in respect of mines and minerals)                                                                                                                                            |                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |
| 137                     | Temporary Possession         | 50 square metres of<br>agricultural land and<br>hedgerow (south of<br>Dyserth Road, B5119) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN | NONE                                                                                                             | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)       |

|                         | Land which is p              | proposed to be subject to: (i)                                             | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                         | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|----------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                            | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                               | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                        | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 137 cont'd              |                              |                                                                            | Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                       |
| 138                     | Temporary Possession         | 54 square metres of<br>agricultural land (south<br>of Dyserth Road, B5119) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                                                                         | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)       |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                     | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                 | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                    |                                                                                                                                                                                                                                         |                                                                                                                   | Category 2                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|                         |                                                             |                                                                                                                                    | Qualitying persons under Regulatio                                                                                                                                                                                                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                | Owners or Reputed Owners                                                                                                                                                                                                                | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                      |
| 139                     | Acquisition of new rights and<br>imposition of restrictions | 158 square metres of<br>agricultural land and<br>hedgerow (south of<br>Dyserth Road, B5119)                                        | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY<br>The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY | NONE                                                                                                              | Frances Mary Wilma Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY<br>The Executor of the Estate of the<br>Late Elwyn Lloyd Thomas<br>Rhydorddwy Fawr<br>Dyserth Road<br>RHYL<br>Denbighshire<br>LL18 4DY | Barclays Security Trustee Limited<br>1 Churchill Place<br>LONDON<br>E14 5HP<br>(as mortgagee for The Executor of the<br>Estate of the Late Elwyn Lloyd<br>Thomas and Frances Mary Wilma<br>Thomas)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights to lay and<br>maintain services)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |
| 140                     | Acquisition of new rights and imposition of restrictions    | 32097 square metres of<br>agricultural land,<br>hedgerow and track<br>(south of Dyserth Road,<br>B5119) and footpath<br>(DE 206 5) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                                                                                                          | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 5)                                                                                                         | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right of access for<br>maintenance)                                                                                                                                                                                                                                                                                                                                                                                     |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri                                                                                                                                                                                                                                     | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                        | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                                                                                                                                                                                | Category 1                                                                                                  |                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                         |                              |                               | Qualifying persons under Regulation                                                                                                                                                                                                                                                            | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                     | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                          | Occupiers                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 140 cont'd              |                              |                               | Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>Unknown<br>(in respect of mines and minerals) |                                                                                                             | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline stated in deed<br>dated 19 May 1955)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline stated in deed<br>dated 19 May 1955)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                       | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                             | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                       |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                                         |                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                  |
|                         |                                                             |                                                                                                                      | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                               | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                  | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                                 | Occupiers                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                            |
| 141                     | Acquisition of new rights and<br>imposition of restrictions | 2528 square metres of<br>agricultural land (south<br>of Dyserth Road, B5119)                                         | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                               | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |
| 142                     | Acquisition of new rights                                   | 3651 square metres of<br>agricultural land and<br>track (south of Dyserth<br>Road, B5119) and<br>footpath (DE 206 5) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                                                                                                                                                                                                                                                                   | NONE                                                                                                               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 5)                                                                                                                                                                                                                 | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right of access for<br>maintenance)                                                                                                                                                           |

|                         | Land which is p                                          | roposed to be subject to: (i)                                                                                                      | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                    |                                                                                                                                                                                                                                                                                                |                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|                         |                                                          |                                                                                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                                                                             | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                | Owners or Reputed Owners                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                               |
| 142 cont'd              |                                                          |                                                                                                                                    | Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                  | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline stated in deed<br>dated 19 May 1955)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |
| 143                     | Acquisition of new rights and imposition of restrictions | 14073 square metres of<br>agricultural land,<br>hedgerow and track<br>(south of Dyserth Road,<br>B5119) and footpath<br>(DE 206 5) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                                                                                                                                                                 | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 5)                                                                                                                                                                                                                 | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right of access for<br>maintenance)                                                                                                                                                                                                                                                                                                                                                                              |

|                         | Land which is p                                          | roposed to be subject to: (i)                                                                | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire                                             | o carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|----------------------------------------------------------|----------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                              | Qualifying persons under Regulatio                                                                                                                                                                                                                                                             | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                          | Owners or Reputed Owners                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                               |
| 143 cont'd              |                                                          |                                                                                              | Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                                                              | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline stated in deed<br>dated 19 May 1955)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |
| 144                     | Acquisition of new rights and imposition of restrictions | 6563 square metres of<br>agricultural land and<br>hedgerow (south of<br>Dyserth Road, B5119) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                                                                                                                                                                 | NONE                                                                                                                                                         | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                                                                                                                                                                                                                  | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right of access for<br>maintenance)                                                                                                                                                                                                                                                                                                                                                                              |

|                         | Land which is p                                             | proposed to be subject to: (i)                                              | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                                      | n 7(1)(a))                                                                                                                                                                                                      |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                             | Qualifying persons under Regulatio                                                                                                                                                                                                                                                             | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                                                                                           | Category 2                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                         | Owners or Reputed Owners                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                           |
| 144 cont'd              |                                                             |                                                                             | Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                  | Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline stated in deed<br>dated 19 May 1955) |
| 145                     | Acquisition of new rights and<br>imposition of restrictions | 206 square metres of<br>agricultural land (south<br>of Dyserth Road, B5119) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                                                                                                                                                                 | NONE                                                                                                             | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                                                                                                                | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                 |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                         | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                                      | n 7(1)(a))                                                                                                                                            |
|-------------------------|----------------------------------------------------------|--------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                        | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and                                                                                                                                                                   |                                                                                                                  |                                                                                                                                                                                                                                               | Category 2                                                                                                                                            |
|                         |                                                          |                                                                                                        |                                                                                                                                                                                                                                                                                                | Procedures) Regulations 2009                                                                                     |                                                                                                                                                                                                                                               |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                    | Owners or Reputed Owners                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 145 cont'd              |                                                          |                                                                                                        | Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                  | Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)       |
| 146                     | Acquisition of new rights and imposition of restrictions | 4674 square metres of<br>agricultural land,<br>hedgerow and ditch<br>(south of Dyserth Road,<br>B5119) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                                                                       | NONE                                                                                                             | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                      | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)       |

|                         | Land which is p                                             | proposed to be subject to: (i                                      | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                 | n 7(1)(a))                                                                                                                                            |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                    |                                                                                                                                                                                                                                      |                                                                                                                  | Colonary 2                                                                                                                                                                               |                                                                                                                                                       |
|                         |                                                             |                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                   | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                      | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                | Owners or Reputed Owners                                                                                                                                                                                                             | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 146 cont'd              |                                                             |                                                                    | Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>Unknown<br>(in respect of mines and minerals)                                 |                                                                                                                  | Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN                                      |                                                                                                                                                       |
| 147                     | Acquisition of new rights and<br>imposition of restrictions | 149 square metres of<br>hedgerow (south of<br>Dyserth Road, B5119) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(as reputed owner)<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(as reputed owner) | NONE                                                                                                             | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF | Unknown                                                                                                                                               |

|                         | Land which is p                                          | proposed to be subject to: (i)                                    | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                                                                                         | n 7(1)(a))                                                                                                                                            |
|-------------------------|----------------------------------------------------------|-------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                           | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                              | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                     | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 147 cont'd              |                                                          |                                                                   | Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN<br>(as reputed owner)<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 SSB<br>(as reputed owner)<br>Unknown |                                                                                                                   | Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Unknown |                                                                                                                                                       |
| 148                     | Acquisition of new rights and imposition of restrictions | 21 square metres of<br>hedgerow (south of<br>Dyserth Road, B5119) | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(as reputed owner)                                                                                                                                                                                                                         | NONE                                                                                                              | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF                                                                                                                                                                                                   | Unknown                                                                                                                                               |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                          | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                      | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                     | Lessees or Tenants                                                                                             | Occupiers                                                                                                                                                                                                                                                                                                                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 148 cont'd              |                              |                              | Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(as reputed owner)<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>(as reputed owner)<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>(as reputed owner)<br>Unknown |                                                                                                                | Iola Morris Kerfoot<br>Henliys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions<br>Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Unknown |                                                                                                                                                       |

|                         | Land which is p                                             | proposed to be subject to: (i)                                             | BOOK OF R<br>powers of compulsory acquisition, (ii)                                                                            | rm DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                                         | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                         |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                            | Qualifying persons under Regulati                                                                                              | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                      | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                        | Owners or Reputed Owners                                                                                                       | Lessees or Tenants                                                                                                                                           | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                              |
| 149                     | Acquisition of new rights and<br>imposition of restrictions | 55 square metres of<br>hedgerow (south of<br>Dyserth Road, B5119)          | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021) |
| 150                     | Acquisition of new rights and imposition of restrictions    | 35 square metres of<br>agricultural land (south<br>of Dyserth Road, B5119) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 SSB | Clwyd<br>LL18 5SB<br>Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH                                                   | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021) |
|                         |                                                             |                                                                            |                                                                                                                                | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB                                                                     |                                                                                       |                                                                                                                                                                                    |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                            | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                            | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                                                         | n 7(1)(a))                                                                                                                                                                         |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                           | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                     |                                                                                                                                 | Category 2                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                       | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants                                                                                                  | Occupiers                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                              |
| 151                     | Acquisition of new rights and<br>imposition of restrictions | 123 square metres of<br>agricultural land (south<br>of Dyserth Road, B5119)<br>and footpath<br>(DE 206 3) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                               | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH                               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 3) | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021) |
|                         |                                                             |                                                                                                           |                                                                                                                                                              | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB                            | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH                                           |                                                                                                                                                                                    |
| 152                     | Acquisition of new rights                                   | 10 square metres of<br>agricultural land (south<br>of Dyserth Road, B5119)                                | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                               | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH                               | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH                                           | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021) |
|                         |                                                             |                                                                                                           |                                                                                                                                                              | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB                            |                                                                                                                                 |                                                                                                                                                                                    |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                      | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                   | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                                                 | o carry out protective works (Regulatio                                                                                                                                                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                     | Category 1                                                                                                                                                                        |                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                         |                              |                                                                                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                  | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                 | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                                                                                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| 153                     | Acquisition of new rights    | 1494 square metres of<br>agricultural land (south<br>of Dyserth Road, B5119)<br>and footpaths<br>(DE 206 4, DE 206 6,<br>DE 206 11) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 4,<br>DE 206 6 and DE 206 11)<br>Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                          |                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                     | Category 1                                                                               |                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                               | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                  | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                 | (Applications: Prescribed Forms and                                                                                                                                                                                              | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                                                           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                       | Occupiers                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |  |
| 154                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 3934 square metres of<br>agricultural land,<br>hedgerow, track and<br>private road (south of<br>Dyserth Road, B5119)<br>and footpaths<br>(DE 206 12, DE 206 13,<br>DE 206 15) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 12,<br>DE 206 13, DE 206 15)<br>Gwilym Howatson<br>Aberkinsey Farm<br>Dyserth<br>Y RHYL<br>LL18 6BL | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                        | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                      | n 7(1)(a))                                                                                                                                                                                                                                       |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                   | Category 1                                                                                                       |                                                                                                                                                              |                                                                                                                                                                                                                                                  |
|                         |                              |                                                                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                          | Category 2                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                   | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                          | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                            |
| 154 cont'd              |                              |                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                  |                                                                                                                                                              | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                    |
| 155                     | Acquisition of new rights    | 1166 square metres of<br>private road (south of<br>Dyserth Road, B5119)<br>and footpaths<br>(DE 204 7,<br>DE 206 15a) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>(as reputed owner)<br>Nicola Shearer Williamson<br>Clarence House<br>Bryniau<br>Dyserth<br>RHYL<br>Denbighshire<br>LL18 6BY<br>(as reputed owner)<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 204 7<br>and DE 206 15a)<br>Unknown | Gwilym Howatson<br>Aberkinsey Farm<br>Dyserth<br>Y RHYL<br>LL18 6BL<br>(in respect of right of access)<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB<br>(in respect of right of access)<br>Unknown |
|                         |                              |                                                                                                                       | Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>(as reputed owner)                                                                                                                                                                                                                                                                                |                                                                                                                  |                                                                                                                                                              |                                                                                                                                                                                                                                                  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                 | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                                                 | o carry out protective works (Regulatio                                                                                                                                                                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                            |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and                                                                                                                               |                                                                                                                                                                                   |                                                                                                                                                                                                                          | Category 2                                                                                                                                                                                                                                                                                                                            |
|                         |                                                             |                                                                                                                                |                                                                                                                                                                                                                                                            | Procedures) Regulations 2009                                                                                                                                                      |                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                            | Owners or Reputed Owners                                                                                                                                                                                                                                   | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                                                                                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                 |
| 155 cont'd              |                                                             |                                                                                                                                | Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>(as reputed owner)<br>Sally Fox Hall<br>The Paddock<br>Dyserth Hall<br>Penisa<br>Dyserth<br>RHYL<br>Denbighshire<br>LL18 6BW<br>(as reputed owner)<br>Unknown |                                                                                                                                                                                   |                                                                                                                                                                                                                          | Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications<br>apparatus)                                                                                                                                                                                                         |
| 156                     | Acquisition of new rights and<br>imposition of restrictions | 3178 square metres of<br>agricultural land and<br>hedgerow (south of<br>Dyserth Road, B5119)<br>and footpath<br>(DE   206   3) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                                                                                                                             | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 3)<br>Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                              | ,<br>BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                    | rm DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                                                              | o carry out protective works (Regulatio                                                                                                                                                                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                             |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                             |                                                                                                                                                                                                                                              | Category 1                                                                                                                                                                        |                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                        |
|                         |                                                             |                                                                                                                             | Qualifying persons under Regulation                                                                                                                                                                                                          | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                         | (Applications: Prescribed Forms and                                                                                                                                                                                      | Category 2                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                         | Owners or Reputed Owners                                                                                                                                                                                                                     | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                                                                                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |
| 157                     | Acquisition of new rights and<br>imposition of restrictions | 5902 square metres of<br>agricultural land and<br>hedgerow (south of<br>Dyserth Road, B5119)<br>and footpath<br>(DE 206 3)  | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                                                                                                               | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 3)<br>Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)  |
| 158                     | Acquisition of new rights and<br>imposition of restrictions | 2128 square metres of<br>agricultural land and<br>shrubbery (south of<br>Dyserth Road, B5119)<br>and footpath<br>(DE 206 3) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 3)<br>Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991) |

|                         | Land which is p              | proposed to be subject to: (i)                                               | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire                                                                 | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                              |                                                                                                                                                                                     | Category 1                                                                                                                                                                        |                                                                                       |                                                                                                                                                                                                                                                                                                                                       |
|                         |                              |                                                                              | Qualifying persons under Regulatio                                                                                                                                                  | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                          | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                          | Owners or Reputed Owners                                                                                                                                                            | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                 |
| 158 cont'd              |                              |                                                                              | Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY |                                                                                                                                                                                   |                                                                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)              |
| 159                     | Temporary Possession         | 1535 square metres of<br>agricultural land (south<br>of Dyserth Road, B5119) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                                                      | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii) ו                                                                                                                                                                                                                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                                                                | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                        | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                           | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                           |
| 160                     | Acquisition of new rights and<br>imposition of restrictions | 84 square metres of<br>hedgerow (south of<br>Dyserth Road, B5119) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                               |                                                                                                                                                                                                                                              |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                               | Qualifying persons under Regulatio                                                                                                                                                                                                           | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                           | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                             |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                           | Owners or Reputed Owners                                                                                                                                                                                                                     | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |  |  |  |
| 160 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                               | Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY                                                                                                                                                       |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 161                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 1552 square metres of<br>agricultural land and<br>shrubbery (south of<br>Dyserth Road, B5119) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                                            |                                                                                                                                                                                                    |                                                                                        |                                                                                                                                  |                                                                                                                                                                                    |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                            | Qualifying persons under Regulatio                                                                                                                                                                 | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                              | Category 2                                                                                                                                                                         |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                                                        | Owners or Reputed Owners                                                                                                                                                                           | Lessees or Tenants                                                                     | Occupiers                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                              |  |  |  |
| 161 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                            | Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY |                                                                                        |                                                                                                                                  | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus)              |  |  |  |
| 162                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 48411 square metres of<br>agricultural land,<br>hedgerow and<br>shrubbery (north of<br>Pentre Lane and south<br>of Dyserth Road, B5119)<br>and footpath<br>(DE   206   44) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD                                                                                         | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 44) | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                           |                                                                                                                                                                                                                                                                                                                       |                                                                                          |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                           |                                                                                                                                                                                                                                                                                                                       | Category 1                                                                               |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                           | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                    | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                       | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 162 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                           | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |  |  |
| 163                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 12537 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre Lane and south<br>of Dyserth Road, B5119) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD                                                                                                                                                                                                            | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH    | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)                                                                                                                                                                                                                                                                                           |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                                      |                                                                                          |                                     |                                                                                                                                                                                                                                                                                                                           |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                   | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009   | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                             | Lessees or Tenants                                                                       | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                     |  |  |  |
| 163 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB |                                     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                     |                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                     |                                                                                                                                                                                                                                                                                                                                           | Category 1                                                                                                                                                                        |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                          | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                           |  |  |
| 164                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 254 square metres of<br>agricultural land (north<br>of Pentre Lane) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                      |                                                                                                                                                                                                                                              |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                      | Qualifying persons under Regulatio                                                                                                                                                                                                           | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                            | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                             |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                  | Owners or Reputed Owners                                                                                                                                                                                                                     | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |  |  |  |
| 164 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                      | Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY                                                                                                                                                       |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 165                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 1876 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre Lane) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                             |                                                                                                                                                                                                    |                                                                                        |                                                                                       |                                                                                                                                                                                    |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                             | Qualifying persons under Regulatio                                                                                                                                                                 | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                         |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                         | Owners or Reputed Owners                                                                                                                                                                           | Lessees or Tenants                                                                     | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                              |  |  |  |
| 165 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                             | Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY |                                                                                        |                                                                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus)              |  |  |  |
| 166                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 11721 square metres of<br>agricultural land,<br>hedgerow and pond<br>(north of Pentre Lane) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD                                                                                         | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH  | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                                         |                                                                                          |                                     |                                                                                                                                                                                                                                                                                                                           |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning                                   | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                | Procedures) Regulations 2009<br>Lessees or Tenants                                       | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                     |  |  |  |
| 166 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm<br>Pentre Farm<br>Lane<br>Rhuddian<br>RHYL<br>LL18 6LY | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB |                                     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                     |                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                     |                                                                                                                                                                                                                                                                                                                                           | Category 1                                                                                                                                                                        |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                          | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| 167                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 358 square metres of<br>agricultural land (north<br>of Pentre Lane) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                     |                                                                                                                                                                                                                                              |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                                           | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                            | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                             |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                 | Owners or Reputed Owners                                                                                                                                                                                                                     | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |  |  |  |
| 167 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                     | Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY                                                                                                                                                       |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 168                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 245 square metres of<br>agricultural land (north<br>of Pentre Lane) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                              | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                         |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                             | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                       |                                                                                                                    |                                                                                       | Category 2                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                         | Owners or Reputed Owners                                                                                                                                                                           | Lessees or Tenants                                                                                                 | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                              |
| 168 cont'd              |                              |                                                                                             | Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY |                                                                                                                    |                                                                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus)              |
| 169                     | Acquisition of new rights    | 4031 square metres of<br>agricultural land, track<br>and hedgerow (north of<br>Pentre Lane) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD                                                                                         | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH                              | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                                      |                                                                                          |                                     |                                                                                                                                                                                                                                                                                                                           |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                   | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009   | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                             | Lessees or Tenants                                                                       | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                     |  |  |  |
| 169 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 55B |                                     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                     |                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                     |                                                                                                                                                                                                                                                                                                                                           | Category 1                                                                                                                                                                        |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                          | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| 170                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 776 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre Lane and south<br>of Bryn Cwnin Road) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                        |                                                                                                                                                                                                                                              |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                        | Qualifying persons under Regulatio                                                                                                                                                                                                           | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                           | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                             |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                    | Owners or Reputed Owners                                                                                                                                                                                                                     | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |  |  |  |
| 170 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                        | Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY                                                                                                                                                       |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 171                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 573 square metres of<br>agricultural land (north<br>of Pentre Lane and<br>south of Bryn Cwnin<br>Road) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                  |                                                                                                                                                                                     |                                                                                                                                                                                   |                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                         |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                  |                                                                                                                                                                                     | Category 1                                                                                                                                                                        |                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                         |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                  | Qualifying persons under Regulatio                                                                                                                                                  | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                          | (Applications: Prescribed Forms and                                                                                                                                                                                           | Category 2                                                                                                                                                                                                                                              |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                              | Owners or Reputed Owners                                                                                                                                                            | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                   |  |  |  |
| 171 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                  | Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY |                                                                                                                                                                                   |                                                                                                                                                                                                                               | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus)                                                                                   |  |  |  |
| 172                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 4649 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre Lane and south<br>of Bryn Cwnin Road) and<br>footpath (DE 206 44) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                                                      | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE   206   44)<br>Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                          | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                                                 | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                        |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                         | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                   |                                                                                       | Category 2                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                     | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                             |
| 172 cont'd              |                              |                                                                                         |                                                                                                                                                              |                                                                                                                                                                                   |                                                                                       | Unknown<br>(in respect to such covenants and<br>easements as may have been<br>imposed thereon before 5 July 2006<br>and are still subsisting and capable of<br>being enforced)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access) |
| 173                     | Temporary Possession         | 156 square metres of<br>agricultural land and<br>hedgerow (south of Bryn<br>Cwnin Road) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                               | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                           |

|                         | Land which is p              | proposed to be subject to: (i)                                                                  | BOOK OF RE powers of compulsory acquisition, (ii)                                                                              | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                                                               | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                        |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                 | Qualifying persons under Regulation                                                                                            | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009                      |                                                                                       |                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                             | Owners or Reputed Owners                                                                                                       | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                             |
| 173 cont'd              |                              |                                                                                                 |                                                                                                                                |                                                                                                                                                                                   |                                                                                       | Unknown<br>(in respect to such covenants and<br>easements as may have been<br>imposed thereon before 5 July 2006<br>and are still subsisting and capable of<br>being enforced)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access) |
| 174                     | Acquisition of new rights    | 1251 square metres of<br>agricultural land, track<br>and woodland (south of<br>Bryn Cwnin Road) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                           |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| 174 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Unknown<br>(in respect to such covenants and<br>easements as may have been<br>imposed thereon before 5 July 2006<br>and are still subsisting and capable of<br>being enforced)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)<br>Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications<br>apparatus) |  |  |

|                         | Land which is j              | proposed to be subject to: (i)                                                                         | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                                                                | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                        | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                        | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                           | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                    | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                           |
| 175                     | Acquisition of new rights    | 227 square metres of<br>agricultural land (north<br>of Pentre Lane and<br>south of Bryn Cwnin<br>Road) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                       |                                                                                                                                                                                                                                              |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                           | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                           | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                             |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                   | Owners or Reputed Owners                                                                                                                                                                                                                     | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |  |  |  |
| 175 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                       | Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY                                                                                                                                                       |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 176                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 43 square metres of<br>agricultural land (north<br>of Pentre Lane and<br>south of Bryn Cwnin<br>Road) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                         |                                                                                                                                                                                                    |                                                                                        |                                                                                                                                  |                                                                                                                                                                                    |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                         | Qualifying persons under Regulatio                                                                                                                                                                 | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                              | Category 2                                                                                                                                                                         |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                     | Owners or Reputed Owners                                                                                                                                                                           | Lessees or Tenants                                                                     | Occupiers                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                              |  |  |  |
| 176 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                         | Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY |                                                                                        |                                                                                                                                  | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus)              |  |  |  |
| 177                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 5302 square metres of<br>agricultural land and<br>wooded area (south of<br>Bryn Cwnin Road) and<br>footpath (DE 206 44) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD                                                                                         | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 44) | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                  |                                                                                                                                                                                                                                                                                                                       |                                                                                          |                                                                                                                      |                                                                                                                                                                                                                                                                                                                           |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                  |                                                                                                                                                                                                                                                                                                                       | Category 1                                                                               |                                                                                                                      |                                                                                                                                                                                                                                                                                                                           |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                  | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                   | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                              | Owners or Reputed Owners                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                       | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                     |  |  |  |
| 177 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                  | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH                                | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |  |  |  |
| 178                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 10 square metres of<br>public road (south of<br>Bryn Cwnin Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown                                                                                                                                                                                       | NONE                                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                                                                                                                                                                                   |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                               | Qualifying persons under Regulatio                                                                                   | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                           | Owners or Reputed Owners                                                                                             | Lessees or Tenants                                                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 178 cont'd              |                              |                                                                               |                                                                                                                      |                                                                                                                  |                                                                                                                      | Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications<br>apparatus)                         |
| 179                     | Acquisition of new rights    | 5 square metres of<br>footway of public road<br>(south of Bryn Cwnin<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                           | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                           | NONE<br>Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications<br>apparatus)                 |
| 180                     | Acquisition of new rights    | 2 square metres of<br>footway of public road<br>(south of Bryn Cwnin<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                          |                                                                                                                                                                                                              |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                         |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                          |                                                                                                                                                                                                              | Category 1                                                                                                                                                                        |                                                                                       |                                                                                                                                                                                                                                                                                                         |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                          | Qualifying persons under Regulatio                                                                                                                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                          | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                              |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                      | Owners or Reputed Owners                                                                                                                                                                                     | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                   |  |  |  |
| 180 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                          | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                         |  |  |  |
| 181                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 1137 square metres of<br>agricultural land and<br>hedgerow (south of Bryn<br>Cwnin Road) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                                                                               | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)<br>Unknown<br>(in respect to such covenants and |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                          |                                                                                                                                                                                                              |                                                                                                                                                                                   |                                                                                       | easements as may have been<br>imposed thereon before 5 July 2006<br>and are still subsisting and capable of<br>being enforced)                                                                                                                                                                          |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                           | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                      | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                                                               | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                          |                                                                                                                                                                                                                                                                                                                                           | Category 1                                                                                                                                                                        |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                         |                              |                                                                          | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                       | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                         | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                      | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                 |
| 181 cont'd              |                              |                                                                          |                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                   |                                                                                       | Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                       |
| 182                     | Temporary Possession         | 1694 square metres of<br>agricultural land (south<br>of Bryn Cwnin Road) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                             |                                                                                                                                                                                                                                              |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                             | Qualifying persons under Regulatio                                                                                                                                                                                                           | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                           | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                             |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                         | Owners or Reputed Owners                                                                                                                                                                                                                     | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |  |  |  |
| 182 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                             | Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY                                                                                                                                                       |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 183                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 447 square metres of<br>track (south of Bryn<br>Cwnin Road) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991) |  |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                                                 | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                                                                              |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                    |                                                                                                                                                                                     | Category 1                                                                                                                                                                        |                                                                                       |                                                                                                                                                                                                                                                         |
|                         |                                                             |                                                                    | Qualifying persons under Regulatio                                                                                                                                                  | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                          | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                | Owners or Reputed Owners                                                                                                                                                            | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                   |
| 183 cont'd              |                                                             |                                                                    | Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY |                                                                                                                                                                                   |                                                                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus)                                                                                   |
| 184                     | Acquisition of new rights and<br>imposition of restrictions | 7765 square metres of<br>wooded area (south of<br>Bryn Cwnin Road) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                                                      | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                   | BOOK OF RE<br>() powers of compulsory acquisition, (ii)                                                                        | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                                                               | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                        |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                  | Qualifying persons under Regulation                                                                                            | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                           | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                              | Owners or Reputed Owners                                                                                                       | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                             |
| 184 cont'd              |                                                          |                                                                  |                                                                                                                                |                                                                                                                                                                                   |                                                                                       | Unknown<br>(in respect to such covenants and<br>easements as may have been<br>imposed thereon before 5 July 2006<br>and are still subsisting and capable of<br>being enforced)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access) |
| 185                     | Acquisition of new rights and imposition of restrictions | 77 square metres of<br>wooded area (south of<br>Bryn Cwnin Road) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                           |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                          | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights to<br>f Denbighshire                                                                | o carry out protective works (Regulatio                                                                                                                                                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                             |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                         |                                                                                                                                                                                                                                              | Category 1                                                                                                                                                                        |                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                        |
|                         |                                                             |                                                                                                                         | Qualifying persons under Regulatio                                                                                                                                                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                          | (Applications: Prescribed Forms and                                                                                                                                                                                       | Category 2                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                     | Owners or Reputed Owners                                                                                                                                                                                                                     | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                  |
| 185 cont'd              |                                                             |                                                                                                                         |                                                                                                                                                                                                                                              |                                                                                                                                                                                   |                                                                                                                                                                                                                           | Unknown<br>(in respect to such covenants and<br>easements as may have been<br>imposed thereon before 5 July 2006<br>and are still subsisting and capable of<br>being enforced)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                      |
| 186                     | Acquisition of new rights and<br>imposition of restrictions | 5551 square metres of<br>agricultural land and<br>wooded area (south of<br>Bryn Cwnin Road) and<br>footpath (DE 206 17) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 17)<br>Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                     |                                                                                                                                                                                                    |                                                                                        |                                                                                                                                                   |                                                                                                                                                                                    |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                     | Qualifying persons under Regulatio                                                                                                                                                                 | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                                               | Category 2                                                                                                                                                                         |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                 | Owners or Reputed Owners                                                                                                                                                                           | Lessees or Tenants                                                                     | Occupiers                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                              |  |  |  |
| 186 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                                     | Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY |                                                                                        |                                                                                                                                                   | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus)              |  |  |  |
| 187                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 8968 square metres of<br>agricultural land and<br>hedgerow (south of Bryn<br>Cwnin Road) and<br>footpaths (DE 206 17,<br>DE 206 18) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD                                                                                         | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 17<br>and DE 206 18) | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                              |                                                                                       |                                                                                                                                                                                                                                                                                                                           |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                             | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                       |                                                                                                                                                                                                                                                                                                                           |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                                                                           | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                     |  |  |  |
| 187 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm<br>Pentre Farm<br>Pentre Farm<br>Rhuddian<br>RHYL<br>LL18 6LY | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB                                                                     | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0LH | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                   |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                 | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                        | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                            | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                             | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| 188                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 517 square metres of<br>agricultural land (south<br>of Bryn Cwnin Road,<br>north of Pentre and east<br>of A525) | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH<br>The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer<br>dated 01 December 2021)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain apparatus) |  |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                               | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                              |                                                                                                                        | Category 1                                                                                                        |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                     |
|                         |                                                             |                                                                                                                              | Qualifying persons under Regulatio                                                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                          | Owners or Reputed Owners                                                                                               | Lessees or Tenants                                                                                                | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                               |
| 188 cont'd              |                                                             |                                                                                                                              | Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY                                 |                                                                                                                   |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                     |
| 189                     | Acquisition of new rights and<br>imposition of restrictions | 338 square metres of<br>agricultural land and<br>hedgerow (south of Bryn<br>Cwnin Road, north of<br>Pentre and east of A525) | Unknown<br>(in respect of mines and minerals)<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                        | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of drainage) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                               |
|-------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                       | Qualifying persons under Regulatio                                                                                                            | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                   | Owners or Reputed Owners                                                                                                                      | Lessees or Tenants                                                                                               | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009    |
| 190                     | Acquisition of new rights and imposition of restrictions | 736 square metres of<br>agricultural land (south<br>of Bryn Cwnin Road,<br>north of Pentre Lane<br>and east of A525)  | Action For Children<br>3 The Boulevard<br>Ascot Road<br>WATFORD<br>Hertfordshire<br>WD18 8AG<br>Unknown<br>(in respect of mines and minerals) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                        | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                                                                     |
| 191                     | Acquisition of new rights and imposition of restrictions | 3050 square metres of<br>agricultural land (south<br>of Bryn Cwnin Road,<br>north of Pentre Lane<br>and east of A525) | Action For Children<br>3 The Boulevard<br>Ascot Road<br>WATFORD<br>Hertfordshire<br>WD18 8AG<br>Unknown<br>(in respect of mines and minerals) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                        | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                                                                     |
| 192                     | Acquisition of new rights and imposition of restrictions | 477 square metres of<br>agricultural land (east of<br>A525)                                                           | Unknown<br>(in respect of mines and minerals)                                                                                                 | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                        | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter) |

|                         | Land which is p                                          | roposed to be subject to: (i)                   | BOOK OF REI<br>powers of compulsory acquisition, (ii) r               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|----------------------------------------------------------|-------------------------------------------------|-----------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                 | Qualifying persons under Regulatio                                    | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                           |                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                             | Owners or Reputed Owners                                              | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                      |
| 192 cont'd              |                                                          |                                                 | Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL |                                                                                                                                                              |                                                                                           | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of drainage) |
| 193                     | Acquisition of new rights and imposition of restrictions | 704 square metres of<br>hedgerow (east of A525) | Unknown<br>(in respect of mines and minerals)                         | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                                                    | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)                                                                                                                                                                                                   |

|                         | Land which is p                                             | proposed to be subject to: (i)                                             | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                  |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                            | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                   |                                                                                           | Category 2                                                                                                                                                                                                                                                                                                  |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                        | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants                                                                                                | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                       |
| 193 cont'd              |                                                             |                                                                            | Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL                                                                                        |                                                                                                                   |                                                                                           | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of drainage)                                                                                                   |
| 194                     | Acquisition of new rights and<br>imposition of restrictions | 15513 square metres of<br>agricultural land and<br>hedgerow (east of A525) | Unknown<br>(in respect of mines and minerals)<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL                                       | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                         | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter) |

|                         | Land which is p              | proposed to be subject to: (i)                                        | BOOK OF REI<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire                                             | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|------------------------------|-----------------------------------------------------------------------|---------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                       | Qualifying persons under Regulatio                      | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                           |                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                   | Owners or Reputed Owners                                | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                      |
| 194 cont'd              |                              |                                                                       |                                                         |                                                                                                                                                              |                                                                                           | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of drainage) |
| 195                     | Temporary Possession         | 2316 square metres of<br>agricultural land (east of<br>A547 and A525) | Unknown<br>(in respect of mines and minerals)           | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                                                    | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)                                                                                                                                                                                                   |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                      | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                     |                                                                                                                        | Category 1                                                                                                       |                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                       |
|                         |                              |                                                                                                                     | Qualifying persons under Regulatio                                                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                 | Owners or Reputed Owners                                                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                 |
| 195 cont'd              |                              |                                                                                                                     | Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL                                                  |                                                                                                                  |                                                                                                                                                                                                                               | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                             |
| 196                     | Acquisition of new rights    | 612 square metres of<br>agricultural land (east of<br>A547 and north of<br>Pentre Lane) and<br>footpath (DE 206 46) | Unknown<br>(in respect of mines and minerals)<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                        | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 46)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of drainage) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                   |                                                                                                                        |                                                                                                                                                              |                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                   | Qualifying persons under Regulatio                                                                                     | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                               | Owners or Reputed Owners                                                                                               | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                  |  |
| 197                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 21252 square metres of<br>agricultural land and<br>hedgerow (east of A547)<br>footpaths (DE 206 20,<br>DE 206 46) | Unknown<br>(in respect of mines and minerals)<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                                                    | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 20<br>and DE 206 46)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of drainage) |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                      | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                          | n 7(1)(a))                                                                                                                                                                                   |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                     |                                                                                                                                               | Category 1                                                                                                       |                                                                                                                                                                                                                                   |                                                                                                                                                                                              |
|                         |                                                             |                                                                                                                                     | Qualifying persons under Regulatio                                                                                                            | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                               | Category 2                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                 | Owners or Reputed Owners                                                                                                                      | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                        |
| 198                     | Acquisition of new rights and<br>imposition of restrictions | 373 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre Lane and east of<br>A547) and footpath<br>(DE 206 20) | Unknown<br>(in respect of mines and minerals)<br>Wendy Rashid<br>14 Oaklands                                                                  | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                        | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 20)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr                                                        | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach |
|                         |                                                             |                                                                                                                                     | WORCESTER<br>Worcestershire<br>WR5 1SL                                                                                                        |                                                                                                                  | Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                                                                                                                                                                   | Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of drainage)                        |
| 199                     | Acquisition of new rights and<br>imposition of restrictions | 615 square metres of<br>agricultural land (north<br>of Pentre Lane and east<br>of A547) and footpath<br>(DE 206 20)                 | Action For Children<br>3 The Boulevard<br>Ascot Road<br>WATFORD<br>Hertfordshire<br>WD18 8AG<br>Unknown<br>(in respect of mines and minerals) | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                        | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE   206   20)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | NONE                                                                                                                                                                                         |

| Category 1 ) of the Infrastructure Planning (Applications: Prescribed Forms and Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ocedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| Lessees or Tenants Occupiers Occupiers Qualifying persons under Regulat<br>(Applications: Prescribed Forms a<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                |
| onesDenbighshire County CouncilAnwyl Construction Company Limr EsgobCounty HallAnwyl Housebed RoadWynnstay RoadClos Dewi SantAPHRUTHINEwloeghshireClwydDEESIDEDLHLL15 1YNConwy(in respect of footpath DE 206 20)CH5 3DT(in respect of rights stated in Trandated 01 December 2021)                                                                                                                                                                                                                                                          |
| odrhyddan Farming Company       Glyn Jones       Centrica PLC         yddan       Plas Yr Esgob       Millstream         yddan       Glascoed Road       Maidenhead Road         dlan       ST. ASAPH       WINDSOR         Denbighshire       Berkshire         SSB       LL17 0LH       SL4 5GD         (in respect of rights stated in Deed dated 17th May 1991)       Dwr Cymru Cyfyngedig Dwr Cymru Welsh Water         Linea       Fortran Road       St. Mellons         CARDIFF       CF3 0LT       (in respect of right to lay and |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                        |                                                                                                                        |                                                                                        |                                                                            |                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|----------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                        | Qualifying persons under Regulatio                                                                                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                          |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                    | Owners or Reputed Owners                                                                                               | Lessees or Tenants                                                                     | Occupiers                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                               |  |
| 200 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                        | Thomas Rowley-Conwy<br>Pentre Farm<br>Pentre Farm Lane<br>Rhuddian<br>RHYL<br>LL18 6LY                                 |                                                                                        |                                                                            |                                                                                                                                                                                                                                                                                                                                                                                     |  |
| 201                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 97 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre Lane and east of<br>A547) | Unknown<br>(in respect of mines and minerals)<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of drainage) |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                          | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                         | Qualifying persons under Regulatio                                                                                               | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                     | Owners or Reputed Owners                                                                                                         | Lessees or Tenants                                                                                               | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                               |
| 202                     | Acquisition of new rights and<br>imposition of restrictions | 122 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre Lane and east of<br>A547) | Unknown<br>(in respect of mines and minerals)<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL           | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                       | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of drainage) |
| 203                     | Acquisition of new rights and imposition of restrictions    | 8 square metres of<br>hedgerow (west of<br>Pentre Lane, Rhyl)                                           | James Vernon<br>The Hall<br>Lygan Y Wern<br>The Nant<br>Pentre Halkyn<br>HOLYWELL<br>Flintshire<br>CH8 8BD<br>(as reputed owner) | NONE                                                                                                             | Unknown                                                                                   | Unknown                                                                                                                                                                                                                                                                                                                                                                             |

|                         | Land which is p                                          | proposed to be subject to: (i)                                             | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                              | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights 1<br>of Denbighshire | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                            |
|-------------------------|----------------------------------------------------------|----------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                            |                                                                                                                                                                                                                                                                                                                                                                                   | Category 1                                                                                                          |                                                                                       |                                                                                                                                                       |
|                         |                                                          |                                                                            | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                               | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                           | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                        | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                          | Lessees or Tenants                                                                                                  | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 203 cont'd              |                                                          |                                                                            | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>(as reputed owner)<br>Ralph Collins<br>Carter Jonas<br>Canon Court North<br>Abbey Lawn<br>SHREWSBURY<br>SY2 5DE<br>(as reputed owner)<br>Unknown<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL<br>(as reputed owner) |                                                                                                                     |                                                                                       |                                                                                                                                                       |
| 204                     | Acquisition of new rights and imposition of restrictions | 2 square metre of<br>hedgerow (west of<br>Pentre Lane and east of<br>A547) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                                                                                                                                                                                                                                                    | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH                               | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | NONE                                                                                                                                                  |

|                         | Land which is p                                          | proposed to be subject to: (i)                                               | BOOK OF RE powers of compulsory acquisition, (ii)                                                                              | rm DEVELOPMENT CONSENT ORDER<br>:FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                            |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                              | Qualifying persons under Regulation                                                                                            | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                              | (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                          | Owners or Reputed Owners                                                                                                       | Lessees or Tenants                                                                                                   | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 204 cont'd              |                                                          |                                                                              |                                                                                                                                | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB                             |                                                                                       |                                                                                                                                                       |
| 205                     | Temporary Possession                                     | 10 square metres of<br>hedgerow (west of<br>Pentre Lane and east of<br>A547) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH                                | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | NONE                                                                                                                                                  |
|                         |                                                          |                                                                              |                                                                                                                                | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB                             |                                                                                       |                                                                                                                                                       |
| 206                     | Acquisition of new rights and imposition of restrictions | 10 square metres of<br>hedgerow (west of<br>Pentre Lane and east of<br>A547) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH                                | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | NONE                                                                                                                                                  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                              | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                            |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                             | Qualifying persons under Regulatio                                                                                                                              | Category 2                                                                                                        |                                                                                       |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                         | Owners or Reputed Owners                                                                                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 206 cont'd              |                                                             |                                                                             |                                                                                                                                                                 | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB                          |                                                                                       |                                                                                                                                                       |
| 207                     | Acquisition of new rights and<br>imposition of restrictions | 2 square metres of<br>hedgerow (west of<br>Pentre Lane and east of<br>A547) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB                                  | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH                             | Glyn Jones<br>Plas Yr Esgob<br>Glascoed Road<br>ST. ASAPH<br>Denbighshire<br>LL17 OLH | NONE                                                                                                                                                  |
|                         |                                                             |                                                                             |                                                                                                                                                                 | The Bodrhyddan Farming Company<br>Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB                          |                                                                                       |                                                                                                                                                       |
| 208                     | Acquisition of new rights and imposition of restrictions    | 4 square metres of<br>hedgerow (west of<br>Pentre Lane and east of<br>A547) | Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>(as reputed owner)<br>Unknown | NONE                                                                                                              | Unknown                                                                               | Unknown                                                                                                                                               |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                             | BOOK OF RE powers of compulsory acquisition, (ii)                                                                      | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                            |                                                                                                                        | Category 1                                                                                                        |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|                         |                                                             |                                                                                            | Qualifying persons under Regulatic                                                                                     | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                        | Owners or Reputed Owners                                                                                               | Lessees or Tenants                                                                                                | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                |
| 208 cont'd              |                                                             |                                                                                            | Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL<br>(as reputed owner)                            |                                                                                                                   |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| 209                     | Acquisition of new rights and<br>imposition of restrictions | 3106 square metres of<br>agricultural land (east of<br>A547 and north of Afon<br>Ffyddion) | Unknown<br>(in respect of mines and minerals)<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                        | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of drainage) |

|                         | Land which is p              | proposed to be subject to: (i)                                | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                |
|-------------------------|------------------------------|---------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                               | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                   | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                             | Lessees or Tenants                                                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009     |
| 210                     | Temporary Possession         | 39 square metres of<br>verge of public road<br>(east of A547) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Owain Grenville Rowley-Conwy<br>Stone House<br>Bodrhyddan Hall<br>Dyserth Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5SB<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Unknown |
| 211                     | Temporary Possession         | 19 square metres of<br>verge of public road<br>(east of A547) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                          | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                           | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)            |

|                         | Land which is j              | proposed to be subject to: (i)                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|--------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                | Owners or Reputed Owners                                                                                                                                                                                                                                                    | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                         |
| 212                     | Temporary Possession         | 751 square metres of<br>grassed area public road<br>(east of A547) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                    | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                   | Qualifying persons under Regulatio                                                                                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                               | Owners or Reputed Owners                                                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                 |
| 212 cont'd              |                              |                                                                                   |                                                                                                                        |                                                                                                                  |                                                                                                                                                                                                                               | Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications<br>apparatus)                                                                                                                                                                                                                                         |
| 213                     | Acquisition of new rights    | 10 square metres of<br>grassed area (east of<br>A547) and footpath<br>(DE 206 46) | Unknown<br>(in respect of mines and minerals)<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                        | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 46)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media<br>and right to enter)<br>Unknown<br>(in respect of right of drainage) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                   | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                                   |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                  | Qualifying persons under Regulatio                                                                                                                                                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                           | g (Applications: Prescribed Forms and                                                                                | Category 2                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                              | Owners or Reputed Owners                                                                                                                                                                                                                                                    | Lessees or Tenants                                                                                              | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                        |
| 214                     | Acquisition of new rights                                | 17 square metres of<br>public road (east of<br>A547)             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |
| 215                     | Temporary Possession                                     | 101 square metres of<br>grassed area (east of<br>A547)           | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                                  | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                           | NONE                                                                                                                                                                                                                                                                                                         |
| 216                     | Acquisition of new rights and imposition of restrictions | 1449 square metres of<br>public road and verge<br>(east of A525) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                                                                        | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                                                                                                               |

|                         | Land which is p                                          | proposed to be subject to: (i)                                  | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|-------------------------|----------------------------------------------------------|-----------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                 | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                             | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                                               | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                          |
| 216 cont'd              |                                                          |                                                                 | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr Bach<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown<br>Wendy Rashid<br>14 Oaklands<br>WORCESTER<br>Worcestershire<br>WR5 1SL<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                                  |                                                                                            | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |
| 217                     | Acquisition of new rights and imposition of restrictions | 2828 square metres of<br>grassed area and public<br>road (A525) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                                                                                                                                                                                                                                                            | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                                                                                                                                                                                                                                                                                                |

|                         | Land which is p              | proposed to be subject to: (i                                   | BOOK OF RE<br>) powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire                                              | to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                                                                                |  |
|-------------------------|------------------------------|-----------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                              |                                                                 | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                             | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                                                                                                                                                                                                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                     |  |
| 218                     | Temporary Possession         | 7 square metres of<br>grassed and wooded<br>area (west of A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                                                                         | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL16 22D | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 26th February 1992)<br>Unknown<br>(in respect of restrictive covenants) |  |

|                         | Land which is p              | proposed to be subject to: (i)                                        | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                        | rm DEVELOPMENT CONSENT ORDER<br>:FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                                                                                                                 | n 7(1)(a))                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|-----------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                       |                                                                                                                                                                                          | Category 1                                                                                                         |                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                        |
|                         |                              |                                                                       | Qualifying persons under Regulation                                                                                                                                                      | on 7(1)(a) of the Infrastructure Plannin;<br>Procedures) Regulations 2009                                          | g (Applications: Prescribed Forms and                                                                                                                                                    | Category 2                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                   | Owners or Reputed Owners                                                                                                                                                                 | Lessees or Tenants                                                                                                 | Occupiers                                                                                                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                  |
| 219                     | Temporary Possession         | 528 square metres of<br>footway and verge (west<br>of A525)           | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                               | NONE                                                                                                               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                               | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)       |
| 220                     | Acquisition of new rights    | 317 square metres of<br>agricultural land and<br>track (west of A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                               | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 26th February 1992)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p                                                                                                              | proposed to be subject to: (i)                              | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights 1<br>f Denbighshire | o carry out protective works (Regulatio                                                                                          | n 7(1)(a))                                                                                                                                                       |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and |                                                             |                                                                                                                                 |                                                                                                                  | (Applications: Prescribed Forms and                                                                                              | Category 2                                                                                                                                                       |
|                         |                                                                                                                              |                                                             |                                                                                                                                 | Procedures) Regulations 2009                                                                                     | 1                                                                                                                                | Qualifying persons under Regulation                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                 | Description of land                                         | Owners or Reputed Owners                                                                                                        | Lessees or Tenants                                                                                               | Occupiers                                                                                                                        | 7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                   |
| 220 cont'd              |                                                                                                                              |                                                             | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD                                               |                                                                                                                  | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD                                                | Unknown<br>(in respect of restrictive covenants)                                                                                                                 |
|                         |                                                                                                                              |                                                             | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD                                          |                                                                                                                  | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD                                           |                                                                                                                                                                  |
|                         |                                                                                                                              |                                                             | Unknown<br>(in respect of mines and minerals)                                                                                   |                                                                                                                  |                                                                                                                                  |                                                                                                                                                                  |
| 221                     | Temporary Possession                                                                                                         | 245 square metres of<br>agricultural land (west of<br>A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Gwadr Fawr | NONE                                                                                                             | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cuurbr Fawr | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 26th February 1992)<br>Unknown |
|                         |                                                                                                                              |                                                             | Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD                                                             |                                                                                                                  | Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD                                                              | (in respect of restrictive covenants)                                                                                                                            |

|                         | Land which is p              | proposed to be subject to: (i)                            | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                           |                                                                                                                                                                                                                              | Category 1                                                                                                        |                                                                                                                                                                             |                                                                                                                                                       |
|                         |                              |                                                           | Qualifying persons under Regulatio                                                                                                                                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                       | Owners or Reputed Owners                                                                                                                                                                                                     | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 221 cont'd              |                              |                                                           | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                   | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD |                                                                                                                                                       |
| 222                     | Acquisition of new rights    | 1 square metre of track<br>(west of A525)                 | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                   | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)         |
| 223                     | Temporary Possession         | 3 square metres of<br>footway and track (west<br>of A525) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                   | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                  | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)       |

|                         | Land which is p              | proposed to be subject to: (i)                             | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                       |
|-------------------------|------------------------------|------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                            |                                                                                                                             | Category 1                                                                                                       |                                                                                            |                                                                                                                                                                                                                                                                                                  |
|                         |                              |                                                            | Qualifying persons under Regulatio                                                                                          | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                        | Owners or Reputed Owners                                                                                                    | Lessees or Tenants                                                                                               | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                            |
| 224                     | Temporary Possession         | 1 square metre of<br>footway (west of A525)                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown | NONE                                                                                                             | Unknown                                                                                    | Unknown                                                                                                                                                                                                                                                                                          |
| 225                     | Acquisition of new rights    | 3 square metres of<br>footway (west of A525)               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown | NONE                                                                                                             | Unknown                                                                                    | Unknown                                                                                                                                                                                                                                                                                          |
| 226                     | Acquisition of new rights    | 25 square metres of<br>footway and track (west<br>of A525) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                  | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights t<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                   | Qualifying persons under Regulatio                                                                                          | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                            |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                               | Owners or Reputed Owners                                                                                                    | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 227                     | Temporary Possession         | 4 square metres of<br>footway (west of A525)                      | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Unknown | NONE                                                                                                                                                         | Unknown                                                                                    | Unknown                                                                                                                                               |
| 228                     | Temporary Possession         | 16 square metres of<br>footway and grassed<br>area (west of A525) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                  | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)       |
| 229                     | Temporary Possession         | 10 square metres of<br>grassed area (west of<br>A525)             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                  | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)       |
| 230                     | Temporary Possession         | 2 square metre of<br>grassed area (west of<br>A525)               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                  | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | NONE                                                                                                                                                  |

|                         | Land which is j              | proposed to be subject to: (i                       | BOOK OF RE<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) right<br>of Denbighshire                               | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                       |
|                         |                              |                                                     | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Lessees or Tenants                                                                                                                             | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 231                     | Temporary Possession         | 1 square metre of<br>grassed area (west of<br>A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(as reputed owner)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(as reputed owner)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL16 4RD<br>(as reputed owner)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Unbighshire<br>LL18 2YD<br>(as reputed owner)<br>Unknown | NONE                                                                                                                                           | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown | Unknown                                                                                                                                               |

|                         | Land which is p              | proposed to be subject to: (i                          | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire                                               | to carry out protective works (Regulatio                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|--------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                        | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                      |                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                    | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                        |
| 232                     | Temporary Possession         | 58 square metres of<br>verge and public road<br>(A525) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                        |                                                                                                                                                                                                                                                                                             |                    |                                                                                                                                                            |                                                                                                                                                                                                                                                                                                   |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                        | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                |                    |                                                                                                                                                            | Category 2                                                                                                                                                                                                                                                                                        |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                    | Owners or Reputed Owners                                                                                                                                                                                                                                                                    | Lessees or Tenants | Occupiers                                                                                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                             |  |  |  |
| 232 conť d              |                                                                                                                                                                                                                                                                                                      |                                                                        | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                                                                                                                     |                    |                                                                                                                                                            |                                                                                                                                                                                                                                                                                                   |  |  |  |
| 233                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 105 square metres of<br>public road (A525) and<br>footpath (DE 206 42) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway) | NONE               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>(in respect of footpath DE 206 42) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)            | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                              | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 1                                                                                                                                     |                                                                                                                      |                                                                                                                                                       |
|                         |                              |                                           | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                      |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                                                                             | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 233 cont'd              |                              |                                           | Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                                                |                                                                                                                      | Unknown                                                                                                                                               |
| 234                     | Temporary Possession         | 97 square metres of<br>public road (A525) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                                                                                                                                                                                                                                                                                                  | NONE                                                                                                                                           | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)        |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) rig                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|-------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Category 1                                                                                                    |                                         |                                                                                                                                                                                                                                                                                                               |
|                         |                              |                               | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                         |
| 234 cont'd              |                              |                               | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                               |                                         | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |

|                         | Land which is p              | proposed to be subject to: (i,                                                      | BOOK OF RE<br>powers of compulsory acquisition, (ii) (                                                                                                                                                                                                                                                                                                                                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                        | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Category 1                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                         |                              |                                                                                     | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                                                                                                                                                                         | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                         | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                            | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Lessees or Tenants                                                                                              | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                           |
| 234 cont'd              |                              |                                                                                     | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 235                     | Acquisition of new rights    | 1159 square metres of<br>private road (west of<br>A525) and footpath<br>(DE 206 42) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(as reputed owner)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(as reputed owner)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner) | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 42)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD | Aaron Haldane<br>The Maisonette<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Ashworth<br>14 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Roberts<br>Flat 1<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                               | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
| 235 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Unknown                            |                                                                                        | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown | Helen McCahill<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>lan Towers<br>11 Cwybr Cottages<br>Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Nixon<br>Flat 3<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Nixon<br>Flat 3<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| 235 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        |                                     | James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulation                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                      | (Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                          | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                        |
| 235 cont'd              |                              |                               |                                                         |                                                                                                             |                                          | Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Martin Nixon<br>Flat 4<br>Cwybr House<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                              |  |  |
| 235 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Peter Baddeley<br>2 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                        | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                             | rm DEVELOPMENT CONSENT ORDER<br>:FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire                                           | s to carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                       |
|-------------------------|------------------------------|-------------------------------------------------------|--------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                       | Qualifying persons under Regulation                                                              | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                   | Owners or Reputed Owners                                                                         | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                            |
| 235 cont'd              |                              |                                                       |                                                                                                  |                                                                                                                                                              |                                                                                                  | Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 GJP<br>(in respect of right of access) |
| 236                     | Acquisition of new rights    | 14 square metres of<br>private road (west of<br>A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | NONE                                                                                                                                                         | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 26th February 1992)                                                                                                                                                                                                                            |

|                         | Land which is p              | proposed to be subject to: (i)                               | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights 1<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                             | n 7(1)(a))                                                                                                                                                                                          |
|-------------------------|------------------------------|--------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                              | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                         |                                                                                                                   |                                                                                                                                                                                                                                                                     | Category 2                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                             | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                               |
| 236 cont'd              |                              |                                                              | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                   | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown<br>(in respect of restrictive covenants) |
| 237                     | Temporary Possession         | 8537 square metres of<br>agricultural land (west of<br>A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH                                                                                                                                                                                                                     | NONE                                                                                                              | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH                                                                                                                                                                    | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 26th February 1992)                                               |

|                         | Land which is p              | proposed to be subject to: (i)                                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                                                                                                                                                                            | n 7(1)(a))                                                                                                                                                                                        |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         | Category 1                   |                                                                                   |                                                                                                                                                                                                                                                                                                                           |                                                                                                                 |                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                   |
|                         |                              |                                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and                                                                                                                                                                                                                                 | Category 2                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                              | Occupiers                                                                                                                                                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                             |
| 237 cont'd              |                              |                                                                                   | Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                 | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants) |
| 238                     | Acquisition of new rights    | 18 square metres of<br>private road (west of<br>A525) and footpath<br>(DE 206 42) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(as reputed owner)                                                                                                                                                                                                    | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 42)                                                                                                                                    | Aaron Haldane<br>The Maisonette<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                      |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REFE<br>) powers of compulsory acquisition, (ii) rig                                                                                                                                                                                                                                                                                     | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                 | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              |                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                                |                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                              | 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                     | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                        | Occupiers                                                                                                                                                                                                                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                    |
| 238 cont'd              |                              |                              | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(as reputed owner)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Unknown |                                                                                                           | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown | Anthony Ashworth<br>14 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Roberts<br>Flat 1<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Helen McCahill<br>13 Cwybr Cottages |
|                         |                              |                              |                                                                                                                                                                                                                                                                                                                                                  |                                                                                                           |                                                                                                                                                                                                                                                                                            | Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| 238 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | lan Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Nixon<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Shaw<br>1 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Shaw<br>1 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|---------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                         | Category 1                                                                                                       |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                         |                              |                                | Qualifying persons under Regulatio                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 238 cont'd              |                              |                                |                                                         |                                                                                                                  |                                         | James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |
| 238 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Martin Nixon<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|--------------------------------|---------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                         | Category 1                                                                                                      |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|                         |                              |                                | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                | Lessees or Tenants                                                                                              | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                  |
| 238 cont'd              |                              |                                |                                                         |                                                                                                                 |                                         | Peter Baddeley<br>2 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights granted by Deed<br>of grant dated 20 December 2017)<br>Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | to carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|------------------------------|--------------------------------|---------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                | Qualifying persons under Regulatio                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                | Lessees or Tenants                                                                                               | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                     |
| 238 cont'd              |                              |                                |                                                         |                                                                                                                  |                                          | Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 GJP<br>(in respect of right of access)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                      | BOOK OF RE<br>powers of compulsory acquisition, (ii) i                                                                                                                                                                                                                                                                                                                                                                                                                        | TT DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                          | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                     | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                           | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                           |
| 239                     | Acquisition of new rights    | 349 square metres of<br>private road (west of<br>A525 and north-east of<br>Cwybr-uchaf) and<br>footpath (DE 206 42) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(as reputed owner)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(as reputed owner)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 42)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD | Aaron Haldane<br>The Maisonette<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Ashworth<br>14 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Roberts<br>Flat 1<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                               | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
| 239 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Unknown                            |                                                                                        | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown | Helen McCahill<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>lan Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Nixon<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Nixon<br>Flat 3<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| 239 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | James Keith Shaw<br>1 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                              |  |  |
| 239 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Martin Nixon<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
| 239 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        |                                     | Peter Baddeley<br>2 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                         | BOOK OF RE powers of compulsory acquisition, (ii)                                                                      | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                                                                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                        |                                                                                                                        | Category 1                                                                                                          |                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                              |
|                         |                              |                                                                                                                                                        | Qualifying persons under Regulatic                                                                                     | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                           | (Applications: Prescribed Forms and                                                                                                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                    | Owners or Reputed Owners                                                                                               | Lessees or Tenants                                                                                                  | Occupiers                                                                                                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                        |
| 239 cont'd              |                              |                                                                                                                                                        |                                                                                                                        |                                                                                                                     |                                                                                                                                                                            | Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 6JP<br>(in respect of right of access)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |
| 240                     | Acquisition of new rights    | 933 square metres of<br>private road (west of<br>A525 and north of<br>Cwybr-uchaf) and<br>footpaths (DE 206 24,<br>DE 206 25, DE 206 31,<br>DE 206 42) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(as reputed owner) | NONE                                                                                                                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 24,<br>DE 206 25, DE 206 31 and<br>DE 206 42) | Aaron Haldane<br>The Maisonette<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                                                 |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REFE<br>) powers of compulsory acquisition, (ii) rig                                                                                                                                                                                                                                                                                     | DEVELOPMENT CONSENT ORDER<br>RENCE - PART 1<br>sht to use the land, and/or (iii) right<br>Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                 | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              |                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                             |                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                              | 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                  | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                     | Occupiers                                                                                                                                                                                                                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                    |
| 240 cont'd              |                              |                              | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(as reputed owner)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Unknown |                                                                                                        | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown | Anthony Ashworth<br>14 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Roberts<br>Flat 1<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Helen McCahill<br>13 Cwybr Cottages |
|                         |                              |                              |                                                                                                                                                                                                                                                                                                                                                  |                                                                                                        |                                                                                                                                                                                                                                                                                            | Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                             |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                       |                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                    | Occupiers                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| 240 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                       | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | lan Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Nixon<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Shaw<br>1 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Shaw<br>1 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| 240 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        |                                     | James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulation                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                      | (Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                          | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                        |
| 240 cont'd              |                              |                               |                                                         |                                                                                                             |                                          | Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Martin Nixon<br>Flat 4<br>Cwybr House<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |
| 240 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Peter Baddeley<br>2 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights granted by Deed<br>of grant dated 20 December 2017)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulatio                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| 240 cont'd              |                              |                               |                                                         |                                                                                                                  |                                         | Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Susan Davies<br>5 Cwybr Cottages<br>Cwybr Cottages<br>Cwybr Cottages<br>Cwybr Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 6JP<br>(in respect of right of access) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                            | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                          | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                           | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                     | Category 1<br>on 7(1)(a) of the Infrastructure Planning                                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                               | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                | Procedures) Regulations 2009<br>Lessees or Tenants                                                              | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                  |
| 241                     | Acquisition of new rights    | 1126 square metres of<br>private road,<br>hardstanding and<br>agricultural land (east of<br>Cwybr-uchaf and north<br>of A525) and footpath<br>(DE 206 24) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 24)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD | Aaron Haldane<br>The Maisonette<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Ashworth<br>14 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Roberts<br>Flat 1<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 26th February 1992) |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) righ<br>Denbighshire | s to carry out protective works (Regulatio                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|-------------------------|------------------------------|------------------------------|-----------------------------------------------------------|----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulation                       | Category 1<br>7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                       | ng (Applications: Prescribed Forms and                                                 | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                       | Occupiers                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 241 cont'd              |                              |                              | Unknown<br>(in respect of mines and minerals)             |                                                                                                          | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Helen McCahill<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|------------------------------------------------------------------------|-------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                             |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
| 241 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                        |                                     | James Keith Nixon<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Massey<br>Flat 2<br>Cwybr House<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|--------------------------------|---------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                         | Category 1                                                                                                       |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                         |                              |                                | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 241 cont'd              |                              |                                |                                                         |                                                                                                                  |                                         | Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Martin Nixon<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Martin Nixon<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
| 241 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                       |  |  |
| 241 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain stated in deed dated 20<br>December 2017)<br>Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access) |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                        | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                              |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                       | Qualifying persons under Regulatic                                                                                                                                                       | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                                                                                                                                                   | Category 2                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                   | Owners or Reputed Owners                                                                                                                                                                 | Lessees or Tenants                                                                                                 | Occupiers                                                                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                   |
| 241 cont'd              |                                                             |                                                                                                                                       |                                                                                                                                                                                          |                                                                                                                    |                                                                                                                                                                                                                                                       | Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 6JP<br>(in respect of right of access)                                                     |
| 242                     | Acquisition of new rights and<br>imposition of restrictions | 43453 square metres of<br>agricultural land (north<br>of A525 and south of<br>Cwybr-uchaf) and<br>footpaths (DE 206 23,<br>DE 206 24) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 23<br>and DE 206 24)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 26th February 1992)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                               | n 7(1)(a))                                                                                                                                                                                    |
|-------------------------|------------------------------|------------------------------|----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulation                                                    | Category 1<br>n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                          | g (Applications: Prescribed Forms and                                                  | Category 2                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                                             | Occupiers                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                         |
| 242 cont'd              |                              |                              | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD      |                                                                                                                | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD   | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain stated in deed dated 20<br>December 2017) |
|                         |                              |                              | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD |                                                                                                                | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                 |
|                         |                              |                              | Unknown<br>(in respect of mines and minerals)                                          |                                                                                                                | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | Unknown<br>(in respect of restrictive covenants)                                                                                                                                              |

|                         | Land which is p              | proposed to be subject to: (i                                     | BOOK OF RE<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                  | rm DEVELOPMENT CONSENT ORDER<br>:FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire           | to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
|-------------------------|------------------------------|-------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                              |                                                                   | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                     | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and |                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                | Procedures) Regulations 2009<br>Lessees or Tenants                                                                           | Occupiers                                                                                                                                                                                                                                                                                                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 243                     | Acquisition of new rights    | 869 square metres of<br>private road and track<br>(north of A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                                         | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | Aaron Haldane<br>The Maisonette<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Ashworth<br>14 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Roberts<br>Flat 1<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 26th February 1992) |  |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulation                       | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 243 cont'd              |                              |                               | Unknown<br>(in respect of mines and minerals)             |                                                                                                                  |                                         | Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Helen McCahill<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Phyle<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |
| 243 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | James Keith Nixon<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF powers of compulsory acquisition, (ii) rig | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|------------------------------|-------------------------------|--------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                        | Category 1                                                                                                    |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                         |                              |                               | Qualifying persons under Regulation                    | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                               | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                              |
| 243 cont'd              |                              |                               |                                                        |                                                                                                               |                                         | Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr House<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Martin Nixon<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REI<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulatio                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                    |
| 243 cont'd              |                              |                               |                                                         |                                                                                                                  |                                         | Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                       |  |  |
| 243 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain stated in deed dated 20<br>December 2017)<br>Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access) |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                               | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                     | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                                 | n 7(1)(a))                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                              | Qualifying persons under Regulatio                                                                                                                                                       | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                             | (Applications: Prescribed Forms and                                                                                                                                                                                                      | Category 2                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                          | Owners or Reputed Owners                                                                                                                                                                 | Lessees or Tenants                                                                                                  | Occupiers                                                                                                                                                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                               |
| 243 cont'd              |                              |                                                                                                              |                                                                                                                                                                                          |                                                                                                                     |                                                                                                                                                                                                                                          | Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 6JP<br>(in respect of right of access) |
| 244                     | Acquisition of new rights    | 731 square metres of<br>track(west of Cwybr-<br>uchaf and north-west of<br>A525) and footpath<br>(DE 206 31) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE   206   31)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed<br>dated 26th February 1992)<br>Unknown<br>(in respect of restrictive covenants)                                           |

|                         | Land which is p              | proposed to be subject to: (i)                                                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                  | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                                                                      | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                   | Qualifying persons under Regulatio                                                                                                      | Category 1<br>n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                           | g (Applications: Prescribed Forms and                                                                                                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                               | Owners or Reputed Owners                                                                                                                | Lessees or Tenants                                                                                              | Occupiers                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 244 cont'd              |                              |                                                                                                   | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan |                                                                                                                 | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire |                                                                                                                                                       |
|                         |                              |                                                                                                   | RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown<br>(in respect of mines and minerals)                                                       |                                                                                                                 | LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD                                                            |                                                                                                                                                       |
| 245                     | Acquisition of new rights    | 5 square metres of<br>agricultural land (south-<br>west of Cwybr-uchaf and<br>north-west of A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(as reputed owner)                  | NONE                                                                                                            | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH                                                              | Unknown                                                                                                                                               |

|                         | Land which is p              | proposed to be subject to: (i)                                                                      | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights 1<br>f Denbighshire            | o carry out protective works (Regulatio                                                                                                                                                                                                                                        | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                               | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and |                                                                                                                                                                                                                                                                                |                                                                                                                                                       |
|                         |                              |                                                                                                     |                                                                                                                                                                                                                                                                                                                                                  | Procedures) Regulations 2009                                                                                                 |                                                                                                                                                                                                                                                                                |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                                           | Occupiers                                                                                                                                                                                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 245 cont'd              |                              |                                                                                                     | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(as reputed owner)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Unknown |                                                                                                                              | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown |                                                                                                                                                       |
| 246                     | Acquisition of new rights    | 535 square metres of<br>agricultural land (south-<br>west of Cwybr-uchaf and<br>north-west of A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH                                                                                                                                                                                                                                                 | NONE                                                                                                                         | Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY                                                                                                                                                                                                                         | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of drainage)                                 |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri                                                                                                                                                                                                                                                           | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulatior                                                                                                                                                                                                                                                                                  | Category 1<br>n 7(1)(a) of the Infrastructure Planning (<br>Procedures) Regulations 2009                      | Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                             | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 246 cont'd              |                              |                              | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown<br>(in respect of mines and minerals) |                                                                                                               |                                         | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of right of drainage)<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of drainage)<br>Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>(in respect of right of drainage)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                              | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulation               | n 7(1)(a))                                                                                                                                            |
|-------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                             | Qualifying persons under Regulatio                                                               | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                    | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                         | Owners or Reputed Owners                                                                         | Lessees or Tenants                                                                                                | Occupiers                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 246 cont'd              |                                                          |                                                                                                                             |                                                                                                  |                                                                                                                   |                                                        | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)       |
| 247                     | Acquisition of new rights and imposition of restrictions | 14555 square metres of<br>agricultural land and<br>watercourse (Afon<br>Ffyddion, north of River<br>Clwyd and west of A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 SUH | NONE                                                                                                              | Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of drainage)                                 |
|                         |                                                          |                                                                                                                             | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD             |                                                                                                                   |                                                        | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of right of drainage)                        |
|                         |                                                          |                                                                                                                             | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD                |                                                                                                                   |                                                        | Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of drainage)                                         |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri                                                                              | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights t<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|------------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              |                                                                                                                                         | Category 1                                                                                                   |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                         |                              |                              | Qualifying persons under Regulation                                                                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                     | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                | Lessees or Tenants                                                                                           | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 247 cont'd              |                              |                              | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown<br>(in respect of mines and minerals) |                                                                                                              |                                         | Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>(in respect of right of drainage)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain electricity cables as<br>contained in a Deed dated 20th<br>December 2017)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                         | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                        | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                        | Qualifying persons under Regulation                                                                                                                                                      | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                             | (Applications: Prescribed Forms and                    | Category 2                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                    | Owners or Reputed Owners                                                                                                                                                                 | Lessees or Tenants                                                                                                  | Occupiers                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                        |
| 247 cont'd              |                              |                                                                        |                                                                                                                                                                                          |                                                                                                                     |                                                        | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for<br>maintenance as contained in an<br>Agreement dated 19th June 1952) |
| 248                     | Temporary Possession         | 7057 square metres of<br>agricultural land (north<br>of Afon Ffyddion) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                                | Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of drainage)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of right of drainage)                                                                                                                      |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                                                                                                                                                              |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation                                                                                                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                                                                                                                                                     | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                               |  |  |
| 248 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Unknown<br>(in respect of mines and minerals) |                                                                                        |                                     | Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of drainage)<br>Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>(in respect of right of drainage)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                             | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                          | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                            |                                                                                  | Category 1                                                                                                       |                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                         |                              |                                                                                                                            | Qualifying persons under Regulatio                                               | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                              | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                        | Owners or Reputed Owners                                                         | Lessees or Tenants                                                                                               | Occupiers                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 248 cont'd              |                              |                                                                                                                            |                                                                                  |                                                                                                                  |                                                                                  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain electricity cables as<br>contained in a Deed dated 20th<br>December 2017)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Wales and West Utilities Limited<br>Wales & Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for<br>maintenance as contained in an<br>Agreement dated 19th June 1952) |
| 249                     | Temporary Possession         | 5304 square metres of<br>agricultural land and<br>watercourse (Afon<br>Ffyddion, north of River<br>Clwyd and west of A525) | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR | NONE                                                                                                             | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                               | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                          | (Applications: Prescribed Forms and                                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                              | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                               |
| 249 cont'd              |                              |                              | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB |                                                                                                                 | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                                                                   | BOOK OF RE powers of compulsory acquisition, (ii) r                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights 1<br>f Denbighshire | to carry out protective works (Regulatio                                                                                         | n 7(1)(a))                                                                                                                                                              |
|-------------------------|----------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                                                                  |                                                                                           | Category 1                                                                                                       |                                                                                                                                  |                                                                                                                                                                         |
|                         |                                                          |                                                                                                                                                                                  | Qualifying persons under Regulatio                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | g (Applications: Prescribed Forms and                                                                                            | Category 2                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                                                              | Owners or Reputed Owners                                                                  | Lessees or Tenants                                                                                               | Occupiers                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                   |
| 249 cont'd              |                                                          |                                                                                                                                                                                  | Unknown<br>(in respect of mines and minerals)                                             |                                                                                                                  |                                                                                                                                  | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain pipeline) |
| 250                     | Acquisition of new rights and imposition of restrictions | 21814 square metres of<br>watercourse (Afon<br>Ffyddion), hedgerow,<br>agricultural land,<br>footway and river bank<br>(north of River Clwyd)<br>and footpath<br>(DE   206   29) | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR          | NONE                                                                                                             | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR                                                 | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)                                   |
|                         |                                                          |                                                                                                                                                                                  | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB |                                                                                                                  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 29) | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)                                               |
|                         |                                                          |                                                                                                                                                                                  | Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY                  |                                                                                                                  | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB                                        | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)                                                  |

|                         | Land which is p              | proposed to be subject to: (i)                                                          | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                  | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                          | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                         |                                                                                                                                                                                                          | Category 1                                                                                                        |                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                         |                              |                                                                                         | Qualifying persons under Regulatio                                                                                                                                                                       | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                              | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                     | Owners or Reputed Owners                                                                                                                                                                                 | Lessees or Tenants                                                                                                | Occupiers                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                   |
| 250 cont'd              |                              |                                                                                         | Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                   |                                                                                  | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain pipeline) |
| 251                     | Temporary Possession         | 264 square metres of<br>agricultural land (north<br>of River Clwyd and west<br>of A525) | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR                                                                                                                         | NONE                                                                                                              | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                                                   |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri                                                                                                                                                                                                                                                                       | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) right:<br>Denbighshire | s to carry out protective works (Regulatio                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                              | Category 2                                                                                                  |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                         | Procedures) Regulations 2009<br>Lessees or Tenants                                                          | Occupiers                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                         |
| 251 cont'd              |                              |                              | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB |                                                                                                             | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain pipeline) |
|                         |                              |                              | Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                                                                                                                    |                                                                                                             |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |

|                         | Land which is p              | proposed to be subject to: (i                                                                                  | BOOK OF RI<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                    | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) right<br>of Denbighshire | s to carry out protective works (Regulatio                                                                                                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                |                                                                                                                                                                                                                                                           | Category 1                                                                                                       |                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                         |                              |                                                                                                                | Qualifying persons under Regulation                                                                                                                                                                                                                       | on 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                          | ng (Applications: Prescribed Forms and                                                                                                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                            | Owners or Reputed Owners                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                             |
| 252                     | Acquisition of new rights    | 726 square metres of<br>agricultural land,<br>hedgerow and track<br>(north of River Clwyd<br>and west of A525) | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY | NONE                                                                                                             | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of way)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage) |
|                         |                              |                                                                                                                | Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL                                                                                                                                                                               |                                                                                                                  |                                                                                                                                                                               | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                            |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                          |                                                                                                                           |                                                                          |                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                          |                                                                                                                           | Category 1                                                               |                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                          | Qualifying persons under Regulatio                                                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                         | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                      | Owners or Reputed Owners                                                                                                  | Lessees or Tenants                                                       | Occupiers                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                   |  |  |  |
| 252 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                          | Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>Unknown<br>(in respect of mines and minerals) |                                                                          |                                                                                                             | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain pipeline) |  |  |  |
| 253                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 1675 square metres of<br>track (north of River<br>Clwyd) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT               | NONE                                                                     | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of way)                                                                                                                                                                                                                                                                                                                                        |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire            | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|------------------------------|------------------------------|-----------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                              |                              | Qualifying persons under Regulation                       | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|                         |                              |                              |                                                           | Procedures) Regulations 2009                                                                                                 |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                                           | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 253 cont'd              |                              |                              | Unknown<br>(in respect of mines and minerals)             |                                                                                                                              |                                         | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of right of way)<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of way)<br>Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>(in respect of right of way)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>(in respect of right of way)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                 |                                                                                                                                                                                                                                                           |                                                                                         |                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                 | Qualifying persons under Regulation                                                                                                                                                                                                                       | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                             | Owners or Reputed Owners                                                                                                                                                                                                                                  | Lessees or Tenants                                                                      | Occupiers                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                          |  |  |  |
| 254                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 77 square metres of<br>track (north of A525 and<br>River Clwyd) | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY | NONE                                                                                    | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of way)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage) |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri                                                                                                                                               | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights t<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              |                                                                                                                                                                                                          | Category 1                                                                                                   |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                     | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                 | Lessees or Tenants                                                                                           | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                         |
| 254 cont'd              |                              |                              | Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>Unknown<br>(in respect of mines and minerals) |                                                                                                              |                                         | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain pipeline) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                          |                                                                          |                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                          | Category 1                                                               |                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                                       | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                            | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                 | Lessees or Tenants                                                       | Occupiers                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 255                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 52 square metres of<br>track and grassed area<br>(north of River Clwyd<br>and south-east of A525)<br>(excluding all interests of<br>the Crown) | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL | NONE                                                                     | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2VH<br>(in respect of right of drainage)<br>(in respect of right of drainage)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                                                                 | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>F Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|------------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulatio                                                                                        | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                  | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 255 cont'd              |                              |                              | Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                  |                                         | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>The King's Most Excellent Majesty In<br>Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(in respect of annual Crown rent)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain pipeline) |

|                         | Land which is <sub>l</sub>   | proposed to be subject to: (i)                                                    | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                   | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                                          | to carry out protective works (Regulatio                                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                   | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                                                                                                  | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                |
| 256                     | Acquisition of new rights    | 290 square metres of<br>track (north of River<br>Clwyd and south-east of<br>A525) | Caroline Teresa Burnside<br>Riverside Cottage<br>Tan Yr Eglwys Road<br>Rhuddlan<br>Y RHYL<br>LL18 2UU<br>(as reputed owner)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(as reputed owner)<br>Joseph Lee Gorman<br>Riverside Cottage<br>Tan Yr Eglwys Road<br>Rhuddlan<br>Y RHYL<br>LL18 2UU<br>(as reputed owner)<br>Unknown | NONE                                                                                                                                                         | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>Unknown | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of access)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of right of access)<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of access) |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|------------------------------|------------------------------|---------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulation                                                                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                      |
| 256 cont'd              |                              |                              | William Gareth Davies<br>Morawel<br>Gwindy Street<br>Rhuddlan<br>RHYL<br>LL18 2US<br>(as reputed owner) |                                                                                                                  |                                         | Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 GHL<br>(in respect of right of access)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications<br>apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                    | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                         | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire                                                                                                                                                                                        | to carry out protective works (Regulatio                                                                                                                                                                                                                                                                 | n 7(1)(a))                                                                                                                                                                                                                                                                                       |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         | Category 1                                                  |                                                                                                                                                   |                                                                                                                                           |                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                  |
|                         |                                                             |                                                                                                                                                   | Qualifying persons under Regulation                                                                                                       | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                                                                                                                                                | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                      | Category 2                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                               | Owners or Reputed Owners                                                                                                                  | Lessees or Tenants                                                                                                                                                                                                                                                                                       | Occupiers                                                                                                                                                                                                                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                            |
| 257                     | Acquisition of new rights and<br>imposition of restrictions | 4323 square metres of<br>watercourse and river<br>(River Clwyd), foreshore,<br>bed and banks thereof<br>(excluding all interests of<br>the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                  | NONE                                                                                                                                                                                                                                                                                                     | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                                                                                                                                                                                 | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights stated in Lease<br>dated 2nd June 2015)<br>Rhyl & Conwy Wildfowlers Club<br>Ty Gwyn<br>Cwm<br>Rhuddlan<br>Y RHYL<br>LL18 5RY<br>(in respect of sporting rights) |
| 258                     | Acquisition of new rights and<br>imposition of restrictions | 340 square metres of<br>grassed area and river<br>banks (south of River<br>Clwyd) (excluding all<br>interests of the Crown)                       | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)<br>Unknown | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder)<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder)<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder) | Unknown                                                                                                                                                                                                                                                                                          |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                              |                                                                                                                                |                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                   |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                              | Qualifying persons under Regulatio                                                                                             | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                                                                                                           | (Applications: Prescribed Forms and                                                                                                                                                                                                                                               | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                          | Owners or Reputed Owners                                                                                                       | Lessees or Tenants                                                                                                                                                                                                                                                                | Occupiers                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 258 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                              |                                                                                                                                | Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder)<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(as reputed leaseholder) | Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder)<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(as reputed leaseholder) |                                                                                                                                                       |  |  |  |
| 259                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 4924 square metres of<br>grassed area and river<br>banks (south of River<br>Clwyd) (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty<br>In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder)                                                                                                                                | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder)                                                                                                                                | Unknown                                                                                                                                               |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-----------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                   | Category 1                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulati | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                               | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                     | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners          | Lessees or Tenants                                                                                                                                                                                                                                                                                                                                                                                                                      | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 259 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Unknown                           | John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder)<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder)<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(as reputed leaseholder) | John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder)<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed leaseholder)<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(as reputed leaseholder) |                                                                                                                                                       |  |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i                                                                            | BOOK OF RI<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                                                                  | rm DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) right<br>of Denbighshire                                            | s to carry out protective works (Regulatio                                                                                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                          |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                          | Qualifying persons under Regulati                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                             |                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                      | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                               |
| 260                     | Acquisition of new rights and<br>imposition of restrictions | 1516 square metres of<br>grassed area and track<br>(south of River Clwyd)<br>and footpath<br>(DE 201 12) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys MolD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>Flintshire<br>CH7 5BP<br>(as reputed owner) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 12)<br>Unknown | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>The Natural Resources Body For<br>Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)<br>Unknown |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                       |                                                                                                                                                                                                                                                  |                                                                                                                |                                                                                                                |                                                                                                                                                                                |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                               | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                         | (Applications: Prescribed Forms and                                                                            | Category 2                                                                                                                                                                     |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                   | Owners or Reputed Owners                                                                                                                                                                                                                         | Lessees or Tenants                                                                                             | Occupiers                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                          |  |  |  |
| 260 conť d              |                                                                                                                                                                                                                                                                                                      |                                                                                       | Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(as reputed owner)<br>Unknown                                                                                                                |                                                                                                                |                                                                                                                |                                                                                                                                                                                |  |  |  |
| 261                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 13531 square metres of<br>agricultural land and<br>hedgerow (south of<br>River Clwyd) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of easement and right of<br>access for maintenance) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                        |                                                                                                                                                                                                                                                                            |                                                                                                                |                                                                                                                                  |                                                                                                                                                                             |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                        | Qualifying persons under Regulatio                                                                                                                                                                                                                                         | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                         | (Applications: Prescribed Forms and                                                                                              | Category 2                                                                                                                                                                  |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                    | Owners or Reputed Owners                                                                                                                                                                                                                                                   | Lessees or Tenants                                                                                             | Occupiers                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                       |  |  |  |
| 261 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                        | Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                |                                                                                                                                  |                                                                                                                                                                             |  |  |  |
| 262                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 52 square metres of<br>track and grassed area<br>(south of River Clwyd)<br>and footpath<br>(DE 201 12) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)                                                                                                                               | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 12) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed<br>dated 20 June 1979) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri                                                                                                                                                                                                                                                                                                                                                                       | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulatio                                                                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                                 |                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                         |                              |                               | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                              | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                    | g (Applications: Prescribed Forms and                                                                                     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                         | Occupiers                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                        |
| 262 cont'd              |                              |                               | John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(as reputed owner)<br>Unknown |                                                                                                            | Unknown<br>E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>The Natural Resources Body For<br>Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)<br>Unknown<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                     |                                                                                                                                                                                                                                                                                              |                                                                                                                                                              |                                                                                                                                             |                                                                                                                                                       |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                           | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                             |                                                                                                                                                       |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                     | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                                   | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 263                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 537 square metres of<br>track (south of River<br>Clwyd) and footpath<br>(DE 201 12) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 12)<br>Unknown | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)        |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                      | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                             | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                                        | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|------------------------------|---------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                     | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                     |                                                                                                                     |                                                                                                                | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                                  | Occupiers                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                |
| 263 cont'd              |                              |                                                                     | Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(as reputed owner)<br>Unknown |                                                                                                                     |                                                                                                                | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Natural Resources Body For<br>Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)<br>Unknown |
| 264                     | Acquisition of new rights    | 158 square metres of<br>agricultural land (south<br>of River Clwyd) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP                                                                                                                                                           | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL      | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                |                                                                                                                |                                                                                                                                                                                                                     |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                     | Category 1                                                                                                     |                                                                                                                |                                                                                                                                                                                                                     |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                 | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                  | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and                                                                            | Category 2                                                                                                                                                                                                          |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                             | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                                             | Occupiers                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                               |  |  |  |
| 264 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                                 | John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH |                                                                                                                |                                                                                                                |                                                                                                                                                                                                                     |  |  |  |
| 265                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 729 square metres of<br>agricultural land (Hafod-<br>Ilwyn Farm north of<br>Abergele Road, A457<br>and south of River<br>Clwyd) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP                                                                                                                                                                                                                              | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of<br>access for maintenance stated in<br>Deed dated 20 June 1979) |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                         |                                                                                                                  |                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                             | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 265 cont'd              |                              |                              | John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                  |                                         | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)        |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                           | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Category 1                                                                                                     |                                                                                                                           |                                                                                                                                                       |
|                         |                              |                                                                                                                                          | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and                                                                                       | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                      | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Lessees or Tenants                                                                                             | Occupiers                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 266                     | Acquisition of new rights    | 40 square metres of<br>agricultural land and<br>hedgerow (Hafod-Ilwyn<br>Farm north of Abergele<br>Road, A457 and east of<br>Marsh Road) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(as reputed owner) | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Unknown | Unknown                                                                                                                                               |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                  | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                                                      | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                                          | o carry out protective works (Regulatio                                                                        | n 7(1)(a))                                                                                                                                                                                                                                                                                 |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                 | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                |                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                             | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                      |
| 266 cont'd              |                              |                                                                                                                 | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Ì                                                                                                                                                            |                                                                                                                |                                                                                                                                                                                                                                                                                            |
| 267                     | Acquisition of new rights    | 72 square metres of<br>hedgerow (Hafod-Ilwyn<br>Farm north of Abergele<br>Road, A457 and east of<br>Marsh Road) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL                                               | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right of access)<br>Frances Marion Davies<br>Bryn Y Mor<br>Dimlands Road<br>LLANTWIT MAJOR<br>South Glamorgan<br>CF61 1YX<br>(in respect of right of access) |

|                         | Land which is j              | proposed to be subject to: (i                                                                                                             | BOOK OF R<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                                                  | rm DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                                       | n 7(1)(a))                                                                                                                                                                     |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Category 1                                                                                                         |                                                                                                                |                                                                                                                                                                                |
|                         |                              |                                                                                                                                           | Qualifying persons under Regulati                                                                                                                                                                                                                                                                                                                                                                                                                                      | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                            | Category 2                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                               | Lessees or Tenants                                                                                                 | Occupiers                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                          |
| 267 cont'd              |                              |                                                                                                                                           | Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                    |                                                                                                                |                                                                                                                                                                                |
| 268                     | Acquisition of new rights    | 378 square metres of<br>agricultural land and<br>hedgerow (Hafod-Ilwyn<br>Farm north of Abergele<br>Road, A457 and east of<br>Marsh Road) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys MOLD<br>Flintshire<br>CH7 5BP | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL     | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of easement and right of<br>access for maintenance) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                             |                                                                                                                                                                                                                                                  |                                                                                                                |                                                                                                                |                                                                                                                                                                                                                                                                                            |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                             | Qualifying persons under Regulatio                                                                                                                                                                                                               | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                         | (Applications: Prescribed Forms and                                                                            | Category 2                                                                                                                                                                                                                                                                                 |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                         | Owners or Reputed Owners                                                                                                                                                                                                                         | Lessees or Tenants                                                                                             | Occupiers                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                      |  |  |  |
| 268 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                             | Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>Unknown<br>(in respect of mines and minerals)                                                                                                |                                                                                                                |                                                                                                                |                                                                                                                                                                                                                                                                                            |  |  |  |
| 269                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 681 square metres of<br>agricultural land and<br>hedgerow (north of<br>Abergele Road, A457) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Frances Marion Davies<br>Bryn Y Mor<br>Dimlands Road<br>LLANTWIT MAJOR<br>South Glamorgan<br>CF61 1YX<br>(in respect of right of access) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                        | n 7(1)(a))                                                                                                                                                                                                                |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                       |                                                                                                                                                                                                                                                  | Category 1                                                                                                       |                                                                                                                |                                                                                                                                                                                                                           |
|                         |                              |                                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                               | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                            | Category 2                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                   | Owners or Reputed Owners                                                                                                                                                                                                                         | Lessees or Tenants                                                                                               | Occupiers                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                     |
| 269 cont'd              |                              |                                                                                       | Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH                        |                                                                                                                  |                                                                                                                |                                                                                                                                                                                                                           |
| 270                     | Temporary Possession         | 18 square metres of<br>hedgerow (Hafod-Ilwyn<br>Farm north of Abergele<br>Road, A457) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL   | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access)<br>Unknown<br>(in respect of rights stated in<br>Conveyance dated 14th July 1924) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                   | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                        | n 7(1)(a))                                                                                                                                                                                                          |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                  | Qualifying persons under Regulatio                                                                                                                                                                                                                                         | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                            | Category 2                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                              | Owners or Reputed Owners                                                                                                                                                                                                                                                   | Lessees or Tenants                                                                                               | Occupiers                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                               |
| 270 cont'd              |                              |                                                                                                  | Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                  |                                                                                                                | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights stated in Deed<br>dated 20th June 1979)                                        |
| 271                     | Temporary Possession         | 483 square metres of<br>agricultural land (Hafod-<br>Ilwyn Farm north of<br>Abergele Road, A457) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP                                                                                                                                                     | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL   | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of<br>access for maintenance stated in<br>Deed dated 20 June 1979) |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                 | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                                                | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                           |
| 271 cont'd              |                              |                              | John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH |                                                                                                                   |                                         | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |
|                         |                              |                              | Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                                                                                                                                       |                                                                                                                   |                                         |                                                                                                                                                                                                                                                                                                 |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                          | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                        | n 7(1)(a))                                                                                                                                                                                                                                                                                                                      |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Category 1                                                                                                       |                                                                                                                |                                                                                                                                                                                                                                                                                                                                 |
|                         |                                                             |                                                                                                         | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                            | Category 2                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                     | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                               | Occupiers                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                           |
| 272                     | Acquisition of new rights and<br>imposition of restrictions | 10631 square metres of<br>agricultural land (north<br>of Abergele Road, A457<br>and east of Gipsy Lane) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL   | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of<br>access for maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p              | proposed to be subject to: (i                                                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii) (                                                                                                                                                                                                                                                                                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire                                              | to carry out protective works (Regulatio                                                                       | n 7(1)(a))                                                                                                                                                                     |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                      | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                     | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                |                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                  | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                               | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                          |
| 272 cont'd              |                              |                                                                                                      | Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                              |                                                                                                                |                                                                                                                                                                                |
| 273                     | Temporary Possession         | 35 square metres of<br>agricultural land (north<br>of Abergele Road, A457<br>and east of Gipsy Lane) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys MoLD<br>Flintshire<br>CH7 5BP | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 SUL                                               | E R Jones & Partners<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of<br>access for maintenance) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                        | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulation                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                       | Qualifying persons under Regulatio                                                                                                                | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                   | Owners or Reputed Owners                                                                                                                          | Lessees or Tenants                                                                                                | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                               |
| 273 cont'd              |                              |                                                                                                                       | Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                   |                                                                                               |                                                                                                                                                                                                                                                                                                                                                     |
| 274                     | Temporary Possession         | 64 square metres of<br>agricultural land and<br>hedgerows (north of<br>Abergele Road, A457<br>and east of Gipsy Lane) | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH                                                     | NONE                                                                                                              | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and<br>right to pass for agricultural purposes<br>only) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                         | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                        | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                  |                                                                                               | Category 2                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                    | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants                                                                                               | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                               |
| 275                     | Acquisition of new rights and<br>imposition of restrictions | 5910 square metres of<br>agricultural land (north<br>of Abergele Road, A457<br>and east of Gipsy Lane) | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 ODH                                                                | NONE                                                                                                             | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and<br>right to pass for agricultural purposes<br>only) |
| 276                     | Temporary Possession                                        | 2256 square metres of<br>agricultural land (north<br>of Abergele Road, A457<br>and east of Gipsy Lane) | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 ODH                                                                | NONE                                                                                                             | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)                                                                                                                                                                                                      |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                       | Qualifying persons under Regulatio                                                            | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                   | Owners or Reputed Owners                                                                      | Lessees or Tenants                                                                                                | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                               |
| 276 conť d              |                              |                                                                                                                       |                                                                                               |                                                                                                                   |                                                                                               | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and<br>right to pass for agricultural purposes<br>only)                                                                                                                                                   |
| 277                     | Temporary Possession         | 358 square metres of<br>agricultural land and<br>hedgerow (north of<br>Abergele Road, A457<br>and east of Gipsy Lane) | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH | NONE                                                                                                              | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 ODH | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and<br>right to pass for agricultural purposes<br>only) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                       | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                  |                                                                                               | Category 2                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                   | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants                                                                                               | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                               |
| 278                     | Acquisition of new rights    | 620 square metres of<br>agricultural land (north<br>of Abergele Road, A457<br>and east of Gipsy Lane) | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH                                                                | NONE                                                                                                             | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 ODH | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and<br>right to pass for agricultural purposes<br>only) |
| 279                     | Temporary Possession         | 27 square metres of<br>hedgerow (east of Gipsy<br>Lane and north of<br>Abergele Road, A45)            | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 ODH                                                                | NONE                                                                                                             | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)                                                                                                                                                                                                      |

|                         | Land which is p              | proposed to be subject to: (i)                                                       | BOOK OF RE powers of compulsory acquisition, (ii) | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------|---------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                      | Qualifying persons under Regulatic                | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                             | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                  | Owners or Reputed Owners                          | Lessees or Tenants                                                                                                  | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                 |
| 279 cont'd              |                              |                                                                                      |                                                   |                                                                                                                     |                                         | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and<br>right to pass for agricultural purposes<br>only)                                                                                                                                                                                                                                     |
| 280                     | Acquisition of new rights    | 562 square metres of<br>private road (Gipsy Lane<br>north of Abergele Road,<br>A457) | Unknown                                           | NONE                                                                                                                | Unknown                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH<br>(in respect of right of access) |

|                         | Land which is p              | roposed to be subject to: (i)                                                      | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                    |                                                                                                                      | Category 1                                                                                                       |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                              |                                                                                    | Qualifying persons under Regulatio                                                                                   | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                | Owners or Reputed Owners                                                                                             | Lessees or Tenants                                                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                         |
| 280 cont'd              |                              |                                                                                    |                                                                                                                      |                                                                                                                  |                                                                                                                      | Police & Crime Commissioner for<br>North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWYN BAY<br>Conwy<br>LL29 8AW<br>(in respect of right of access)<br>R G B Webster & Sons Limited<br>Banks House<br>1 Paradise Street<br>RHYL<br>Denbighshire<br>LL18 3LW<br>(in respect of right of access)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of access)<br>Unknown |
| 281                     | Acquisition of new rights    | 51 square metres of<br>public road (Gipsy Lane<br>north of Abergele Road,<br>A457) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                                |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                       |                                                                                                                                       |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                         | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                    | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                | (Applications: Prescribed Forms and                                                                                                   | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                     | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                                                                    | Occupiers                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 281 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                         | Police & Crime Commissioner for<br>North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWYN BAY<br>Conwy<br>LL29 8AW<br>(in respect of subsoil beneath<br>public highway)<br>R G B Webster & Sons Limited<br>Banks House<br>1 Paradise Street<br>RHYL<br>Denbighshire<br>LL18 3LW<br>(in respect of subsoil beneath<br>public highway)<br>Unknown |                                                                                                                                       | Unknown                                                                                                                               | Unknown                                                                                                                                               |  |  |  |
| 282                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 19597 square metres of<br>agricultural land (north<br>of Abergele Road, A457<br>and east of Gipsy Lane) | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH                                                                                                                                                                                                                                                                 | The Executor of the Estate of the<br>Late Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH | The Executor of the Estate of the<br>Late Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Raymond Ivor<br>Beech)                                            |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                    | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                                        |                                                                                                                                 |                                                                                                                                                                                                                                                                                                              |
|                         |                              |                                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                               | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                             | Category 2                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                                | Occupiers                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                        |
| 282 cont'd              |                              |                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                   |                                                                                                                                 | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH<br>(in respect of right to pass and right<br>to lay and maintain a water pipeline)                                                                                                                             |
| 283                     | Temporary Possession         | 302 square metres of<br>verge of public road<br>(south of Abergele Road,<br>A547) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Mark Gerard Brian Webster<br>Argraig<br>Tremeirchion<br>LLANELWY<br>LL17 OUN<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Frank William Webster<br>39 Llys Dyffryn<br>LLANELWY<br>LL17 OSX<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown |

|                         | Land which is p              | proposed to be subject to: (i)                                                    | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                   | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                             | (Applications: Prescribed Forms and                                                                                             | Category 2                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Lessees or Tenants                                                                                                  | Occupiers                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                        |
| 284                     | Temporary Possession         | 400 square metres of<br>verge of public road<br>(north of Abergele Road,<br>A547) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Police & Crime Commissioner for<br>North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWYN BAY<br>Conwy<br>LL29 8AW<br>(in respect of subsoil beneath half<br>width of public highway)<br>R G B Webster & Sons Limited<br>Banks House<br>1 Paradise Street<br>RHYL<br>Denbighshire<br>LL18 3LW<br>(in respect of subsoil beneath half<br>width of public highway)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                            | BOOK OF RE powers of compulsory acquisition, (ii) r                                                   | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                     |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                           | Qualifying persons under Regulatio                                                                    | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                       |                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                       | Owners or Reputed Owners                                                                              | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                          |
| 284 cont'd              |                                                             |                                                                                                                                           | Unknown                                                                                               |                                                                                                                                                              |                                                                                                                                       |                                                                                                                                                                                                                                                                                                |
| 285                     | Acquisition of new rights and<br>imposition of restrictions | 7031 square metres of<br>agricultural land (Hafod-<br>llwyn Farm north of<br>Abergele Road, A457<br>and east of Gipsy Lane)               | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | The Executor of the Estate of the<br>Late Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH                        | The Executor of the Estate of the<br>Late Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Raymond Ivor<br>Beech)<br>Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH<br>(in respect of right to pass and right<br>to lay and maintain a water pipeline) |
| 286                     | Acquisition of new rights                                   | 644 square metres of<br>agricultural land and<br>hedgerow (Hafod-Ilwyn<br>Farm north of Abergele<br>Road, A457 and east of<br>Gipsy Lane) | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | The Executor of the Estate of the<br>Late Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH                        | The Executor of the Estate of the<br>Late Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Raymond Ivor<br>Beech)                                                                                                                                                                                     |

|                         | Land which is p              | proposed to be subject to: (i)                                                      | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                          | n 7(1)(a))                                                                                                                                                                                                                                                                                                           |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                     |                                                                                                                                                                                                                                                                                              | Category 1                                                                                                       |                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                      |
|                         |                              |                                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                              | Category 2                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                     | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                |
| 286 cont'd              |                              |                                                                                     |                                                                                                                                                                                                                                                                                              |                                                                                                                  |                                                                                                                                                                                                                                                  | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH<br>(in respect of right to pass and right<br>to lay and maintain a water pipeline)                                                                                                                                     |
| 287                     | Acquisition of new rights    | 120 square metres of<br>track (Hafod-Ilwyn Farm<br>north of Abergele Road,<br>A457) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner) | NONE                                                                                                             | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of access)<br>The Executor of the Estate of the Late<br>Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH<br>(in respect of right of access) |

|                         | Land which is p              | proposed to be subject to: (i)                                                          | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire                                             | o carry out protective works (Regulation                                                                                                                                                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                         | Qualifying persons under Regulatio                                                                                                                                                                                                                                               | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                                                           |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                     | Owners or Reputed Owners                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 287 cont'd              |                              |                                                                                         | Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(as reputed owner)<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(as reputed owner)<br>Unknown |                                                                                                                                                              | Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH | Unknown                                                                                                                                               |
| 288                     | Acquisition of new rights    | 6 square metres of<br>public road (Hafod-Ilwyn<br>Farm north of Abergele<br>Road, A457) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(in respect of subsoil beneath half<br>width of public highway)                                                                                        | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                      | Unknown                                                                                                                                               |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulation | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Category 1<br>n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                          | g (Applications: Prescribed Forms and     | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                             | Occupiers                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 288 cont'd              |                              |                               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                                | Unknown                                   |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                   |                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                              |                                                                                                                                 |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                 |                                                                                                                                                       |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                          | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 288 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                   | Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                                                                                                                                    |                                                                                                                                                              |                                                                                                                                 |                                                                                                                                                       |  |  |  |
| 289                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 208 square metres of<br>verge of public road<br>(north of Abergele Road,<br>A457) | David Thomas Charles Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown | Unknown                                                                                                                                               |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire                                              | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                         |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Lessees or Tenants                                                                                                                                           | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 289 cont'd              |                              |                              | John Frederick Lloyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(in respect of subsoil beneath half<br>width of public highway)<br>Meigh Enterprises Limited<br>H.T.M Business Park<br>Abergele Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                                                                              |                                         |                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i)                                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                         | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                   | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 289 cont'd              |                              |                                                                                   | Robert Mervyn Llwyd Owens<br>Caerwys Hall<br>Crossways Road<br>Pen Y Cefn<br>Caerwys<br>MOLD<br>Flintshire<br>CH7 5BP<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Michael Llewelyn Owens<br>Bryn Gwyn Hall<br>Pen Y Cefn Road<br>Caerwys<br>MOLD<br>CH7 5BH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                   |                                                                                                                      |                                                                                                                                                       |
| 290                     | Temporary Possession         | 172 square metres of<br>verge of public road<br>(south of Abergele Road,<br>A457) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                                                                                                                                                                       | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)        |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>i) powers of compulsory acquisition, (ii) ri                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                 |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                               | g (Applications: Prescribed Forms and    | Category 2                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Procedures) Regulations 2009<br>Lessees or Tenants                                                            | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009      |
| 290 cont'd              |                              |                               | Fabian Jeremy Morgan Evans<br>Alfra<br>Abergele Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Gary Paul Hughes<br>Plumb Workz<br>Unit 6-7<br>Coed Park<br>Abergele Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UE<br>(in respect of subsoil beneath half<br>width of public highway)<br>Loni Jane Evans<br>Alfra<br>Abergele Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UE<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                               | Unknown                                  | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown |

|                         | Land which is p                                             | proposed to be subject to: (i                                           | BOOK OF RE<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                   | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                                        | n 7(1)(a))                                                                                                                                                                                                                                                                                        |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                         |                                                                                                                                                                                                                                                                                                          | Category 1                                                                                                      |                                                                                                                                 |                                                                                                                                                                                                                                                                                                   |
|                         |                                                             |                                                                         | Qualifying persons under Regulation                                                                                                                                                                                                                                                                      | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                         | g (Applications: Prescribed Forms and                                                                                           | Category 2                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                     | Owners or Reputed Owners                                                                                                                                                                                                                                                                                 | Lessees or Tenants                                                                                              | Occupiers                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                             |
| 290 cont'd              |                                                             |                                                                         | Mark Gerard Brian Webster<br>Argraig<br>Tremeirchion<br>LLANELWY<br>LL17 OUN<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Frank William Webster<br>39 Llys Dyffryn<br>LLANELWY<br>LL17 OSX<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                 |                                                                                                                                 |                                                                                                                                                                                                                                                                                                   |
| 291                     | Acquisition of new rights and<br>imposition of restrictions | 904 square metres of<br>public road and verges<br>(Abergele Road, A457) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Mark Gerard Brian Webster<br>Argraig<br>Tremeirchion<br>LLANELWY<br>LL17 OUN<br>(in respect of subsoil beneath half<br>width of public highway)                                  | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                               | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                      | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                              | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                          | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                          | Lessees or Tenants                                                                                                | Occupiers                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 291 cont'd              |                              |                                                                                              | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Frank William Webster<br>39 Llys Dyffryn<br>LLANELWY<br>LL17 OSX<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                   |                                                                                                              | Unknown                                                                                                                                               |
| 292                     | Acquisition of new rights    | 1200 square metres of<br>agricultural land and<br>hedgerow (south of<br>Abergele Road, A457) | Mark Gerard Brian Webster<br>Argraig<br>Tremeirchion<br>LLANELWY<br>LL17 OUN<br>Robert Frank William Webster<br>39 Llys Dyffryn<br>LLANELWY<br>LL17 OSX                                                                                                                                                                           | S.R. Beech & Sons LLP<br>102 Bowen Court<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJE      | S.R. Beech & Sons LLP<br>102 Bowen Court<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJE | NONE                                                                                                                                                  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                   | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                  | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                              |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                  |                                                                                                                                                                                                                                       | Category 1                                                                                                       |                                                                                                                                  |                                                                                                                                                                                                                                                                                         |
|                         |                                                             |                                                                                                                                  | Qualifying persons under Regulatio                                                                                                                                                                                                    | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                              | Category 2                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                              | Owners or Reputed Owners                                                                                                                                                                                                              | Lessees or Tenants                                                                                               | Occupiers                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                   |
| 293                     | Acquisition of new rights and<br>imposition of restrictions | 19992 square metres of<br>agricultural land and<br>hedgerow (south of<br>Abergele Road, A457)<br>and footpath<br>(DE   206   40) | Mark Gerard Brian Webster<br>Argraig<br>Tremeirchion<br>LLANELWY<br>LL17 OUN                                                                                                                                                          | S.R. Beech & Sons LLP<br>102 Bowen Court<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJE     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 40) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain electric lines and right to fell<br>trees that would obstruct the electric<br>lines as contained in a Deed dated<br>21st July 1994) |
|                         |                                                             |                                                                                                                                  | Robert Frank William Webster<br>39 Llys Dyffryn<br>LLANELWY<br>LL17 OSX                                                                                                                                                               |                                                                                                                  | S.R. Beech & Sons LLP<br>102 Bowen Court<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJE                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                           |
| 294                     | Acquisition of new rights and<br>imposition of restrictions | 7754 square metres of<br>agricultural land and<br>hedgerow (south of<br>Abergele Road, A457<br>and north of<br>Bodelwyddan Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>The Executor of the Estate of the<br>Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL | NONE                                                                                                             | Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY                                                                           | NONE                                                                                                                                                                                                                                                                                    |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                    | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         | Category 1                   |                                                                                                                                   |                                                                                                                                                                                                                                       |                                                                                                                   |                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                 |
|                         |                              |                                                                                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                    | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                                                                                                   | Category 2                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                               | Owners or Reputed Owners                                                                                                                                                                                                              | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                           |
| 294 cont'd              |                              |                                                                                                                                   | Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                         |                                                                                                                   |                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                 |
| 295                     | Acquisition of new rights    | 32 square metres of<br>agricultural land and<br>hedgerows (south of<br>Abergele Road, A457<br>and north of<br>Bodelwyddan Road)   | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH                               | NONE                                                                                                              | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH               | NONE                                                                                                                                                                                                                                                                                            |
| 296                     | Acquisition of new rights    | 2073 square metres of<br>agricultural land (north<br>of Bodelwyddan Road)<br>and footpaths<br>(DE 201 8, DE 206 38,<br>DE 206 41) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>The Executor of the Estate of the<br>Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 8,<br>DE 206 38 and DE 206 41)<br>Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                    | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | to carry out protective works (Regulatio                                                                                                                                                                | n 7(1)(a))                                                                                                                                            |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                   |                                                                                                                                                                                                         | Category 1                                                                                                         |                                                                                                                                                                                                         |                                                                                                                                                       |
|                         |                                                             |                                                                                                                                   | Qualifying persons under Regulation                                                                                                                                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                           | (Applications: Prescribed Forms and                                                                                                                                                                     | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                               | Owners or Reputed Owners                                                                                                                                                                                | Lessees or Tenants                                                                                                 | Occupiers                                                                                                                                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 296 cont'd              |                                                             |                                                                                                                                   | Unknown<br>(in respect of mines and minerals)                                                                                                                                                           |                                                                                                                    |                                                                                                                                                                                                         |                                                                                                                                                       |
| 297                     | Acquisition of new rights and<br>imposition of restrictions | 6129 square metres of<br>agricultural land and<br>hedgerows (south of<br>Abergele Road, A457<br>and north of<br>Bodelwyddan Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | NONE                                                                                                               | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | NONE                                                                                                                                                  |
| 298                     | Acquisition of new rights                                   | 12 square metres of<br>verge of public road<br>(north of Bodelwyddan<br>Road)                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                    | NONE                                                                                                               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                    | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)        |

|                         | Land which is p              | proposed to be subject to: (i)                                                | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                               | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                           |                                                                                                                   |                                                                                            | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                               | Lessees or Tenants                                                                                                | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 298 cont'd              |                              |                                                                               | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>The Executor of the Estate of the<br>Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                   |                                                                                            | Unknown                                                                                                                                               |
| 299                     | Acquisition of new rights    | 10 square metres of<br>verge of public road<br>(north of Bodelwyddan<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Unknown<br>(in respect of mines and minerals)                                                                                                                                                                                                                                            | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | NONE                                                                                                                                                  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                     | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                       | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                 | n 7(1)(a))                                                                                                                                                           |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         | Category 1                   |                                                                                                                    |                                                                                                                                                                                                         |                                                                                                                     |                                                                                                                                                                                                         |                                                                                                                                                                      |
|                         |                              |                                                                                                                    | Qualifying persons under Regulation                                                                                                                                                                     | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                           | (Applications: Prescribed Forms and                                                                                                                                                                     | Category 2                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                | Owners or Reputed Owners                                                                                                                                                                                | Lessees or Tenants                                                                                                  | Occupiers                                                                                                                                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                |
| 300                     | Acquisition of new rights    | 18 square metres of<br>verge of public road<br>(north of Bodelwyddan<br>Road)                                      | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown                                                                         | NONE                                                                                                                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                    | Unknown                                                                                                                                                              |
| 301                     | Acquisition of new rights    | 40 square metres of<br>agricultural land (north<br>of Bodelwyddan Road)                                            | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | NONE                                                                                                                | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | NONE                                                                                                                                                                 |
| 302                     | Acquisition of new rights    | 42 square metres of<br>agricultural land and<br>hedgerow (north of<br>Bodelwyddan Road) and<br>footpath (DE 206 4) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH                                                                                                        | NONE                                                                                                                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 4)                                                                         | Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right<br>to lay and maintain services) |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri                                                                                                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulation                                                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                                                                                                                                                     | Category 1                                                                                                  |                                                                                                                                                                                                                                                                               |                                                                                                                                                       |
|                         |                              |                               | Qualifying persons under Regulation                                                                                                                                                                                                                                 | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                     | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                         | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                                                                                                                                                            | Lessees or Tenants                                                                                          | Occupiers                                                                                                                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 302 cont'd              |                              |                               | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD |                                                                                                             | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD |                                                                                                                                                       |
|                         |                              |                               |                                                                                                                                                                                                                                                                     |                                                                                                             | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD                                                                                                                                                                                        |                                                                                                                                                       |

|                         | Land which is                | proposed to be subject to: (i                                                             | BOOK OF RI<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                  | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) right<br>of Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                              | n 7(1)(a))                                                                                                                                                           |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                         | Category 1                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                      |
|                         |                              |                                                                                           | Qualifying persons under Regulati                                                                                                                                                                                                                                                                                                                                       | on 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                          | ng (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                  | Category 2                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                                                                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                |
| 303                     | Temporary Possession         | 174 square metres of<br>agricultural land and<br>hedgerows (north of<br>Bodelwyddan Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                             | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right<br>to lay and maintain services) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                      | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                  | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire                                | s to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                                                       | n 7(1)(a))                                                                                                                                                           |  |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                              |                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                    | Category 1                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                      |  |
|                         |                              |                                                                                                                     | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                                                                                 | Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                      |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                           | Lessees or Tenants                                                                                                                             | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                |  |
| 304                     | Acquisition of new rights    | 231 square metres of<br>agricultural land and<br>hedgerow (north of<br>Bodelwyddan Road) and<br>footpath (DE 206 4) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL16 4RD | NONE                                                                                                                                           | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 4)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD | Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right<br>to lay and maintain services) |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                             | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                            | Qualifying persons under Regulatio                                                                                                                                                                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                                                                                                                                     | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                        | Owners or Reputed Owners                                                                                                                                                                                | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 304 cont'd              |                              |                                                                                                            |                                                                                                                                                                                                         |                                                                                                                   | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD                                                                                                                                                  |                                                                                                                                                       |
| 305                     | Acquisition of new rights    | 19 square metres of<br>private road and verge<br>(north of Bodelwyddan<br>Road) and footpath<br>(DE 206 4) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE   206   4)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | NONE                                                                                                                                                  |
|                         |                              |                                                                                                            |                                                                                                                                                                                                         |                                                                                                                   | Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH                                                                                                                                     |                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i                                       | BOOK OF RE<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                 | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|---------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                     | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                    | Category 1<br>on 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                            | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                 | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                               | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                               |
| 306                     | Acquisition of new rights    | 472 square metres of<br>private road (north of<br>Bodelwyddan Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Kelly Proffitt<br>Y Wern<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DS<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Toni Mayne<br>Y Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | NONE                                                                                                              | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Kelly Proffitt<br>Y Wern<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DS<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Toni Mayne<br>Y Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right to enter to lay and<br>maintain water pipes)<br>The Executor of the Estate of the Late<br>George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of right of way) |

|                         | Land which is                | proposed to be subject to: (i                                             | BOOK OF R<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                 | rm DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) right<br>of Denbighshire | s to carry out protective works (Regulatio             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|---------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                           |                                                                                                                                                                                                                                                                                                                                                                                       | Category 1                                                                                                        |                                                        |                                                                                                                                                       |
|                         |                              |                                                                           | Qualifying persons under Regulati                                                                                                                                                                                                                                                                                                                                                     | on 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                           | ng (Applications: Prescribed Forms and                 | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                                                | Occupiers                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 307                     | Temporary Possession         | 4958 square metres of<br>agricultural land (north<br>of Bodelwyddan Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                              | Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY | NONE                                                                                                                                                  |

|                         | Land which is p              | proposed to be subject to: (i,                                            | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                  | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                                                           | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|---------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                           | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                   | g (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                              | Category 2                                                                                                                                            |
|                         |                              |                                                                           |                                                                                                                                                                                                                                                                                                                                                                                       | Procedures) Regulations 2009                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                                                                                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 308                     | Temporary Possession         | 6339 square metres of<br>agricultural land (north<br>of Bodelwyddan Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                              | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)         |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                       | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulation                                                                                                                                                                                                                                                                                                                                                                        | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                         | Category 1                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                         |                                                             |                                                                                                                      | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                                                                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                              | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                  | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                | Lessees or Tenants                                                                                              | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                   |
| 309                     | Acquisition of new rights and<br>imposition of restrictions | 9496 square metres of<br>agricultural land and<br>hedgerow (north of<br>Bodelwyddan Road) and<br>footpath (DE 206 4) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 206 4)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right<br>to lay and maintain services)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                 | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|--------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Category 1                                                                                                      |                                                                                                                      |                                                                                                                                                                                                                                                                                                              |
|                         |                              |                                                                                | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                            | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                               | Lessees or Tenants                                                                                              | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                        |
| 309 cont'd              |                              |                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                 | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD                               |                                                                                                                                                                                                                                                                                                              |
| 310                     | Temporary Possession         | 862 square metres of<br>verge of public road<br>(south of Bodelwyddan<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                   |                                                                                                                                                                                                                                                                                                                              |                                                                          |                                                                                                                      |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                   |                                                                                                                                                                                                                                                                                                                              | Category 1                                                               |                                                                                                                      |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                     | Lessees or Tenants                                                       | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 310 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                   | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                          |                                                                                                                      |                                                                                                                                                       |  |  |  |
| 311                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 17 square metres of<br>public road (south of<br>Bodelwyddan Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                                                                                   | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                 |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                              | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                  | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                             | Qualifying persons under Regulatio                                                                                   | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                         | Owners or Reputed Owners                                                                                             | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                             |
| 312                     | Temporary Possession         | 252 square metres of<br>verge of public road<br>(north of Bodelwyddan<br>Road)              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                           | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to enter, right to<br>construct, place, lay, inspect,<br>maintain, clean, repair, renew,<br>enlarge, replace, conduct, use and<br>manage water pipes and right to<br>pass)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |
| 313                     | Temporary Possession         | 102 square metres of<br>hedgerow and verge of<br>public road (north of<br>Bodelwyddan Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                           |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REFI<br>) powers of compulsory acquisition, (ii) rig                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                                    |                                         |                                                                                                                                                       |
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and     | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 313 cont'd              |                              |                              | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                               |                                         |                                                                                                                                                       |

|                         | Land which is <sub>l</sub>   | proposed to be subject to: (i                                                            | BOOK OF RE<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                  | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio               | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                          | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                     | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                    | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                      | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                | Lessees or Tenants                                                                                                | Occupiers                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 313 cont'd              |                              |                                                                                          | Unknown                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                   |                                                        |                                                                                                                                                       |
| 314                     | Temporary Possession         | 203 square metres of<br>agricultural land and<br>hedgerow (south of<br>Bodelwyddan Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                              | Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)       |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights t<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                            |  |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                             |                                                                               | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                     | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                      |                                                                                                                                                       |  |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                               | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |
| 315                     | Temporary Possession                                        | 7 square metres of verge<br>of public road (south of<br>Bodelwyddan Road)     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                                                                                                                                                                                                             | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | NONE                                                                                                                                                  |  |
| 316                     | Acquisition of new rights and<br>imposition of restrictions | 14 square metres of<br>verge of public road<br>(north of Bodelwyddan<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |  |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                            | n 7(1)(a))                                                                                                                                                                                                                                                                                                         |
|-------------------------|----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                    |                                                                                                                                                                                                                                                                                                                              | Category 1                                                                                                       |                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                    |
|                         |                                                          |                                                                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                | Category 2                                                                                                                                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                     | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                              |
| 316 cont'd              |                                                          |                                                                                                                    | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                  |                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                    |
| 317                     | Acquisition of new rights and imposition of restrictions | 523 square metres of<br>agricultural land,<br>hedgerows, private road<br>and verges (north of<br>Bodelwyddan Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Kelly Proffitt<br>Y Wern<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DS                                                                                                                                           | NONE                                                                                                             | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Kelly Proffitt<br>Y Wern<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DS | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Executor of the Estate of the Late<br>George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of right of way) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                  | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                       | rm DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                                                                                                                         | n 7(1)(a))                                                                                                                                                                      |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                 | Qualifying persons under Regulation                                                                                                                                                                                                        | Category 2                                                                                                         |                                                                                                                                                                                                  |                                                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                             | Owners or Reputed Owners                                                                                                                                                                                                                   | Lessees or Tenants                                                                                                 | Occupiers                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                           |
| 317 cont'd              |                                                             |                                                                                 | Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 SUH<br>Toni Mayne<br>Y Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 SUH                                           |                                                                                                                    | Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Toni Mayne<br>Y Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH |                                                                                                                                                                                 |
| 318                     | Acquisition of new rights and<br>imposition of restrictions | 5 square metres of<br>private road and verges<br>(north of Bodelwyddan<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as reputed owner)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(as reputed owner) | NONE                                                                                                               | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH   | The Executor of the Estate of the Late<br>George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of right of access)<br>Unknown |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                  | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                             | n 7(1)(a))                                                                                                                                            |  |  |
|-------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                          |                                                                                 |                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                                       |                                                                                                                                                                                                                                                                     |                                                                                                                                                       |  |  |
|                         |                                                          |                                                                                 | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                               | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                 | Category 2                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                             | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 318 cont'd              |                                                          |                                                                                 | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(as reputed owner)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(as reputed owner)<br>Unknown |                                                                                                                  | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD |                                                                                                                                                       |  |  |
| 319                     | Acquisition of new rights and imposition of restrictions | 9 square metres of<br>private road and verges<br>(north of Bodelwyddan<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                                                                                                       | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                          | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of access)                   |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                |  |  |
| 319 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of access)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of access)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of access)<br>The Executor of the Estate of the Late<br>George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of right of access) |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                    | BOOK OF RE powers of compulsory acquisition, (ii)                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                             |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                   | Qualifying persons under Regulatio                                                                                   | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                               | Owners or Reputed Owners                                                                                             | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                  |
| 320                     | Acquisition of new rights and<br>imposition of restrictions | 272 square metres of<br>verge of public road<br>(north of Bodelwyddan<br>Road)    | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                           | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |
| 321                     | Acquisition of new rights and imposition of restrictions    | 167 square metres of<br>verge of public road<br>(south of Bodelwyddan<br>Road)    | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                           | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                          |
| 322                     | Acquisition of new rights and imposition of restrictions    | 521 square metres of<br>public road and<br>hedgerow (part of<br>Bodelwyddan Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                         |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REFI<br>) powers of compulsory acquisition, (ii) rig                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulation | n 7(1)(a))                                                                                                                                                                                                                                                                                                  |
|-------------------------|------------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                                    |                                          |                                                                                                                                                                                                                                                                                                             |
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 7(1)(a) of the Infrastructure Planning (<br>Procedures) Regulations 2009                                      | Applications: Prescribed Forms and       | Category 2                                                                                                                                                                                                                                                                                                  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                            | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                       |
| 322 cont'd              |                              |                              | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                               |                                          | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                  |                                                                                                                                                                                                                                                                                             |                                                                                        |                                                                                                                      |                                                                                                                                                                                                                                                                                                   |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                  | Qualifying persons under Regulatio                                                                                                                                                                                                                                                          | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                        |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                              | Owners or Reputed Owners                                                                                                                                                                                                                                                                    | Lessees or Tenants                                                                     | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                             |  |  |  |
| 322 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                  | The Executor of the Estate of the<br>Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                                                                             |                                                                                        |                                                                                                                      |                                                                                                                                                                                                                                                                                                   |  |  |  |
| 323                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 67 square metres of<br>public road (part of<br>Bodelwyddan Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |  |  |  |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REFE<br>) powers of compulsory acquisition, (ii) rig                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | DEVELOPMENT CONSENT ORDER<br>RENCE - PART 1<br>ht to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulation | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Category 1                                                                                                |                                          |                                                                                                                                                       |
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                    | (Applications: Prescribed Forms and      | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Lessees or Tenants                                                                                        | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 323 cont'd              |                              |                              | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>The Executor of the Estate of the<br>Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                           |                                          | Unknown                                                                                                                                               |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                        |                                                                           |                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                        | Category 1                                                                |                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                          | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                    | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                    | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                      | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                               | Lessees or Tenants                                                        | Occupiers                                                                                                                                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 323 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                          | Unknown                                                                                                                                                                                                                                                                                                                                                                                |                                                                           |                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                       |  |  |  |
| 324                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 835 square metres of<br>agricultural land and<br>hedgerow (north of<br>Bodelwyddan Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Kelly Proffitt<br>Y Wern<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DS<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Toni Mayne<br>Y Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | NONE                                                                      | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Kelly Proffitt<br>Y Wern<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DS<br>Richard David Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Toni Mayne<br>Y Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH | NONE                                                                                                                                                  |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                 | BOOK OF RE powers of compulsory acquisition, (ii)                                                                    | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                     |
|-------------------------|------------------------------|--------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                |                                                                                                                      | Category 1                                                                                                        |                                                                                                                      |                                                                                                                                                                                                                                                                |
|                         |                              |                                                                                | Qualifying persons under Regulation                                                                                  | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                            | Owners or Reputed Owners                                                                                             | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                          |
| 325                     | Temporary Possession         | 124 square metres of<br>verge of public road<br>(north of Bodelwyddan<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                           | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of way)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |
| 326                     | Temporary Possession         | 4 square metres of<br>public road (part of<br>Bodelwyddan Road)                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                           | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of way)                                                                                                                          |
| 327                     | Temporary Possession         | 207 square metres of<br>verge of public road<br>(north of Bodelwyddan<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                                                                                                                        |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                           |                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                                                                                                                                                                                                                                                                                                      |                                                                                           | Qualifying persons under Regulatio                                                                                                                                                                             | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                       | Owners or Reputed Owners                                                                                                                                                                                       | Lessees or Tenants                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Occupiers                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
|                                                                                                                                                                                                                                                                                                      |                                                                                           | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
| Temporary Possession                                                                                                                                                                                                                                                                                 | 1728 square metres of<br>agricultural land and<br>hedgerow (north of<br>Bodelwyddan Road) | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Unknown<br>(in respect of mines and minerals)                                                         | The Executor of the Estate of the<br>Late Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH                                                                                                                                                                                                                                                                                                                                                                                                                                                          | The Executor of the Estate of the<br>Late Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain water pipeline)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(in respect of right to an easement as                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |  |
|                                                                                                                                                                                                                                                                                                      | Extent of acquisition or use                                                              | Extent of acquisition or use       Description of land         Description of land       Description of land         Temporary Possession       1728 square metres of agricultural land and hedgerow (north of | Extent of acquisition or use       Description of land       Qualifying persons under Regulation         Extent of acquisition or use       Description of land       Owners or Reputed Owners         Raymond Ivor Beech       Bryn Carrog       Bodelwyddan Road         Rhuddlan       RHYL       Denbighshire         L118 SUH       (in respect of subsoil beneath half         remporary Possession       1728 square metres of agricultural land and hedgerow (north of Bodelwyddan Road)         Rhuddlan       RHYL         Denbighshire       L118 SUH         Unknown       Unknown | BOOK OF REFERENCE - PART 1         Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to Locunty of Denbighshire         Extent of acquisition or use       Description of land       Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning Procedures) Regulations 2009         Extent of acquisition or use       Description of land       Owners or Reputed Owners       Lessees or Tenants         Raymond Ivor Beech Bryn Carrog Bodelwyddan Road Rhuddlan RHYL       Raymond Ivor Beech Bryn Carrog Bodelwyddan Road Rhuddlan RHYL       The Executor of the Estate of the Late Richard Doad Bryn Carrog Bodelwyddan Road Rhuddlan RHYL         Temporary Possession       1728 square metres of agricultural land and Bryn Carrog Bodelwyddan Road Rhuddlan RHYL Denbighshire LL18 SUH (in respect of subsoil beneath half width of public highway)       The Executor of the Estate of the Late Richard Doad Bryn Carrog Farm Bodelwyddan Road Rhuddlan RHYL Denbighshire LL18 SUH LL3 SUH LL3 SUH LL3 SUH         Unknown       Unknown       Bodelwyddan Road Rhuddlan RHYL Bohighshire LL18 SUH | Intersection of the subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation County of Denbighshire         Extent of acquisition or use       Description of land       Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009         Extent of acquisition or use       Description of land       Owners or Reputed Owners       Lessees or Tenants       Occupiers         Raymond Ivor Beech       Bryn Carrog       Bodelwyddan Road       Rhvul       Denbighshire       III S UH       Intersection of the Estate of the Late Richard Dod       Bryn Carrog         Temporary Possession       1728 square metres of Bodelwyddan Road       Raymond Ivor Beech       Bryn Carrog       Bodelwyddan Road       Bryn Carrog       Bodelwyddan Road       Bryn Carrog       Bodelwyddan Road       The Executor of the Estate of the Late Richard Dod       Bryn Carrog Form       Bryn Carrog Form |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                 |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                    |                                                                                                                                                                                                                                                                                                                                     | Category 1                                                                                                       |                                                                                                                      |                                                                                                                                                            |
|                         |                              |                                                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                  | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009      |
| 328 cont'd              |                              |                                                                                                    |                                                                                                                                                                                                                                                                                                                                     |                                                                                                                  |                                                                                                                      | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Raymond Ivor<br>Beech)<br>Unknown<br>(in respect of right of drainage) |
| 329                     | Temporary Possession         | 272 square metres of<br>verge of public road<br>(north of Bodelwyddan<br>Road)                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                                                                                          | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of way)                      |
| 330                     | Temporary Possession         | 4 square metres of<br>public road (part of<br>Nant-y-Faenol Road,<br>south of Bodelwyddan<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>The Executor of the Estate of the<br>Late Wynford Davies<br>Bryn Awel<br>Bodelwyddan Road<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                    |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                             |                                                                                                                                                                                |                                                                                                           |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                             | Qualifying persons under Regulatio                                                                                                                                             | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                    | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                         | Owners or Reputed Owners                                                                                                                                                       | Lessees or Tenants                                                                                        | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                        |  |  |  |
| 331                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 1305 square metres of<br>agricultural land and<br>hedgerows (west of<br>Nant-y-Faenol Road) | The Executor of the Estate of the<br>Late Wynford Davies<br>Bryn Awel<br>Bodelwyddan Road<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Unknown<br>(in respect of mines and minerals) | Huw Parry Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Huw Parry Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL            | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(in respect of right to lay and<br>maintain electricity cables)<br>Unknown<br>(in respect of right of way) |  |  |  |
| 332                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 86 square metres of<br>hedgerow and track<br>(west of Nant-y-Faenol<br>Road)                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                           | NONE                                                                                                      | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                               |                                                                                                                                                                                                                                                                                             |                                                                          |                                                                                                                      |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                               |                                                                                                                                                                                                                                                                                             | Category 1                                                               |                                                                                                                      |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                               | Qualifying persons under Regulatio                                                                                                                                                                                                                                                          | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                           | Owners or Reputed Owners                                                                                                                                                                                                                                                                    | Lessees or Tenants                                                       | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 332 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                               | The Executor of the Estate of the<br>Late Wynford Davies<br>Bryn Awel<br>Bodelwyddan Road<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway )<br>Unknown                                                                                |                                                                          |                                                                                                                      |                                                                                                                                                       |  |  |  |
| 333                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 154 square metres of<br>hedgerow (north and<br>east of Nant-y-Faenol<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                        |                                     |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------|-------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning | (Applications: Prescribed Forms and | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Procedures) Regulations 2009<br>Lessees or Tenants     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 333 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                        |                                     |                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                        |                                                        |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|--------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                         | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                   | Category 1<br>on 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009 | ng (Applications: Prescribed Forms and                 | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                     | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                     | Occupiers                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 334                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 975 square metres of<br>agricultural land (north<br>and east of Nant-y-<br>Faenol Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                   | Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY | NONE                                                                                                                                                  |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                         |                                                        |                                                        |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------|--------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                         | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                                                                      | Category 1<br>on 7(1)(a) of the Infrastructure Plannin | g (Applications: Prescribed Forms and                  | Category 2                                                                                                                                            |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                         | Procedures) Regulations 2009                           | U ( FF                                                 |                                                                                                                                                       |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                     | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                | Lessees or Tenants                                     | Occupiers                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 335                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 11216 square metres of<br>agricultural land (north<br>and east of Nant-y-<br>Faenol Road, south of<br>Bodelwyddan Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                   | Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)         |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                            |                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                        |                                                        |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|--------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                            | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                   | Category 1<br>on 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                  | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                        | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                     | Occupiers                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 336                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 146 square metres of<br>agricultural land (north<br>of Nant-y-Faenol Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                   | Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY | NONE                                                                                                                                                  |  |  |  |

|                         | Land which is j              | proposed to be subject to: (i)                                                  | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|---------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Category 1                                                                                                      |                                                                                                                      |                                                                                                                                                       |
|                         |                              |                                                                                 | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                             | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                              | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 337                     | Acquisition of new rights    | 19 square metres of<br>verge of public road<br>(north of Nant-y-Faenol<br>Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |

|                         | Land which is j              | proposed to be subject to: (i)                                                            | BOOK OF RE<br>powers of compulsory acquisition, (ii) ו                                                                                                                                                                                                                                      | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                           | Qualifying persons under Regulatio                                                                                                                                                                                                                                                          | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                      |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                    | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 337 cont'd              |                              |                                                                                           | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                                                                                                                        |                                                                                                                                                              |                                                                                                                      |                                                                                                                                                       |
| 338                     | Temporary Possession         | 37 square metres of<br>agricultural land and<br>hedgerow (north of<br>Nant-y-Faenol Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                        |                                     |                                                                                                                                                       |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |
| 338 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                        |                                     |                                                                                                                                                       |  |  |

|                         | Land which is j              | proposed to be subject to: (i,                                            | BOOK OF R<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                 | rm DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) right<br>of Denbighshire | s to carry out protective works (Regulatio             | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|---------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                           | Qualifying persons under Regulati                                                                                                                                                                                                                                                                                                                                                     | Category 1<br>on 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                             | ng (Applications: Prescribed Forms and                 | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                              | Lessees or Tenants                                                                                                | Occupiers                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 339                     | Temporary Possession         | 39 square metres of<br>agricultural land (north<br>of Nant-y-Faenol Road) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD | NONE                                                                                                              | Dylan Jones<br>Clywedog<br>Rhewl<br>RUTHIN<br>LL15 2TY | NONE                                                                                                                                                  |

|                         | Land which is p              | proposed to be subject to: (i)                                                              | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                             |                                                                                                                                                                                                                                                                                                                                                                 | Category 1                                                                                                       |                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                          |
|                         |                              |                                                                                             | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                              | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                    | Category 2                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                         | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                        | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                    |
| 340                     | Temporary Possession         | 149 square metres of<br>agricultural land and<br>hedgerow (south of<br>Nant-y-Faenol Road)  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                                                                   | Unknown                                                                                                                                                                                                                                                                                                                                  |
| 341                     | Temporary Possession         | 1557 square metres of<br>agricultural land and<br>hedgerow (south of<br>Nant-y-Faenol Road) | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Unknown<br>(in respect of mines and minerals)                                                                                                                                                      | NONE                                                                                                             | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Huw Parry Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed<br>dated 12th December 2016)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for The Executor of the<br>Estate of the Late Elizabeth Alice<br>Jones) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                         | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                     |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                        |                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                  |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                |
|                         |                                                             |                                                                        | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                   | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                    | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                             | Lessees or Tenants                                                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                          |
| 341 cont'd              |                                                             |                                                                        |                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                  |                                                                                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain cables)<br>The Executor of the Estate of the Late<br>Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for<br>maintenance of pipes) |
| 342                     | Acquisition of new rights and<br>imposition of restrictions | 420 square metres of<br>public road and verges<br>(Nant-y-Faenol Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                                                                                                                                                                                                                                        |

|                         | Land which is p              | roposed to be subject to: (i | BOOK OF REFI<br>) powers of compulsory acquisition, (ii) rig                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                                    |                                         |                                                                                                                                                       |
|                         |                              |                              | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                        | (Applications: Prescribed Forms and     | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 342 cont'd              |                              |                              | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                                               |                                         |                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                          |                                                                                                                      |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 1                                                               |                                                                                                                      | Category 2                                                                                                                                            |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                            | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                  | oute8017 _                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                        | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Lessees or Tenants                                                       | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 342 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                            | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                          |                                                                                                                      |                                                                                                                                                       |  |  |  |
| 343                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 2 square metres of verge<br>of public road (west of<br>Nant-y-Faenol Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of subsoil beneath half<br>width of public highway)<br>The Executor of the Estate of the<br>Late Wynford Davies<br>Bryn Awel<br>Bodelwyddan Road<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unebighshire | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                               | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                              | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                              | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                        | Lessees or Tenants                                                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 344                     | Acquisition of new rights    | 23 square metres of<br>hedgerow and verge of<br>public road (south of<br>Nant-y-Faenol Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |
| 345                     | Temporary Possession         | 17 square metres of<br>verge of public road<br>(south of Nant-y-Faenol<br>Road)              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                                                                                                                                                            | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |

|                         | Land which is p              | proposed to be subject to: (i)                                             | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|----------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                            |                                                                                                                                                                                                                                         | Category 1                                                                                                       |                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                         |                              |                                                                            | Qualifying persons under Regulatio                                                                                                                                                                                                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                    | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                        | Owners or Reputed Owners                                                                                                                                                                                                                | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                   |
| 345 cont'd              |                              |                                                                            | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                  |                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 346                     | Acquisition of new rights    | 103 square metres of<br>agricultural land (south<br>of Nant-y-Faenol Road) | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Unknown<br>(in respect of mines and minerals)                              | NONE                                                                                                             | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Huw Parry Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed<br>dated 12th December 2016)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for The Executor of the<br>Estate of the Late Elizabeth Alice<br>Jones)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain cables) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                          |                                                                                                                                                                                                            |                                                                          |                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                          |                                                                                                                                                                                                            | Category 1                                                               |                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                          | Qualifying persons under Regulatio                                                                                                                                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                                                                                                                                                                    | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                      | Owners or Reputed Owners                                                                                                                                                                                   | Lessees or Tenants                                                       | Occupiers                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| 346 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                          |                                                                                                                                                                                                            |                                                                          |                                                                                                                                                                                                                                                                        | The Executor of the Estate of the Late<br>Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for<br>maintenance of pipes)                                                                                                                                                                                                                                                                                         |  |  |  |
| 347                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 15477 square metres of<br>agricultural land (west of<br>Little Pengwern, south<br>of Nant-y-Faenol Road) | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                     | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Huw Parry Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed<br>dated 12th December 2016)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for The Executor of the<br>Estate of the Late Elizabeth Alice<br>Jones)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain cables) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                        | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                       |                                                                                                                                                                                                            | Category 1                                                                                                       |                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                         |                              |                                                                                                                                       | Qualifying persons under Regulatio                                                                                                                                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                    | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                   | Owners or Reputed Owners                                                                                                                                                                                   | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                   |
| 347 cont'd              |                              |                                                                                                                                       |                                                                                                                                                                                                            |                                                                                                                  |                                                                                                                                                                                                                                                                        | The Executor of the Estate of the Late<br>Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for<br>maintenance of pipes)                                                                                                                                                                                                                                                                                         |
| 348                     | Temporary Possession         | 2398 square metres of<br>agricultural land and<br>hedgerows (west of<br>Little Pengwern, south<br>and west of Nant-y-<br>Faenol Road) | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Huw Parry Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed<br>dated 12th December 2016)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for The Executor of the<br>Estate of the Late Elizabeth Alice<br>Jones)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain cables) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                        | BOOK OF RE powers of compulsory acquisition, (ii)                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | to carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                      |
|-------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                       | Qualifying persons under Regulatio                                                      | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                            | Category 2                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                   | Owners or Reputed Owners                                                                | Lessees or Tenants                                                                                                | Occupiers                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                           |
| 348 cont'd              |                                                          |                                                                                                                                       |                                                                                         |                                                                                                                   |                                                                                                | The Executor of the Estate of the Late<br>Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for<br>maintenance of pipes) |
| 349                     | Acquisition of new rights and imposition of restrictions | 2525 square metres of<br>woodland (west of Little<br>Pengwern and south-<br>west of Nant-y-Faenol<br>Road)                            | Gareth Jones<br>44 Tynewydd Road<br>Y RHYL<br>LL18 3SR<br>(as reputed owner)<br>Unknown | NONE                                                                                                              | Gareth Jones<br>44 Tynewydd Road<br>Y RHYL<br>LL18 3SR<br>Unknown                              | Unknown                                                                                                                                                                                                         |
| 350                     | Temporary Possession                                     | 286 square metres of<br>woodland (west of Little<br>Pengwern, south-west<br>of Nant-y-Faenol Road)                                    | Gareth Jones<br>44 Tynewydd Road<br>Y RHYL<br>LL18 3SR<br>(as reputed owner)<br>Unknown | NONE                                                                                                              | Gareth Jones<br>44 Tynewydd Road<br>Y RHYL<br>LL18 3SR<br>Unknown                              | Unknown                                                                                                                                                                                                         |
| 351                     | Temporary Possession                                     | 11635 square metres of<br>agricultural land and<br>hedgerows (west of<br>Little Pengwern and<br>south-west of Nant-y-<br>Faenol Road) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 0DX                 | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY                    | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and<br>maintain electricity lines)                                   |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                          |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|--------------------------------------------------------------------------|-------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    | Category 1                                                               |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                       | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                  |  |  |
| 351 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                          |                                     | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of<br>service media as contained in a<br>Conveyance dated 18th November<br>1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                              |                                                                                                   | Qualifying persons under Regulatio                                                                                                                                                                         | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                               | Owners or Reputed Owners                                                                                                                                                                                   | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                   |  |
| 351 cont'd              |                              |                                                                                                   |                                                                                                                                                                                                            |                                                                                                                                                              |                                                                                                                                                                                                                                                                        | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Deed dated 26th August 1957)                                                                                                                                                                                                                                                                            |  |
| 352                     | Temporary Possession         | 5480 square metres of<br>agricultural land (west of<br>Little Pengwern and<br>Nant-y-Faenol Road) | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                                                                         | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Huw Parry Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed<br>dated 12th December 2016)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for The Executor of the<br>Estate of the Late Elizabeth Alice<br>Jones)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain cables) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                    |                                                                                                                                                                                                            |                                                                          |                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                    |                                                                                                                                                                                                            | Category 1                                                               |                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                    | Qualifying persons under Regulatio                                                                                                                                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                                                                                                                                                                    | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                | Owners or Reputed Owners                                                                                                                                                                                   | Lessees or Tenants                                                       | Occupiers                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| 352 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                    |                                                                                                                                                                                                            |                                                                          |                                                                                                                                                                                                                                                                        | The Executor of the Estate of the Late<br>Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for<br>maintenance of pipes)                                                                                                                                                                                                                                                                                         |  |  |  |
| 353                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 22471 square metres of<br>agricultural land (west of<br>Little Pengwern and<br>Nant-y-Faenol Road) | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                     | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Huw Parry Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed<br>dated 12th December 2016)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for The Executor of the<br>Estate of the Late Elizabeth Alice<br>Jones)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain cables) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | Category 1                   |                                                                                                               |                                                                                                                                                                                                            |                                                                                                                  |                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
|                         |                              |                                                                                                               | Qualifying persons under Regulatio                                                                                                                                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                    | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                           | Owners or Reputed Owners                                                                                                                                                                                   | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                       |  |
| 353 cont'd              |                              |                                                                                                               |                                                                                                                                                                                                            |                                                                                                                  |                                                                                                                                                                                                                                                                        | The Executor of the Estate of the Late<br>Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for<br>maintenance of pipes)                                                                                                                                                                                                                                                                             |  |
| 354                     | Acquisition of new rights    | 2603 square metres of<br>agricultural land (south<br>of Little Pengwern and<br>west of Nant-y-Faenol<br>Road) | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | The Executor of the Estate of the<br>Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>Huw Parry Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed<br>dated 12th December 2016)<br>HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for The Executor of the<br>Estate of the Late Elizabeth Alice<br>Jones)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                             |                                                                                                                      |                                                                                        |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                             | Qualifying persons under Regulatio                                                                                   | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                         | Owners or Reputed Owners                                                                                             | Lessees or Tenants                                                                     | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| 354 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                             |                                                                                                                      |                                                                                        |                                                                                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain cables)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Executor of the Estate of the Late<br>Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for<br>maintenance of pipes) |  |  |  |
| 355                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 11 square metres of<br>public road (Nant-y-<br>Faenol Road) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | NONE                                                                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                           | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                  | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                          |                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                  |                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                       |
|                         |                              |                                                                          | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                    | Category 2                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                      | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                              | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                 |
| 355 cont'd              |                              |                                                                          | James Dominic Sebastian Bellis<br>Woodhaven<br>Nant Y Faenol Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UL<br>(in respect of subsoil beneath half<br>width of public highway)<br>Serena Athene Bellis<br>Woodhaven<br>Nant Y Faenol Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UL<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                  |                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                       |
| 356                     | Acquisition of new rights    | 26 square metres of<br>agricultural land (west of<br>Nant-y-Faenol Road) | James Dominic Sebastian Bellis<br>Woodhaven<br>Nant Y Faenol Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UL<br>Serena Athene Bellis<br>Woodhaven<br>Nant Y Faenol Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UL                                                                                                                                                  | NONE                                                                                                             | James Dominic Sebastian Bellis<br>Woodhaven<br>Nant Y Faenol Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UL<br>Serena Athene Bellis<br>Woodhaven<br>Nant Y Faenol Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UL | Barclays Bank UK PLC<br>1 Churchill Place<br>LONDON<br>E14 5HP<br>(as mortgagee for James Dominic<br>Sebastian Bellis and Serena Athene<br>Bellis)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |

|                         | Land which is j              | proposed to be subject to: (i                                                                                                               | BOOK OF R<br>) powers of compulsory acquisition, (ii                    | arm DEVELOPMENT CONSENT ORDEF<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) righ<br>of Denbighshire | ants to carry out protective works (Regulatio                                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                             |                                                                         | Category 1                                                                                                        |                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                         |                              |                                                                                                                                             | Qualifying persons under Regulat                                        | ion 7(1)(a) of the Infrastructure Planr<br>Procedures) Regulations 2009                                           | ing (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                         | Owners or Reputed Owners                                                | Lessees or Tenants                                                                                                | Occupiers                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 357                     | Temporary Possession         | 236 square metres of<br>agricultural land and<br>hedgerow (north of<br>North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY                    | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and<br>maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of<br>service media as contained in a<br>Conveyance dated 18th November<br>1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                  |                                                                         |                                                                                                |                                                                                                |                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                  |                                                                         | Category 1                                                                                     |                                                                                                |                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                  | Qualifying persons under Regulatio                                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                       | (Applications: Prescribed Forms and                                                            | Category 2                                                                                                                                                                                                                                                                                                                      |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                              | Owners or Reputed Owners                                                | Lessees or Tenants                                                                             | Occupiers                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                           |  |  |  |
| 357 conťd               |                                                                                                                                                                                                                                                                                                      |                                                                                                  |                                                                         |                                                                                                |                                                                                                | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Deed dated 26th August 1957)                                                                                                    |  |  |  |
| 358                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 1504 square metres of<br>track (north of Tyddyn<br>Isaf Farm and North<br>Wales Expressway, A55) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 0DX | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and<br>maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                                                                                              |           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                                                                                           | Occupiers | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| 358 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                                                                                              |           | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of<br>service media as contained in a<br>Conveyance dated 18th November<br>1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Deed dated 26th August 1957) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                                                                |                                                                         |                                                                                                |                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                |                                                                         | Category 1                                                                                     |                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                | Qualifying persons under Regulatic                                      | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                      | (Applications: Prescribed Forms and                                                            | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                                                                            | Owners or Reputed Owners                                                | Lessees or Tenants                                                                             | Occupiers                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 359                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 35572 square metres of<br>agricultural land,<br>woodland and<br>hedgerows (north-east<br>of Tyddyn Isaf Farm,<br>north of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 0DX | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and<br>maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain water pipeline as contained<br>in a Deed dated 5th September 1967) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| 359 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of<br>service media as contained in a<br>Conveyance dated 18th November<br>1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Deed dated 26th August 1957) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                          |                                                                         |                                                                                                |                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                          | Qualifying persons under Regulatio                                      | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009        | (Applications: Prescribed Forms and                                                                                                                                                                                               | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                                      | Owners or Reputed Owners                                                | Lessees or Tenants                                                                             | Occupiers                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| 360                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 19135 square metres of<br>agricultural land (north<br>of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) and footpath<br>(DE 201 7) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 7)<br>P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and<br>maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain water pipeline as contained<br>in a Deed dated 5th September 1967) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| 360 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of<br>service media as contained in a<br>Conveyance dated 18th November<br>1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Deed dated 26th August 1957) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                             |                                                                         |                                                                                                |                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                             | Qualifying persons under Regulatio                                      | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009        | (Applications: Prescribed Forms and                                                            | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                         | Owners or Reputed Owners                                                | Lessees or Tenants                                                                             | Occupiers                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |
| 361                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 828 square metres of<br>agricultural land and<br>hedgerow (north of<br>North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and<br>maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of<br>service media as contained in a<br>Conveyance dated 18th November<br>1958) |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                              | BOOK OF RE powers of compulsory acquisition, (ii)                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                        | n 7(1)(a))                                                                                                                                                                                                                                                                                                                      |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                             |                                                                         | Category 1                                                                                                        |                                                                                                |                                                                                                                                                                                                                                                                                                                                 |
|                         |                                                             |                                                                                                                                             | Qualifying persons under Regulatic                                      | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                            | Category 2                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                         | Owners or Reputed Owners                                                | Lessees or Tenants                                                                                                | Occupiers                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                           |
| 361 cont'd              |                                                             |                                                                                                                                             |                                                                         |                                                                                                                   |                                                                                                | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Deed dated 26th August 1957)                                                                                                    |
| 362                     | Acquisition of new rights and<br>imposition of restrictions | 559 square metres of<br>agricultural land and<br>hedgerow (north of<br>North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY                    | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and<br>maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                             |                                                                                                              |                                                                                        |                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                             | Qualifying persons under Regulatio                                                                           | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                          | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                                         | Owners or Reputed Owners                                                                                     | Lessees or Tenants                                                                     | Occupiers                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                               |  |  |  |
| 362 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                             |                                                                                                              |                                                                                        |                                                                                                              | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of<br>service media as contained in a<br>Conveyance dated 18th November<br>1958)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Deed dated 26th August 1957) |  |  |  |
| 363                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 77 square metres of<br>hedgerow (north of<br>North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) (excluding all<br>interests of the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY | NONE                                                                                   | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)                                                                                                                                                                                                                                                                           |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                        | Category 1                                                                                                     |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                         |                              |                               | Qualifying persons under Regulatio                                                                                                     | n 7(1)(a) of the Infrastructure Plannin;<br>Procedures) Regulations 2009                                       | g (Applications: Prescribed Forms and                                                 | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                               | Lessees or Tenants                                                                                             | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 363 cont'd              |                              |                               | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                                               | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                     | n 7(1)(a))                                                                                                                                                                                                                                                                                                                |
|-------------------------|----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                                              | Qualifying persons under Regulatic                                                                           | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                         | (Applications: Prescribed Forms and                                                                          | Category 2                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                                          | Owners or Reputed Owners                                                                                     | Lessees or Tenants                                                                                              | Occupiers                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                     |
| 363 cont'd              |                                                          |                                                                                                                                                              |                                                                                                              |                                                                                                                 |                                                                                                              | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991)                                                                                     |
| 364                     | Acquisition of new rights and imposition of restrictions | 151 square metres of<br>hedgerow (north of<br>North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) (excluding all<br>interests of the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY | NONE                                                                                                            | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)                                                                                                                                 |
|                         |                                                          |                                                                                                                                                              | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY                        |                                                                                                                 | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY                        | The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                               |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-----------------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio            | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                      | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
| 364 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Unknown<br>(in respect of mines and minerals) |                                                                                        |                                     | Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991) |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i                                                                                                                                          | BOOK OF RE<br>) powers of compulsory acquisition, (ii)                                                                                                                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) righ<br>of Denbighshire                                               | ts to carry out protective works (Regulatic                                                                                                                                                           | on 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                                        | Qualifying persons under Regulation                                                                                                                                                                                                                    | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                                    | Owners or Reputed Owners                                                                                                                                                                                                                               | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 365                     | Acquisition of new rights and<br>imposition of restrictions | 2597 square metres of<br>agricultural land (north<br>of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) (excluding all<br>interests of the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                                                                         | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights t<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|-----------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulatior                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                       | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                           | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| 365 cont'd              |                              |                               |                                                          |                                                                                                              |                                         | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain overhead electricity cables<br>as contained in a Deed dated 17th<br>October 2003)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                               |                                                                         |                                                                                                |                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                               | Qualifying persons under Regulatic                                      | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009        | (Applications: Prescribed Forms and                                                            | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                           | Owners or Reputed Owners                                                | Lessees or Tenants                                                                             | Occupiers                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |
| 366                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 16 square metres of<br>agricultural land (north<br>of Tyddyn Isaf Farm and<br>North Wales<br>Expressway, A55) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and<br>maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of<br>service media as contained in a<br>Conveyance dated 18th November<br>1958) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                               |                                                                          |                                                                                                           |                                                                                                                                                                                                                              |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                               | Category 1                                                               |                                                                                                           |                                                                                                                                                                                                                              |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                             | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                            | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                       | Category 2                                                                                                                                                                                                                   |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                         | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                      | Lessees or Tenants                                                       | Occupiers                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                        |  |  |  |
| 366 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                               |                                                                          |                                                                                                           | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Deed dated 26th August 1957) |  |  |  |
| 367                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 2 square metres of<br>agricultural land and<br>hedgerows (north-east<br>of Tyddyn Isaf Farm and<br>north of North Wales<br>Expressway, A55) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(as reputed owner)<br>Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(as reputed owner)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(as reputed owner)<br>Unknown | NONE                                                                     | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY<br>Unknown | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                                                     |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                               |                                                                         |                                                                                                |                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                               | Qualifying persons under Regulatic                                      | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009        | (Applications: Prescribed Forms and                                                            | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                           | Owners or Reputed Owners                                                | Lessees or Tenants                                                                             | Occupiers                                                                                      | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| 368                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 393 square metres of<br>agricultural land (north-<br>east of Tyddyn Isaf Farm<br>and north of North<br>Wales Expressway, A55) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | P Grindal & J G Jones<br>Ty Mawr<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UY | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and<br>maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and<br>maintain water pipeline as contained<br>in a Deed dated 5th September 1967) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| 368 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of<br>service media as contained in a<br>Conveyance dated 18th November<br>1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and<br>maintain gas pipeline as contained in<br>a Deed dated 26th August 1957) |  |  |  |

|                         | Land which is                | proposed to be subject to: (i)                                                                                                                                                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                   | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) righ<br>of Denbighshire | ts to carry out protective works (Regulatic                                                                                                                                                           | on 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                                                   | Qualifying persons under Regulation                                                                                                                                                                                                                    | Category 1<br>on 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                            | ing (Applications: Prescribed Forms and                                                                                                                                                               | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                                               | Owners or Reputed Owners                                                                                                                                                                                                                               | Lessees or Tenants                                                                                              | Occupiers                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 369                     | Temporary Possession         | 10 square metres of<br>agricultural land (east of<br>Tyddyn Isaf Farm, north<br>of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) (excluding all<br>interests of the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                            | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                                                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                                                       | Qualifying persons under Regulatio                                                                           | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                                                   | Owners or Reputed Owners                                                                                     | Lessees or Tenants                                                                                                | Occupiers                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                 |
| 369 cont'd              |                              |                                                                                                                                                                                                       |                                                                                                              |                                                                                                                   |                                                                                                                                 | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991) |
| 370                     | Temporary Possession         | 27217 square metres of<br>agricultural land (north<br>of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) and footpath<br>(DE 201 7) (excluding<br>all interests of the<br>Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 7) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)                                                                                                                                                                                                                                                                             |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                                                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                        | Category 1                                                                                                     |                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                         |                              |                               | Qualifying persons under Regulation                                                                                                    | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                        | g (Applications: Prescribed Forms and                                                                                                                                                                 | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                               | Lessees or Tenants                                                                                             | Occupiers                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 370 cont'd              |                              |                               | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                          |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|--------------------------------------------------------------------------|-------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    | Category 1                                                               |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                       | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| 370 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                          |                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain overhead electricity cables<br>as contained in a Deed dated 17th<br>October 2003)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991) |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                                                                                                    | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) right<br>of Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                                                                                   | Qualifying persons under Regulatio                                                                                                                                                                    | Category 1<br>on 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                           | ng (Applications: Prescribed Forms and                                                                                                                                                                                                          | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                                                                               | Owners or Reputed Owners                                                                                                                                                                              | Lessees or Tenants                                                                                              | Occupiers                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                  |
| 371                     | Acquisition of new rights    | 830 square metres of<br>agricultural land (east of<br>Tyddyn Isaf Farm, north<br>of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) and footpath<br>(DE  201 7) (excluding<br>all interests of the<br>Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 7)<br>Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958) |
|                         |                              |                                                                                                                                                                                                                                   | Unknown<br>(in respect of mines and minerals)                                                                                                                                                         |                                                                                                                 | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY                                                                                                                                                           | Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010)                                                                                                                                                                                                                                                                                                                        |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
| 371 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain overhead electricity cables<br>as contained in a Deed dated 17th<br>October 2003) |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                                                           | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                                          | Qualifying persons under Regulatio                                                                                                                                                                    | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                                                                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                                      | Owners or Reputed Owners                                                                                                                                                                              | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                               |
| 371 cont'd              |                              |                                                                                                                                                                                          |                                                                                                                                                                                                       |                                                                                                                   |                                                                                                                                                                                                       | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991)                                                                                                                               |
| 372                     | Acquisition of new rights    | 1429 square metres of<br>agricultural land, tracks<br>and hardstanding (at<br>Tyddyn Isaf Farm, north<br>of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | NONE                                                                                                              | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus) |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|--------------------------------|-----------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                           | Category 1                                                                                                        |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                         |                              |                                | Qualifying persons under Regulatio                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                                | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 372 cont'd              |                              |                                | Unknown<br>(in respect of mines and minerals)             |                                                                                                                   |                                         | The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                 | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                |                                                                                                              | Category 1                                                                                                       |                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|                         |                              |                                                                                                                                | Qualifying persons under Regulatio                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                          | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                            | Owners or Reputed Owners                                                                                     | Lessees or Tenants                                                                                               | Occupiers                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 372 cont'd              |                              |                                                                                                                                |                                                                                                              |                                                                                                                  |                                                                                                              | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain overhead electricity cables<br>as contained in a Deed dated 17th<br>October 2003)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991) |
| 373                     | Acquisition of new rights    | 786 square metres of<br>private road (north of<br>North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY | NONE                                                                                                             | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                                                                                      | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                                                                                     | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                  |                                                                                                                                 | Category 2                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                                                                                 | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants                                                                                               | Occupiers                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                 |
| 373 cont'd              |                                                          |                                                                                                                                                                                                     | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals)                       |                                                                                                                  | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY                                           | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain cables and right of access)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in<br>Transfer dated 11 March 1991) |
| 374                     | Acquisition of new rights and imposition of restrictions | 10879 square metres of<br>agricultural land (north<br>of North Wales<br>Expressway, A55 and<br>east of Tyddyn Isaf<br>Farm) and footpath<br>(DE 201 7) (excluding<br>all interests of the<br>Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY                                                 | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 7) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)                                                                                                                                                                             |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri                                                                               | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulatio                                                                                                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                        | Category 1                                                                                                  |                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                         |                              |                               | Qualifying persons under Regulation                                                                                                    | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                     | g (Applications: Prescribed Forms and                                                                                                                                                                 | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                               | Lessees or Tenants                                                                                          | Occupiers                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 374 cont'd              |                              |                               | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) |                                                                                                             | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                          |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|--------------------------------------------------------------------------|-------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                    | Category 1                                                               |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                       | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| 374 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                          |                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain overhead electricity cables<br>as contained in a Deed dated 17th<br>October 2003)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991) |  |  |

|                         | Land which is j              | proposed to be subject to: (i)                                                                                                                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii) ا                                                                                                                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) righ<br>ff Denbighshire | ts to carry out protective works (Regulatic                                                                                                                                                           | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                    | Qualifying persons under Regulation                                                                                                                                                                                                                    | Category 1<br>on 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                           | ing (Applications: Prescribed Forms and                                                                                                                                                               | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                | Owners or Reputed Owners                                                                                                                                                                                                                               | Lessees or Tenants                                                                                             | Occupiers                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 375                     | Temporary Possession         | 56 square metres of<br>agricultural land (north<br>of North Wales<br>Expressway, A55 and<br>east of Tyddyn Isaf<br>Farm) (excluding all<br>interests of the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                           | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulatior                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                        | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| 375 cont'd              |                              |                               |                                                          |                                                                                                               |                                         | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain overhead electricity cables<br>as contained in a Deed dated 17th<br>October 2003)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991) |

|                         | Land which is j                                             | proposed to be subject to: (i)                                                                                                                                                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                  | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) righ<br>of Denbighshire | ts to carry out protective works (Regulatio                                                                                                                                                                                                     | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                                                                       | Qualifying persons under Regulation                                                                                                                                                                   | Category 1<br>on 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                            | ing (Applications: Prescribed Forms and                                                                                                                                                                                                         | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                                                                   | Owners or Reputed Owners                                                                                                                                                                              | Lessees or Tenants                                                                                              | Occupiers                                                                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                  |
| 376                     | Acquisition of new rights and<br>imposition of restrictions | 16642 square metres of<br>agricultural land (north<br>of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) and footpath<br>(DE[201 7) (excluding<br>all interests of the<br>Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 7)<br>Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958) |
|                         |                                                             |                                                                                                                                                                                                       | Unknown<br>(in respect of mines and minerals)                                                                                                                                                         |                                                                                                                 | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY                                                                                                                                                           | Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010)                                                                                                                                                                                                                                                                                                                        |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
| 376 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain overhead electricity cables<br>as contained in a Deed dated 17th<br>October 2003) |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                            | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                           |                                                                                                                                                        | Category 1                                                                                                       |                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                         |                              |                                                                                                                                           | Qualifying persons under Regulatio                                                                                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                       | Owners or Reputed Owners                                                                                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                 |
| 376 cont'd              |                              |                                                                                                                                           |                                                                                                                                                        |                                                                                                                  |                                                                                                       | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991)                                                                                                                                                                                                                 |
| 377                     | Acquisition of new rights    | 122 square metres of<br>private road and verges<br>(north of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of restrictive covenants)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of restrictive covenants)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for<br>maintenance) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                                                         | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                               |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                                                        | Qualifying persons under Regulatio                                                                                                                                              | Category 2                                                                                                       |                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                                                    | Owners or Reputed Owners                                                                                                                                                        | Procedures) Regulations 2009<br>Lessees or Tenants                                                               | Occupiers                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                    |
| 378                     | Acquisition of new rights and imposition of restrictions | 2220 square metres of<br>agricultural land (north<br>of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) (excluding all<br>interests of the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan | NONE                                                                                                             | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and<br>maintain cables and right of access<br>stated in Deed dated 17 October<br>2003)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON |
|                         |                                                          |                                                                                                                                                                        | Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals)                                                                                                             |                                                                                                                  | Y RHYL<br>LL18 5UY                                                                                                                                                              | Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in<br>Transfer dated 11 March 1991)                                                    |
| 379                     | Acquisition of new rights                                | 122 square metres of<br>agricultural land (north<br>of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) (excluding all<br>interests of the Crown)  | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY                                                                    | NONE                                                                                                             | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY                                                                    | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958)                                                                                                                |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                        | Category 1                                                                                                     |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                         |                              |                               | Qualifying persons under Regulatio                                                                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | g (Applications: Prescribed Forms and                                                 | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                               | Lessees or Tenants                                                                                             | Occupiers                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 379 cont'd              |                              |                               | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | The Executor of the Estate of the Late<br>Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for<br>maintenance and right to free<br>passage of service media as<br>contained in a Conveyance dated<br>18th November 1958)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(as beneficiary of an Option<br>Agreement dated 29th June 2010)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for<br>maintenance as contained in a<br>Conveyance dated 18th November<br>1958) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                                        | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                                                                                  | n 7(1)(a))                                                                                                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                       |                                                                                                                                                                                                                                                        | Category 1                                                                                                       |                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                       |
|                         |                              |                                                                                                                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                      | Category 2                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                   | Owners or Reputed Owners                                                                                                                                                                                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                 |
| 379 cont'd              |                              |                                                                                                                                                                       |                                                                                                                                                                                                                                                        |                                                                                                                  |                                                                                                                                                                                                                                                                                                                                          | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain as<br>contained in a Transfer dated 11th<br>March 1991) |
| 380                     | Acquisition of new rights    | 1383 square metres of<br>private road and verges<br>(north of North Wales<br>Expressway, A55) and<br>footpath (DE 201 7)<br>(excluding all interests of<br>the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 7)<br>Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in<br>Transfer dated 11 March 1991)                                                    |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                                       | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                                                                                                                                 | n 7(1)(a))                                                                                                                                                                         |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                                      | Qualifying persons under Regulatio                                                                                                                                                                                                                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                                                      | Category 2                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                                  | Owners or Reputed Owners                                                                                                                                                                                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                              |
| 381                     | Acquisition of new rights and<br>imposition of restrictions | 945 square metres of<br>private road and verges<br>(north of North Wales<br>Expressway, A55) and<br>footpath (DE 201 7)<br>(excluding all interests of<br>the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 7)<br>Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in<br>Transfer dated 11 March 1991) |
| 382                     | Acquisition of new rights and imposition of restrictions    | 249 square metres of<br>verge (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                                          | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP                                                                                                                                                  | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                                                                                                                                     | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of restrictive covenants)                              |

|                         | Land which is p                                          | roposed to be subject to: (i)                                                                               | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                   |
|-------------------------|----------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                             | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                  |                                                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                         | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                        |
| 382 cont'd              |                                                          |                                                                                                             | Unknown<br>(in respect of mines and minerals)                                                                                                                |                                                                                                                  | North and Mid Wales Trunk Road<br>Agency<br>Unit 5<br>Llys Britannia<br>Ffordd Y Parc<br>Parc Menai<br>BANGOR<br>LL57 4BN<br>(as maintainer of North Wales<br>Expressway) | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of restrictive covenants)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for<br>maintenance) |
| 383                     | Acquisition of new rights and imposition of restrictions | 147 square metres of<br>verge (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                 | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                      | NONE                                                                                                                                                                                                                                                                                         |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                   | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                                           | n 7(1)(a))                                                                                                                                                                                                                                                                                                                            |
|-------------------------|----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                    | Qualifying persons under Regulatio                                                       | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                               | Category 2                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                | Owners or Reputed Owners                                                                 | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                 |
| 383 cont'd              |                                                          |                                                                                                                    |                                                                                          |                                                                                                                  | North and Mid Wales Trunk Road<br>Agency<br>Unit 5<br>Llys Britannia<br>Ffordd Y Parc<br>Parc Menai<br>BANGOR<br>LL57 4BN<br>(as maintainer of North Wales<br>Expressway)                                                                                                                         |                                                                                                                                                                                                                                                                                                                                       |
| 384                     | Acquisition of new rights and imposition of restrictions | 1481 square metres of<br>public road (North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>North and Mid Wales Trunk Road<br>Agency<br>Unit 5<br>Llys Britannia<br>Ffordd Y Parc<br>Parc Menai<br>BANGOR<br>LL57 4BN<br>(as maintainer of North Wales<br>Expressway) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right to enter and right<br>to service media)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of right to enter and right<br>to service media) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                              | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                          | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                                           | n 7(1)(a))                                                                                                                                                                                                                                                                                                                            |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         | Category 1                                                  |                                                                                                             |                                                                                                                               |                                                                                                                     |                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                       |
|                         |                                                             |                                                                                                             | Qualifying persons under Regulatic                                                                                            | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                           | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                               | Category 2                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                         | Owners or Reputed Owners                                                                                                      | Lessees or Tenants                                                                                                  | Occupiers                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                 |
| 385                     | Acquisition of new rights and<br>imposition of restrictions | 194 square metres of<br>verge (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                      | NONE                                                                                                                | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                                                                                                                          | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right to enter and right<br>to service media)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of right to enter and right<br>to service media) |
| 386                     | Acquisition of new rights and<br>imposition of restrictions | 359 square metres of<br>verge (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>North and Mid Wales Trunk Road<br>Agency<br>Unit 5<br>Llys Britannia<br>Ffordd Y Parc<br>Parc Menai<br>BANGOR<br>LL57 4BN<br>(as maintainer of North Wales<br>Expressway) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                         |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                   |                                                                                                                                                                                                                                                           |                                                                          |                                                                                                                                                                                                                                                                                                   |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                   |                                                                                                                                                                                                                                                           | Category 1                                                               |                                                                                                                                                                                                                                                                                                   | Category 2                                                                                                                                            |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                               | category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                               | Owners or Reputed Owners                                                                                                                                                                                                                                  | Lessees or Tenants                                                       | Occupiers                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 387                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 782 square metres of<br>public road (North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath<br>public highway) | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>North and Mid Wales Trunk Road<br>Agency<br>Unit 5<br>Llys Britannia<br>Ffordd Y Parc<br>Parc Menai<br>BANGOR<br>LL57 4BN<br>(as maintainer of North Wales<br>Expressway) | Unknown                                                                                                                                               |  |  |  |
| 388                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 39 square metres of<br>verge (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)        | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                                                                                              | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                                                                                              | NONE                                                                                                                                                  |  |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                  | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                                                                                                                                                                                          | n 7(1)(a))                                                                                                                                            |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                 | Qualifying persons under Regulatio                                           | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                               | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                             | Owners or Reputed Owners                                                     | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 388 cont'd              |                                                             |                                                                                                                                 |                                                                              |                                                                                                                  | North and Mid Wales Trunk Road<br>Agency<br>Unit 5<br>Llys Britannia<br>Ffordd Y Parc<br>Parc Menai<br>BANGOR<br>LL57 4BN<br>(as maintainer of North Wales<br>Expressway)                                                                                                                         |                                                                                                                                                       |
| 389                     | Acquisition of new rights and<br>imposition of restrictions | 145 square metres of<br>public road and verges<br>(North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>North and Mid Wales Trunk Road<br>Agency<br>Unit 5<br>Llys Britannia<br>Ffordd Y Parc<br>Parc Menai<br>BANGOR<br>LL57 4BN<br>(as maintainer of North Wales<br>Expressway) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)       |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                               | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                         | Category 1<br>n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                           | g (Applications: Prescribed Forms and                                                                                | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Lessees or Tenants                                                                                              | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 390                     | Acquisition of new rights and<br>imposition of restrictions | 5 square metres of<br>public road and verges<br>(North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath half<br>width of public highway)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |
| 391                     | Acquisition of new rights and imposition of restrictions    | 4 square metres of<br>verges (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                    | National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as reputed owner)                                                                                                                                                                                                                                                                                                                                                                 | NONE                                                                                                            | National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)         |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                            | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                          | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                           | Qualifying persons under Regulatio                                                                                                                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                              | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                       | Owners or Reputed Owners                                                                                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                          |
| 391 cont'd              |                                                             |                                                                                                           | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(as reputed owner)<br>Unknown                 |                                                                                                                  | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>Unknown | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 392                     | Acquisition of new rights and<br>imposition of restrictions | 562 square metres of<br>verge(North Wales<br>Expressway,<br>A55)(excluding all<br>interests of the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP            | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for<br>maintenance) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                 | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                                                      | n 7(1)(a))                                                                                                                                                 |  |  |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                             |                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Category 1                                                                                                      |                                                                                                                                                                                                                                                                                                              |                                                                                                                                                            |  |  |
|                         |                                                             |                                                                                                                                | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                             | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and                                                                                                                                                                                                                                                                          | Category 2                                                                                                                                                 |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                            | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                              | Occupiers                                                                                                                                                                                                                                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009      |  |  |
| 392 cont'd              |                                                             |                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                 |                                                                                                                                                                                                                                                                                                              | Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunications<br>apparatus)                                    |  |  |
| 393                     | Acquisition of new rights and<br>imposition of restrictions | 21 square metres of<br>public road and verges<br>(North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of subsoil beneath half<br>width of public highway)<br>The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>North and Mid Wales Trunk Road<br>Agency<br>Unit 5<br>Llys Britannia<br>Ffordd Y Parc<br>Parc Menai<br>BANGOR<br>LL57 4BN<br>(as maintainer of North Wales<br>Expressway)<br>Unknown | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                   | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                      | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                                                                                           | n 7(1)(a))                                                                                                                                                                                                       |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                  | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                  |                                                                                                                                                                                                                                                                                                   | Category 2                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                              | Owners or Reputed Owners                                                                                                                                     | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                            |
| 394                     | Acquisition of new rights and<br>imposition of restrictions | 14 square metres of<br>public road (North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown) | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>Unknown<br>(in respect of mines and minerals)                                | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>North and Mid Wales Trunk Road<br>Agency<br>Unit 5<br>Llys Britannia<br>Ffordd Y Parc<br>Parc Menai<br>BANGOR<br>LL57 4BN<br>(as maintainer of North Wales<br>Expressway) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect to rights granted by Deed<br>dated 6 June 1958 which also<br>contains restrictive covenants) |
| 395                     | Acquisition of new rights and imposition of restrictions    | 175 square metres of<br>verges (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)     | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                 | NONE                                                                                                             | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                                                                                                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect to rights granted by Deed<br>dated 6 June 1958 which also<br>contains restrictive covenants) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                 | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                               | n 7(1)(a))                                                                                                                                                                                                                                                               |
|-------------------------|----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                |                                                                                                                                                            | Category 1                                                                                                       |                                                                                                       |                                                                                                                                                                                                                                                                          |
|                         |                                                          |                                                                                                                                | Qualifying persons under Regulatio                                                                                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                   | Category 2                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                            | Owners or Reputed Owners                                                                                                                                   | Lessees or Tenants                                                                                               | Occupiers                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                    |
| 395 cont'd              |                                                          |                                                                                                                                | Unknown<br>(in respect of mines and minerals)                                                                                                              |                                                                                                                  |                                                                                                       | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunications<br>apparatus) |
| 396                     | Acquisition of new rights and imposition of restrictions | 104 square metres of<br>public road and verge<br>(North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                             | NONE                                                                                                  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                                                                                                 |
| 397                     | Acquisition of new rights and imposition of restrictions | 67 square metres of<br>verges (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                    | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP                                                      | NONE                                                                                                             | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                                                                                                          |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                    | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|----------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                   |                                                                                                       | Category 1                                                                                                        |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|                         |                                                          |                                                                                                                   | Qualifying persons under Regulatio                                                                    | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                               | Owners or Reputed Owners                                                                              | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                  |
| 397 cont'd              |                                                          |                                                                                                                   | Unknown<br>(in respect of mines and minerals)                                                         |                                                                                                                   |                                                                                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for<br>maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunications<br>apparatus) |
| 398                     | Acquisition of new rights and imposition of restrictions | 418 square metres of<br>public road (North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>Denbighshire | to carry out protective works (Regulatio                                                                                                                                  | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulation                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                      | g (Applications: Prescribed Forms and                                                                                                                                     | Category 2                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                          | Occupiers                                                                                                                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                     |
| 398 cont'd              |                              |                               | Unknown<br>(in respect of mines and minerals)            |                                                                                                             | North and Mid Wales Trunk Road<br>Agency<br>Unit 5<br>Llys Britannia<br>Ffordd Y Parc<br>Parc Menai<br>BANGOR<br>LL57 4BN<br>(as maintainer of North Wales<br>Expressway) | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of restrictive covenants)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET |
|                         |                              |                               |                                                          |                                                                                                             |                                                                                                                                                                           | (in respect of right of access for<br>maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                       |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                         | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                                                                   | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulation                                                 | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|----------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                        |                                                                                                                                           | Category 1                                                                                                        |                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|                         |                                                          |                                                                                                                                        | Qualifying persons under Regulatio                                                                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                    | Owners or Reputed Owners                                                                                                                  | Lessees or Tenants                                                                                                | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                |
| 398 cont'd              |                                                          |                                                                                                                                        |                                                                                                                                           |                                                                                                                   |                                                                                          | Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunications<br>apparatus)                                                                                                                                                                                                                                                                                                                                                              |
| 399                     | Acquisition of new rights and imposition of restrictions | 86 square metres of<br>public road and verge<br>(south of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                              | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and<br>maintain gas pipeline)<br>Kent Periscopes Limited<br>Dowlish Ford<br>ILMINSTER<br>TA19 0PF<br>(in respect of right to service media<br>and right to enter)<br>North Wales Fire Authority<br>Site G<br>Ffordd Salesbury<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JJ<br>(in respect of rights stated in Lease<br>dated 26th April 2005) |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|--------------------------------|----------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                          | Category 1                                                                                                     |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                         |                              |                                | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                             | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 399 cont'd              |                              |                                |                                                          |                                                                                                                |                                         | NW (BBP) Limited<br>3 More London Riverside<br>LONDON<br>SE1 2AQ<br>(in respect of rights stated in Lease<br>dated 9th October 2002)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Police & Crime Commissioner for<br>North Wales<br>North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWYN BAY<br>Conwy<br>LL29 8AW<br>(in respect of rights stated in Lease<br>dated 1st March 2000)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access along<br>roadways and right to lay and<br>maintain gas pipeline) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulation                      | Category 1                                                                                                    | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Procedures) Regulations 2009<br>Lessees or Tenants                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 399 cont'd              |                              |                               |                                                          |                                                                                                               |                                         | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Star Units Limited<br>San Remo<br>Trinity Square<br>Llandudno<br>Conwy<br>LL30 2RB<br>(as beneficiary of an Option<br>Agreement dated 27th January 2021)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access along<br>roadways and right to free passage of<br>service media)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access along<br>roadways) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                  | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|----------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                               |                                                                                          | Category 1                                                                                                       |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                         |                                                          |                                                                                                                               | Qualifying persons under Regulatio                                                       | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                           | Owners or Reputed Owners                                                                 | Lessees or Tenants                                                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                           |
| 399 cont'd              |                                                          |                                                                                                                               |                                                                                          |                                                                                                                  |                                                                                                                      | WGIF (Jersey) Trustee I Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease<br>dated 26th April 2005)<br>WGIF (Jersey) Trustee II Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease<br>dated 26th April 2005)<br>Wrexham Glyndwr University<br>Plas Coch<br>Mold Road<br>WREXHAM<br>LL11 2AW<br>(in respect of rights stated in Lease<br>dated 9th July 2013) |
| 400                     | Acquisition of new rights and imposition of restrictions | 622 square metres of<br>public road (south of<br>North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and<br>maintain gas pipeline)                                                                                                                                                                                                                                                                                                                                                                                   |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                               |                                                                                       |                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-----------------------------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio            | Category 1<br>n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                    | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                      | Lessees or Tenants                                                                    | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
| 400 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Unknown<br>(in respect of mines and minerals) |                                                                                       | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ | Kent Periscopes Limited<br>Dowlish Ford<br>ILMINSTER<br>TA19 OPF<br>(in respect of right to service media<br>and right to enter)<br>North Wales Fire Authority<br>Site G<br>Ffordd Salesbury<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJJ<br>(in respect of rights stated in Lease<br>dated 26th April 2005)<br>NW (BBP) Limited<br>3 More London Riverside<br>LONDON<br>SE1 2AQ<br>(in respect of rights stated in Lease<br>dated 9th October 2002)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
| 400 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Police & Crime Commissioner for<br>North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWYN BAY<br>Conwy<br>LL29 8AW<br>(in respect of rights stated in Lease<br>dated 1st March 2000)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access along<br>roadways and right to lay and<br>maintain gas pipeline)<br>Star Units Limited<br>San Remo<br>Trinity Square<br>Llandudno<br>Conwy<br>LL30 2RB<br>(as beneficiary of an Option<br>Agreement dated 27th January 2021) |  |  |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|--------------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                          | Category 1                                                                                                    |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                              |                                | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                      | Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 400 cont'd              |                              |                                |                                                          |                                                                                                               |                                         | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access along<br>roadways and right to free passage of<br>service media)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access along<br>roadways)<br>WGIF (Jersey) Trustee I Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease<br>dated 26th April 2005)<br>WGIF (Jersey) JE1 2LH<br>(in respect of rights stated in Lease<br>dated 26th April 2005) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                                               | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                   | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                  | n 7(1)(a))                                                                                                                                            |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                                              | Qualifying persons under Regulatio                                                       | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                      | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                                          | Owners or Reputed Owners                                                                 | Lessees or Tenants                                                                                               | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 400 cont'd              |                                                             |                                                                                                                                                                              |                                                                                          |                                                                                                                  |                                                                                          | Wrexham Glyndwr University<br>Plas Coch<br>Mold Road<br>WREXHAM<br>LL11 2AW<br>(in respect of rights stated in Lease<br>dated 9th July 2013)          |
| 401                     | Acquisition of new rights and<br>imposition of restrictions | 583 square metres of<br>agricultural land,<br>hedgerow and verge of<br>public road (south of<br>North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ | NONE                                                                                                             | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY       | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and<br>maintain gas pipeline)         |
|                         |                                                             |                                                                                                                                                                              | Unknown<br>(in respect of mines and minerals)                                            |                                                                                                                  | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of bridleway 201 10)                        |
|                         |                                                             |                                                                                                                                                                              |                                                                                          |                                                                                                                  | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                              |                                                                          |                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                              | Category 1                                                               |                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                        | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                                                    | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                     | Lessees or Tenants                                                       | Occupiers                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| 401 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                              |                                                                          |                                                                                                                                 | Star Units Limited<br>San Remo<br>Trinity Square<br>Llandudno<br>Conwy<br>LL30 2RB<br>(as beneficiary of an Option<br>Agreement dated 27th January 2021)                                                                                                                                                                                                                                                                                                                                                                             |  |  |  |
| 402                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 10 square metres of<br>public road (Old A55<br>Faenol Bropor, south of<br>North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to<br>maintain drainage and right of<br>support)<br>HSBC Bank PLC<br>8 Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998) |  |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                             | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                   | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                              |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                            |                                                                                                                                           | Category 1                                                                                                       |                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                         |
|                         |                                                             |                                                                                                                                                            | Qualifying persons under Regulatio                                                                                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                        | Owners or Reputed Owners                                                                                                                  | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                   |
| 402 cont'd              |                                                             |                                                                                                                                                            | Unknown<br>(in respect of mines and minerals)                                                                                             |                                                                                                                  |                                                                                                                                                                               | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service<br>media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage) |
| 403                     | Acquisition of new rights and<br>imposition of restrictions | 905 square metres of<br>agricultural land,<br>hedgerow and verge<br>(south of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and<br>maintain gas pipeline)<br>Star Units Limited<br>San Remo<br>Trinity Square<br>Llandudno<br>Conwy<br>LL30 2RB<br>(as beneficiary of an Option<br>Agreement dated 27th January 2021)                               |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                               | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                   | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                  | n 7(1)(a))                                                                                                                                                                                      |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                              | Qualifying persons under Regulatio                                                       | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                      | Category 2                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                          | Owners or Reputed Owners                                                                 | Lessees or Tenants                                                                                               | Occupiers                                                                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                           |
| 403 cont'd              |                              |                                                                                                                                                              |                                                                                          |                                                                                                                  | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ |                                                                                                                                                                                                 |
| 404                     | Temporary Possession         | 1479 square metres of<br>agricultural land, verges<br>and hedgerow (south of<br>North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ | NONE                                                                                                             | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY       | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and<br>maintain gas pipeline)                                                   |
|                         |                              |                                                                                                                                                              | Unknown<br>(in respect of mines and minerals)                                            |                                                                                                                  | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of bridleway 201 10)                                                                  |
|                         |                              |                                                                                                                                                              |                                                                                          |                                                                                                                  | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ | North Wales Fire Authority<br>Site G<br>Ffordd Salesbury<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJJ<br>(in respect of rights stated in Lease<br>dated 26th April 2005) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulation                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                        | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 404 cont'd              |                              |                               |                                                          |                                                                                                               |                                         | NW (BBP) Limited<br>3 More London Riverside<br>LONDON<br>SE1 2AQ<br>(in respect of rights stated in Lease<br>dated 9th October 2002)<br>Police & Crime Commissioner for<br>North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWYN BAY<br>Conwy<br>LL29 8AW<br>(in respect of rights stated in Lease<br>dated 1st March 2000)<br>Star Units Limited<br>San Remo<br>Trinity Square<br>Llandudno<br>Conwy<br>LL30 2RB<br>(as beneficiary of an Option<br>Agreement dated 27th January 2021)<br>WGIF (Jersey) Trustee I Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease<br>dated 26th April 2005) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                    | Qualifying persons under Regulatio                                                                                                                                            | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                | Owners or Reputed Owners                                                                                                                                                      | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                  |
| 404 cont'd              |                              |                                                                                                                                                    |                                                                                                                                                                               |                                                                                                                   |                                                                                                                                                                               | WGIF (Jersey) Trustee II Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease<br>dated 26th April 2005)<br>Wrexham Glyndwr University<br>Plas Coch<br>Mold Road<br>WREXHAM<br>LL11 2AW<br>(in respect of rights stated in Lease<br>dated 9th July 2013)                 |
| 405                     | Temporary Possession         | 77 square metres of<br>agricultural land and<br>hedgerow (south of<br>North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | NONE                                                                                                              | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain) |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                                     |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                              |                               | Qualifying persons under Regulation                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                             | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 405 cont'd              |                              |                               | Unknown<br>(in respect of mines and minerals)              |                                                                                                                |                                         | HSBC Bank PLC<br>& Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to<br>maintain drainage and right of<br>support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 &FZ<br>(in respect of right of support and<br>right to free passage of service<br>media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                      | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                                                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                     |                                                                                                                                                                                                                                | C.I                                                                                                            |                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                         |                              |                                                                                                                                     | Qualifying persons under Regulatio                                                                                                                                                                                             | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                        | g (Applications: Prescribed Forms and                                                                                                                                         | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                 | Owners or Reputed Owners                                                                                                                                                                                                       | Lessees or Tenants                                                                                             | Occupiers                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 406                     | Temporary Possession         | 509 square metres of<br>agricultural land (south<br>of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                           | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain)<br>HSBC Bank PLC<br>8 Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to<br>maintain drainage and right of<br>support) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                   | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                       | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                  | Qualifying persons under Regulatio                                                                                                                                            | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                              | Owners or Reputed Owners                                                                                                                                                      | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                  |
| 406 cont'd              |                              |                                                                                                                                                  |                                                                                                                                                                               |                                                                                                                  |                                                                                                                                                                               | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service<br>media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage)                |
| 407                     | Acquisition of new rights    | 54 square metres of<br>track and verge (Faenol-<br>Bropor, south of North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | NONE                                                                                                             | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain) |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                                    |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|                         |                              |                               | Qualifying persons under Regulation                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                      | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 407 cont'd              |                              |                               | Unknown<br>(in respect of mines and minerals)              |                                                                                                               |                                         | HSBC Bank PLC<br>8 Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to<br>maintain drainage and right of<br>support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service media<br>as contained in a Conveyance dated<br>11th March 1991)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of support and<br>right to free passage of service media<br>as contained in a Conveyance dated<br>11th March 1991)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                                               | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                 | n 7(1)(a))                                                                                                                                                                     |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                                              | Qualifying persons under Regulatio                                                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                     | Category 2                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                                          | Owners or Reputed Owners                                                                | Lessees or Tenants                                                                                               | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                          |
| 407 cont'd              |                                                             |                                                                                                                                                                              |                                                                                         |                                                                                                                  |                                                                                         | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage)                                               |
| 408                     | Acquisition of new rights and<br>imposition of restrictions | 499 square metres of<br>agricultural land,<br>hedgerow and verge<br>(Faenol-Bropor, south of<br>North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY      | NONE                                                                                                             | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY      | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                |
|                         |                                                             |                                                                                                                                                                              | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY |                                                                                                                  | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)           |
|                         |                                                             |                                                                                                                                                                              | Unknown<br>(in respect of mines and minerals)                                           |                                                                                                                  |                                                                                         | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain) |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|--------------------------------|---------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                         | Category 1                                                                                                    |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                              |                                | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                      | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 408 cont'd              |                              |                                |                                                         |                                                                                                               |                                         | HSBC Bank PLC<br>& Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to<br>maintain drainage and right of<br>support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 &FZ<br>(in respect of right of support and<br>right to free passage of service<br>media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                    | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                                                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                   |                                                                                                                                                                                                                                | Category 1                                                                                                     |                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                         |                                                             |                                                                                                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                             | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                               | Owners or Reputed Owners                                                                                                                                                                                                       | Lessees or Tenants                                                                                             | Occupiers                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 409                     | Acquisition of new rights and<br>imposition of restrictions | 120 square metres of<br>track and verge (Faenol-<br>Bropor, south of North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                           | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain)<br>HSBC Bank PLC<br>8 Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to<br>maintain drainage and right of<br>support) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                                               | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                        | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                                              |                                                                                                                                                                               | Category 1                                                                                                        |                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                        |
|                         |                                                             |                                                                                                                                                                              | Qualifying persons under Regulatio                                                                                                                                            | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                                          | Owners or Reputed Owners                                                                                                                                                      | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                  |
| 409 cont'd              |                                                             |                                                                                                                                                                              |                                                                                                                                                                               |                                                                                                                   |                                                                                                                                                                               | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service<br>media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage)                |
| 410                     | Acquisition of new rights and<br>imposition of restrictions | 386 square metres of<br>agricultural land,<br>hedgerow and verge<br>(Faenol-Bropor, south of<br>North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | NONE                                                                                                              | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>) powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                                     |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                         |                              |                               | Qualifying persons under Regulation                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                             | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 410 cont'd              |                              |                               | Unknown<br>(in respect of mines and minerals)              |                                                                                                                |                                         | HSBC Bank PLC<br>8 Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to<br>maintain drainage and right of<br>support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service media<br>as contained in a Conveyance dated<br>11th March 1991)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service media<br>as contained in a Conveyance dated<br>11th March 1991)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                               | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                 | n 7(1)(a))                                                                                                                                                                     |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                              | Qualifying persons under Regulatio                                                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                     | Category 2                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                          | Owners or Reputed Owners                                                                | Lessees or Tenants                                                                                               | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                          |
| 410 cont'd              |                              |                                                                                                                                                              |                                                                                         |                                                                                                                  |                                                                                         | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage)                                               |
| 411                     | Temporary Possession         | 1198 square metres of<br>agricultural land (Faenol-<br>bropor farm, south of<br>North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY      | NONE                                                                                                             | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY      | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)           |
|                         |                              |                                                                                                                                                              | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY |                                                                                                                  | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain) |
|                         |                              |                                                                                                                                                              | Unknown<br>(in respect of mines and minerals)                                           |                                                                                                                  |                                                                                         | HSBC Bank PLC<br>8 Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)                  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                                                         | BOOK OF RE powers of compulsory acquisition, (ii)                                  | TT DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights to<br>f Denbighshire | to carry out protective works (Regulatio                                           | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                                        |                                                                                    | Category 1                                                                                                          |                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                         |                              |                                                                                                                                                                                        | Qualifying persons under Regulatic                                                 | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                           | (Applications: Prescribed Forms and                                                | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                                    | Owners or Reputed Owners                                                           | Lessees or Tenants                                                                                                  | Occupiers                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                        |
| 411 cont'd              |                              |                                                                                                                                                                                        |                                                                                    |                                                                                                                     |                                                                                    | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to<br>maintain drainage and right of<br>support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service<br>media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage) |
| 412                     | Acquisition of new rights    | 312 square metres of<br>agricultural land and<br>private road (east of<br>Faenol-bropor farm,<br>south of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | NONE                                                                                                                | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)                                                                                                                                                                                                                                                                                                                                                         |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r                                                                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                                                                                                          | Category 2                                                                                                     |                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|                         |                              |                               | Qualifying persons under Regulation                                                                                                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                                                                                                 | Lessees or Tenants                                                                                             | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| 412 cont'd              |                              |                               | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) |                                                                                                                | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain)<br>HSBC Bank PLC<br>8 Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to<br>maintain drainage and right of<br>support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service<br>media) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                                             | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire                                             | o carry out protective works (Regulatio                                                 | n 7(1)(a))                                                                                                                                                                     |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                                            | Qualifying persons under Regulatio                                                      | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                         |                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                                        | Owners or Reputed Owners                                                                | Lessees or Tenants                                                                                                                                           | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                          |
| 412 cont'd              |                                                             |                                                                                                                                                                            |                                                                                         |                                                                                                                                                              |                                                                                         | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage)                                               |
| 413                     | Acquisition of new rights and<br>imposition of restrictions | 26731 square metres of<br>agricultural land, track,<br>hedgerows (Faenol-<br>Bropor, south of North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY      | NONE                                                                                                                                                         | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY      | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)                                |
|                         |                                                             |                                                                                                                                                                            | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY |                                                                                                                                                              | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)           |
|                         |                                                             |                                                                                                                                                                            | Unknown<br>(in respect of mines and minerals)                                           |                                                                                                                                                              |                                                                                         | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain) |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|------------------------------|--------------------------------|---------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                         | Category 1                                                                                                       |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|                         |                              |                                | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 413 cont'd              |                              |                                |                                                         |                                                                                                                  |                                         | HSBC Bank PLC<br>8 Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to<br>maintain drainage and right of<br>support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service media<br>as contained in a Conveyance dated<br>11th March 1991)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of support and<br>right to free passage of service media<br>as contained in a Conveyance dated<br>11th March 1991)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                                                        | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                           | n 7(1)(a))                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                                       | Qualifying persons under Regulatio                                                                                                                                | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                                                                                               | Category 2                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                                   | Owners or Reputed Owners                                                                                                                                          | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                           |
| 413 cont'd              |                              |                                                                                                                                                                                       |                                                                                                                                                                   |                                                                                                                   |                                                                                                                                                                   | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage)                                                                                                                                                |
| 414                     | Temporary Possession         | 2872 square metres of<br>agricultural land, ditch<br>and hedgerow (east of<br>Faenol-Bropor, south of<br>North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL | NONE                                                                                                              | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP   |
|                         |                              |                                                                                                                                                                                       | LL18 5UY<br>Unknown<br>(in respect of mines and minerals)                                                                                                         |                                                                                                                   | LL18 5UY                                                                                                                                                          | <ul> <li>(in respect of right of access for maintenance of fence and drain)</li> <li>HSBC Bank PLC</li> <li>8 Canada Square</li> <li>LONDON</li> <li>E14 5HQ</li> <li>(as mortgagee for John Berwyn Evans and Eirian Evans stated in Charge dated 28th January 1998)</li> </ul> |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                                                         | BOOK OF RE<br>powers of compulsory acquisition, (ii)                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | to carry out protective works (Regulatio                                           | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009           |                                                                                    |                                                                                                                   | (Applications: Prescribed Forms and                                                | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                                                    | Owners or Reputed Owners                                                           | Lessees or Tenants                                                                                                | Occupiers                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                        |
| 414 cont'd              |                                                          |                                                                                                                                                                        |                                                                                    |                                                                                                                   |                                                                                    | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to<br>maintain drainage and right of<br>support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service<br>media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage) |
| 415                     | Acquisition of new rights and imposition of restrictions | 11946 square metres of<br>agricultural land and<br>hedgerows (Faenol-<br>Bropor, south of North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | NONE                                                                                                              | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                                                                               |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                                                                          |                                                                          |                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                                                                                                                          | Category 1                                                               |                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation                                                                                                      | n 7(1)(a) of the Infrastructure Plannin;<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                   | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                                                                 | Lessees or Tenants                                                       | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 415 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>Unknown<br>(in respect of mines and minerals) |                                                                          | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain)<br>HSBC Bank PLC<br>8 Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to<br>maintain drainage and right of<br>support) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                                                                                                   | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                             | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                                                                                                  |                                                                                                                                                                               | Category 1                                                                                                        |                                                                                                                                                                                                                       | Catagany 2                                                                                                                                                                                                                                                                                                                              |
|                         |                              |                                                                                                                                                                                                                                  | Qualifying persons under Regulatio                                                                                                                                            | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                                                                                                   | Category 2                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                                                                                              | Owners or Reputed Owners                                                                                                                                                      | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                   |
| 415 cont'd              |                              |                                                                                                                                                                                                                                  |                                                                                                                                                                               |                                                                                                                   |                                                                                                                                                                                                                       | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service<br>media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage) |
| 416                     | Acquisition of Freehold      | 10443 square metres of<br>woodland, agricultural<br>land and hedgerow<br>(south of North Wales<br>Expressway, A55 and<br>west of William Morgan<br>Road) and footpath<br>(DE 201 9) (excluding<br>all interests of the<br>Crown) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 201 9)<br>Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of<br>drainage and right of support)                  |

|                         | Land which is p              | proposed to be subject to: (i | BOOK OF REF<br>) powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire                | to carry out protective works (Regulatio                                                | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|------------------------------|-------------------------------|-----------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                              |                               | Qualifying persons upder Pegulatio                        | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and |                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
|                         |                              |                               |                                                           | Procedures) Regulations 2009                                                                                                 |                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                  | Lessees or Tenants                                                                                                           | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| 416 cont'd              |                              |                               | Unknown<br>(in respect of mines and minerals)             |                                                                                                                              | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for<br>maintenance of fence and drain)<br>HSBC Bank PLC<br>8 Canada Square<br>LONDON<br>E14 5HQ<br>(as mortgagee for John Berwyn Evans<br>and Eirian Evans stated in Charge<br>dated 28th January 1998)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to<br>maintain drainage and right of<br>support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and<br>right to free passage of service<br>media) |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                       | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                  | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire                                             | o carry out protective works (Regulatio                                                 | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                      | Qualifying persons under Regulatio                                                      | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                         |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                                  | Owners or Reputed Owners                                                                | Lessees or Tenants                                                                                                                                           | Occupiers                                                                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 416 cont'd              |                              |                                                                                                                                                      |                                                                                         |                                                                                                                                                              |                                                                                         | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage)                      |
| 417                     | Acquisition of Freehold      | 325521 square metres<br>of agricultural land and<br>private roads (north of<br>Glascoed Road, B5381,<br>and south of North<br>Wales Expressway, A55) | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY      | NONE                                                                                                                                                         | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY      | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)        |
|                         |                              |                                                                                                                                                      | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY |                                                                                                                                                              | John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)       |
|                         |                              |                                                                                                                                                      | Unknown<br>(in respect of mines and minerals)                                           |                                                                                                                                                              |                                                                                         | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)         |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF RE<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|------------------------------|--------------------------------|--------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                | Qualifying persons under Regulatio                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                               | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 417 cont'd              |                              |                                |                                                        |                                                                                                                  |                                         | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to<br>maintain drainage and right to<br>support as contained in a Conveyance<br>dated 11th March 1991)<br>Unknown<br>(in respect of right of drainage as<br>contained in a Conveyance dated<br>20th March 1919)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to<br>maintain drainage as contained in a<br>Conveyance dated 11th March 1991)<br>Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunication<br>apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                         | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                             |                                                                        | Qualifying persons under Regulatio                                                                                                                                                                                                                                                  | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                      |                                                                                                                                                                                                                                                                                                   |  |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                    | Owners or Reputed Owners                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                             |  |
| 418                     | Acquisition of new rights and<br>imposition of restrictions | 50 square metres of<br>verge of public road<br>(Glascoed Road, B5381)  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Lowri Owen Jones<br>Dwyfor<br>Ruthin Road<br>DINBYCH<br>LL16 3EU<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown              | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Unknown               |  |
| 419                     | Acquisition of new rights and<br>imposition of restrictions | 201 square metres of<br>verge of public road<br>(Glascoed Road, B5381) | Charlotte Anne Southern<br>Bryn Celyn Lodge<br>Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DF<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                           |                                                                                                                                                                                                                                                                                                                                |                                                                          |                                                                                            |                                                                                                                                                                                                                |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                           |                                                                                                                                                                                                                                                                                                                                | Category 1                                                               |                                                                                            |                                                                                                                                                                                                                |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                           | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                             | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                     |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                       | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                       | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                          |  |  |
| 419 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                           | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>(in respect of subsoil beneath half<br>width of public highway)<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                          |                                                                                            | Unknown<br>Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunication<br>apparatus)                                                                              |  |  |
| 420                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 74 square metres of<br>pavement (Glascoed<br>Road, B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                                                                                     | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Alun Lloyd Davies<br>6 Ffordd Glascoed<br>ABERGELE<br>LL22 9DW<br>(in respect of right of access)<br>Amy Louise Evans<br>5 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access) |  |  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REFE<br>powers of compulsory acquisition, (ii) rig | DEVELOPMENT CONSENT ORDER<br>RENCE - PART 1<br>sht to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulation | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|-------------------------------|------------------------------------------------------------|------------------------------------------------------------------------------------------------------------|------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                            | Category 1                                                                                                 |                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                              |                               | Qualifying persons under Regulation                        | 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                     | (Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                   | Lessees or Tenants                                                                                         | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 420 cont'd              |                              |                               |                                                            |                                                                                                            |                                          | Martin Griffiths<br>5 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)<br>Carrie Hargraves<br>8 Ffordd Glascoed<br>ABERGELE<br>LL22 9DW<br>(in respect of right of access)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Ethan James Homer<br>4 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)<br>Hayley Elizabeth Doroshenko-Nuttall<br>4 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF RE<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|-------------------------------|--------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulatio                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                               | Lessees or Tenants                                                                                               | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 420 cont'd              |                              |                               |                                                        |                                                                                                                  |                                         | Lynn Alison Hughes<br>3 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right to use common<br>sewers and right to enter to maintain<br>sewers and drains)<br>Meirick Lloyd Davies<br>1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)<br>Nesta Davies<br>1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)<br>Nesta Davies<br>1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                     | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                    | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|----------------------------------------------------------|--------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                    | Qualifying persons under Regulatio                                                         | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                | Owners or Reputed Owners                                                                   | Lessees or Tenants                                                                                                | Occupiers                                                                                  | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                    |
| 420 cont'd              |                                                          |                                                                    |                                                                                            |                                                                                                                   |                                                                                            | The Executor of the Estate of the Late<br>Ian Charles Hughes<br>3 Ffordd Glascoed<br>CONWY<br>Conwy<br>LL22 9DW<br>(in respect of right to use common<br>sewers and right to enter to maintain<br>sewers and drains)<br>William Owen Morris<br>7 Ffordd Glascoed<br>ABERGELE<br>LL22 9DW<br>(in respect of right of access)                                              |
| 421                     | Acquisition of new rights and imposition of restrictions | 48 square metres of<br>pavement (south of<br>Glascoed Road, B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Meirick Lloyd Davies<br>1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)<br>Nesta Davies<br>1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                     | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                                             | n 7(1)(a))                                                                                                                                            |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                                                                | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                          | g (Applications: Prescribed Forms and                                                                                | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                | Owners or Reputed Owners                                                                                                                                                                                                                                                          | Lessees or Tenants                                                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 421 cont'd              |                                                          |                                                                                    |                                                                                                                                                                                                                                                                                   |                                                                                                                  |                                                                                                                      | Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications<br>apparatus)                         |
| 422                     | Acquisition of new rights and imposition of restrictions | 1 square metre of verge<br>of public road (south of<br>Glascoed Road, B5381)       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                                                                                                                                                                                        | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN                           | Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                 |
| 423                     | Acquisition of new rights and imposition of restrictions | 101 square metres of<br>verge of public road<br>(south of Glascoed Road,<br>B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                      |

|                         | Land which is j              | proposed to be subject to: (i)                                        | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                 |
|-------------------------|------------------------------|-----------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                | Category 1                                                                                                        |                                                                                                                      |                                                                                                                                                            |
|                         |                              |                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                             | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                   | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                       | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009      |
| 423 cont'd              |                              |                                                                       | Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                                                                                                                                                                                                                                                                           |                                                                                                                   |                                                                                                                      |                                                                                                                                                            |
| 424                     | Temporary Possession         | 33 square metres of<br>verge of public road<br>(Glascoed Road, B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown |

|                         | Land which is p                                             | proposed to be subject to: (i)                                           | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                                             | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 1                                                                                                      |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                                                             |                                                                          | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                                         | g (Applications: Prescribed Forms and                                                                                | Category 2                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                      | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Lessees or Tenants                                                                                              | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                         |
| 424 cont'd              |                                                             |                                                                          | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                 |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                               |
| 425                     | Acquisition of new rights and<br>imposition of restrictions | 509 square metres of<br>public road and verges<br>(Glascoed Road, B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>(in respect of subsoil beneath half<br>width of public highway)<br>Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown<br>Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunication<br>apparatus)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                   | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                  |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 1                                                                                                        |                                                                                                                      |                                                                                                                                                                                                                                                                                                             |
|                         |                                                             |                                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                       | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                  |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                 | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                       |
| 425 cont'd              |                                                             |                                                                                    | Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                                                                                                                                                                                                                                                                                     |                                                                                                                   |                                                                                                                      |                                                                                                                                                                                                                                                                                                             |
| 426                     | Acquisition of new rights and<br>imposition of restrictions | 278 square metres of<br>verge of public road<br>(south of Glascoed Road,<br>B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Lois Oldfield Williams<br>Groesffordd Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                  | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                          | n 7(1)(a))                                                                                                                                            |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                 | Qualifying persons under Regulatio                                                                                                                                               | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                              | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                             | Owners or Reputed Owners                                                                                                                                                         | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 426 cont'd              |                                                             |                                                                                 | Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown             |                                                                                                                  |                                                                                                                                                                                  | Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                 |
| 427                     | Acquisition of new rights and<br>imposition of restrictions | 490 square metres of<br>agricultural land (south<br>of Glascoed Road,<br>B5381) | Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                             | Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)       |

|                         | Land which is p                                             | proposed to be subject to: (i                                                   | ,<br>BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                         | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) righ<br>of Denbighshire | ts to carry out protective works (Regulatio                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                 | Qualifying persons under Regulatic                                                                                                                | Category 1<br>on 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                            | ng (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                             | Owners or Reputed Owners                                                                                                                          | Lessees or Tenants                                                                                              | Occupiers                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 428                     | Acquisition of new rights and<br>imposition of restrictions | 258 square metres of<br>agricultural land (south<br>of Glascoed Road,<br>B5381) | Lois Oldfield Williams<br>Groesffordd Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                            | Lois Oldfield Williams<br>Groesffordd Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to inspect,<br>maintain, protect, use, replace,<br>remove or render unusable a gas<br>pipeline and right to pass)<br>Duncan Robertson<br>Llantysilio Hall<br>Llantysilio<br>LLANGOLLEN<br>LL20 8BT<br>(in respect of right to enter and right<br>to all game)<br>David Watkin Williams-Wynn Bt<br>3 Upper Brook Street<br>OSWESTRY<br>SY11 2TB<br>(In respect of right to enter and right<br>to all game)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                             |                                                                                                                                                                                                                                                                                                        |                                                                          |                                                                                                                      |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                             |                                                                                                                                                                                                                                                                                                        | Category 1                                                               |                                                                                                                      |                                                                                                                                                       |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                             | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                         | Owners or Reputed Owners                                                                                                                                                                                                                                                                               | Lessees or Tenants                                                       | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 428 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                             |                                                                                                                                                                                                                                                                                                        |                                                                          |                                                                                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)         |  |  |  |
| 429                     | Acquisition of new rights                                                                                                                                                                                                                                                                            | 3 square metres of<br>public road (Glascoed<br>Road, B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Lois Oldfield Williams<br>Groesffordd Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |  |  |  |
| 430                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 7 square metres of<br>public road (Glascoed<br>Road, B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                                                                                                   | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                         |                                                                                                                      |                                                                                                                                                          |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                         | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                          | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                               |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                     | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                    | Lessees or Tenants                                                                      | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009    |  |  |  |
| 430 conť d              |                                                                                                                                                                                                                                                                                                      |                                                                         | Lois Oldfield Williams<br>Groesffordd Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                                                                                                                                                                                                                                                              |                                                                                         |                                                                                                                      | Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                    |  |  |  |
| 431                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 104 square metres of<br>public road and verge<br>(Glascoed Road, B5381) | Brenda Davies<br>21 Clwyd Avenue<br>PRESTATYN<br>Denbighshire<br>LL19 9NG<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Lois Oldfield Williams<br>Groesffordd Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                    | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown |  |  |  |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | Category 1                                                                                                       |                                                                                                                      |                                                                                                                                                       |
|                         |                                                          |                                                                                   | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                               | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 431 cont'd              |                                                          |                                                                                   | Memoria - Denbighshire Memorial<br>Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of subsoil beneath half<br>width of public highway)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                  |                                                                                                                      |                                                                                                                                                       |
| 432                     | Acquisition of new rights and imposition of restrictions | 71 square metres of<br>verge of public road<br>(north of Glascoed Road,<br>B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                     |                                                                                        |                                     |                                                                                                                                                                                                                                                 |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation                                                                                                                                                                                                                                                                                 | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                      |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                           |  |  |  |
| 432 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Eirian Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>(in respect of subsoil beneath half<br>width of public highway)<br>John Berwyn Evans<br>Faenol Bropor<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>LL18 5UY<br>(in respect of subsoil beneath half<br>width of public highway) |                                                                                        |                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                                                                        |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Unknown                                                                                                                                                                                                                                                                                                             |                                                                                        |                                     | Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunication<br>apparatus)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |  |  |  |

|                         | Land which is p              | proposed to be subject to: (i)                                                                | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights 1<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                          | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                               | Qualifying persons under Regulatio                                                                                                                                               | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                              | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                           | Owners or Reputed Owners                                                                                                                                                         | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 433                     | Temporary Possession         | 3604 square metres of<br>agricultural land (south<br>of Glascoed Road,<br>B5381)              | Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                             | Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                                                                  |
| 434                     | Temporary Possession         | 1423 square metres of<br>agricultural land and<br>hedgerow (south of<br>Glascoed Road, B5381) | Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                             | Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                                                                  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                              | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                          | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                             | Qualifying persons under Regulatio                                                                                                                                               | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                              | Category 2                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                         | Owners or Reputed Owners                                                                                                                                                         | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                  |
| 435                     | Acquisition of new rights and<br>imposition of restrictions | 7765 square metres of<br>agricultural land and<br>hedgerow (south of<br>Glascoed Road, B5381)                                               | Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                             | Howard Glyn Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Mair Edmund Hughes<br>Marli Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                       |
| 436                     | Acquisition of new rights                                   | 1226 square metres of<br>agricultural land and<br>hedgerows (west of<br>Memorail Park and<br>Crematorium, south of<br>Glascoed Road, B5381) | Lois Oldfield Williams<br>Groesffordd Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Unknown<br>(in respect of mines and minerals)                                | NONE                                                                                                             | Lois Oldfield Williams<br>Groesffordd Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR                                                                                 | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to inspect,<br>maintain, protect, use, replace,<br>remove or render unusable a gas<br>pipeline and right to pass)<br>Duncan Robertson<br>Llantysilio Hall<br>Llantysilio<br>LLANGOLLEN<br>LL20 8BT<br>(in respect of right to enter and right<br>to all game) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights 1<br>Denbighshire | to carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|-------------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulation                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                        | (Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                            | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 436 cont'd              |                              |                               |                                                          |                                                                                                               |                                          | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay a water<br>main and right to enter as contained<br>in a Deed dated 12th January 1984)<br>David Watkin Williams-Wynn Bt<br>3 Upper Brook Street<br>OSWESTRY<br>SY11 2TB<br>(In respect of right to enter and right<br>to all game)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                               | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                          | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                              |                                                                                                                                                   |                                                                                                                   |                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|                         |                                                             |                                                                                                                              | Qualifying persons under Regulation                                                                                                               | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                              | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                          | Owners or Reputed Owners                                                                                                                          | Lessees or Tenants                                                                                                | Occupiers                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                          |
| 436 cont'd              |                                                             |                                                                                                                              |                                                                                                                                                   |                                                                                                                   |                                                                                                  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                               |
| 437                     | Acquisition of new rights and<br>imposition of restrictions | 9776 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre-Mawr and south<br>of Glascoed Road,<br>B5381) | Lois Oldfield Williams<br>Groesffordd Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                              | Lois Oldfield Williams<br>Groesffordd Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to inspect,<br>maintain, protect, use, replace,<br>remove or render unusable a gas<br>pipeline and right to pass)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay a water<br>main and right to enter as contained<br>in a Deed dated 12th January 1984) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
| 437 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                               | BOOK OF R<br>powers of compulsory acquisition, (ii)                                                                                                                                             | ITM DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) righ<br>of Denbighshire | ts to carry out protective works (Regulatio                                                                                                                                                     | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                              | Qualifying persons under Regulati                                                                                                                                                               | Category 1<br>on 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                              | ng (Applications: Prescribed Forms and                                                                                                                                                          | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                          | Owners or Reputed Owners                                                                                                                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 438                     | Acquisition of new rights and<br>imposition of restrictions | 5595 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre-Mawr and south<br>of Glascoed Road,<br>B5381) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                              | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>Memoria - Denbighshire Memorial<br>Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support,<br>and maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights granted by deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966) |

|                         | Land which is j              | proposed to be subject to: (i)                                                                                              | BOOK OF RI<br>powers of compulsory acquisition, (ii)                                                                                                                                            | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) right<br>of Denbighshire             | s to carry out protective works (Regulatio                                                                                                                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                              |                                                                                                                             | Qualifying persons under Regulati                                                                                                                                                               | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and |                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                         | Owners or Reputed Owners                                                                                                                                                                        | Procedures) Regulations 2009<br>Lessees or Tenants                                                                           | Occupiers                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 439                     | Acquisition of new rights    | 156 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre-Mawr and south<br>of Glascoed Road,<br>B5381) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                                         | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>Memoria - Denbighshire Memorial<br>Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support,<br>and maintenance)<br>Unknown<br>(in respect of rights granted by deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966) |  |
| 440                     | Acquisition of new rights    | 8 square metres of verge<br>of public road (north of<br>Pentre-Mawr and south<br>of Glascoed Road,<br>B5381)                | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                            | NONE                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                            | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Land which is                                                               | proposed to be subject to: (i)                                                                               | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|-----------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastruc |                                                                                                              |                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                  | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use                                                | Description of land                                                                                          | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                         | Procedures) Regulations 2009<br>Lessees or Tenants                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 440 cont'd              |                                                                             |                                                                                                              | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                  |                                                                                                                      |                                                                                                                                                       |
| 441                     | Temporary Possession                                                        | 3 square metres of verge<br>of public road (north of<br>Pentre-Mawr and south<br>of Glascoed Road,<br>B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)                                                       | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                             | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                            |                                                                                                                                                                                                 | Category 1                                                                                                        |                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|                         |                              |                                                                                                                            | Qualifying persons under Regulatio                                                                                                                                                              | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                        | Owners or Reputed Owners                                                                                                                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                            |
| 441 cont'd              |                              |                                                                                                                            | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                      |                                                                                                                   |                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 442                     | Temporary Possession         | 63 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre-Mawr and south<br>of Glascoed Road,<br>B5381) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                              | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>Memoria - Denbighshire Memorial<br>Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support,<br>and maintenance)<br>Unknown<br>(in respect of rights granted by deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                          |                                                                                                                      |                                                                                                                                                       |  |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 1                                                               |                                                                                                                      |                                                                                                                                                       |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                               | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | ; (Applications: Prescribed Forms and                                                                                | Category 2                                                                                                                                            |  |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Lessees or Tenants                                                       | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |  |
| 443                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 40 square metres of<br>verge of public road<br>(north of Pentre-Mawr<br>and south of Glascoed<br>Road, B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>UL22 9DR | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |  |  |  |  |
| 444                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 258 square metres of<br>public road (road from<br>Glascoed Road past<br>Waen Meredydd)                        | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of subsoil beneath half<br>width of public highway)                                                                                                                                                                                                                                                                                                                                                                                        | NONE                                                                     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |  |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                        |                                                                                                 |                                                                                                                                                       |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                             | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and                                                             | Category 2                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                         | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Lessees or Tenants                                                                     | Occupiers                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |  |  |  |
| 444 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Ul22 9DR |                                                                                        |                                                                                                 |                                                                                                                                                       |  |  |  |
| 445                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                             | 635 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre-Mawr and south<br>of Glascoed Road,<br>B5381) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR                                                                                                                                                                                                                                                                                                                                                                                                                                                          | NONE                                                                                   | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)                  |  |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                 | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                | Qualifying persons under Regulatio                                                                                                                                                                                                                                                         | Category 1<br>n 7(1)(a) of the Infrastructure Planning                                                           | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                            | Owners or Reputed Owners                                                                                                                                                                                                                                                                   | Procedures) Regulations 2009<br>Lessees or Tenants                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                    |
| 445 cont'd              |                                                             |                                                                                                                | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR                                                                                                                                                                                               |                                                                                                                  | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR                         | Memoria - Denbighshire Memorial<br>Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support,<br>and maintenance)<br>Unknown<br>(in respect of rights granted by deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966) |
| 446                     | Acquisition of new rights and<br>imposition of restrictions | 103 square metres of<br>verge of public road<br>(north of Pentre-Mawr<br>and south of Glascoed<br>Road, B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                                                                                                                                                                                                                                                                  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                             | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                            |                                                                                                                                                                                                 | Category 1                                                                                                        |                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|                         |                                                             |                                                                                                                            | Qualifying persons under Regulatio                                                                                                                                                              | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                        | Owners or Reputed Owners                                                                                                                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                            |
| 446 cont'd              |                                                             |                                                                                                                            | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                      |                                                                                                                   |                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 447                     | Acquisition of new rights and<br>imposition of restrictions | 99 square metres of<br>agricultural land and<br>hedgerow (north of<br>Pentre-Mawr and south<br>of Glascoed Road,<br>B5381) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                              | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>Memoria - Denbighshire Memorial<br>Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support,<br>and maintenance)<br>Unknown<br>(in respect of rights granted by deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|-------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                               | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Category 2                                                                                                       |                                                                                                                      |                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Lessees or Tenants                                                                                               | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 448                     | Acquisition of new rights and<br>imposition of restrictions | 2 square metres of<br>agricultural land (north<br>of Pentre-Mawr and<br>south of Glascoed Road,<br>B5381)     | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |
| 449                     | Temporary Possession                                        | 16 square metres of<br>verge of public road<br>(north of Pentre-Mawr<br>and south of Glascoed<br>Road, B5381) | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | NONE                                                                                                             | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |

|                         | Land which is j              | proposed to be subject to: (i)                                                                                 | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                                                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                            |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                |                                                                                                                                                                                                                                                                                                                                                  | Category 1                                                                                                        |                                                                                                                      |                                                                                                                                                       |
|                         |                              |                                                                                                                | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                               | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                            | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                         | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |
| 449 cont'd              |                              |                                                                                                                | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown |                                                                                                                   |                                                                                                                      |                                                                                                                                                       |
| 450                     | Temporary Possession         | 143 square metres of<br>verge of public road<br>(north of Pentre-Mawr<br>and south of Glascoed<br>Road, B5381) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)                                                        | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                               |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                 | BOOK OF RE powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                |                                                                                                                                                                                                                                                                                                      |                                                                                                                   |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                            |
|                         |                              | Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                                                                                                                                                                                                      |                                                                                                                   |                                                                                                                      | Category 2                                                                                                                                                                                                                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                            | Owners or Reputed Owners                                                                                                                                                                                                                                                                             | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                      |
| 450 cont'd              |                              |                                                                                                                                                | Unknown                                                                                                                                                                                                                                                                                              |                                                                                                                   |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                            |
| 451                     | Temporary Possession         | 188 square metres of<br>verge of public road<br>(north of Pentre-Mawr<br>and south of Glascoed<br>Road, B5381)                                 | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                                                                                                                                                                                                                    |
| 452                     | Temporary Possession         | 3016 square metres of<br>agricultural land (north<br>of Pentre-Mawr and<br>south of Glascoed Road,<br>B5381)                                   | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY                                                                                                                                                                                                               | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                     | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                        | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                               | BOOK OF RE powers of compulsory acquisition, (ii) r                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                              |                                                                                                | Category 1                                                                                                       |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|                         |                                                             |                                                                                                              | Qualifying persons under Regulatio                                                             | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                          | Owners or Reputed Owners                                                                       | Lessees or Tenants                                                                                               | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 452 cont'd              |                                                             |                                                                                                              |                                                                                                |                                                                                                                  |                                                                                               | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 453                     | Acquisition of new rights and<br>imposition of restrictions | 1990 square metres of<br>agricultural land (north<br>of Pentre-Mawr and<br>south of Glascoed Road,<br>B5381) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                    | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed to fright of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                |                                                                                                                                                           |                                                                                               |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                | Qualifying persons under Regulation                                                                                                                       | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009        | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                            | Owners or Reputed Owners                                                                                                                                  | Lessees or Tenants                                                                            | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
| 454                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 2903 square metres of<br>agricultural land (north<br>of Pentre-Mawr and<br>south of Glascoed Road,<br>B5381)   | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY                                                            | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                        | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed to fright of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |  |  |  |
| 455                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 245 square metres of<br>verge of public road<br>(north of Pentre-Mawr<br>and south of Glascoed<br>Road, B5381) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                          | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                     | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                    |                                                                                                                                 | Category 1                                                                                                        |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                         |                                                             |                                                                                    | Qualifying persons under Regulatio                                                                                              | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                | Owners or Reputed Owners                                                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                     |
| 455 cont'd              |                                                             |                                                                                    | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown |                                                                                                                   |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 456                     | Acquisition of new rights and<br>imposition of restrictions | 514 square metres of<br>agricultural land and<br>hedge (south of Waen<br>Meredydd) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY                                  | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                     | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                         | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                        |                                                                                        | Category 1                                                                                                       |                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|                         |                                                             |                                                                                                                                        | Qualifying persons under Regulatio                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                    | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 456 cont'd              |                                                             |                                                                                                                                        |                                                                                        |                                                                                                                  |                                                                                                                                                                                                                                      | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 457                     | Acquisition of new rights and<br>imposition of restrictions | 290 square metres of<br>agricultural land (north<br>of Pentre-Mawr and<br>south of Glascoed Road,<br>B5381) and footpath<br>(DE 105 7) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                    | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE   105   7) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed to fright of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire                                           | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                             |                                                                       | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                   | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                      |                                                                                                                                                                                                                                                                                                                              |  |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                   | Owners or Reputed Owners                                                                                                                                                                                                                                                                             | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                        |  |
| 458                     | Acquisition of new rights and<br>imposition of restrictions | 137 square metres of<br>road from Glascoed road<br>past Waen Meredydd | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Unknown                                                                                                                                                                                                                                                                                                                      |  |
| 459                     | Acquisition of new rights and imposition of restrictions    | 158 square metres of<br>agricultural land (south<br>of Waen Meredydd) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY                                                                                                                                                                                                               | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                                                                | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                        | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                        | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                   | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights 1<br>of Denbighshire | to carry out protective works (Regulatio                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|-------------------------------------------------------------|-----------------------------------------------------------------------|----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                       |                                                                                        | Category 1                                                                                                          |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                         |                                                             |                                                                       | Qualifying persons under Regulation                                                    | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                           | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                   | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                                                  | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                     |
| 459 cont'd              |                                                             |                                                                       |                                                                                        |                                                                                                                     |                                                                                               | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |
| 460                     | Acquisition of new rights and<br>imposition of restrictions | 239 square metres of<br>agricultural land (south<br>of Waen Meredydd) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                       | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)                                                                                              |

|                         | Land which is p              | proposed to be subject to: (i)                                        | BOOK OF RE powers of compulsory acquisition, (ii)                                      | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|------------------------------|-----------------------------------------------------------------------|----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                       |                                                                                        |                                                                                                                     |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                         |                              |                                                                       | Qualifying persons under Regulatic                                                     | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                           | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                   | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                                                  | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                     |
| 460 cont'd              |                              |                                                                       |                                                                                        |                                                                                                                     |                                                                                               | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |
| 461                     | Acquisition of new rights    | 524 square metres of<br>agricultural land (south<br>of Waen Meredydd) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                       | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)                                                                                             |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|--------------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                          | Category 1                                                                                                    |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                              |                                | Qualifying persons under Regulatior                      | n 7(1)(a) of the Infrastructure Planning (<br>Procedures) Regulations 2009                                    | Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 461 cont'd              |                              |                                |                                                          |                                                                                                               |                                         | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                                                   | BOOK OF R<br>powers of compulsory acquisition, (ii)                                            | rm DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) right<br>of Denbighshire | s to carry out protective works (Regulatio                                                                                                                                                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                                                  |                                                                                                | Category 1                                                                                                        |                                                                                                                                                                                                                                  | Cottogory 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                         |                                                             |                                                                                                                                                                  | Qualifying persons under Regulati                                                              | on 7(1)(a) of the Infrastructure Planni<br>Procedures) Regulations 2009                                           | ng (Applications: Prescribed Forms and                                                                                                                                                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                                              | Owners or Reputed Owners                                                                       | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 462                     | Acquisition of new rights and<br>imposition of restrictions | 24500 square metres of<br>agricultural land and<br>private road (north of<br>Pentre-Mawr and south<br>of Glascoed Road,<br>B5381) and footpath<br>(DE   105   7) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                     | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 105 7) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables) |

|                         | Land which is p              | proposed to be subject to: (i)                                                       | BOOK OF RE powers of compulsory acquisition, (ii) r                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                      |                                                                                                | Category 1                                                                                                       |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                         |                              |                                                                                      | Qualifying persons under Regulatio                                                             | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                  | Owners or Reputed Owners                                                                       | Lessees or Tenants                                                                                               | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 462 cont'd              |                              |                                                                                      |                                                                                                |                                                                                                                  |                                                                                               | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed to dated 26th January 1996) |
| 463                     | Temporary Possession         | 13597 square metres of<br>agricultural land and<br>hedge (south of Waen<br>Meredydd) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                    | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                   |

|                         | Land which is p              | proposed to be subject to: (i)                                                       | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                                                                             | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                      | Qualifying persons under Regulatio                                                                                                                                                              | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                                                                                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                  | Owners or Reputed Owners                                                                                                                                                                        | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                     |
| 463 cont'd              |                              |                                                                                      |                                                                                                                                                                                                 |                                                                                                                  |                                                                                                                                                                                                 | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |
| 464                     | Temporary Possession         | 17129 square metres of<br>agricultural land and<br>hedge (south of Waen<br>Meredydd) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                             | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed<br>dated 6th November 2009, and of<br>restrictive covenants)                                                                                           |

|                         | Land which is p                                             | proposed to be subject to: (i)                                          | BOOK OF RE powers of compulsory acquisition, (ii)                                              | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                      | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                         | Qualifying persons under Regulatio                                                             | Category 1<br>on 7(1)(a) of the Infrastructure Planning                                                          | g (Applications: Prescribed Forms and                                                         | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                     | Owners or Reputed Owners                                                                       | Procedures) Regulations 2009<br>Lessees or Tenants                                                               | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                         |
| 464 cont'd              |                                                             |                                                                         |                                                                                                |                                                                                                                  |                                                                                               | Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966)                                                                                                                                                                                                                                                                                                   |
| 465                     | Acquisition of new rights and<br>imposition of restrictions | 11550 square metres of<br>agricultural land (south<br>of Waen Meredydd) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                    | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                               | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                                                                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                      |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                              |                                                                                                                                                                                                 | Category 1                                                                                                       |                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                 |
|                         |                                                             |                                                                                              | Qualifying persons under Regulatio                                                                                                                                                              | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                                                                             | Category 2                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                          | Owners or Reputed Owners                                                                                                                                                                        | Lessees or Tenants                                                                                               | Occupiers                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                           |
| 465 cont'd              |                                                             |                                                                                              |                                                                                                                                                                                                 |                                                                                                                  |                                                                                                                                                                                                 | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)                                                                                                                    |
| 466                     | Acquisition of new rights and<br>imposition of restrictions | 10905 square metres of<br>agricultural land and<br>hedge (south of Ffordd<br>Richard Davies) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                             | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed<br>dated 6th November 2009, and of<br>restrictive covenants) |
|                         |                                                             |                                                                                              |                                                                                                                                                                                                 |                                                                                                                  |                                                                                                                                                                                                 | Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966)                                                                                                                     |

|                         | Land which is p              | proposed to be subject to: (i)                                                 | BOOK OF RE powers of compulsory acquisition, (ii)                                      | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|-------------------------|------------------------------|--------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                | Qualifying persons under Regulation                                                    | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                             | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                            | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                                                  | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                          |
| 466 cont'd              |                              |                                                                                |                                                                                        |                                                                                                                     |                                                                                               | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                                                                                                                                |
| 467                     | Acquisition of new rights    | 40 square metres of<br>agricultural land and<br>hedge (south Waen<br>Meredydd) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                       | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                                            | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                                                                        | Category 1<br>Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                | Owners or Reputed Owners                                                                                                                                                                                                                                                                  | Lessees or Tenants                                                                                                                                           | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                     |
| 467 cont'd              |                              |                                                                                                                                    |                                                                                                                                                                                                                                                                                           |                                                                                                                                                              |                                                                                                                      | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |
| 468                     | Acquisition of new rights    | 48 square metres of<br>public road (road from<br>Glascoed Road past<br>Waen Meredydd)<br>(excluding all interests of<br>the Crown) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | NONE                                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                          |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                 | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                                                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                               | n 7(1)(a))                                                                                                                                                                                                                                                                                                     |
|-------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                |                                                                                                                                                                                                                                                   | Category 1                                                                                                        |                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                |
|                         |                              |                                                                                                                | Qualifying persons under Regulatio                                                                                                                                                                                                                | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                                                                                   | Category 2                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                            | Owners or Reputed Owners                                                                                                                                                                                                                          | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                             | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                          |
| 468 cont'd              |                              |                                                                                                                | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown                                                                            |                                                                                                                   |                                                                                                                                                                                                       | Unknown                                                                                                                                                                                                                                                                                                        |
| 469                     | Acquisition of new rights    | 304 square metres of<br>track (south of Ffordd<br>Richard Davies)<br>(excluding all interests of<br>the Crown) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY<br>(as reputed owner)<br>The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as reputed owner)<br>Unknown | NONE                                                                                                              | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>Unknown | National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of easements granted by<br>Deed of grant dated 15 July 2013)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |
| 470                     | Acquisition of new rights    | 46 square metres of<br>agricultural land and<br>track (south of Ffordd<br>Richard Davies)                      | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR                                                                                                                                                   | NONE                                                                                                              | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR                                                                                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                 |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                              | BOOK OF RE powers of compulsory acquisition, (ii)                                               | rm DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights<br>of Denbighshire | to carry out protective works (Regulatio                                                        | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                             | Qualifying persons under Regulatic                                                              | Category 1<br>on 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009                            | g (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                         | Owners or Reputed Owners                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| 470 cont'd              |                                                          |                                                                                             | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR    |                                                                                                                   | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR    | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed<br>dated 6th November 2009, and of<br>restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Ceed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966) |
| 471                     | Acquisition of new rights and imposition of restrictions | 1394 square metres of<br>agricultural land and<br>hedge (south of Ffordd<br>Richard Davies) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                              | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                    | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                   | Qualifying persons under Regulatio                                                              | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                               | Owners or Reputed Owners                                                                        | Lessees or Tenants                                                                                               | Occupiers                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| 471 cont'd              |                                                          |                                                                                   | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR    |                                                                                                                  | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR    | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed<br>dated 6th November 2009, and of<br>restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966) |
| 472                     | Acquisition of new rights and imposition of restrictions | 3007 square metres of<br>agricultural land (south<br>of Ffordd Richard<br>Davies) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                             | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |

|                         | Land which is p                                             | roposed to be subject to: (i)                                                                                                                | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights<br>f Denbighshire | to carry out protective works (Regulatio                                                        | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                              |                                                                                                 | Category 1                                                                                                     |                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|                         |                                                             |                                                                                                                                              | Qualifying persons under Regulatio                                                              | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                          | Owners or Reputed Owners                                                                        | Lessees or Tenants                                                                                             | Occupiers                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| 472 cont'd              |                                                             |                                                                                                                                              | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR    |                                                                                                                | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR    | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed<br>dated 6th November 2009, and of<br>restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Leed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966) |
| 473                     | Acquisition of new rights and<br>imposition of restrictions | 10383 square metres of<br>agricultural land and<br>hedgerow (west of<br>Electricity Distribution<br>Site, south of Ffordd<br>Richard Davies) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                           | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                              |                                                                                       |                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|----------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio                                                           | Category 1<br>n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                     | Lessees or Tenants                                                                    | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |
| 473 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR |                                                                                       | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed<br>dated 6th November 2009, and of<br>restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |  |  |

|                         | Land which is j              | proposed to be subject to: (i                                                                                                     | BOOK OF R<br>) powers of compulsory acquisition, (ii)                                                                                                                                           | ITM DEVELOPMENT CONSENT ORDER<br>EFERENCE - PART 1<br>right to use the land, and/or (iii) righ<br>of Denbighshire | ts to carry out protective works (Regulatio                                                                                                                                                     | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                   |                                                                                                                                                                                                 | Category 1                                                                                                        |                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|                         |                              |                                                                                                                                   | Qualifying persons under Regulati                                                                                                                                                               | on 7(1)(a) of the Infrastructure Plann<br>Procedures) Regulations 2009                                            | ing (Applications: Prescribed Forms and                                                                                                                                                         | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                               | Owners or Reputed Owners                                                                                                                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 474                     | Temporary Possession         | 2441 square metres of<br>agricultural land and<br>hedgerow (west of<br>Electricity Distribution<br>Site, north of Hendy<br>Gorse) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                              | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed<br>dated 6th November 2009, and of<br>restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966)<br>Vodafone Limited<br>Vodafone Limited<br>Vodafone Limited<br>RG14 2FN<br>(in respect of telecommunication<br>apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                                 | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                                | Qualifying persons under Regulatio                                                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                            | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                        |
| 474 cont'd              |                              |                                                                                                                                                |                                                                                        |                                                                                                                  |                                                                                               | National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                   |
| 475                     | Temporary Possession         | 15344 square metres of<br>grassland area and<br>shrubbery and plyon<br>(south of Bodelwyddan<br>Substation, south of<br>Ffordd Richard Davies) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                    | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain water pipeline)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF RE<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | to carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|--------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                | Qualifying persons under Regulatio                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                               | Lessees or Tenants                                                                                               | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 475 cont'd              |                              |                                |                                                        |                                                                                                                  |                                          | Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication<br>apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                      | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|---------------------------------------------------------------------|----------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                     |                                                                                        | Category 1                                                                                                       |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                         |                              |                                                                     | Qualifying persons under Regulatio                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                 | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 475 cont'd              |                              |                                                                     |                                                                                        |                                                                                                                  |                                                                                               | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and<br>maintain gas pipelines)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed to January 1996) |
| 476                     | Temporary Possession         | 204 square metres of<br>agricultural land (north<br>of Hendy Gorse) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                    | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain water pipeline)                                                                                                                                                                                                                                                                                                                                                                                                                            |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF RE<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | to carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|--------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                | Qualifying persons under Regulatio                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                               | Lessees or Tenants                                                                                               | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 476 cont'd              |                              |                                |                                                        |                                                                                                                  |                                          | Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication<br>apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                    | BOOK OF RE powers of compulsory acquisition, (ii) r                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                            | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|-----------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                   |                                                                                        | Category 1                                                                                                       |                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                         |                              |                                                                                   | Qualifying persons under Regulatio                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                               | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                                               | Occupiers                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                   |
| 476 cont'd              |                              |                                                                                   |                                                                                        |                                                                                                                  |                                                                                                    | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |
| 477                     | Temporary Possession         | 2889 square metres of<br>agricultural land and<br>hedge (north of Hendy<br>Gorse) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU               | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and<br>maintain water pipeline)                                                                                                                                                                                                                                                                                                                                                                              |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF RE<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | to carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|-------------------------------|--------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulatio                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                           | (Applications: Prescribed Forms and      | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                               | Lessees or Tenants                                                                                               | Occupiers                                | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 477 cont'd              |                              |                               |                                                        |                                                                                                                  |                                          | Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication<br>apparatus) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulatio                      | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                                | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                   |
| 477 cont'd              |                              |                               |                                                         |                                                                                                                   |                                         | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                                                           | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                                                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                                                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|-------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                                                          |                                                                                                                                                                                                 | Category 1                                                                                                        |                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|                         |                                                             |                                                                                                                                          | Qualifying persons under Regulatio                                                                                                                                                              | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                                                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                                                      | Owners or Reputed Owners                                                                                                                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                                                                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| 478                     | Acquisition of new rights and imposition of restrictions    | 168 square metres of<br>agricultural land (west of<br>Electricity Distribution<br>Site, south of Ffordd<br>Richard Davies)               | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY                                                                                                          | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR                      | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR                                                                                                    | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 479                     | Acquisition of new rights and<br>imposition of restrictions | 2209 square metres of<br>agricultural land and<br>hedge (west of<br>Electricity Distribution<br>Site, south of Ffordd<br>Richard Davies) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                              | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed<br>dated 6th November 2009, and of<br>restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                              | BOOK OF RE powers of compulsory acquisition, (ii) r                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                  |
|-------------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                             |                                                                                                | Category 1                                                                                                        |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                             |
|                         |                                                          |                                                                                                                                             | Qualifying persons under Regulatio                                                             | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                  |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                         | Owners or Reputed Owners                                                                       | Lessees or Tenants                                                                                                | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                       |
| 479 cont'd              |                                                          |                                                                                                                                             |                                                                                                |                                                                                                                   |                                                                                               | Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication<br>apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |
| 480                     | Acquisition of new rights and imposition of restrictions | 3332 square metres of<br>agricultural land and<br>hedgerow (west of<br>Electricity Distribution<br>Site, south of Ffordd<br>Richard Davies) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                     | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain water pipeline)                                                                                                                                                                                                                                                  |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) r | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|------------------------------|-------------------------------|---------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               |                                                         | Category 1                                                                                                     |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                         |                              |                               | Qualifying persons under Regulation                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                       | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                                | Lessees or Tenants                                                                                             | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 480 cont'd              |                              |                               |                                                         |                                                                                                                |                                         | National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and<br>maintain gas pipelines) |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                                                                          | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                         | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|----------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                                                                         |                                                                                                 | Category 1                                                                                                        |                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                         |                                                          |                                                                                                                                         | Qualifying persons under Regulatio                                                              | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                             | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                                                                     | Owners or Reputed Owners                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                                       | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 480 cont'd              |                                                          |                                                                                                                                         |                                                                                                 |                                                                                                                   |                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed to dated 26th January 1996) |
| 481                     | Acquisition of new rights and imposition of restrictions | 174 square metres of<br>agricultural land and<br>hedge (west of<br>Electricity Distribution<br>Site, south of Ffordd<br>Richard Davies) | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | NONE                                                                                                              | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | HSBC UK Bank PLC<br>1 Centenary Square<br>BIRMINGHAM<br>B1 1HQ<br>(as mortgagee for Robert Bryn Davies<br>and Mona Elizabeth Davies)                                                                                                                                                                                                                                                                                                                                                                                                                                            |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                                                                              |                                                                         |                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                                                                              | Category 1                                                              |                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio                                                           | n 7(1)(a) of the Infrastructure Plannin<br>Procedures) Regulations 2009 | g (Applications: Prescribed Forms and                                                        | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners                                                                     | Lessees or Tenants                                                      | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |
| 481 cont'd              |                                                                                                                                                                                                                                                                                                      |                     | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR |                                                                         | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed<br>dated 6th November 2009, and of<br>restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed<br>of grant dated 19th September 1979,<br>and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by<br>conveyance dated 14th June 1966)<br>Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication<br>apparatus)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                                           |                                                                                                |                                                                                               |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                                           | Qualifying persons under Regulatio                                                             | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009        | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                                       | Owners or Reputed Owners                                                                       | Lessees or Tenants                                                                            | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| 482                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 90 square metres of<br>agricultural land (west of<br>Electricity Distribution<br>Site, south of Ffordd<br>Richard Davies) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain water pipeline)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                          |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|--------------------------------------------------------------------------|-------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                               |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                       | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |
| 482 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                          |                                     | Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication<br>apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and<br>maintain gas pipelines)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                           |                                                                                        |                                                                                              |                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                           | Qualifying persons under Regulatio                                                     | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009      | (Applications: Prescribed Forms and                                                          | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                       | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                           | Occupiers                                                                                    | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                           |  |  |
| 482 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                                           |                                                                                        |                                                                                              |                                                                                              | National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                      |  |  |
| 483                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 4800 square metres of<br>agricultural land (north<br>of Electricity Distribution<br>Site) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |  |  |

|                         | Land which is p                                             | proposed to be subject to: (i)                                                                               | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>'ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                           | n 7(1)(a))                                                                                                                                                                                                                                                                                                                     |
|-------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                             |                                                                                                              | Qualifying persons under Regulatio                                                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                            | (Applications: Prescribed Forms and                                               | Category 2                                                                                                                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Extent of acquisition or use                                | Description of land                                                                                          | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                                                | Occupiers                                                                         | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                          |
| 483 cont'd              |                                                             |                                                                                                              |                                                                                        |                                                                                                                   |                                                                                   | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication<br>apparatus)      |
| 484                     | Acquisition of new rights and<br>imposition of restrictions | 9467 square metres of<br>substation (Electricity<br>Distribution Site, south<br>of Ffordd Richard<br>Davies) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | National Grid Electricity<br>Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH                                 | National Grid Electricity<br>Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables) |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF powers of compulsory acquisition, (ii) ri | m DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------|--------------------------------|-------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                | Qualifying persons under Regulation                   | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                         | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                              | Lessees or Tenants                                                                                             | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 484 cont'd              |                              |                                |                                                       |                                                                                                                |                                         | Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication<br>apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and<br>maintain gas pipelines)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                                           |                                                                                                |                                                                                               |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                                           | Qualifying persons under Regulation                                                            | Category 1<br>on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009       | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                                       | Owners or Reputed Owners                                                                       | Lessees or Tenants                                                                            | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
| 485                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 2378 square metres of<br>agricultural land and<br>hedgerow (south of<br>Electricity Distribution<br>Site) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain water pipeline)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulation | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |
| 485 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                                        |                                     | Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication<br>apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                           |                                                                                        |                                                                                               |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------|----------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                           |                                                                                        | Category 1                                                                                    |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                      |                                                                           | Qualifying persons under Regulatio                                                     | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                      | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                       | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                            | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                       |  |  |  |
| 485 cont'd              |                                                                                                                                                                                                                                                                                                      |                                                                           |                                                                                        |                                                                                               |                                                                                               | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                |  |  |  |
| 486                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                          | 189 square metres of<br>ditch (south of Electricity<br>Distribution Site) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain water pipeline)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables) |  |  |  |

|                         | Land which is p                                          | proposed to be subject to: (i)                                          | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                         | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                       | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|----------------------------------------------------------|-------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                         |                                                                                                | Category 1                                                                                                       |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|                         |                                                          |                                                                         | Qualifying persons under Regulatio                                                             | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                           | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                     | Owners or Reputed Owners                                                                       | Lessees or Tenants                                                                                               | Occupiers                                                                                     | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                               |
| 486 cont'd              |                                                          |                                                                         |                                                                                                |                                                                                                                  |                                                                                               | Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996) |
| 487                     | Acquisition of new rights and imposition of restrictions | 232 square metres of<br>hardstanding (Electricity<br>Distribution Site) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS                    | Arthur Elwy Morris Owen<br>Bodysgaw Isaf<br>Llannefydd<br>DENBIGH<br>Denbighshire<br>LL16 5DS | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and<br>maintain water pipeline)                                                                                                                                                                                                                                                                                                                                                                          |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                    |                                                                                        |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------|----------------------------------------------------------------------------------------|-------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatio | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners           | Lessees or Tenants                                                                     | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
| 487 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                    |                                                                                        |                                     | Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |  |

|                         | Land which is p                                          | proposed to be subject to: (i)                                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii)                                           | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>of Denbighshire | o carry out protective works (Regulatio                                                            | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                          |                                                                                    |                                                                                                | Category 1                                                                                                         |                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                         |                                                          |                                                                                    | Qualifying persons under Regulatio                                                             | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Number on<br>Land Plans | Extent of acquisition or use                             | Description of land                                                                | Owners or Reputed Owners                                                                       | Lessees or Tenants                                                                                                 | Occupiers                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                   |
| 487 cont'd              |                                                          |                                                                                    |                                                                                                |                                                                                                                    |                                                                                                    | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and<br>access for maintenance stated in<br>Deed dated 26th January 1996)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |
| 488                     | Acquisition of new rights and imposition of restrictions | 29 square metres of<br>private road (south of<br>Electricity Distribution<br>Site) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU                 | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU | National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                    |

|                         | Land which is p              | proposed to be subject to: (i)                                             | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                                 | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire                              | o carry out protective works (Regulatio                                                            | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|----------------------------------------------------------------------------|----------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                            |                                                                                        | Category 1                                                                                                                                     |                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                 |
|                         |                              |                                                                            | Qualifying persons under Regulatio                                                     | Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and<br>Procedures) Regulations 2009 |                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                 |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                        | Owners or Reputed Owners                                                               | Lessees or Tenants                                                                                                                             | Occupiers                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                           |
| 488 cont'd              |                              |                                                                            |                                                                                        |                                                                                                                                                |                                                                                                    | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and<br>maintain gas pipelines) |
| 489                     | Acquisition of new rights    | 1012 square metres of<br>track (south of Electricity<br>Distribution Site) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU                                             | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU | National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and<br>maintain electricity cables)                                                             |

|                         | Land which is p              | proposed to be subject to: (i)                                    | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|------------------------------|-------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                   |                                                                                                                            | Category 1                                                                                                       |                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                         |                              |                                                                   | Qualifying persons under Regulatio                                                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                               | Owners or Reputed Owners                                                                                                   | Lessees or Tenants                                                                                               | Occupiers                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                           |
| 489 cont'd              |                              |                                                                   |                                                                                                                            |                                                                                                                  |                                                                           | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and<br>maintain gas pipelines)                                                                                                                                                                                                                                                                                                                                |
| 490                     | Acquisition of new rights    | 1302 square metres of<br>track (south of Glascoed<br>Road, B5381) | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                             | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of shooting rights and<br>restrictive covenants)<br>Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and<br>maintain cable)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access to<br>maintain water pipeline) |

|                         | Land which is p              | roposed to be subject to: (i) | BOOK OF REF powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights to<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|------------------------------|-------------------------------|-------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|-----------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                               | Qualifying persons under Regulatior                   | Category 1<br>n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                        | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land           | Owners or Reputed Owners                              | Lessees or Tenants                                                                                            | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 490 cont'd              |                              |                               |                                                       |                                                                                                               |                                         | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>(in respect of rights stated in Deed<br>dated 6th November 2009)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed<br>dated 6th November 2009)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |

|                         | Land which is p              | proposed to be subject to: (i)                                                                                                       | BOOK OF RE powers of compulsory acquisition, (ii)                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulation                                                                                         | n 7(1)(a))                                                                                                                                                                                    |
|-------------------------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                                                                      |                                                                                                                      | Category 1                                                                                                        |                                                                                                                                  |                                                                                                                                                                                               |
|                         |                              |                                                                                                                                      | Qualifying persons under Regulatio                                                                                   | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                              | Category 2                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                                                                                  | Owners or Reputed Owners                                                                                             | Lessees or Tenants                                                                                                | Occupiers                                                                                                                        | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                         |
| 490 cont'd              |                              |                                                                                                                                      |                                                                                                                      |                                                                                                                   |                                                                                                                                  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                 |
| 491                     | Acquisition of new rights    | 33 square metres of<br>private road (south of<br>Glascoed Road, B5381,<br>and west of Lon Coed<br>Esgob) and footpath<br>(DE 208 32) | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU<br>(as reputed owner)                      | NONE                                                                                                              | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU                                                        | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)                                                    |
|                         |                              |                                                                                                                                      | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(as reputed owner) |                                                                                                                   | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY                                   | ESP Electricity Limited<br>E S P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalls Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of access) |
|                         |                              |                                                                                                                                      |                                                                                                                      |                                                                                                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of footpath DE 208 32) | Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(in respect of right to install and<br>maintain cables)       |

|                         | Land which is p              | proposed to be subject to: (i)                                    | BOOK OF REF<br>powers of compulsory acquisition, (ii) r                                                                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights to<br>f Denbighshire | o carry out protective works (Regulatio                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|------------------------------|-------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | Category 1                   |                                                                   |                                                                                                                            |                                                                                                                   |                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|                         |                              |                                                                   | Qualifying persons under Regulatio                                                                                         | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                               | Owners or Reputed Owners                                                                                                   | Lessees or Tenants                                                                                                | Occupiers                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 491 conťd               |                              |                                                                   |                                                                                                                            |                                                                                                                   |                                                                           | National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed<br>dated 6th November 2009)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                                       |  |
| 492                     | Acquisition of new rights    | 1147 square metres of<br>track (south of Glascoed<br>Road, B5381) | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU<br>Unknown<br>(in respect of mines and minerals) | NONE                                                                                                              | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)<br>ESP Electricity Limited<br>E S P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalls Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of access)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(in respect of rights stated in Deed<br>dated 6th November 2009) |  |

|                         | Land which is p              | proposed to be subject to: (i) | BOOK OF REF<br>powers of compulsory acquisition, (ii) ri | n DEVELOPMENT CONSENT ORDER<br>ERENCE - PART 1<br>ght to use the land, and/or (iii) rights t<br>Denbighshire | o carry out protective works (Regulatio | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-------------------------|------------------------------|--------------------------------|----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                |                                                          | Category 1                                                                                                   |                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|                         |                              |                                | Qualifying persons under Regulation                      | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                     | (Applications: Prescribed Forms and     | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land            | Owners or Reputed Owners                                 | Lessees or Tenants                                                                                           | Occupiers                               | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 492 cont'd              |                              |                                |                                                          |                                                                                                              |                                         | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed<br>dated 6th November 2009)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants) |

|                         | Land which is                | proposed to be subject to: (i                                 | BOOK OF RE<br>i) powers of compulsory acquisition, (ii)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) right:<br>f Denbighshire | s to carry out protective works (Regulatio                                                                           | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|---------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | T                                                                                                               |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                              |                                                               | Qualifying persons under Regulatic                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | n 7(1)(a) of the Infrastructure Plannir<br>Procedures) Regulations 2009                                         | g (Applications: Prescribed Forms and                                                                                | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                           | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Lessees or Tenants                                                                                              | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                         |
| 493                     | Temporary Possession         | 177 square metres of<br>public road (Glascoed<br>Road, B5381) | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Eryl James Gomer Davies<br>Derwen Deg<br>Glascoed Road<br>ST. ASAPH<br>LL17 0LH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Nesta Wyn Davies<br>Derwen Deg<br>Glascoed Road<br>ST. ASAPH<br>LL17 0LH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Nesta Wyn Davies<br>Derwen Deg<br>Glascoed Road<br>ST. ASAPH<br>LL17 0LH<br>(in respect of subsoil beneath half<br>width of public highway)<br>Unknown | NONE                                                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown |

|                         | Land which is p              | proposed to be subject to: (i)                                                  | BOOK OF REI<br>powers of compulsory acquisition, (ii) r                                                                            | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                                            | n 7(1)(a))                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------|---------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                                 |                                                                                                                                    | Category 1                                                                                                       |                                                                                                                                    |                                                                                                                                                                                                                                                                                              |
|                         |                              |                                                                                 | Qualifying persons under Regulatio                                                                                                 | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                                                                                | Category 2                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                             | Owners or Reputed Owners                                                                                                           | Lessees or Tenants                                                                                               | Occupiers                                                                                                                          | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                        |
| 494                     | Temporary Possession         | 71 square metres of<br>hedgerow (south of<br>Glascoed Road, B5381)              | Eryl James Gomer Davies<br>Derwen Deg<br>Glascoed Road<br>ST. ASAPH<br>LL17 OLH<br>Nesta Wyn Davies<br>Derwen Deg<br>Glascoed Road | NONE                                                                                                             | Eryl James Gomer Davies<br>Derwen Deg<br>Glascoed Road<br>ST. ASAPH<br>LL17 OLH<br>Nesta Wyn Davies<br>Derwen Deg<br>Glascoed Road | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of rights contained in a<br>Conveyance dated 20 March 1919)                                                          |
|                         |                              |                                                                                 | ST. ASAPH<br>LL17 OLH<br>Unknown<br>(in respect of mines and minerals)                                                             |                                                                                                                  | ST. ASAPH<br>LL17 OLH                                                                                                              | ,                                                                                                                                                                                                                                                                                            |
| 495                     | Temporary Possession         | 15 square metres of<br>track and hedgerow<br>(south of Glascoed Road,<br>B5381) | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU<br>Unknown<br>(in respect of mines and minerals)         | NONE                                                                                                             | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU                                                          | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)<br>ESP Electricity Limited<br>E S P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalls Road<br>LEATHERHEAD<br>Surroy |
|                         |                              |                                                                                 |                                                                                                                                    |                                                                                                                  |                                                                                                                                    | Surrey<br>KT22 7BA<br>(in respect of right of access)                                                                                                                                                                                                                                        |

|                         | Land which is p              | proposed to be subject to: (i)                               | BOOK OF RE powers of compulsory acquisition, (ii) r                                                                                          | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                                                              | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|------------------------------|--------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                              |                                                                                                                                              | Category 1                                                                                                        |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|                         |                              |                                                              | Qualifying persons under Regulatio                                                                                                           | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                          | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                          | Owners or Reputed Owners                                                                                                                     | Lessees or Tenants                                                                                                | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                |
| 495 cont'd              |                              |                                                              |                                                                                                                                              |                                                                                                                   |                                                                                                                      | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>(in respect of rights stated in Deed<br>dated 6th November 2009)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed<br>dated 6th November 2009)<br>Unknown<br>(in respect of restrictive covenants)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |
| 496                     | Acquisition of new rights    | 60 square metres of<br>public road (Glascoed<br>Road, B5381) | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU<br>(in respect of subsoil beneath half<br>width of public highway) | NONE                                                                                                              | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                                                                                       |

|                         | Land which is p              | proposed to be subject to: (i)                                   | BOOK OF RE powers of compulsory acquisition, (ii)                                                                               | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>right to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|------------------------------|------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                  |                                                                                                                                 | Category 1                                                                                                        |                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                              |                                                                  | Qualifying persons under Regulatio                                                                                              | on 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                              | Owners or Reputed Owners                                                                                                        | Lessees or Tenants                                                                                                | Occupiers                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                         |
| 496 cont'd              |                              |                                                                  | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Unknown |                                                                                                                   |                                                                           | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>Unknown                                                                                                                                                                                                                                                                                                                                                    |
| 497                     | Acquisition of new rights    | 194 square metres of<br>track (south of Glascoed<br>Road, B5381) | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU                                                       | NONE                                                                                                              | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and<br>maintain cable)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>ESP Electricity Limited<br>E S P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalls Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of way) |

|                         | Land which is p              | proposed to be subject to: (i)                                     | BOOK OF RE<br>powers of compulsory acquisition, (ii) r                    | m DEVELOPMENT CONSENT ORDER<br>FERENCE - PART 1<br>ight to use the land, and/or (iii) rights t<br>f Denbighshire | o carry out protective works (Regulatio                                   | n 7(1)(a))                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|------------------------------|--------------------------------------------------------------------|---------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                              |                                                                    |                                                                           | Category 1                                                                                                       |                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                         |                              |                                                                    | Qualifying persons under Regulatio                                        | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009                                         | (Applications: Prescribed Forms and                                       | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Number on<br>Land Plans | Extent of acquisition or use | Description of land                                                | Owners or Reputed Owners                                                  | Lessees or Tenants                                                                                               | Occupiers                                                                 | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                              |
| 497 cont'd              |                              |                                                                    |                                                                           |                                                                                                                  |                                                                           | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>(in respect of rights stated in Deed<br>dated 6th November 2009)<br>National Grid Electricity Transmission<br>PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed<br>dated 6th November 2009)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus) |
| 498                     | Temporary Possession         | 13 square metres of<br>hedgerow (south of<br>Glascoed Road, B5381) | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU | NONE                                                                                                             | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and<br>maintain cable)                                                                                                                                                                                                                                                                                                                      |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                     |                                     |                                                                          |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------|--------------------------------------------------------------------------|-------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                                                                                                                                                                                                                      |                     |                                     | Category 1                                                               |                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                         |                                                                                                                                                                                                                                                                                                      |                     | Qualifying persons under Regulatior | n 7(1)(a) of the Infrastructure Planning<br>Procedures) Regulations 2009 | (Applications: Prescribed Forms and | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land | Owners or Reputed Owners            | Lessees or Tenants                                                       | Occupiers                           | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 498 cont'd              |                                                                                                                                                                                                                                                                                                      |                     |                                     |                                                                          |                                     | ESP Electricity Limited<br>E S P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalls Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of way)<br>Gwynt Y Mor Offshore Wind Farm<br>Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(in respect of rights stated in Deed<br>dated 6th November 2009)<br>Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>(in respect of rights stated in Deed<br>dated 6th November 2009)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 1<br>Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))<br>County of Denbighshire |                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                        |                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                                                                                                                                                                                                                      |                                                                                    | Qualifying persons under Regulatio                                                                                                                                                                                                                                                                                                                                                                                                     | Category 1<br>n 7(1)(a) of the Infrastructure Planning | (Applications: Prescribed Forms and                                                                                  | Category 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                         | Description of land                                                                | Owners or Reputed Owners                                                                                                                                                                                                                                                                                                                                                                                                               | Procedures) Regulations 2009<br>Lessees or Tenants     | Occupiers                                                                                                            | Qualifying persons under Regulation<br>7 (1)(a) of the Infrastructure Planning<br>(Applications: Prescribed Forms and<br>Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 499                     | Temporary Possession                                                                                                                                                                                                                                                                                 | 137 square metres of<br>verge of public road<br>(south of Glascoed Road,<br>B5381) | Betty May Jones<br>31 Tan Y Bryn<br>ST. ASAPH<br>Denbighshire<br>LL17 OPU<br>(in respect of subsoil beneath half<br>width of public highway)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority)<br>Linda Louise Griffiths<br>Cottage Villa<br>Glascoed Road<br>LLANELWY<br>LL17 0LH<br>(in respect of subsoil beneath full<br>width of public highway)<br>Unknown | NONE                                                   | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(as highway authority) | Diamond Transmission Partners BBE<br>Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication<br>apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                      |                                                                                                                                                                                                                                          |  |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                         |                                                                                                                                                                                                                                          |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                          | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                           |  |  |
| 1                       | 217151 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                   |  |  |
| 2                       | 111434 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                   |  |  |
| 4                       | 155 square metres of foreshore land seaward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown)         | Unknown                                                                                                                                                                                                                                  |  |  |
| 7                       | 34 square metres of foreshore land seaward of the Mean High Water at Rhyl<br>(excluding all interests of the Crown)                          | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)                                                          |  |  |
|                         |                                                                                                                                              | Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                                                                                          |  |  |
| 8                       | 4615 square metres of footway (north of Rhyl Coast Road, A548) and footpath<br>(DE 207 11)                                                   | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>The Natural Resources Body For Wales<br>Cambria House |  |  |
|                         |                                                                                                                                              | 29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)                                                                                                                                                                |  |  |
|                         |                                                                                                                                              | Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                                                                                          |  |  |

|                         | Awel v Môr Offst                                                                                                                 | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                           |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                                                  | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                         |
|                         |                                                                                                                                  | County of Denbighshire                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Description of Land                                                                                                              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                     |
| 9                       | 2 square metres of footway (north of Rhyl Coast Road, A548)                                                                      | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883) |
| 10                      | 113 square metres of foreshore landward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                            |
| 11                      | 12 square metres of slipway seaward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown)     | Unknown                                                                                                                                                                                                                                                                                            |
| 12                      | 86 square metres of slipway landward of Mean High Water north of Rhyl (excluding all interests of the Crown)                     | Unknown                                                                                                                                                                                                                                                                                            |
| 13                      | 345 square metres of footway and grassed area (north of Garford Road)                                                            | The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 OTP<br>(in respect of right of access)                                                                                                                                                                 |
| 14                      | 347 square metres of footway, hardstanding and grassed area (north of Garford Road)                                              | The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 OTP<br>(in respect of right of access)                                                                                                                                                                 |
| 15                      | 3125 square metres of hardstanding and grassed area (north of Brynhedydd Bay)<br>and footpath (DE 207 11)                        | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883) |
| 16                      | 2728 square metres of private road, buildings and hardstanding at Rhyl Golf Club,<br>Rhyl                                        | AP Wireless II (UK) Limited<br>A P Wireless Ltd<br>16-18 Conduit Street<br>LICHFIELD<br>WS13 6JR<br>(in respect of rights stated in Lease dated 28th September 2012)                                                                                                                               |

|                         | Awel y                                                    | y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                       |
|-------------------------|-----------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans | Description of Land                                       | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                   |
| 16 cont'd               |                                                           | Cornerstone Telecommunications Infrastructure Limited<br>Hive 2<br>1530 Arlington Business Park<br>Theale<br>READING<br>Berkshire<br>RG7 4SA<br>(in respect of rights granted by Lease dated 17th November 2017) |
|                         |                                                           | EE Limited<br>Trident Place<br>Mosquito Way<br>HATFIELD<br>Hertfordshire<br>AL10 9BW<br>(in respect of rights stated in Lease dated 31st March 2004)                                                             |
|                         |                                                           | Hutchison 3G UK Limited<br>450 Longwater Avenue<br>READING<br>RG2 6GF<br>(in respect of rights stated in Lease dated 31st March 2004)                                                                            |
| 17                      | 12 square metres of hardstanding at Rhyl Golf Club, Rhyl  | EE Limited<br>Trident Place<br>Mosquito Way<br>HATFIELD<br>Hertfordshire<br>AL10 9BW<br>(in respect of rights stated in Lease dated 31st March 2004)<br>Hutchison 3G UK Limited                                  |
|                         |                                                           | 450 Longwater Avenue<br>READING<br>RG2 6GF<br>(in respect of rights stated in Lease dated 31st March 2004)                                                                                                       |
| 18                      | 206 square metres of grassed area at Rhyl Golf Club, Rhyl | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of right of access)                                                                                    |

|                         | Awel y Môr Offsl                                                                                                                    | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                           |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                                                     | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                         |
|                         |                                                                                                                                     | County of Denbighshire                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Description of Land                                                                                                                 | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                     |
| 19                      | 15 square metres of grassed area at Rhyl Golf Club, Rhyl                                                                            | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of right of access)                                                                                                                                                                      |
| 20                      | 123 square metres of private road at Rhyl Golf Club, Rhyl                                                                           | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of right of access)                                                                                                                                                                      |
| 21                      | 483 square metres of footway of public road (north of Rhyl Coast Road, A548)                                                        | Unknown                                                                                                                                                                                                                                                                                            |
| 22                      | 152117 square metres of grassed area and hardstanding (Rhyl Golf Club, north of Rhyl Coast Road, A548) and footpath (DE   207   11) | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883) |
| 23                      | 34 square metres of foreshore landward of the Mean High Water lying to the north of Rhyl, (excluding all interests of the Crown)    | Unknown                                                                                                                                                                                                                                                                                            |
| 24                      | 4159 square metres of foreshore landward of the Mean High Water and footway<br>(north of Rhyl Coast Road, A548)                     | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883) |
| 25                      | 15 square metres of hardstanding (north of Rhyl Coast Road, A548)                                                                   | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)                                                                                                                    |

|                         | Awel y Môr Offs                                                                                                                       | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                             |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                      |  |
|                         |                                                                                                                                       | County of Denbighshire                                                                                                                                                                                                                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                                                                                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                       |  |
| 25 cont'd               |                                                                                                                                       | The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                                                                                |  |
| 26                      | Plot No Longer Included In DCO                                                                                                        | Plot No Longer Included In DCO                                                                                                                                                                                                                                                                                                                                       |  |
| 27                      | 1209 square metres of foreshore land seaward of the Mean High Water at Rhyl<br>(excluding all interests of the Crown)                 | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                                   |  |
| 28                      | 16 square metres of foreshore land seaward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown)   | Rhyl Golf Club Limited         Rhyl Golf Club Limited         Rhyl Coast Road         RHYL         Clwyd         LL18 3RE         (in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)         Unknown         (in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883) |  |
| 29                      | 466 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                                                                                              |  |
| 30                      | 8349 square metres of foreshore landward of the Mean High Water and footway<br>lying to the north of Rhyl                             | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                                   |  |

|                         | Awel y Môr Offsh                                                                                                                           | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                           |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                       |                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                                                                                                        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                     |  |
| 31                      | 218 square metres of foreshore land seaward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown)       | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883) |  |
| 32                      | 6074 square metres of foreshore land seaward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown)      | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883) |  |
| 33                      | 63 square metres of footway (north of Rhyl Coast Road, A548)                                                                               | Unknown                                                                                                                                                                                                                                                                                            |  |
| 34                      | 3534 square metres of foreshore landward of the Mean High Water and footway (north of Rhyl Coast Road, A548)                               | Unknown                                                                                                                                                                                                                                                                                            |  |
| 35                      | 9198 square metres of foreshore land seaward of the Mean High Water and footway at Prestatyn (excluding all interests of the Crown)        | Unknown                                                                                                                                                                                                                                                                                            |  |
| 36                      | 160 square metres of footway (north of Rhyl Coast Road, A548)                                                                              | Unknown                                                                                                                                                                                                                                                                                            |  |
| 37                      | 275 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn                                             | Unknown                                                                                                                                                                                                                                                                                            |  |
| 38                      | 379 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)  | Unknown                                                                                                                                                                                                                                                                                            |  |
| 39                      | 3649 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter)                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                            |                                                                                                                                                                                                                                                                                                                             |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                              |  |
| 39 cont'd               |                                                                                                                                                            | Unknown<br>(in respect of right of access)                                                                                                                                                                                                                                                                                  |  |
| 40                      | 1228 square metres of foreshore landward of the Mean High Water and<br>hardstanding lying to the north of Prestatyn and footpath (DE 205 48, DE 205 49)    | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of access) |  |
| 41                      | 1013 square metres of footway (north of Rhyl Coast Road, A548) and footpath<br>(DE 205 48)                                                                 | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of access) |  |
| 42                      | 35 square metres of footway (north of Rhyl Coast Road, A548)                                                                                               | Unknown                                                                                                                                                                                                                                                                                                                     |  |
| 43                      | 67 square metres of footway and foreshore landward of the Mean High Water<br>(north of Rhyl Coast Road, A548) and footpath (DE 205 49)                     | Unknown                                                                                                                                                                                                                                                                                                                     |  |
| 44                      | 47 square metres of hardstanding and foreshore land seaward of the Mean High Water (north of Rhyl Coast Road, A548) (excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                                                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                        |                                                                                                                                                                                                                               |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                                              | County of Denbighshire                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                                                                          | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                |  |
| 45                      | 183 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)    | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                          |  |
|                         |                                                                                                                                              | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                   |  |
| 46                      | 28705 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn, (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                        |  |
| 47                      | 73 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)         | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Unknown                                                                             |  |
| 48                      | 4312 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)   | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                        |  |
| 49                      | 7 square metres of footway seaward of the Mean High Water (north of Victoria<br>Road, A548) (excluding all interests of the Crown)           | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934) |  |
|                         |                                                                                                                                              | Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                            |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                      |                                                                                                                                                                                             |  |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | County of Denbighshire                                                                                                                     |                                                                                                                                                                                             |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                              |  |  |
| 50                      | 17734 square metres of footway landward of the Mean High Water (north of Victoria Road, A548) and footpath (DE 205 49)                     | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                        |  |  |
|                         |                                                                                                                                            | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951) |  |  |
| 51                      | 5853 square metres of footway (north of Victoria Road, A548) and footpath<br>(DE 205 49)                                                   | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                        |  |  |
|                         |                                                                                                                                            | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951) |  |  |
| 52                      | 3329 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)     | Unknown                                                                                                                                                                                     |  |  |
| 53                      | 56 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)       | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Unknown                                           |  |  |
| 54                      | 1968 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                      |  |  |

|                         | Awel v Môr Offst                                                                                                                           | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                   |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         | ,                                                                                                                                          | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                 |
|                         |                                                                                                                                            | County of Denbighshire                                                                                                                                                                                                                     |
| Number on<br>Land Plans | Description of Land                                                                                                                        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                             |
| 55                      | 1175 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                     |
| 56                      | 167 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)      | Unknown                                                                                                                                                                                                                                    |
| 57                      | 2173 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                     |
| 58                      | 1839 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                     |
| 59                      | 174 square metres of foreshore land seaward of the Mean High Water lying north of Prestatyn (excluding all interests of the Crown)         | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                     |
| 60                      | 26 square metres of foreshore landward of the Mean High Water lying north of Prestatyn (excluding all interests of the Crown)              | Unknown                                                                                                                                                                                                                                    |
| 61                      | 15 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)       | Unknown                                                                                                                                                                                                                                    |
| 62                      | 795 square metres of footway and foreshore landward of the Mean High Water<br>(north of Victoria Road, A548)                               | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants) |

|                         | Awel y Môr Offst                                                                                                 | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                         |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                             |                                                                                                                                                                                                                                                                                                                                  |  |  |
| Number on<br>Land Plans | Description of Land                                                                                              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                   |  |  |
| 62 cont'd               |                                                                                                                  | Unknown<br>(in respect of rights granted by Deed of easement dated 7th March 2005)                                                                                                                                                                                                                                               |  |  |
| 63                      | 155 square metres of slipway at Ffrith Beach landward of Mean High Water (north of Victoria Road, A548)          | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of easement dated 7th March 2005) |  |  |
| 64                      | 94 square metres of hardstanding and foreshore landward of the Mean High Water<br>(north of Victoria Road, A548) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of easement dated 7th March 2005) |  |  |
| 65                      | 25 square metres of hardstanding and foreshore landward of the Mean High Water<br>(north of Victoria Road, A548) | Unknown                                                                                                                                                                                                                                                                                                                          |  |  |
| 66                      | 17 square metres of footway (north of Victoria Road, A548)                                                       | Unknown                                                                                                                                                                                                                                                                                                                          |  |  |
| 67                      | 53 square metres of grassed area (north of Victoria Road, A548)                                                  | Unknown                                                                                                                                                                                                                                                                                                                          |  |  |
| 68                      | 3178 square metres of grassed area (north of Victoria Road, A548)                                                | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed of easement dated 7th March 2005)                                                                                                                                           |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                         |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                          |  |  |
| 68 cont'd               |                                                                                                                 | Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of easement dated 7th March 2005)                                                                                                                                  |  |  |
| 69                      | 3442 square metres of grassed area (north of Victoria Road, A548)                                               | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA             |  |  |
|                         |                                                                                                                 | <ul> <li>(in respect of right of way)</li> <li>Unknown</li> <li>(in respect of restrictive covenants in Conveyance dated 14 February 1934)</li> <li>Unknown</li> <li>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)</li> </ul> |  |  |
| 69a                     | Plot No Longer Included In DCO                                                                                  | Plot No Longer Included In DCO                                                                                                                                                                                                                                          |  |  |
| 70                      | 45 square metres of footway (north of Victoria Road, A548)                                                      | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                    |  |  |
|                         |                                                                                                                 | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                             |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
| Number on<br>Land Plans | Description of Land                                                                                 | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                             |  |  |
| 71                      | 55 square metres of grassed area (north of Victoria Road, A548)                                     | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                        |  |  |
| 72                      | 891 square metres of grassed area and track (north of Victoria Road, A548)                          | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951) |  |  |
| 73                      | 57 square metres of grassed area at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                        |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         | BOOK OF REFERENCE - PART 2                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
|                         | County of Denbighshire                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Description of Land                                                                            | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                             |  |  |  |
| 74                      | 66 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)                                                                                                                                                                                                                                                              |  |  |  |
|                         |                                                                                                | Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                                                                                                                                                                                                                                         |  |  |  |
| 75                      | 2624 square metres of grassed area (north of Victoria Road, A548)                              | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951) |  |  |  |
| 76                      | 469 square metres of private road and hardstanding (north of Ferguson Avenue)                  | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                                                                                                                                                                                                                       |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                   |                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                      |                                                                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                       | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                       |  |
| 76 cont'd               |                                                                                                           | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)                                                                                                                                                                                                                                                                                                |  |
|                         |                                                                                                           | Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                                                                                                                                   |  |
| 77                      | 22 square metres of private road and verge (north of Ferguson Avenue)                                     | Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                                     |  |
| 78                      | 10 square metres of public road and verge (Ferguson Avenue)                                               | Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                                     |  |
| 79                      | 52 square metres of public road (Ferguson Avenue)                                                         | Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                                     |  |
| 80                      | 24 square metres of private road (north of Ferguson Avenue)                                               | Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                                     |  |
| 81                      | 1729 square metres of grassed area and footway (north of Victoria Road, A548 and west of Ferguson Avenue) | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934) |  |
|                         |                                                                                                           | Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                                                                                                                                   |  |
| 82                      | 225 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria Road, A548)              | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                                                                                                                 |  |
|                         |                                                                                                           | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)                                                                                                                                                                                                                                                                                                |  |
|                         |                                                                                                           | Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                                                                                                                                   |  |

| Awel y Môr Offs                                                                                  | shore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|--------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
| Description of Land                                                                              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| 1746 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|                                                                                                  | (in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| 27 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548)   | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|                                                                                                  | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 19 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548)   | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|                                                                                                  | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 16 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria Road, A548)      | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|                                                                                                  | Description of Land         1746 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria Road, A548)         27 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria Road, A548)         19 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria Road, A548)         19 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria Road, A548)         19 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria Road, A548)         16 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria Road, A548) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                         |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                          |  |
| 86 cont'd               |                                                                                                                 | Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                      |  |
| 87                      | 30836 square metres of grassed area (Rhyl Golf Club, north of Rhyl Coast Road,<br>A548)                         | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)                 |  |
| 88                      | 14607 square metres of grassed area and footway (Rhyl Golf Club, north of Rhyl<br>Coast Road, A548)             | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)                 |  |
| 89                      | 361 square metres of public road (Rhyl Coast Road, A548)                                                        | Unknown                                                                                                                                                 |  |
| 90                      | 5 square metres of hardstanding (north of Rhyl Coast Road, A548)                                                | Unknown                                                                                                                                                 |  |
| 91                      | 1363 square metres of public road (Rhyl Coast Road, A548)                                                       | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                  |  |
| 92                      | 232 square metres of footway of public road (south of Rhyl Coast Road, A548)                                    | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                  |  |
| 93                      | 116 square metres of footway of public road (south of Rhyl Coast Road, A548)                                    | Unknown                                                                                                                                                 |  |
| 94                      | 11 square metres of electricity substation (west of Sherwood Avenue and south of Rhyl Coast Road, A548)         | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for maintenance) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                      |                                                                                                                                                                                                                                                                         |  |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                   |                                                                                                                                                                                                                                                                         |  |  |
|                         | County of Denbighshire                                                                                       |                                                                                                                                                                                                                                                                         |  |  |
| Number on<br>Land Plans | Description of Land                                                                                          | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                          |  |  |
| 94 cont'd               |                                                                                                              | Iain Alexander Douglas Blair Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as contained in a Conveyance dated 30th November 1906)<br>Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN |  |  |
| 0.5                     |                                                                                                              | (in respect of restrictive covenants as contained in a Conveyance dated 16th October 1933)                                                                                                                                                                              |  |  |
| 95                      | 808 square metres of footway of public road (south of Rhyl Coast Road, A548)                                 | Unknown                                                                                                                                                                                                                                                                 |  |  |
| 96                      | 18635 square metres of buildings, caravans, hardstanding and grassed area (Robin<br>Hood Holiday Camp, Rhyl) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for maintenance)                                                                                                                 |  |  |
|                         |                                                                                                              | lain Alexander Douglas Blair Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as contained in a Conveyance dated 30th November 1906)                                                                                   |  |  |
|                         |                                                                                                              | Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(in respect of restrictive covenants as contained in a Conveyance dated 16th October 1933)                                                                                            |  |  |
|                         |                                                                                                              | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights stated in Lease dated 14 May 1959)                                                                                                        |  |  |
|                         |                                                                                                              | Unknown<br>(in respect of right of way as contained in a Conveyance dated 10th December 1935)                                                                                                                                                                           |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire        |                                                                                                                                                                                       |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                    | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                        |  |
| 97                      | 22566 square metres of buildings, caravans, hardstanding and grassed area (Robin<br>Hood Holiday Camp, Rhyl)           | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for maintenance)                               |  |
|                         |                                                                                                                        | Iain Alexander Douglas Blair Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as contained in a Conveyance dated 30th November 1906) |  |
|                         |                                                                                                                        | Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(in respect of restrictive covenants as contained in a Conveyance dated 16th October 1933)          |  |
|                         |                                                                                                                        | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                         |  |
|                         |                                                                                                                        | Unknown<br>(in respect of right of way as contained in a Conveyance dated 10th December 1935)                                                                                         |  |
| 98                      | 7994 square metres of private road, buildings, caravans, hardstanding and grassed area (Robin Hood Holiday Camp, Rhyl) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for maintenance)                               |  |
|                         |                                                                                                                        | lain Alexander Douglas Blair Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as contained in a Conveyance dated 30th November 1906) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                           |                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|---------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                        |                                                                                                                                                                                                                                                                                                                                                                                         |  |
|                         |                                                                                                   | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                               | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                          |  |
| 98 cont'd               |                                                                                                   | Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(in respect of restrictive covenants as contained in a Conveyance dated 16th October 1933)                                                                                                                                                                                                            |  |
|                         |                                                                                                   | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights stated in Lease dated 14 May 1959)<br>Unknown                                                                                                                                                                                                             |  |
|                         |                                                                                                   | (in respect of right of way as contained in a Conveyance dated 10th December 1935)                                                                                                                                                                                                                                                                                                      |  |
| 99                      | 2590 square metres of railway (Prestatyn and Rhyl) and hedgerow                                   | Unknown                                                                                                                                                                                                                                                                                                                                                                                 |  |
| 100                     | 27103 square metres of agricultural land (north of The Cut and south of Rhyl Coast<br>Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964)                     |  |
| 101                     | 2759 square metres of agricultural land (north of The Cut and south of Rhyl Coast<br>Road, A548)  | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |  |

|                         | Awel y Môr Offsl                                                                                                                 | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                                                  | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                                                    |
| Number on<br>Land Plans | Description of Land                                                                                                              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                          |
| 102                     | 15 square metres of agricultural land (north of The Cut and south of Rhyl Coast<br>Road, A548)                                   | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                            |
|                         |                                                                                                                                  | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |
| 103                     | 323 square metres of agricultural land (north of The Cut and south of Rhyl Coast<br>Road, A548)                                  | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted)                                                                                |
|                         |                                                                                                                                  | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |
| 104                     | 417 square metres of hedgerow, track and watercourse (The Cut) (north of Dyserth Road, B5119 and south of Rhyl Coast Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                            |
|                         |                                                                                                                                  | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |

|                         | Awel v Môr Offst                                                                                                                    | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                 |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                                          |                                                                                                                                                                                                                          |  |
|                         |                                                                                                                                     | County of Denbighshire                                                                                                                                                                                                   |  |
| Number on<br>Land Plans | Description of Land                                                                                                                 | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                           |  |
| 105                     | 8131 square metres of agricultural land and watercourse (The Cut) (north of Dyserth Road, B5119 and south of Rhyl Coast Road, A548) | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959) |  |
| 106                     | 259 square metres of agricultural land (north of Dyserth Road, B5119 and south of The Cut)                                          | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959) |  |
| 107                     | 17 square metres of watercourse (The Cut) and banks (north of Dyserth Road, B5119)                                                  | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959) |  |
| 108                     | 1 square metre of agricultural land (north of Dyserth Road, B5119 and south of The<br>Cut)                                          | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959) |  |
| 109                     | 1489 square metres of agricultural land (north of Dyserth Road, B5119 and south of The Cut)                                         | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959) |  |
| 110                     | 470 square metres of hedgerow (north of Dyserth Road, B5119 and south of The Cut)                                                   | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959) |  |

|                         | Awel y Môr Offsl                                                                                                   | nore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land                                                                                                | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                          |  |
| 111                     | 2039 square metres of track (north of Dyserth Road, B5119)                                                         | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                                                                                                            |  |
| 112                     | 6479 square metres of agricultural land and track (north of Dyserth Road, B511 and south of Rhyl Coast Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |  |
| 113                     | 2 square metres of agricultural land (north of Dyserth Road, B5119)                                                | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                              |  |
| 114                     | 681 square metres of agricultural land (north of Dyserth Road, B5119)                                              | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                                                                                                            |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                          |  |
| 115                     | 449 square metres of agricultural land (north of Dyserth Road, B5119)                                           | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                                                                                                            |  |
| 116                     | 3362 square metres of track (north of Dyserth Road, B5119)                                                      | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |  |
| 117                     | 2 square metres of wooded area (north of Dyserth Road, B5119)                                                   | Diamond Transmission Partners BBE LimitedMid City Place71 High HolbornLONDONWC1V 6BA(in respect of right to lay and maintain cable)Wales and West Utilities LimitedWales & West HouseSpooner CloseCoedkernewCASNEWYDDNP10 8FZ(in respect of gas apparatus)                                                                                                                              |  |
| 118                     | 21327 square metres of agricultural land and hedgerow (north of Dyserth Road, B5119)                            | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                                                                                                            |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                             |  |
| 119                     | 974 square metres of agricultural land (north of Dyserth Road, B5119)                                           | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                               |  |
| 120                     | 9265 square metres of agricultural land (north of Dyserth Road, B5119)                                          | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                               |  |
| 121                     | 2650 square metres of agricultural land, hedgerow and track (north of Dyserth Road, B5119)                      | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                               |  |
| 122                     | 2 square metres of wooded area (north of Dyserth Road, B5119)                                                   | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |
| 123                     | 843 square metres of agricultural land (south of Dyserth Road, B5119)                                           | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights to lay and maintain services)                                                                                                                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                               |  |
| 124                     | 113 square metres of wooded area (north of Dyserth Road, B5119)                                                 | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)      |  |
|                         |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)              |  |
| 125                     | 56 square metres of verge of public road (north of Dyserth Road, B5119)                                         | Unknown                                                                                                                                                      |  |
| 126                     | 21 square metres of track (north of Dyserth Road, B5119)                                                        | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017) |  |
| 127                     | 106 square metres of hedgerow (north of Dyserth Road, B5119)                                                    | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017) |  |
| 128                     | 305 square metres of agricultural land (north of Dyserth Road, B5119)                                           | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017) |  |
| 129                     | 593 square metres of agricultural land (north of Dyserth Road, B5119)                                           | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017) |  |

|                         | Awel v Môr Offs                                                                               | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|                         |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                           | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| 130                     | 150 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119)            | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights to lay and maintain apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 131                     | 444 square metres of public road (Dyserth Road, B5119) (excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| 133                     | 677 square metres of agricultural land and hedgerow (south of Dyserth Road,<br>B5119)         | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of right to take water and use water pipe)<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of right to take water and use water pipe)<br>Jason Wayne Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to septic tank)<br>Jeremy Charles Salisbury<br>Salisbury & Co Business Solutions Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>(in respect of right to take water and use water pipe) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
|-------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | County of Denbighshire                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
| Number on<br>Land Plans | Description of Land                                                                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
| 133 cont'd              |                                                                                       | Pamela Elaine Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to septic tank)<br>Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |
| 134                     | 5 square metres of hedgerow (south of Dyserth Road, B5119)                            | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of right to take water and use water pipe)<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of right to take water and use water pipe)<br>Jason Wayne Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to septic tank)<br>Jeremy Charles Salisbury<br>Salisbury & Co Business Solutions Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN<br>(in respect of right to take water and use water pipe) |  |  |

|                         | Awel v Môr Offst                                                                                | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                  |  |
|-------------------------|-------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                      |                                                                                                                                                                                                                                                                                           |  |
|                         | County of Denbighshire                                                                          |                                                                                                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                            |  |
| 134 cont'd              |                                                                                                 | Pamela Elaine Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to septic tank)<br>Unknown                                                                                                                                                      |  |
| 135                     | 19 square metres of verge of public road (south of Dyserth Road, B5119) and footpath (DE 206 5) | Unknown                                                                                                                                                                                                                                                                                   |  |
| 136                     | 16 square metres of verge of public road (south of Dyserth Road, B5119) and footpath (DE 206 5) | Richard Gratton<br>8 Eldon Drive<br>ABERGELE<br>LL22 7BY<br>(in respect of right to passage of service media)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline) |  |
| 137                     | 50 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119)               | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                           |  |
| 138                     | 54 square metres of agricultural land (south of Dyserth Road, B5119)                            | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                           |  |
| 139                     | 158 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119)              | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights to lay and maintain services)                                                                                                                               |  |

|                         | Awel y Môr Off                                                                                                         | shore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                    |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                             |                                                                                                                                                                                                                                                                                              |  |
|                         |                                                                                                                        | County of Denbighshire                                                                                                                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                                                                                    | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                               |  |
| 140                     | 32097 square metres of agricultural land, hedgerow and track (south of Dyserth<br>Road, B5119) and footpath (DE 206 5) | Centrica PLCMillstreamMaidenhead RoadWINDSORBerkshireSL4 5GD(in respect of right of access for maintenance)Wales and West Utilities LimitedWales & West HouseSpooner CloseCoedkernewCASNEWYDDNP10 8FZ(in respect of right to lay and maintain gas pipeline stated in deed dated 19 May 1955) |  |
| 141                     | 2528 square metres of agricultural land (south of Dyserth Road, B5119)                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                              |  |
| 142                     | 3651 square metres of agricultural land and track (south of Dyserth Road, B5119)<br>and footpath (DE 206 5)            | Centrica PLCMillstreamMaidenhead RoadWINDSORBerkshireSL4 5GD(in respect of right of access for maintenance)Wales and West Utilities LimitedWales & West HouseSpooner CloseCoedkernewCASNEWYDDNP10 8FZ(in respect of right to lay and maintain gas pipeline stated in deed dated 19 May 1955) |  |
| 143                     | 14073 square metres of agricultural land, hedgerow and track (south of Dyserth Road, B5119) and footpath (DE 206 5)    | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right of access for maintenance)                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2      |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|--------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                        | County of Denbighshire<br>Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act<br>2008                                                                                                                                                                                      |  |
| 143 cont'd              |                                                                                            | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline stated in deed dated 19 May 1955)                                                                                                                                        |  |
| 144                     | 6563 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119)        | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right of access for maintenance)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline stated in deed dated 19 May 1955) |  |
| 145                     | 206 square metres of agricultural land (south of Dyserth Road, B5119)                      | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                  |  |
| 146                     | 4674 square metres of agricultural land, hedgerow and ditch (south of Dyserth Road, B5119) | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                  |  |
| 147                     | 149 square metres of hedgerow (south of Dyserth Road, B5119)                               | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 148                     | 21 square metres of hedgerow (south of Dyserth Road, B5119)                                | Unknown                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                              |                                                                                                                                                                                 |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                 |                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                                                  | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                  |  |
| 149                     | 55 square metres of hedgerow (south of Dyserth Road, B5119)                                                          | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 150                     | 35 square metres of agricultural land (south of Dyserth Road, B5119)                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 151                     | 123 square metres of agricultural land (south of Dyserth Road, B5119) and footpath<br>(DE 206 3)                     | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 152                     | 10 square metres of agricultural land (south of Dyserth Road, B5119)                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 153                     | 1494 square metres of agricultural land (south of Dyserth Road, B5119) and footpaths (DE 206 4, DE 206 6, DE 206 11) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                   |  |

|            | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                                    |                                                                                                                                                                                 |  |
|------------|----------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on  | Description of Land                                                                                                                                      | County of Denbighshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act                                |  |
| Land Plans | ··· [·· · · ·                                                                                                                                            | 2008                                                                                                                                                                            |  |
| 153 cont'd |                                                                                                                                                          | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)              |  |
|            |                                                                                                                                                          | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 154        | 3934 square metres of agricultural land, hedgerow, track and private road (south of Dyserth Road, B5119) and footpaths (DE 206 12, DE 206 13, DE 206 15) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                   |  |
|            |                                                                                                                                                          | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)              |  |
|            |                                                                                                                                                          | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                |  |
| 155                     | 1166 square metres of private road (south of Dyserth Road, B5119) and footpaths<br>(DE 204 7, DE 206 15a)       | Gwilym Howatson<br>Aberkinsey Farm<br>Dyserth<br>Y RHYL<br>LL18 6BL<br>(in respect of right of access)<br>The Bodrhyddan Farming Company Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB<br>(in respect of right of access)<br>Unknown |  |
| 156                     | 3178 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119) and footpath (DE 206 3)     | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                               |  |
| 157                     | 5902 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119) and footpath (DE 206 3)     | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                               |  |
| 158                     | 2128 square metres of agricultural land and shrubbery (south of Dyserth Road, B5119) and footpath (DE 206 3)    | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                 |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                      |  |
| 158 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                  |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                     |  |
| 159                     | 1535 square metres of agricultural land (south of Dyserth Road, B5119)                                          | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                     |  |
| 160                     | 84 square metres of hedgerow (south of Dyserth Road, B5119)                                                     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 SGD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                               | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                         |  |
| 160 cont'd              |                                                                                                                                                   | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                                                                                                                        |  |
| 161                     | 1552 square metres of agricultural land and shrubbery (south of Dyserth Road, B5119)                                                              | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 SGD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of right s tated in Transfer dated 01 December 2021) |  |
| 162                     | 48411 square metres of agricultural land, hedgerow and shrubbery (north of Pentre Lane and south of Dyserth Road, B5119) and footpath (DE 206 44) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                 |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                  |  |
| 163 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)              |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 163                     | 12537 square metres of agricultural land and hedgerow (north of Pentre Lane and south of Dyserth Road, B5119)   | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                   |  |
|                         |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)              |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                 |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                  |  |
| 164                     | 254 square metres of agricultural land (north of Pentre Lane)                                                   | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                   |  |
|                         |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)              |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 165                     | 1876 square metres of agricultural land and hedgerow (north of Pentre Lane)                                     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                   |  |
|                         |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain apparatus)              |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                         |  |
| 165 cont'd              |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                                                                                                                        |  |
| 166                     | 11721 square metres of agricultural land, hedgerow and pond (north of Pentre<br>Lane)                           | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CHS 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 167                     | 358 square metres of agricultural land (north of Pentre Lane)                                                   | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2 |                                                                                                                                                                                                |  |
|-------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                   | County of Denbighshire<br>Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act<br>2008                                    |  |
| 167 cont'd              |                                                                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                             |  |
|                         |                                                                                       | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                |  |
| 168                     | 245 square metres of agricultural land (north of Pentre Lane)                         | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water |  |
|                         |                                                                                       | Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                                                              |  |
|                         |                                                                                       | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                 |                                                                                                                                                                                                                         |  |
|-------------------------|---------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                    |                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land                                                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                          |  |
| 169                     | 4031 square metres of agricultural land, track and hedgerow (north of Pentre Lane)                      | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                           |  |
|                         |                                                                                                         | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House |  |
|                         |                                                                                                         | Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                              |  |
| 170                     | 776 square metres of agricultural land and hedgerow (north of Pentre Lane and south of Bryn Cwnin Road) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                           |  |
|                         |                                                                                                         | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain apparatus)                                                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                               | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                         |  |
| 170 cont'd              |                                                                                                                                   | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                                                                                                                        |  |
| 171                     | 573 square metres of agricultural land (north of Pentre Lane and south of Bryn<br>Cwnin Road)                                     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 172                     | 4649 square metres of agricultural land and hedgerow (north of Pentre Lane and south of Bryn Cwnin Road) and footpath (DE 206 44) | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                               |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                        |                                                                                                                                                                       |  |
|-------------------------|--------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                           |                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                                            | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                        |  |
| 172 cont'd              |                                                                                | Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                           |  |
|                         |                                                                                | Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced) |  |
|                         |                                                                                | Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                       |  |
| 173                     | 156 square metres of agricultural land and hedgerow (south of Bryn Cwnin Road) | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)                                              |  |
|                         |                                                                                | Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                           |  |
|                         |                                                                                | Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced) |  |
|                         |                                                                                | Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2         |                                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                        |                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                                                           | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                        |  |
| 174                     | 1251 square metres of agricultural land, track and woodland (south of Bryn Cwnin<br>Road)     | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin   |  |
|                         |                                                                                               | Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                                                      |  |
|                         |                                                                                               | Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced) |  |
|                         |                                                                                               | Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                       |  |
| 175                     | 227 square metres of agricultural land (north of Pentre Lane and south of Bryn<br>Cwnin Road) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                         |  |
|                         |                                                                                               | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                         |  |
| 175 cont'd              |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                                                                                                                        |  |
| 176                     | 43 square metres of agricultural land (north of Pentre Lane and south of Bryn Cwnin<br>Road)                    | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 177                     | 5302 square metres of agricultural land and wooded area (south of Bryn Cwnin Road) and footpath (DE 206 44)     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                   |  |
| 177 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                            |  |
| 178                     | 10 square metres of public road (south of Bryn Cwnin Road)                                                      | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 180                     | 2 square metres of footway of public road (south of Bryn Cwnin Road)                                            | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 181                     | 1137 square metres of agricultural land and hedgerow (south of Bryn Cwnin Road)                                 | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)<br>Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                |  |
| 181 cont'd              |                                                                                                                 | Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                                                                                                                                                                                                                                                               |  |
| 182                     | 1694 square metres of agricultural land (south of Bryn Cwnin Road)                                              | Centrica PLCMillstreamMaidenhead RoadWINDSORBerkshireSL4 5GD(in respect of rights stated in Deed dated 17th May 1991)Dwr Cymru CyfyngedigDwr Cymru Welsh WaterLineaFortran RoadSt. MellonsCARDIFFCF3 0LT(in respect of right to lay and maintain apparatus)Anwyl Construction Company LimitedAnwyl HouseClos Dewi SantEwloeDESIDEConwyCH5 3DT(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 183                     | 447 square metres of track (south of Bryn Cwnin Road)                                                           | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                                                                                 |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| Number on<br>Land Plans | Description of Land                                                                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| 183 cont'd              |                                                                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         |                                                                                       | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                                                                                                                                                                     |  |
| 184                     | 7765 square metres of wooded area (south of Bryn Cwnin Road)                          | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)<br>Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                        |  |
| 185                     | 77 square metres of wooded area (south of Bryn Cwnin Road)                                                      | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)                                              |  |
|                         |                                                                                                                 | Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                           |  |
|                         |                                                                                                                 | Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced) |  |
|                         |                                                                                                                 | Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                       |  |
| 186                     | 5551 square metres of agricultural land and wooded area (south of Bryn Cwnin Road) and footpath (DE 206 17)     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                         |  |
|                         |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                         |  |
| 186 cont'd              |                                                                                                                         | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                                                                                                                        |  |
| 187                     | 8968 square metres of agricultural land and hedgerow (south of Bryn Cwnin Road)<br>and footpaths (DE 206 17, DE 206 18) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 188                     | 517 square metres of agricultural land (south of Bryn Cwnin Road, north of Pentre and east of A525)                     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                          |                                                                                                                                                                                                                                                                                             |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                       |                                                                                                                                                                                                                                                                                             |  |
|                         | County of Denbighshire                                                                                           |                                                                                                                                                                                                                                                                                             |  |
| Number on<br>Land Plans | Description of Land                                                                                              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                              |  |
| 188 cont'd              |                                                                                                                  | Dwr Cymru Cyfyngedig         Dwr Cymru Welsh Water         Linea         Fortran Road         St. Mellons         CARDIFF         CF3 0LT         (in respect of right to lay and maintain apparatus)         Anwyl Construction Company Limited         Anwyl House         Clos Dewi Sant |  |
|                         |                                                                                                                  | Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                    |  |
| 189                     | 338 square metres of agricultural land and hedgerow (south of Bryn Cwnin Road, north of Pentre and east of A525) | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                       |  |
|                         |                                                                                                                  | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                      |  |
|                         |                                                                                                                  | Unknown<br>(in respect of right of drainage)                                                                                                                                                                                                                                                |  |
| 192                     | 477 square metres of agricultural land (east of A525)                                                            | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                        |  |
| 192 cont'd              |                                                                                                                 | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                                |  |
| 193                     | 704 square metres of hedgerow (east of A525)                                                                    | Paul Raymond Cook         Cwybr Bach         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2YD         (in respect of right to service media and right to enter)         Sandra Louise Cook         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2YD         Rhuddlan         RHYL         Denbighshire         LL18 2YD         (in respect of right to service media and right to enter)         Unknown         (in respect of right of drainage) |  |
| 194                     | 15513 square metres of agricultural land and hedgerow (east of A525)                                            | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                                                                                                                                                                                                                                 |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                  |  |
| 194 cont'd              |                                                                                                                 | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                          |  |
| 195                     | 2316 square metres of agricultural land (east of A547 and A525)                                                 | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage) |  |
| 196                     | 612 square metres of agricultural land (east of A547 and north of Pentre Lane) and footpath (DE 206 46)         | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire      |                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                  | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                  |  |
| 196 cont'd              |                                                                                                                      | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                          |  |
| 197                     | 21252 square metres of agricultural land and hedgerow (east of A547) footpaths<br>(DE 206 20, DE 206 46)             | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage) |  |
| 198                     | 373 square metres of agricultural land and hedgerow (north of Pentre Lane and east of A547) and footpath (DE 206 20) | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                                                                                           |  |

|                         | Awel y Môr Offs                                                                                                          | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                                                      | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                         |  |
| 198 cont'd              |                                                                                                                          | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                                 |  |
| 200                     | 2829 square metres of agricultural land and hedgerow (north of Pentre Lane and<br>east of A547) and footpath (DE 206 20) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 201                     | 97 square metres of agricultural land and hedgerow (north of Pentre Lane and east of A547)                               | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                                                                                                                                                                                                                                  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                        |                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                      |  |
|                         |                                                                                                | County of Denbighshire                                                                                                                                                                                                                                                                                                                                          |  |
| Number on<br>Land Plans | Description of Land                                                                            | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                  |  |
| 201 cont'd              |                                                                                                | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                          |  |
| 202                     | 122 square metres of agricultural land and hedgerow (north of Pentre Lane and east<br>of A547) | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage) |  |
| 203                     | 8 square metres of hedgerow (west of Pentre Lane, Rhyl)                                        | Unknown                                                                                                                                                                                                                                                                                                                                                         |  |
| 208                     | 4 square metres of hedgerow (west of Pentre Lane and east of A547)                             | Unknown                                                                                                                                                                                                                                                                                                                                                         |  |
| 209                     | 3106 square metres of agricultural land (east of A547 and north of Afon Ffyddion)              | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|--------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| Number on<br>Land Plans | Description of Land                                                      | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                    |  |
| 209 cont'd              |                                                                          | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                                                                                                                            |  |
| 210                     | 39 square metres of verge of public road (east of A547)                  | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 212                     | 751 square metres of grassed area public road (east of A547)             | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 213                     | 10 square metres of grassed area (east of A547) and footpath (DE 206 46) | Paul Raymond Cook         Cwybr Bach         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2YD         (in respect of right to service media and right to enter)         Sandra Louise Cook         Cwybr         Cwybr Bach         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2YD         (in respect of right to service media and right to enter)         UNROWN         (in respect of right of drainage) |  |
| 214                     | 17 square metres of public road (east of A547)                           | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 216                     | 1449 square metres of public road and verge (east of A525)               | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 218                     | 7 square metres of grassed and wooded area (west of A525)                | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)                                                                                                                                                                                                                                                                                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                    |                                                                                                                                                                                                        |  |
|-------------------------|----------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                       |                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                         |  |
| 218 cont'd              |                                                                            | Unknown<br>(in respect of restrictive covenants)                                                                                                                                                       |  |
| 220                     | 317 square metres of agricultural land and track (west of A525)            | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Unknown<br>(in respect of restrictive covenants) |  |
| 221                     | 245 square metres of agricultural land (west of A525)                      | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Unknown<br>(in respect of restrictive covenants) |  |
| 224                     | 1 square metre of footway (west of A525)                                   | Unknown                                                                                                                                                                                                |  |
| 225                     | 3 square metres of footway (west of A525)                                  | Unknown                                                                                                                                                                                                |  |
| 227                     | 4 square metres of footway (west of A525)                                  | Unknown                                                                                                                                                                                                |  |
| 231                     | 1 square metre of grassed area (west of A525)                              | Unknown                                                                                                                                                                                                |  |
| 232                     | 58 square metres of verge and public road (A525)                           | Unknown                                                                                                                                                                                                |  |
| 233                     | 105 square metres of public road (A525) and footpath (DE 206 42)           | Unknown                                                                                                                                                                                                |  |
| 234                     | 97 square metres of public road (A525)                                     | Unknown                                                                                                                                                                                                |  |
| 235                     | 1159 square metres of private road (west of A525) and footpath (DE 206 42) | Aaron Haldane<br>The Maisonette<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                           |  |

|                         | А                          | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|----------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|                         |                            | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
| 235 cont'd              |                            | Anthony Ashworth 14 Gwybr Cottages Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Anthony Roberts Filat 1 Cwybr House Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Christopher George Roberts 1 Cwybr Uchaf Farm Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Helen McCahll 13 Gwybr Cottages Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Helen McCahll 13 Gwybr Cottages Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Ian Towers 1 Cwybr Cottages Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Ian Towers 1 Cwybr Cottages Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Ian Towers 1 Cwybr Cottages Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Ian Towers 1 Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Ian Towers 1 Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Ian Towers 1 Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Ian Towers 1 Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Ian Towers 1 Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Ian Towers 1 Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Ian Towers 1 Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) Iac Christopher Mousley 9 Cwybr Cottages Cwybr Rhuddlan Y RYVL LLIS 27D (In respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
| 235 cont'd              |                                                                                                                 | James Keith Nixon<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)       |  |
|                         |                                                                                                                 | James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)            |  |
|                         |                                                                                                                 | John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)               |  |
|                         |                                                                                                                 | Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)             |  |
|                         |                                                                                                                 | Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|                         | Description of Land                                                                                             | 2008       Mark Nicholls       6 Cwybr Cottages       Cwybr       Rhuddian       Y RHYL       LL18 2VD       (in respect of right of access)       Martin Nixon       Flat 4       Cwybr House       Cwybr       Rhuddian       Y RHYL       LL18 2VD       (in respect of right of access)       Martin Nixon       Flat 4       Cwybr House       Cwybr       Rhuddian       Y RHYL       LL18 2VD       (in respect of right of access)       Matt Hession       2 Cwybr Uchaf Farm       Cwybr       Rhuddian       Y RHYL       LL18 2VD       (in respect of right of access)       Peter Baddeley       2 Cwybr Villa       Cwybr       Rhuddian       Y RHYL       LL18 2VD       (in respect of right of access)       Peter Baddeley       2 Cwybr Villa       Cwybr       Rhuddian       Y RHYL       LL18 2VD       (in respect of right of access)       Shau Williams |  |
|                         |                                                                                                                 | 6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 235 cont'd              |                                                                                                                 | Stephen Allen NicksonFlat 3Cwybr HouseCwybrRhuddlanY RHYLLL18 2YD(In respect of right of access)Susan DaviesS Cwybr CottagesCwybrRhuddlanY RHYLLL18 2YD(In respect of right of access)Susan DaviesS Cwybr CottagesCwybrRhuddlanY RHYLLL18 2YD(In respect of right of access)UnknownVictoria Jones2 Cwybr Uchaf FarmCwybrRhuddlanY RHYLLL18 2YD(In respect of right of access)Uitkoria Jones2 Cwybr Uchaf FarmCwybrRhuddlanY RHYLLL18 2YD(In respect of right of access)William Richard Basketfield4 Copthorne PlaceKents HillMILTON KEYNESMK7 6JP(In respect of right of access) |  |
| 236                     | 14 square metres of private road (west of A525)                                                                 | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                                                                           |  |

|                         | Awel y Môr O                                                             | ffshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|--------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                          | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Description of Land                                                      | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 237                     | 8537 square metres of agricultural land (west of A525)                   | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                                                                                           |
| 238                     | 18 square metres of private road (west of A525) and footpath (DE 206 42) | Aaron HaldaneThe MaisonetteCwybr HouseCwybrRhuddlanY RHVLLL18 2YD(in respect of right of access)Anthony Ashworth14 Cwybr CottagesCwybrRhuddlanY RHVLLL18 2YD(in respect of right of access)Anthony RobertsFlat 1Cwybr HouseCwybrCwybr HouseCwybrChuddlanY RHVLLL18 2YD(in respect of right of access)Anthony RobertsFlat 1CwybrRhuddlanY RHVLLL18 2YD(in respect of right of access)Christopher George Roberts1 Cwybr Uchaf FarmCwybrRhuddlanY RHVLLL18 2YD(in respect of right of access)Christopher George Roberts1 Cwybr Uchaf FarmCwybrRhuddlanY RHVLLL18 2YD(in respect of right of access) |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-----------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans                                                                                         | Description of Land | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 238 cont'd                                                                                                      |                     | Helen McCahill         13 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         lan Towers         11 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Jack Christopher Mousley         9 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         Kotlan         Y RHYL         LL18 2YD         Rhuddlan         Y RHYL         LL18 2YD         Khuddlan         Y RHYL         LL18 2YD |
|                                                                                                                 |                     | <ul> <li>(in respect of right of access)</li> <li>James Keith Nixon</li> <li>Flat 3</li> <li>Cwybr House</li> <li>Cwybr</li> <li>Rhuddlan</li> <li>Y RHYL</li> <li>LL18 2YD</li> <li>(in respect of right of access)</li> <li>James Keith Shaw</li> <li>1 Cwybr Villa</li> <li>Cwybr</li> <li>Rhuddlan</li> <li>Y RHYL</li> <li>LL18 2YD</li> <li>(in respect of right of access)</li> </ul>                                                                                                                                                                                  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 238 cont'd              |                                                                                                                 | James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Kristy Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Kristy Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Kristy Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access) |  |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER |                            |                                                                                                                                |  |
|---------------------------------------------------------|----------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
|                                                         | BOOK OF REFERENCE - PART 2 |                                                                                                                                |  |
|                                                         |                            | County of Denbighshire                                                                                                         |  |
| Number on<br>Land Plans                                 | Description of Land        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
|                                                         | Description of Land        |                                                                                                                                |  |
|                                                         |                            | North Cheshire Trading Estate                                                                                                  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| 238 cont'd              |                                                                                                                 | Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Susan Davies<br>S Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 6JP |  |
| 239                     | 349 square metres of private road (west of A525 and north-east of Cwybr-uchaf)<br>and footpath (DE 206 42)      | Aaron Haldane<br>The Maisonette<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |

|                         | Α                          | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|----------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|                         |                            | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| 239 cont'd              |                            | Anthony Ashworth<br>14 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Anthony Roberts<br>Filat 1<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Helen MrCahili<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Helen MrCahili<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Helen MrCahili<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Iax Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Iax Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Iax Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Iax Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Iax Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>Iax Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>IA CW Pr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>IA CW Pr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>IA CW Pr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>IA CW Pr<br>Rhuddian<br>Y RHYL<br>LLB 2YD<br>(In respect of right of access)<br>IA CW Pr<br>IA CHAR<br>IA CHAR |  |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on Description of Land                                                                                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 239 cont'd                                                                                                      | James Keith Nixon<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Shaw<br>1 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| 239 cont'd              |                                                                                                                 | Kirsty Massey         Flat 2         Cwybr House         Cwybr         Rhuddlan         Y RHVL         L118 2YD         (In respect of right of access)         Mark Nicholls         6 Cwybr Cottages         Cwybr         Rhuddlan         Y RHVL         L18 2YD         (In respect of right of access)         Marki Nicholls         Flat 4         Cwybr House         Cwybr House         Cwybr House         Cwybr House         Cwybr House         Cwybr Rhuddlan         Y RHVL         L18 2YD         (In respect of right of access)         Mattin Nixon         Flat 4         Cwybr House         Cwybr Rhuddlan         Y RHVL         L18 2YD         (In respect of right of access)         Matt Hession         2 Cwybr UchaF Farm         Cwybr         Rhuddlan         Y RHVL         L18 2YD         (In respect of right of access)         Pater Baddeley         2 Cwybr Villa         Cwybr         < |  |
|                         |                                                                                                                 | Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
| 239 cont'd              |                                                                                                                 | Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)               |  |
|                         |                                                                                                                 | Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)   |  |
|                         |                                                                                                                 | Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                 |  |
|                         |                                                                                                                 | Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)  |  |
|                         |                                                                                                                 | William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 6JP<br>(in respect of right of access)  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| 240                     | 933 square metres of private road (west of A525 and north of Cwybr-uchaf) and<br>footpaths (DE 206 24, DE 206 25, DE 206 31, DE 206 42) | Aaron Haldane         The Maisonette         Cwybr House         Cwybr Rhuddlan         Y RHYL         L118 2YD         (in respect of right of access)         Anthony Ashworth         14 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         L118 2YD         (in respect of right of access)         Anthony Ashworth         14 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         L118 2YD         (in respect of right of access)         Anthony Roberts         Flat 1         Cwybr House         Cwybr         Rhuddlan         Y RHYL         L118 2YD         (in respect of right of access)         Christopher George Roberts         1 Cwybr Uchaf Farm         Cwybr         Rhuddlan         Y RHYL         L118 2YD         (in respect of right of access)         Helen MCCahill         13 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         L118 2YD         (in respect of r |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
| 240 cont'd              |                                                                                                                 | lan Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                  |  |
|                         |                                                                                                                 | Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)     |  |
|                         |                                                                                                                 | James Keith Nixon<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)       |  |
|                         |                                                                                                                 | James Keith Shaw<br>1 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                |  |
|                         |                                                                                                                 | James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)            |  |

|                          | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|--------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans  | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| Land Plans<br>240 cont'd |                                                                                                                 | John Appleton<br>1 Cowybr Cottages<br>Cowybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(In respect of right of access)<br>Keith Jones<br>Flat 4<br>Cowybr House<br>Cowybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(In respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cowybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(In respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cowybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(In respect of right of access)<br>Mark Nicholls<br>6 Cowybr Cottages<br>Cowybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(In respect of right of access)<br>Mark Nicholls<br>6 Cowybr Cottages<br>Cowybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(In respect of right of access)<br>Mark Nicholls<br>6 Cowybr Cottages<br>Cowybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(In respect of right of access)<br>Mark Nicholls<br>6 Cowybr House<br>Cowybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(In respect of right of access)<br>Mark Nicholls<br>7 RHYL<br>7 RHYL |  |
|                          |                                                                                                                 | LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER |                                                                                                                                                                                |  |
|-------------------------|---------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                 |  |
| 240 cont'd              |                                                         | Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Peter Baddeley                                             |  |
|                         |                                                         | 2 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                    |  |
|                         |                                                         | Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                               |  |
|                         |                                                         | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights granted by Deed of grant dated 20 December 2017) |  |
|                         |                                                         | Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                   |  |
|                         |                                                         | Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                 |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                            |                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                  |  |
| 240 cont'd              |                                                                                                                                            | Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 6JP<br>(in respect of right of access)                                                                                                                  |  |
| 241                     | 1126 square metres of private road, hardstanding and agricultural land (east of<br>Cwybr-uchaf and north of A525) and footpath (DE 206 24) | Aaron HaldaneThe MaisonetteCwybr HouseCwybrRhuddlanY RHYLLL18 2YD(in respect of right of access)Anthony Ashworth14 Cwybr CottagesCwybrRhuddlanY RHYLLL18 2YD(in respect of right of access)Anthony RobertsFlat 1Cwybr HouseCwybrRhuddlanY RHYLLL18 2YD(in respect of right of access)Anthony RobertsFlat 1Cwybr HouseCwybrRhuddlanY RHYLLL18 2YD(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| 241 cont'd              |                                                                                                                 | Centrica PLC<br>Milistream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(In respect of rights stated in Deed dated 26th February 1992)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(In respect of right of access)<br>Helen McCahill<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(In respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(In respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(In respect of right of access)<br>Ian Towers<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(In respect of right of access)<br>Ian Zowers<br>Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(In respect of right of access)<br>Ian Zowers<br>Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(In respect of right of access)<br>Ian Zowers<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(In respect of right of access) |  |

|                         | Awel y Môr O           | ffshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                        | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|                         | County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| Number on<br>Land Plans | Description of Land    | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| 241 cont'd              |                        | James Keith Nixon<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Hat 2<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| 241 cont'd              |                                                                                                                 | Kirsty Massey.         Flat 2         Cwybr House         Cwybr         Rhuddlan         Y RHYL         LL18 2VD         (in respect of right of access)         Mark Nicholls         6 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 2VD         (in respect of right of access)         Mark Nicholls         6 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 2VD         (in respect of right of access)         Martin Nixon         Flat 4         Cwybr         Rhuddlan         Y RHYL         LL18 2VD         (in respect of right of access)         Matt Hession         2 Cwybr Uchaf Farm         Cwybr         Rhuddlan         Y RHYL         LL18 2VD         (in respect of right of access)         Matt Hession         2 Cwybr Uchaf Farm         Cwybr         Rhuddlan         Y RHYL         LL18 2VD         (in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| 241 cont'd              |                                                                                                                 | SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to lay and maintain stated in deed dated 20 December 2017)         Stephen Allen Nickson         Flat 3         Cwybr House         Cwybr         Rhuddlan         Y RHYL         LL18 2VD         (in respect of right of access)         Susan Davies         S Cwybr         S Cwybr         Rhuddlan         Y RHYL         L118 2VD         (in respect of right of access)         Susan Davies         S Cwybr         Rhuddlan         Y RHYL         L118 2VD         (in respect of right of access)         Unknown         (in respect of right of access)         Unknown |  |
|                         |                                                                                                                 | <ul> <li>(in respect of right of access)</li> <li>Victoria Jones</li> <li>2 Cwybr Uchaf Farm</li> <li>Cwybr</li> <li>Rhuddlan</li> <li>Y RHYL</li> <li>LL18 2YD</li> <li>(in respect of right of access)</li> <li>William Richard Basketfield</li> <li>4 Copthorne Place</li> <li>Kents Hill</li> <li>MILTON KEYNES</li> <li>MK7 6JP</li> <li>(in respect of right of access)</li> </ul>                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offsl                                                                                                          | nore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| Number on<br>Land Plans | Description of Land                                                                                                       | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| 242                     | 43453 square metres of agricultural land (north of A525 and south of Cwybr-uchaf)<br>and footpaths (DE 206 23, DE 206 24) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain stated in deed dated 20 December 2017)                                                                                                                                                                                                                                                                                     |  |
|                         |                                                                                                                           | Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 243                     | 869 square metres of private road and track (north of A525)                                                               | Aaron Haldane         The Maisonette         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Anthony Ashworth         14 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Anthony Ashworth         14 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Anthony Roberts         Flat 1         Cwybr House         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access) |  |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|-----------------------------------------------------------------------------------------------------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans                                                                                         | Description of Land | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| 243 cont'd                                                                                                      |                     | Centrica PLC<br>Milistream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2YD<br>(in respect of right of access)<br>Helen McCahill<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Iak Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Iak Christopher Mousley<br>9 Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| 243 cont'd              |                                                                                                                 | James Keith Nixon<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr Mouse<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr Mouse<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| 243 cont'd              |                                                                                                                 | Kirsty Massey       Flat 2       Cwybr       Rhuddlan       Y RH/L       LL18 2YD       (in respect of right of access)       Mark Nicholls       6 Cwybr Cottages       Cwybr       Rhuddlan       Y RHYL       LL18 2YD       (in respect of right of access)       Mark Nicholls       6 Cwybr Cottages       Cwybr       Rhuddlan       Y RHYL       LL18 2YD       (in respect of right of access)       Marti Nixon       Flat 4       Cwybr       Cwybr       Rhuddlan       Y RHYL       LL18 2YD       (in respect of right of access)       Matt Hession       2 Cwybr Uchaf Farm       Cwybr       Rhuddlan       Y RHYL       LL18 2YD       (in respect of right of access)       Matt Hession       2 Cwybr Uchaf Farm       Cwybr       Rhuddlan       Y RHYL       LL18 2YD       (in respect of right of access)       Shaun Williams       6 Cwybr Cottages       Cwybr       Rhuddlan       Y RHYL       LL18 2YD       LL18 2YD |  |
|                         |                                                                                                                 | (in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|---------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| 243 cont'd              |                                                                                       | SP Manweb PLC<br>SP menton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain stated in deed dated 20 December 2017)<br>Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2YD<br>(in respect of right of access)<br>Susan Davies<br>S Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access)<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr |  |
|                         |                                                                                       | Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         |                                                                                       | William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 6JP<br>(in groups of sight of agons)                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|                         |                                                                                       | (in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|-------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| Number on<br>Land Plans | Description of Land                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                               |  |
| 244                     | 731 square metres of track(west of Cwybr-uchaf and north-west of A525) and footpath (DE 206 31) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                       |  |
| 245                     | 5 square metres of agricultural land (south-west of Cwybr-uchaf and north-west of A525)         | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| 246                     | 535 square metres of agricultural land (south-west of Cwybr-uchaf and north-west<br>of A525)    | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of drainage)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of right of drainage)<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of drainage)<br>Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>(in respect of right of drainage) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire    |                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                    |  |
| 246 cont'd              |                                                                                                                    | Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |
| 247                     | 14555 square metres of agricultural land and watercourse (Afon Ffyddion, north of<br>River Clwyd and west of A525) | Arthur Horron Wynne DaviesTrescawen FarmLLANGEFNIGwyneddLL77 7UR(in respect of right of drainage)Hugh Henry Wynne DaviesMorfa CwybrMorfa CwybrRHYLDenbighshireLL18 2YB(in respect of right of drainage)Mary WilliamsCefn CestyllRhydlanfairBETWS-Y-COEDLL24 0LY(in respect of right of drainage)                                                                                                                  |  |

|                         | Awel y M                                                         | Iôr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                  | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Description of Land                                              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 247 cont'd              |                                                                  | Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>(in respect of right of drainage)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 19th June 1952) |
| 248                     | 7057 square metres of agricultural land (north of Afon Ffyddion) | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of drainage)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                                                                                                                                                          |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                          |                                                                                                                                                                                                                      |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                                         |                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                            | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                       |  |
| 248 cont'd              |                                                                                                                | Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of drainage)                                                                                                        |  |
|                         |                                                                                                                | Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>(in respect of right of drainage)                                                                                                     |  |
|                         |                                                                                                                | Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>(in respect of right of drainage)                                                                                                        |  |
|                         |                                                                                                                | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017) |  |
|                         |                                                                                                                | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 19th June 1952)  |  |
| 249                     | 5304 square metres of agricultural land and watercourse (Afon Ffyddion, north of River Clwyd and west of A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)                                                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 249 cont'd              |                                                                                                                                                         | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHVL<br>Denbighshire<br>LL18 2VD<br>(in respect of right of drainage)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2VD<br>(in respect of right of drainage)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain pipeline) |  |
| 250                     | 21814 square metres of watercourse (Afon Ffyddion), hedgerow, agricultural land, footway and river bank (north of River Clwyd) and footpath (DE 206 29) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                                                                                                                                                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 250 cont'd              |                                                                                                                 | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>Wales and West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain pipeline) |  |
| 251                     | 264 square metres of agricultural land (north of River Clwyd and west of A525)                                  | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                                                                                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                               |  |
| 251 cont'd              |                                                                                                                 | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>L118 2YD<br>(in respect of right of drainage)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>L116 4RD<br>(in respect of right of drainage)<br>Sandra Archdale<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain pipeline) |  |
| 252                     | 726 square metres of agricultural land, hedgerow and track (north of River Clwyd<br>and west of A525)           | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                                                                                                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2 |                                                                                                                                                                      |  |
|-------------------------|---------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                       | County of Denbighshire                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                       |  |
| 252 cont'd              |                                                                                       | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)                                            |  |
|                         |                                                                                       | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)                                               |  |
|                         |                                                                                       | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)                                          |  |
|                         |                                                                                       | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain pipeline) |  |
| 253                     | 1675 square metres of track (north of River Clwyd)                                    | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of way)                                                     |  |
|                         |                                                                                       | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of right of way)                                            |  |

|                         | Awely                                              | Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|----------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|                         | County of Denbighshire                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| 253 conťd               |                                                    | Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of way)<br>Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>(in respect of right of way)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 2YB<br>(in respect of right of way)                                                                                                                                                                                                                                                                                                                                  |  |
| 254 77 squar            | re metres of track (north of A525 and River Clwyd) | Dwr Cymru Cyfyngedig         Dwr Cymru Welsh Water         Linea         Fortran Road         St. Mellons         CARDIFF         CF3 0LT         (in respect of right of way)         Helen Owen Proffitt         Fferm         Bodelwyddan Road         RHYL         Denbighshire         LL18 SUH         (in respect of right of drainage)         Janet Johnson         Cwybr Fawr         Cwybr         Rhuddlan         RHYL         Denbighshire         L18 SUH         (in respect of right of drainage)         Janet Johnson         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 27D         (in respect of right of drainage) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                 |                                                                                                                                                                      |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                       |  |
| 254 cont'd              |                                                                                                                                 | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)                                               |  |
|                         |                                                                                                                                 | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)                                          |  |
|                         |                                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain pipeline) |  |
| 255                     | 52 square metres of track and grassed area (north of River Clwyd and south-east of A525) (excluding all interests of the Crown) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)                                |  |
|                         |                                                                                                                                 | Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)                                            |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 255 cont'd              |                                                                                                                 | Rachel Georgina Hughes         Brook House Farm         DENBIGH         Denbighshire         LL16 4RD         (in respect of right of drainage)         Sandra Archdale         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2YD         (in respect of right of drainage)         The King's Most Excellent Majesty In Right Of His Crown         1 St. James's Market         LONDON         SW1Y 4AH         (in respect of annual Crown rent)         Wales and West Utilities Limited         Wales and West Utilities Limited         Wales and West House         Spooner Close         Coedkernew         CASNEWYDD         NP10 8FZ         (in respect of right to lay and maintain pipeline) |  |
| 256                     | 290 square metres of track (north of River Clwyd and south-east of A525)                                        | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                            |                                                                                                                                                                                                                                             |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                                         |                                                                                                                                                                                                                                             |  |
|                         | County of Denbighshire                                                                                                             |                                                                                                                                                                                                                                             |  |
| Number on<br>Land Plans | Description of Land                                                                                                                | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                              |  |
| 256 cont'd              |                                                                                                                                    | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of right of access)<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of access) |  |
|                         |                                                                                                                                    | Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>(in respect of right of access)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries                                                                      |  |
|                         |                                                                                                                                    | RHYL<br>LL18 2YB<br>(in respect of right of access)<br>Unknown                                                                                                                                                                              |  |
| 257                     | 4323 square metres of watercourse and river (River Clwyd), foreshore, bed and banks thereof (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights stated in Lease dated 2nd June 2015)<br>Rhyl & Conwy Wildfowlers Club<br>Ty Gwyn<br>Cwm<br>Rhuddlan<br>Y RHYL |  |
|                         |                                                                                                                                    | LL18 5RY<br>(in respect of sporting rights)                                                                                                                                                                                                 |  |
| 258                     | 340 square metres of grassed area and river banks (south of River Clwyd) (excluding all interests of the Crown)                    | Unknown                                                                                                                                                                                                                                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                               |                                                                                                                                                                                                                                                                                                                           |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                 | County of Denbighshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                     |  |
| 259                     | 4924 square metres of grassed area and river banks (south of River Clwyd)<br>(excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                                                   |  |
| 260                     | 1516 square metres of grassed area and track (south of River Clwyd) and footpath<br>(DE 201 12)                     | The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)<br>Unknown                                                                                                                                                                             |  |
| 261                     | 13531 square metres of agricultural land and hedgerow (south of River Clwyd)                                        | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of access for maintenance)                                                                                                                                               |  |
| 262                     | 52 square metres of track and grassed area (south of River Clwyd) and footpath<br>(DE 201 12)                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of rights granted by Deed dated 20 June 1979)<br>The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 OTP<br>(in respect of right of access)<br>Unknown |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire           |                                                                                                                                                                                                               |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                       | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                |  |
| 263                     | 537 square metres of track (south of River Clwyd) and footpath (DE 201 12)                                                | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)                                                                |  |
|                         |                                                                                                                           | The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)<br>Unknown                                                                 |  |
|                         |                                                                                                                           | Unknown                                                                                                                                                                                                       |  |
| 264                     | 158 square metres of agricultural land (south of River Clwyd)                                                             | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right of access)                                                                |  |
| 265                     | 729 square metres of agricultural land (Hafod-Ilwyn Farm north of Abergele Road, A457 and south of River Clwyd)           | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of access for maintenance stated in Deed dated 20 June 1979) |  |
| 266                     | 40 square metres of agricultural land and hedgerow (Hafod-Ilwyn Farm north of Abergele Road, A457 and east of Marsh Road) | Unknown                                                                                                                                                                                                       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire            |                                                                                                                                                                                                                                                                                            |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                             |  |
| 267                     | 72 square metres of hedgerow (Hafod-Ilwyn Farm north of Abergele Road, A457<br>and east of Marsh Road)                     | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Frances Marion Davies<br>Bryn Y Mor<br>Dimlands Road<br>LLANTWIT MAJOR<br>South Glamorgan<br>CF61 1YX<br>(in respect of right of access) |  |
| 268                     | 378 square metres of agricultural land and hedgerow (Hafod-Ilwyn Farm north of Abergele Road, A457 and east of Marsh Road) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of access for maintenance)                                                                                                                |  |
| 269                     | 681 square metres of agricultural land and hedgerow (north of Abergele Road, A457)                                         | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Frances Marion Davies<br>Bryn Y Mor<br>Dimlands Road<br>LLANTWIT MAJOR<br>South Glamorgan<br>CF61 1YX<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                           |  |
| 270                     | 18 square metres of hedgerow (Hafod-Ilwyn Farm north of Abergele Road, A457)                                    | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access)                                                                                                                  |  |
|                         |                                                                                                                 | Unknown<br>(in respect of rights stated in Conveyance dated 14th July 1924)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of rights stated in Deed dated 20th June 1979) |  |
| 271                     | 483 square metres of agricultural land (Hafod-Ilwyn Farm north of Abergele Road,<br>A457)                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of access for maintenance stated in Deed dated 20 June 1979)                                            |  |
| 272                     | 10631 square metres of agricultural land (north of Abergele Road, A457 and east of Gipsy Lane)                  | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of access for maintenance)                                                                              |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                      |                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                         |                                                                                                                                                                                                                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                                                          | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                |  |
| 273                     | 35 square metres of agricultural land (north of Abergele Road, A457 and east of Gipsy Lane)                  | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of access for maintenance)                                                                                                                                                                   |  |
| 274                     | 64 square metres of agricultural land and hedgerows (north of Abergele Road, A457<br>and east of Gipsy Lane) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only) |  |
| 275                     | 5910 square metres of agricultural land (north of Abergele Road, A457 and east of Gipsy Lane)                | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only)                                                                                                                                                   |  |
| 276                     | 2256 square metres of agricultural land (north of Abergele Road, A457 and east of Gipsy Lane)                | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right of access)                                                                                                                                                                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                      |                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                   |                                                                                                                                                                                                                                                                                                                                               |  |
|                         | County of Denbighshire                                                                                       |                                                                                                                                                                                                                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                                                          | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                |  |
| 276 cont'd              |                                                                                                              | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only)                                                                                                                                                   |  |
| 277                     | 358 square metres of agricultural land and hedgerow (north of Abergele Road, A457<br>and east of Gipsy Lane) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only) |  |
| 278                     | 620 square metres of agricultural land (north of Abergele Road, A457 and east of Gipsy Lane)                 | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only)                                                                                                                                                   |  |
| 279                     | 27 square metres of hedgerow (east of Gipsy Lane and north of Abergele Road, A45)                            | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right of access)                                                                                                                                                                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|                         | County of Denbighshire                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                         | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| 279 cont'd              |                                                                             | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only)                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 280                     | 562 square metres of private road (Gipsy Lane north of Abergele Road, A457) | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 ODH<br>(in respect of right of access)<br>Police & Crime Commissioner for North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWYN BAY<br>COUWYN BAY<br>COUWYN BAY<br>CONWY<br>LL29 SAW<br>(in respect of right of access)<br>R G B Webster & Sons Limited<br>Banks House<br>1 Paradise Street<br>RHYL<br>Denbighshire<br>LL18 3LW<br>(in respect of right of access)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddan<br>RHYL<br>Denbighshire<br>LL15 UH<br>(in respect of right of access) |  |
| 281                     | 51 square metres of public road (Gipsy Lane north of Abergele Road, A457)   | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                    |                                                                                                                                                                               |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                       |                                                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                                                                        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                |  |
| 282                     | 19597 square metres of agricultural land (north of Abergele Road, A457 and east of Gipsy Lane)                             | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH<br>(in respect of right to pass and right to lay and maintain a water pipeline) |  |
| 283                     | 302 square metres of verge of public road (south of Abergele Road, A547)                                                   | Unknown                                                                                                                                                                       |  |
| 284                     | 400 square metres of verge of public road (north of Abergele Road, A547)                                                   | Unknown                                                                                                                                                                       |  |
| 285                     | 7031 square metres of agricultural land (Hafod-Ilwyn Farm north of Abergele Road,<br>A457 and east of Gipsy Lane)          | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 ODH<br>(in respect of right to pass and right to lay and maintain a water pipeline) |  |
| 286                     | 644 square metres of agricultural land and hedgerow (Hafod-Ilwyn Farm north of Abergele Road, A457 and east of Gipsy Lane) | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 ODH<br>(in respect of right to pass and right to lay and maintain a water pipeline) |  |
| 287                     | 120 square metres of track (Hafod-Ilwyn Farm north of Abergele Road, A457)                                                 | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of access)                                      |  |

|                         | Awel v Môr Offs                                                                                               | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                       |
|-------------------------|---------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                               | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                     |
|                         |                                                                                                               | County of Denbighshire                                                                                                                                                                                                                                                         |
| Number on<br>Land Plans | Description of Land                                                                                           | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                 |
| 287 cont'd              |                                                                                                               | The Executor of the Estate of the Late Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH<br>(in respect of right of access)<br>Unknown                                                                                               |
| 288                     | 6 square metres of public road (Hafod-Ilwyn Farm north of Abergele Road, A457)                                | Unknown                                                                                                                                                                                                                                                                        |
| 289                     | 208 square metres of verge of public road (north of Abergele Road, A457)                                      | Unknown                                                                                                                                                                                                                                                                        |
| 290                     | 172 square metres of verge of public road (south of Abergele Road, A457)                                      | Unknown                                                                                                                                                                                                                                                                        |
| 291                     | 904 square metres of public road and verges (Abergele Road, A457)                                             | Unknown                                                                                                                                                                                                                                                                        |
| 293                     | 19992 square metres of agricultural land and hedgerow (south of Abergele Road, A457) and footpath (DE 206 40) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain electric lines and right to fell trees that would obstruct the electric lines as contained in a<br>Deed dated 21st July 1994) |
| 298                     | 12 square metres of verge of public road (north of Bodelwyddan Road)                                          | Unknown                                                                                                                                                                                                                                                                        |
| 300                     | 18 square metres of verge of public road (north of Bodelwyddan Road)                                          | Unknown                                                                                                                                                                                                                                                                        |
| 302                     | 42 square metres of agricultural land and hedgerow (north of Bodelwyddan Road)<br>and footpath (DE 206 4)     | Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right to lay and maintain services)                                                                                                              |
| 303                     | 174 square metres of agricultural land and hedgerows (north of Bodelwyddan Road)                              | Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right to lay and maintain services)                                                                                                              |

|                         | Awel y Môr Offsl                                                                                            | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                        |                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                                         | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                         |  |
| 304                     | 231 square metres of agricultural land and hedgerow (north of Bodelwyddan Road)<br>and footpath (DE 206 4)  | Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right to lay and maintain services)                                                                                                                                                                                                      |  |
| 306                     | 472 square metres of private road (north of Bodelwyddan Road)                                               | Rhyl Town Council         Town Clerk         Civic Offices         Wellington Road         RHYL         LL18 1LE         (in respect of right to enter to lay and maintain water pipes)         The Executor of the Estate of the Late George Edward Brookes         Ystrad         PENBIGH         Denbighshire         LL16 4RL         (in respect of right of way) |  |
| 309                     | 9496 square metres of agricultural land and hedgerow (north of Bodelwyddan<br>Road) and footpath (DE 206 4) | Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right to lay and maintain services)                                                                                                                                                                                                      |  |
| 310                     | 862 square metres of verge of public road (south of Bodelwyddan Road)                                       | Unknown                                                                                                                                                                                                                                                                                                                                                                |  |
| 313                     | 102 square metres of hedgerow and verge of public road (north of Bodelwyddan Road)                          | Unknown                                                                                                                                                                                                                                                                                                                                                                |  |
| 316                     | 14 square metres of verge of public road (north of Bodelwyddan Road)                                        | Unknown                                                                                                                                                                                                                                                                                                                                                                |  |
| 317                     | 523 square metres of agricultural land, hedgerows, private road and verges (north of Bodelwyddan Road)      | The Executor of the Estate of the Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of right of way)                                                                                                                                                                                                         |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                           |  |
| 318                     | 5 square metres of private road and verges (north of Bodelwyddan Road)                                          | The Executor of the Estate of the Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of right of access)<br>Unknown                                                                                                                                                                                             |  |
| 319                     | 9 square metres of private road and verges (north of Bodelwyddan Road)                                          | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHVL<br>Denbighshire<br>LL18 SUH<br>(in respect of right of access)<br>Janet Johnson<br>Cwybr<br>Rhuddlan<br>RHVL<br>Denbighshire<br>LL18 ZYD<br>(in respect of right of access)<br>Rachel Georgina Hughes<br>Brock House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                 |  |
| 319 cont'd              |                                                                                                                 | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of access)<br>The Executor of the Estate of the Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of right of access) |  |
| 322                     | 521 square metres of public road and hedgerow (part of Bodelwyddan Road)                                        | Unknown                                                                                                                                                                                                                                                                                        |  |
| 323                     | 67 square metres of public road (part of Bodelwyddan Road)                                                      | Unknown                                                                                                                                                                                                                                                                                        |  |
| 325                     | 124 square metres of verge of public road (north of Bodelwyddan Road)                                           | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of way)                                                                                                                                                          |  |
| 326                     | 4 square metres of public road (part of Bodelwyddan Road)                                                       | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of way)                                                                                                                                                          |  |
| 327                     | 207 square metres of verge of public road (north of Bodelwyddan Road)                                           | Unknown                                                                                                                                                                                                                                                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                 |                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                             |  |
| 328                     | 1728 square metres of agricultural land and hedgerow (north of Bodelwyddan<br>Road)                             | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)                                                    |  |
|                         |                                                                                                                 | Gwynt Y Mor Offshore Wind Farm Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SN5 6PB<br>(in respect of right to an easement as contained in a Deed of Grant dated 8th September 2010) |  |
|                         |                                                                                                                 | Unknown<br>(in respect of right of drainage)                                                                                                                                                                               |  |
| 329                     | 272 square metres of verge of public road (north of Bodelwyddan Road)                                           | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of way)                                                                                      |  |
| 330                     | 4 square metres of public road (part of Nant-y-Faenol Road, south of Bodelwyddan Road)                          | Unknown                                                                                                                                                                                                                    |  |
| 331                     | 1305 square metres of agricultural land and hedgerows (west of Nant-y-Faenol Road)                              | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Unknown<br>(in repect of right of way)                                   |  |
| 332                     | 86 square metres of hedgerow and track (west of Nant-y-Faenol Road)                                             | Unknown                                                                                                                                                                                                                    |  |

|                         | Awel y Môr Offsh                                                                      | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                       | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Description of Land                                                                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 333                     | 154 square metres of hedgerow (north and east of Nant-y-Faenol Road)                  | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 337                     | 19 square metres of verge of public road (north of Nant-y-Faenol Road)                | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 338                     | 37 square metres of agricultural land and hedgerow (north of Nant-y-Faenol Road)      | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 340                     | 149 square metres of agricultural land and hedgerow (south of Nant-y-Faenol Road)     | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 341                     | 1557 square metres of agricultural land and hedgerow (south of Nant-y-Faenol<br>Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes) |
| 342                     | 420 square metres of public road and verges (Nant-y-Faenol Road)                      | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 343                     | 2 square metres of verge of public road (west of Nant-y-Faenol Road)                  | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 344                     | 23 square metres of hedgerow and verge of public road (south of Nant-y-Faenol Road)   | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 345                     | 17 square metres of verge of public road (south of Nant-y-Faenol Road)                | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                        |  |
| 346                     | 103 square metres of agricultural land (south of Nant-y-Faenol Road)                                            | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)                                                                               |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)                                                                                              |  |
|                         |                                                                                                                 | The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes)                                             |  |
| 347                     | 15477 square metres of agricultural land (west of Little Pengwern, south of Nant-y-Faenol Road)                 | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON |  |
|                         |                                                                                                                 | Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)                                                                                                                                                                            |  |

| <b></b>                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         |                                                                                                                          | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                                                                                      | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 347 cont'd              |                                                                                                                          | The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes)                                                                                                                                                                                                                                                                                                                                        |  |
| 348                     | 2398 square metres of agricultural land and hedgerows (west of Little Pengwern,<br>south and west of Nant-y-Faenol Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes) |  |
| 349                     | 2525 square metres of woodland (west of Little Pengwern and south-west of Nant-<br>y-Faenol Road)                        | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 350                     | 286 square metres of woodland (west of Little Pengwern, south-west of Nant-y-<br>Faenol Road)                            | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 351                     | 11635 square metres of agricultural land and hedgerows (west of Little Pengwern and south-west of Nant-y-Faenol Road)    | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)                                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                             |  |
| 351 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                             |  |
|                         |                                                                                                                 | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958) |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                              |  |
|                         |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)     |  |
| 352                     | 5480 square metres of agricultural land (west of Little Pengwern and Nant-y-Faenol Road)                        | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)                                                    |  |

| <b></b>                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 352 cont'd              |                                                                                                       | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes)                                                                                                                                                                            |  |
| 353                     | 22471 square metres of agricultural land (west of Little Pengwern and Nant-y-Faenol Road)             | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WCIV 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes) |  |
| 354                     | 2603 square metres of agricultural land (south of Little Pengwern and west of Nant-<br>y-Faenol Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)                                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire              |                                                                                                                                                                                                                                   |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                          | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                    |  |
| 354 cont'd              |                                                                                                                              | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>The Executor of the Estate of the Late Edwin Richard Jones            |  |
|                         |                                                                                                                              | Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes)                                                                                       |  |
| 355                     | 11 square metres of public road (Nant-y-Faenol Road)                                                                         | Unknown                                                                                                                                                                                                                           |  |
| 357                     | 236 square metres of agricultural land and hedgerow (north of North Wales<br>Expressway, A55 and west of Nant-y-Faenol Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea |  |
|                         |                                                                                                                              | Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                              |  |
|                         |                                                                                                                              | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                      |                                                                                                                                                                                                                            |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                                                                                   |                                                                                                                                                                                                                            |  |
|                         | County of Denbighshire                                                                                                                                                       |                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                                          | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                             |  |
| 357 cont'd              |                                                                                                                                                                              | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)     |  |
| 358                     | 1504 square metres of track (north of Tyddyn Isaf Farm and North Wales Expressway, A55)                                                                                      | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)                                                    |  |
|                         |                                                                                                                                                                              | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                             |  |
|                         |                                                                                                                                                                              | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958) |  |
|                         |                                                                                                                                                                              | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)     |  |
| 359                     | 35572 square metres of agricultural land, woodland and hedgerows (north-east of<br>Tyddyn Isaf Farm, north of North Wales Expressway, A55 and west of Nant-y-Faenol<br>Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                       | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 359 cont'd              |                                                                                                                                           | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline as contained in a Deed dated 5th September 1967)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957) |  |
| 360                     | 19135 square metres of agricultural land (north of North Wales Expressway, A55<br>and west of Nant-y-Faenol Road) and footpath (DE 201 7) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline as contained in a Deed dated 5th September 1967)                                                                                                                                                                                                                                                                              |  |

|                         | Awel v Môr Of                                                                                                                | ffshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                              | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|                         | County of Denbighshire                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land                                                                                                          | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 360 cont'd              |                                                                                                                              | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)                                                                                                    |  |
| 361                     | 828 square metres of agricultural land and hedgerow (north of North Wales<br>Expressway, A55 and west of Nant-y-Faenol Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958) |  |

|                         | Awel y Môr Offsl                                                                                                                          | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                   |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                      |                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                                                       | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                             |  |
| 361 cont'd              |                                                                                                                                           | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)     |  |
| 362                     | 559 square metres of agricultural land and hedgerow (north of North Wales Expressway, A55 and west of Nant-y-Faenol Road)                 | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)                                                    |  |
|                         |                                                                                                                                           | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                             |  |
|                         |                                                                                                                                           | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958) |  |
|                         |                                                                                                                                           | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)     |  |
| 363                     | 77 square metres of hedgerow (north of North Wales Expressway, A55 and west of Nant-y-Faenol Road) (excluding all interests of the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)                                           |  |

|            | Awel y Môr Offsh                                                                                                                              | nore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|------------|-----------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on  | Description of Land                                                                                                                           | County of Denbighshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| Land Plans | Description of Land                                                                                                                           | 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 363 cont'd |                                                                                                                                               | The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |
| 364        | 151 square metres of hedgerow (north of North Wales Expressway, A55 and west of<br>Nant-y-Faenol Road) (excluding all interests of the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)                                                                                                                                                                                                                                                                    |

|                         | Awel v Môr Offst                                                                                                                                        | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                                                         | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|                         | County of Denbighshire                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 364 cont'd              |                                                                                                                                                         | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991)                                                                                                                                                                                                                                                                      |  |
| 365                     | 2597 square metres of agricultural land (north of North Wales Expressway, A55 and<br>west of Nant-y-Faenol Road) (excluding all interests of the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 DDX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 SUL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958) |  |

|                         | Awel y Môr Offs                                                                                      | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                      | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                                                                                                      | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Description of Land                                                                                  | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 365 cont'd              |                                                                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)<br>The Secretary Of State For Wales<br>William Margare Hauge                                                                                                                                                                                                                                                                                                                                                                                                                            |
|                         |                                                                                                      | William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 366                     | 16 square metres of agricultural land (north of Tyddyn Isaf Farm and North Wales<br>Expressway, A55) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)<br>Wales and West Holise<br>Spooner Close<br>Coedkernew<br>CASNEWWDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans | Description of Land                                                                                                                                                                | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 367                     | 2 square metres of agricultural land and hedgerows (north-east of Tyddyn Isaf Farm and north of North Wales Expressway, A55)                                                       | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 368                     | 393 square metres of agricultural land (north-east of Tyddyn Isaf Farm and north of<br>North Wales Expressway, A55)                                                                | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline as contained in a Deed dated 5th September 1967)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)<br>Wales and West Utilities Limited<br>Wales & West House |
|                         |                                                                                                                                                                                    | Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 369                     | 10 square metres of agricultural land (east of Tyddyn Isaf Farm, north of North<br>Wales Expressway, A55 and west of Nant-y-Faenol Road) (excluding all interests of<br>the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |

| <b> </b>                | Awel y Môr Offsh                                                                                                                                                                    | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                                                                                                     | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| Number on<br>Land Plans | Description of Land                                                                                                                                                                 | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| 369 cont'd              |                                                                                                                                                                                     | The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UV<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |
| 370                     | 27217 square metres of agricultural land (north of North Wales Expressway, A55<br>and west of Nant-y-Faenol Road) and footpath (DE 201 7) (excluding all interests of<br>the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)                                                                                                                                                                                                                                                                    |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans | Description of Land                                                                                                                                                                                         | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 370 cont'd              |                                                                                                                                                                                                             | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |
| 371                     | 830 square metres of agricultural land (east of Tyddyn Isaf Farm, north of North<br>Wales Expressway, A55 and west of Nant-y-Faenol Road) and footpath (DE 201 7)<br>(excluding all interests of the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)                                                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans | Description of Land                                                                                                                                                    | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 371 cont'd              |                                                                                                                                                                        | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |
| 372                     | 1429 square metres of agricultural land, tracks and hardstanding (at Tyddyn Isaf<br>Farm, north of North Wales Expressway, A55) (excluding all interests of the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)                                                                                                                                                                                       |

|                         | Awel v Môr Offs                                                                                                                                                                   | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                      |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                                                                                                   | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                    |
|                         |                                                                                                                                                                                   | County of Denbighshire                                                                                                                                                                                                                                                                                        |
| Number on<br>Land Plans | Description of Land                                                                                                                                                               | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                |
| 372 cont'd              |                                                                                                                                                                                   | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>SP Manweb PLC                                                                          |
|                         |                                                                                                                                                                                   | 3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)                                                                                               |
|                         |                                                                                                                                                                                   | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991)                                                                                  |
| 373                     | 786 square metres of private road (north of North Wales Expressway, A55)<br>(excluding all interests of the Crown)                                                                | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991)                                                                                                                               |
| 374                     | 10879 square metres of agricultural land (north of North Wales Expressway, A55<br>and east of Tyddyn Isaf Farm) and footpath (DE 201 7) (excluding all interests of<br>the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)                                                                                                                              |
|                         |                                                                                                                                                                                   | The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans | Description of Land                                                                                                                              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 374 cont'd              |                                                                                                                                                  | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |
| 375                     | 56 square metres of agricultural land (north of North Wales Expressway, A55 and east of Tyddyn Isaf Farm) (excluding all interests of the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 0DX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)                                                                                                                                                                                       |

|                                                                                                                                                                                  | ore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|                                                                                                                                                                                  | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| Description of Land                                                                                                                                                              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|                                                                                                                                                                                  | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991)                                                                                                                                                 |
| 6642 square metres of agricultural land (north of North Wales Expressway, A55<br>nd west of Nant-y-Faenol Road) and footpath (DE 201 7) (excluding all interests of<br>he Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire |
| n                                                                                                                                                                                | 642 square metres of agricultural land (north of North Wales Expressway, A55<br>d west of Nant-y-Faenol Road) and footpath (DE 201 7) (excluding all interests of                                                                                                                                                                                                                                                                                                                                                                                                                                   |

| l                       | Awel y Môr Offst                                                                                                                                     | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         | Awayinoi oloi                                                                                                                                        | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                         |                                                                                                                                                      | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number on<br>Land Plans | Description of Land                                                                                                                                  | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                   |
| 376 cont'd              |                                                                                                                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |
| 377                     | 122 square metres of private road and verges (north of North Wales Expressway,<br>A55) (excluding all interests of the Crown)                        | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of restrictive covenants)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of restrictive covenants)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for maintenance)               |
| 378                     | 2220 square metres of agricultural land (north of North Wales Expressway, A55 and west of Nant-y-Faenol Road) (excluding all interests of the Crown) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables and right of access stated in Deed dated 17 October 2003)                                                                                                                                                                                                                                                |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                    | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| 378 cont'd              |                                                                                                                                                        | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| 379                     | 122 square metres of agricultural land (north of North Wales Expressway, A55 and<br>west of Nant-y-Faenol Road) (excluding all interests of the Crown) | Catherine Nerys Thomas 1 Oak Close ST. ASAPH Clwyd LL17 ODX (in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958) The Executor of the Estate of the Late Elizabeth Alice Jones Pengwern Farm Nant Y Faenol Road Bodelwyddan RHYL Denbighshire LL18 SUL (in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th November 1958) Jane Elizabeth Grindal Ty Mawr Farm St. Asaph Road Bodelwyddan RHYL Denbigshire LL18 SUV (in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958) The Secretary Of State For Wales William Morgan House 6-7 Central Square CARDIFF (F10 1EP (in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 380                     | 1383 square metres of private road and verges (north of North Wales Expressway, A55) and footpath (DE 201 7) (excluding all interests of the Crown)    | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                     |  |
| 381                     | 945 square metres of private road and verges (north of North Wales Expressway, A55) and footpath (DE 201 7) (excluding all interests of the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991)                                                                                                                                                                                                                                                                    |  |
| 382                     | 249 square metres of verge (North Wales Expressway, A55) (excluding all interests of the Crown)                                                    | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of restrictive covenants)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of restrictive covenants)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for maintenance) |  |
| 384                     | 1481 square metres of public road (North Wales Expressway, A55) (excluding all interests of the Crown)                                             | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right to enter and right to service media)                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire   |                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                               | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                  |  |
| 384 cont'd              |                                                                                                                   | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of right to enter and right to service media)                                                                                                                                                                              |  |
| 385                     | 194 square metres of verge (North Wales Expressway, A55) (excluding all interests of the Crown)                   | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right to enter and right to service media)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of right to enter and right to service media) |  |
| 386                     | 359 square metres of verge (North Wales Expressway, A55) (excluding all interests of the Crown)                   | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                   |  |
| 387                     | 782 square metres of public road (North Wales Expressway, A55) (excluding all interests of the Crown)             | Unknown                                                                                                                                                                                                                                                                                                                         |  |
| 390                     | 5 square metres of public road and verges (North Wales Expressway, A55)<br>(excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                                                         |  |
| 391                     | 4 square metres of verges (North Wales Expressway, A55) (excluding all interests of the Crown)                    | Unknown                                                                                                                                                                                                                                                                                                                         |  |

|                         | Awel y Môr Offst                                                                                                | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                   |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                                 | BOOK OF REFERENCE - PART 2                                                                                                                                                                                 |
|                         |                                                                                                                 | County of Denbighshire                                                                                                                                                                                     |
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                             |
| 392                     | 562 square metres of verge(North Wales Expressway, A55)(excluding all interests of the Crown)                   | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for maintenance)                                                    |
| 393                     | 21 square metres of public road and verges (North Wales Expressway, A55) (excluding all interests of the Crown) | Unknown                                                                                                                                                                                                    |
| 394                     | 14 square metres of public road (North Wales Expressway, A55) (excluding all interests of the Crown)            | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect to rights granted by Deed dated 6 June 1958 which also contains restrictive covenants) |
| 395                     | 175 square metres of verges (North Wales Expressway, A55) (excluding all interests of the Crown)                | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect to rights granted by Deed dated 6 June 1958 which also contains restrictive covenants) |
| 396                     | 104 square metres of public road and verge (North Wales Expressway, A55) (excluding all interests of the Crown) | Unknown                                                                                                                                                                                                    |
| 397                     | 67 square metres of verges (North Wales Expressway, A55) (excluding all interests of the Crown)                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for maintenance)                                                    |
| 398                     | 418 square metres of public road (North Wales Expressway, A55) (excluding all interests of the Crown)           | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of restrictive covenants)                                                      |

|                         | Awel y Môr Offst                                                                                                           | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                                            | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Number on<br>Land Plans | Description of Land                                                                                                        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 398 cont'd              |                                                                                                                            | Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of restrictive covenants)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for maintenance)                                                                                                                                                                                                                                                                                                                       |
| 399                     | 86 square metres of public road and verge (south of North Wales Expressway, A55)<br>(excluding all interests of the Crown) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and maintain gas pipeline)<br>Kent Periscopes Limited<br>Dowlish Ford<br>ILMINSTER<br>TA19 0PF<br>(in respect of right to service media and right to enter)<br>North Wales Fire Authority<br>Site G<br>Ffordd Salesbury<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJ<br>(in respect of rights stated in Lease dated 26th April 2005)<br>NW (BBP) Limited<br>3 More London Riverside<br>LONDON<br>SE1 2AQ<br>(in respect of rights stated in Lease dated 9th October 2002) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER |                                                                                                                                                                                     |  |
|-------------------------|---------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                              |                                                                                                                                                                                     |  |
|                         | County of Denbighshire                                  |                                                                                                                                                                                     |  |
| Number on<br>Land Plans | Description of Land                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                      |  |
|                         | Description of Land                                     |                                                                                                                                                                                     |  |
|                         |                                                         | Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)                                                                     |  |
|                         |                                                         | WGIF (Jersey) Trustee II Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                           |                                                                                                                                                                                                                                                                             |  |  |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                        |                                                                                                                                                                                                                                                                             |  |  |  |
|                         | County of Denbighshire                                                                                            |                                                                                                                                                                                                                                                                             |  |  |  |
| Number on<br>Land Plans | Description of Land                                                                                               | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                              |  |  |  |
| 399 cont'd              |                                                                                                                   | Wrexham Glyndwr University<br>Plas Coch<br>Mold Road<br>WREXHAM<br>LL11 2AW<br>(in respect of rights stated in Lease dated 9th July 2013)                                                                                                                                   |  |  |  |
| 400                     | 622 square metres of public road (south of North Wales Expressway, A55)<br>(excluding all interests of the Crown) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and maintain gas pipeline)<br>Kent Periscopes Limited<br>Dowlish Ford<br>ILMINSTER<br>TA19 0PF<br>(in respect of right to service media and right to enter) |  |  |  |
|                         |                                                                                                                   | North Wales Fire Authority<br>Site G<br>Ffordd Salesbury<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JJ<br>(in respect of rights stated in Lease dated 26th April 2005)                                                                                |  |  |  |
|                         |                                                                                                                   | NW (BBP) Limited<br>3 More London Riverside<br>LONDON<br>SE1 2AQ<br>(in respect of rights stated in Lease dated 9th October 2002)                                                                                                                                           |  |  |  |
|                         |                                                                                                                   | Police & Crime Commissioner for North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWYN BAY<br>Conwy<br>LL29 8AW<br>(in respect of rights stated in Lease dated 1st March 2000)                                                                        |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2 |                                                                                                                                                                                                   |  |
|-------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                   | County of Denbighshire<br>Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act<br>2008                                       |  |
| 400 cont'd              |                                                                                       | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access along roadways and right to lay and maintain gas pipeline) |  |
|                         |                                                                                       | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access along roadways and right to free passage of service media)       |  |
|                         |                                                                                       | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access along roadways)                                  |  |
|                         |                                                                                       | WGIF (Jersey) Trustee I Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)                |  |
|                         |                                                                                       | WGIF (Jersey) Trustee II Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)               |  |
|                         |                                                                                       | Wrexham Glyndwr University<br>Plas Coch<br>Mold Road<br>WREXHAM<br>LL11 2AW<br>(in respect of rights stated in Lease dated 9th July 2013)                                                         |  |

|                         | Awel y Môr Offst                                                                                                                                           | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                                                                            | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Description of Land                                                                                                                                        | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 401                     | 583 square metres of agricultural land, hedgerow and verge of public road (south of<br>North Wales Expressway, A55) (excluding all interests of the Crown) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and maintain gas pipeline)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of bridleway 201 10)                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 402                     | 10 square metres of public road (Old A55 Faenol Bropor, south of North Wales<br>Expressway, A55) (excluding all interests of the Crown)                    | The Secretary Of State For Wales<br>William Morgan House<br>G-7 Central Square<br>CARDIFF<br>CF10 IEP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage) |

|                         | Awel v Môr Offs                                                                                                                               | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                               | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|                         | County of Denbighshire                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                           | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |  |
| 403                     | 905 square metres of agricultural land, hedgerow and verge (south of North Wales Expressway, A55) (excluding all interests of the Crown)      | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and maintain gas pipeline)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
| 404                     | 1479 square metres of agricultural land, verges and hedgerow (south of North<br>Wales Expressway, A55) (excluding all interests of the Crown) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and maintain gas pipeline)<br>Denbighshire County Council<br>County Hail<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of bridleway 201 10)<br>North Wales Fire Authority<br>Site G<br>Ffordd Salesbury<br>St. Asaph Business Park<br>ST. Asaph Business Park |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
| Number on<br>Land Plans | Number on Description of Land Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning A |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| 404 cont'd              |                                                                                                                                                       | WGIF (Jersey) Trustee I Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)                                                                                                                                                                                                                                                                                                                                              |  |  |
|                         |                                                                                                                                                       | WGIF (Jersey) Trustee II Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)                                                                                                                                                                                                                                                                                                                                             |  |  |
|                         |                                                                                                                                                       | Wrexham Glyndwr University<br>Plas Coch<br>Mold Road<br>WREXHAM<br>LL11 2AW<br>(in respect of rights stated in Lease dated 9th July 2013)                                                                                                                                                                                                                                                                                                                                                                                       |  |  |
| 405                     | 77 square metres of agricultural land and hedgerow (south of North Wales<br>Expressway, A55) (excluding all interests of the Crown)                   | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         |                                                                                                                         | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                                                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 405 cont'd              |                                                                                                                         | Wales and West Utilities LimitedWales & West HouseSpooner CloseCoedkernewCASNEWYDDNP10 8FZ(in respect of right of support and right to free passage of service media)Wild Ground3-4 Plas Power RoadTanyfronWREXHAMLL11 55Z(in respect of right of access to maintain drainage)                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 406                     | 509 square metres of agricultural land (south of North Wales Expressway, A55)<br>(excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media) |  |

|                         | Awel y Môr C                                                                                                                        | Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                     |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                |                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                 | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                   |  |
| 406 cont'd              |                                                                                                                                     | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                                                                    |  |
| 407                     | 54 square metres of track and verge (Faenol-Bropor, south of North Wales<br>Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)                                                                                |  |
|                         |                                                                                                                                     | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                                                                      |  |
|                         |                                                                                                                                     | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to maintain drainage and right of support)                                                                      |  |
|                         |                                                                                                                                     | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media as contained in a Conveyance dated 11th March 1991) |  |
|                         |                                                                                                                                     | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                 |                                                                                                                                                                                               |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                                                              |                                                                                                                                                                                               |  |
|                         | County of Denbighshire                                                                                                                                  |                                                                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                |  |
| 408                     | 499 square metres of agricultural land, hedgerow and verge (Faenol-Bropor, south of North Wales Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)                             |  |
|                         |                                                                                                                                                         | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                   |  |
|                         |                                                                                                                                                         | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to maintain drainage and right of support)                   |  |
|                         |                                                                                                                                                         | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media) |  |
|                         |                                                                                                                                                         | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                 |  |
| 409                     | 120 square metres of track and verge (Faenol-Bropor, south of North Wales Expressway, A55) (excluding all interests of the Crown)                       | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                         |                                                                                                                                                                                               |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                |  |
| 409 cont'd              |                                                                                                                                                         | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                   |  |
|                         |                                                                                                                                                         | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support)                   |  |
|                         |                                                                                                                                                         | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media) |  |
|                         |                                                                                                                                                         | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                 |  |
| 410                     | 386 square metres of agricultural land, hedgerow and verge (Faenol-Bropor, south of North Wales Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)                             |  |
|                         |                                                                                                                                                         | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                   |  |

|                         | Awel y Môr Offst                                                                                                                             | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                          | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 410 cont'd              |                                                                                                                                              | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media as contained in a Conveyance dated 11th March 1991)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage) |  |
| 411                     | 1198 square metres of agricultural land (Faenol-bropor farm, south of North Wales<br>Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support)                                  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         |                                                                                                                                                                      | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                                  | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 411 cont'd              |                                                                                                                                                                      | Wales and West Utilities LimitedWales & West HouseSpooner CloseCoedkernewCASNEWYDDNP10 8FZ(in respect of right of support and right to free passage of service media)Wild Ground3-4 Plas Power RoadTanyfronWREXHAMLL11 5SZ(in respect of right of access to maintain drainage)                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 412                     | 312 square metres of agricultural land and private road (east of Faenol-bropor farm,<br>south of North Wales Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0IB<br>(in respect of right of access to maintain drainage and right of support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media) |  |

|                         | Awel y Môr Offsh                                                                                                                                        | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                         |  |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                                                              |                                                                                                                                                                                                                                                  |  |  |
|                         | County of Denbighshire                                                                                                                                  |                                                                                                                                                                                                                                                  |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                   |  |  |
| 412 cont'd              |                                                                                                                                                         | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                                                                    |  |  |
| 413                     | 26731 square metres of agricultural land, track, hedgerows (Faenol-Bropor, south of North Wales Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)                                                                                |  |  |
|                         |                                                                                                                                                         | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                                                                      |  |  |
|                         |                                                                                                                                                         | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to maintain drainage and right of support)                                                                      |  |  |
|                         |                                                                                                                                                         | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media as contained in a Conveyance dated 11th March 1991) |  |  |
|                         |                                                                                                                                                         | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                                                                    |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                          |                                                                                                                                                                                               |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                                                                       |                                                                                                                                                                                               |  |
|                         | County of Denbighshire                                                                                                                                           |                                                                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                |  |
| 414                     | 2872 square metres of agricultural land, ditch and hedgerow (east of Faenol-Bropor, south of North Wales Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)                             |  |
|                         |                                                                                                                                                                  | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                   |  |
|                         |                                                                                                                                                                  | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to maintain drainage and right of support)                   |  |
|                         |                                                                                                                                                                  | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media) |  |
|                         |                                                                                                                                                                  | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                 |  |
| 415                     | 11946 square metres of agricultural land and hedgerows (Faenol-Bropor, south of North Wales Expressway, A55) (excluding all interests of the Crown)              | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                     |                                                                                                                                                                                               |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                                        |                                                                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                                                                         | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                |  |
| 415 cont'd              |                                                                                                                                                                                                             | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                   |  |
|                         |                                                                                                                                                                                                             | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to maintain drainage and right of support)                   |  |
|                         |                                                                                                                                                                                                             | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media) |  |
|                         |                                                                                                                                                                                                             | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                 |  |
| 416                     | 10443 square metres of woodland, agricultural land and hedgerow (south of North<br>Wales Expressway, A55 and west of William Morgan Road) and footpath (DE 201 9)<br>(excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)                             |  |
|                         |                                                                                                                                                                                                             | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                       |                                                                                                                                                                                                                                                                                                                                                                           |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                            |  |
| 416 cont'd              |                                                                                                                                       | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to maintain drainage and right of support)<br>Wales and West Utilities Limited                                                                                                                                                           |  |
|                         |                                                                                                                                       | Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media)                                                                                                                                                                                                                 |  |
|                         |                                                                                                                                       | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                                                                                                                                                                                             |  |
| 417                     | 325521 square metres of agricultural land and private roads (north of Glascoed Road, B5381, and south of North Wales Expressway, A55) | TRB Limited         1 TRB Drive         St. Asaph Business Park         ST. ASAPH         Denbighshire         LL17 OJB         (in respect of right of access to maintain drainage and right to support as contained in a Conveyance dated 11th March 1991)         Unknown         (in respect of right of drainage as contained in a Conveyance dated 20th March 1919) |  |
|                         |                                                                                                                                       | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage as contained in a Conveyance dated 11th March 1991)                                                                                                                                                                                          |  |
| 418                     | 50 square metres of verge of public road (Glascoed Road, B5381)                                                                       | Unknown                                                                                                                                                                                                                                                                                                                                                                   |  |
| 419                     | 201 square metres of verge of public road (Glascoed Road, B5381)                                                                      | Unknown                                                                                                                                                                                                                                                                                                                                                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|                         | Description of Land 74 square metres of pavement (Glascoed Road, B5381) | 2008       Alun Lloyd Davies       6 Ffordd Glascoed       ABERGELE       LL22 9DW       (In respect of right of access)       Amy Louise Evans       5 Ffordd Glascoed       ABERGELE       Conwy       LL22 9DW       (in respect of right of access)       Martin Griffiths       5 Ffordd Glascoed       ABERGELE       Conwy       LL22 9DW       (in respect of right of access)       Martin Griffiths       5 Ffordd Glascoed       ABERGELE       Conwy       LL22 9DW       (in respect of right of access)       Carrie Hargraves       8 Ffordd Glascoed       ABERGELE       LL22 9DW       (in respect of right of access)       Ethan James Homer       4 Ffordd Glascoed       ABERGELE       Conwy       LL22 9DW       (in respect of right of access)       Ethan James Homer       4 Ffordd Glascoed       ABERGELE       Conwy       LL22 9DW       (in respect of right of access)       Ethan James Homer       4 Ffordd Glascoed       ABERGELE       Conwy       LL22 9DW       (in respect of right of access)    < |  |
|                         |                                                                         | 4 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|                         |                                                                         | Meirick Lloyd Davies<br>1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |

| <b>—</b>                | Awel y Môr Ot                                                             | ffshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|-------------------------|---------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                           | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|                         |                                                                           | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Number on<br>Land Plans | Description of Land                                                       | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                              |
| 420 cont'd              | 48 square metres of pavement (south of Glascoed Road, B5381)              | Nesta Davies         1 Ffordd Glascoed         ABERGELE         Conwy         LL22 9DW         (in respect of right of access)         The Executor of the Estate of the Late Ian Charles Hughes         3 Ffordd Glascoed         CONWY         Conwy         LL22 9DW         (in respect of right to use common sewers and right to enter to maintain sewers and drains)         William Owen Morris         7 Ffordd Glascoed         ABERGELE         LL22 9DW         (in respect of right of access) |
| 421                     | 48 square metres of pavement (south of Glascoed Road, B5381)              | 1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)<br>Nesta Davies<br>1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                              |
| 423                     | 101 square metres of verge of public road (south of Glascoed Road, B5381) | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 424                     | 33 square metres of verge of public road (Glascoed Road, B5381)           | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 425                     | 509 square metres of public road and verges (Glascoed Road, B5381)        | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 426                     | 278 square metres of verge of public road (south of Glascoed Road, B5381) | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 428                     | 258 square metres of agricultural land (south of Glascoed Road, B5381)    | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to inspect, maintain, protect, use, replace, remove or render unusable a gas pipeline and right to pass)                                                                                                                                                                                                                                                                                           |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                 |                                                                                                                                                                                                                                                                                                                                                                                                           |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                            |  |
| 428 cont'd              |                                                                                                                                 | Duncan Robertson<br>Llantysilio Hall<br>Llantysilio<br>LLANGOLLEN<br>LL20 8BT<br>(in respect of right to enter and right to all game)<br>David Watkin Williams-Wynn Bt<br>3 Upper Brook Street<br>OSWESTRY<br>SY11 2TB<br>(In respect of right to enter and right to all game)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |  |
| 429                     | 3 square metres of public road (Glascoed Road, B5381)                                                                           | Unknown                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 430                     | 7 square metres of public road (Glascoed Road, B5381)                                                                           | Unknown                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 431                     | 104 square metres of public road and verge (Glascoed Road, B5381)                                                               | Unknown                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 432                     | 71 square metres of verge of public road (north of Glascoed Road, B5381)                                                        | Unknown                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 436                     | 1226 square metres of agricultural land and hedgerows (west of Memorail Park and<br>Crematorium, south of Glascoed Road, B5381) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to inspect, maintain, protect, use, replace, remove or render unusable a gas pipeline and right to pass)<br>Duncan Robertson<br>Llantysilio Hall<br>Llantysilio<br>LLANGOLLEN<br>LL20 8BT<br>(in respect of right to enter and right to all game)                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| 436 cont'd              |                                                                                                                                | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay a water main and right to enter as contained in a Deed dated 12th January 1984)<br>David Watkin Williams-Wynn Bt<br>3 Upper Brook Street<br>OSWESTRY<br>SY11 2TB<br>(In respect of right to enter and right to all game)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus) |  |
| 437                     | 9776 square metres of agricultural land and hedgerow (north of Pentre-Mawr and<br>south of Glascoed Road, B5381)               | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to inspect, maintain, protect, use, replace, remove or render unusable a gas pipeline and right to pass)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay a water main and right to enter as contained in a Deed dated 12th January 1984)                                                |  |

| <u> </u>                | Awel v Môr Offs                                                                                               | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                  |
|-------------------------|---------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                                               | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                |
|                         |                                                                                                               | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Description of Land                                                                                           | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                            |
| 437 cont'd              |                                                                                                               | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                   |
| 438                     | 5595 square metres of agricultural land and hedgerow (north of Pentre-Mawr and south of Glascoed Road, B5381) | Memoria - Denbighshire Memorial Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support, and maintenance)<br>Unknown<br>(in respect of rights granted by deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |
| 439                     | 156 square metres of agricultural land and hedgerow (north of Pentre-Mawr and south of Glascoed Road, B5381)  | Memoria - Denbighshire Memorial Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support, and maintenance)<br>Unknown<br>(in respect of rights granted by deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |
| 440                     | 8 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)              | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |
| 441                     | 3 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)              | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |
| 442                     | 63 square metres of agricultural land and hedgerow (north of Pentre-Mawr and south of Glascoed Road, B5381)   | Memoria - Denbighshire Memorial Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support, and maintenance)                                                                                                                                                                                                       |

|                         | Awel y Môr Offs                                                                                              | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                           |  |
|                         | County of Denbighshire                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                                                                          | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                            |  |
| 442 cont'd              |                                                                                                              | Unknown<br>(in respect of rights granted by deed of grant dated 19th September 1979, and of restrictive covenants)                                                                                                                                                                                                                                                                        |  |
|                         |                                                                                                              | Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966)                                                                                                                                                                                                                                                                                                              |  |
| 443                     | 40 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)            | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 444                     | 258 square metres of public road (road from Glascoed Road past Waen Meredydd)                                | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 445                     | 635 square metres of agricultural land and hedgerow (north of Pentre-Mawr and south of Glascoed Road, B5381) | Memoria - Denbighshire Memorial Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support, and maintenance)<br>Unknown<br>(in respect of rights granted by deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 446                     | 103 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)           | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 447                     | 99 square metres of agricultural land and hedgerow (north of Pentre-Mawr and south of Glascoed Road, B5381)  | Memoria - Denbighshire Memorial Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support, and maintenance)<br>Unknown<br>(in respect of rights granted by deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 448                     | 2 square metres of agricultural land (north of Pentre-Mawr and south of Glascoed Road, B5381)                | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 449                     | 16 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)            | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 450                     | 143 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)           | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                     |  |
| 451                     | 188 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)              | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 452                     | 3016 square metres of agricultural land (north of Pentre-Mawr and south of Glascoed Road, B5381)                | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE |  |
|                         |                                                                                                                 | Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                                           |  |
| 453                     | 1990 square metres of agricultural land (north of Pentre-Mawr and south of Glascoed Road, B5381)                | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                               |  |

|                         | Awel y Môr Of                                                                                      | fshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                             |  |  |  |
|-------------------------|----------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         | - ,                                                                                                | BOOK OF REFERENCE - PART 2                                                                                                                                                                             |  |  |  |
|                         | County of Denbighshire                                                                             |                                                                                                                                                                                                        |  |  |  |
| Number on<br>Land Plans | Description of Land                                                                                | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                         |  |  |  |
| 453 cont'd              |                                                                                                    | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |  |  |
| 454                     | 2903 square metres of agricultural land (north of Pentre-Mawr and south of Glascoed Road, B5381)   | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                          |  |  |  |
|                         |                                                                                                    | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)    |  |  |  |
|                         |                                                                                                    | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |  |  |
| 455                     | 245 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381) | Unknown                                                                                                                                                                                                |  |  |  |
| 456                     | 514 square metres of agricultural land and hedge (south of Waen Meredydd)                          | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                               |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire         |                                                                                                                                                                                                        |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                     | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                         |  |
| 456 cont'd              |                                                                                                                         | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)    |  |
|                         |                                                                                                                         | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |
| 457                     | 290 square metres of agricultural land (north of Pentre-Mawr and south of Glascoed Road, B5381) and footpath (DE 105 7) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                          |  |
|                         |                                                                                                                         | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)    |  |
|                         |                                                                                                                         | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |
| 458                     | 137 square metres of road from Glascoed road past Waen Meredydd                                                         | Unknown                                                                                                                                                                                                |  |

| <b></b>                 | Awel v N                                                        | /lôr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
|-------------------------|-----------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 2                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
|                         | County of Denbighshire                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
| Number on<br>Land Plans | Description of Land                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
| 459                     | 158 square metres of agricultural land (south of Waen Meredydd) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
|                         |                                                                 | ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
| 460                     | 239 square metres of agricultural land (south of Waen Meredydd) | SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to install and maintain electricity cables)         Robert Bryn Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         LL22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)         Mona Elizabeth Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         LL22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)         Mona Elizabeth Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         LL22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |  |

|                         | Awel y Môr Offs                                                                                                                               | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|                         |                                                                                                                                               | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| Number on<br>Land Plans | Description of Land                                                                                                                           | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 461                     | 524 square metres of agricultural land (south of Waen Meredydd)                                                                               | SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to install and maintain electricity cables)         Robert Bryn Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         Ll22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)         Mona Elizabeth Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         Ll22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)         Mona Elizabeth Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         Ll22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |
| 462                     | 24500 square metres of agricultural land and private road (north of Pentre-Mawr<br>and south of Glascoed Road, B5381) and footpath (DE 105 7) | SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to install and maintain electricity cables)         Robert Bryn Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         Ll22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)         Mona Elizabeth Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         Ll22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)         Mona Elizabeth Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         Ll22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                       |  |
| 463                     | 13597 square metres of agricultural land and hedge (south of Waen Meredydd)                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                        |  |
|                         |                                                                                                                 | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                                  |  |
|                         |                                                                                                                 | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                               |  |
| 464                     | 17129 square metres of agricultural land and hedge (south of Waen Meredydd)                                     | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 465                     | 11550 square metres of agricultural land (south of Waen Meredydd)                                               | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                             |                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                     | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                            |  |
| Number                  |                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                 | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                  |  |
| 465 conť d              |                                                                                     | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                             |  |
|                         |                                                                                     | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                          |  |
| 466                     | 10905 square metres of agricultural land and hedge (south of Ffordd Richard Davies) | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)                                                                                                                                                                                                  |  |
|                         |                                                                                     | Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966)                                                                                                                                                                              |  |
| 467                     | 40 square metres of agricultural land and hedge (south Waen Meredydd)               | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

|                         | Awel y Môr Offsl                                                                                                       | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                             |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                   |                                                                                                                                                                                                                                                                                                                                                                                      |  |
|                         |                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                                    | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                       |  |
| 467 cont'd              |                                                                                                                        | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                               |  |
| 468                     | 48 square metres of public road (road from Glascoed Road past Waen Meredydd)<br>(excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                                                                                                              |  |
| 469                     | 304 square metres of track (south of Ffordd Richard Davies) (excluding all interests of the Crown)                     | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of easements granted by Deed of grant dated 15 July 2013)                                                                                                                                                                                                                              |  |
| 470                     | 46 square metres of agricultural land and track (south of Ffordd Richard Davies)                                       | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 471                     | 1394 square metres of agricultural land and hedge (south of Ffordd Richard Davies)                                     | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                  |                                                                                                                                                                                                    |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                     |  |
| 472                     | 3007 square metres of agricultural land (south of Ffordd Richard Davies)                                                         | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)                     |  |
|                         |                                                                                                                                  | Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 473                     | 10383 square metres of agricultural land and hedgerow (west of Electricity<br>Distribution Site, south of Ffordd Richard Davies) | National Grid Electricity Transmission PLC         1-3 Strand         LONDON         WC2N 5EH         (in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants) |  |
|                         |                                                                                                                                  | Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 474                     | 2441 square metres of agricultural land and hedgerow (west of Electricity Distribution Site, north of Hendy Gorse)               | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)<br>Unknown          |  |
|                         |                                                                                                                                  | (in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966)            |  |
| 475                     | 15344 square metres of grassland area and shrubbery and plyon (south of Bodelwyddan Substation, south of Ffordd Richard Davies)  | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)                            |  |

|                         | Awel y Môr Offs     | shore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|-------------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                     | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                         |                     | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Description of Land | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 475 cont'd              |                     | Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshine Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Colse<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and maintain gas pipelines)<br>Robert Bryn Davies<br>Pentre Mavr Farm<br>Groesffordd Marli<br>ABERGELE<br>Convy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |

|                         | Awel y I                                             | Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                       |  |
|-------------------------|------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                  | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                         |  |
| 476                     |                                                      | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)                                |  |
|                         |                                                      | Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                         |  |
|                         |                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                               |  |
|                         |                                                      | Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                                 |  |
|                         |                                                      | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)    |  |
|                         |                                                      | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                         |  |
| 477                     | 2889 square metres of agricultural land and hedge (north of Hendy Gorse)                                        | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)                                |  |
|                         |                                                                                                                 | Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                         |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                               |  |
|                         |                                                                                                                 | Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                                 |  |
|                         |                                                                                                                 | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)    |  |
|                         |                                                                                                                 | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire                 |                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                    |  |
| 478                     | 168 square metres of agricultural land (west of Electricity Distribution Site, south of Ffordd Richard Davies)                  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                                                                                                                                                                                                                                          |  |
| 479                     | 2209 square metres of agricultural land and hedge (west of Electricity Distribution<br>Site, south of Ffordd Richard Davies)    | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966)                              |  |
| 480                     | 3332 square metres of agricultural land and hedgerow (west of Electricity<br>Distribution Site, south of Ffordd Richard Davies) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire          |                                                                                                                                                                                                        |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                      | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                         |  |
| 480 cont'd              |                                                                                                                          | Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                                 |  |
|                         |                                                                                                                          | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and maintain gas pipelines)                          |  |
|                         |                                                                                                                          | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)    |  |
|                         |                                                                                                                          | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |
| 481                     | 174 square metres of agricultural land and hedge (west of Electricity Distribution Site, south of Ffordd Richard Davies) | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)                         |  |
|                         |                                                                                                                          | Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966)     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                      |  |
| 482                     | 90 square metres of agricultural land (west of Electricity Distribution Site, south of Ffordd Richard Davies)   | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)                             |  |
|                         |                                                                                                                 | Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                      |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                            |  |
|                         |                                                                                                                 | Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                              |  |
|                         |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and maintain gas pipelines)                       |  |
|                         |                                                                                                                 | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                             |                                                                                                                                                                                                                                                                                                                                                           |  |
|-------------------------|-----------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                                                          |                                                                                                                                                                                                                                                                                                                                                           |  |
|                         |                                                                                                     | County of Denbighshire                                                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                                                                 | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                            |  |
| 482 cont'd              |                                                                                                     | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                    |  |
| 483                     | 4800 square metres of agricultural land (north of Electricity Distribution Site)                    | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                                                                                                                                                                                  |  |
| 484                     | 9467 square metres of substation (Electricity Distribution Site, south of Ffordd<br>Richard Davies) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and maintain gas pipelines) |  |
| 485                     | 2378 square metres of agricultural land and hedgerow (south of Electricity<br>Distribution Site)    | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                 |  |

|                         | A                   | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|---------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                     | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on<br>Land Plans | Description of Land | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|                         |                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 485 cont'd              |                     | SP Manweb PLC<br>3 Frenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 38T<br>(in respect of right to install and maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marii<br>ABERGELE<br>Comwy<br>LL22 JDR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marii<br>ABERGELE<br>Comwy<br>LL22 JDR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                       | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|                         | re metres of ditch (south of Electricity Distribution Site)                           | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain water pipeline)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY<br>(in respect of right of access)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marii<br>ABERGELE<br>Corwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

|                         | Awel v                                                            | Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|                         |                                                                   | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                               | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 486 cont'd              |                                                                   | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 487                     | 232 square metres of hardstanding (Electricity Distribution Site) | Dwr Cymru Cyfyngedig         Dwr Cymru Welsh Water         Linea         Fortran Road         St. Mellons         CARDIFF         CF3 0LT         (in respect of right to lay and maintain water pipeline)         Richard Thomas Owen Williams         Cefn Estate Office         ST. ASAPH         Denbighshire         LL17 0EY         (in respect of right of access)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to install and maintain electricity cables)         Timothy Michael Bell         Cefn Estate Office         ST. ASAPH         Denbighshire         L127 0EY         (in respect of right to install and maintain electricity cables)         Timothy Michael Bell         Cefn Estate Office         ST. ASAPH         Denbighshire         L127 0EY         (in respect of right of access)         Robert Bryn Davies         Pentre Mawr Farm         Groesffordd Marii         ABERGELE         Conwy |  |
|                         |                                                                   | LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |

|                         | Awel v Mô                                                                 | r Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|---------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                           | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                         |                                                                           | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Number on<br>Land Plans | Description of Land                                                       | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 487 cont'd              |                                                                           | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                                                                                                   |
| 488                     | 29 square metres of private road (south of Electricity Distribution Site) | National Grid Electricity Transmission PLC         1-3 Strand         LONDON         WC2N 5EH         (in respect of right of access)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to install and maintain electricity cables)         Wales and West Utilities Limited         Wales & West House         Spooner Close         Coedkernew         CASNEWYDD         NP10 8FZ         (in respect of right to install and maintain gas pipelines) |
| 489                     | 1012 square metres of track (south of Electricity Distribution Site)      | National Grid Electricity Transmission PLC         1-3 Strand         LONDON         WC2N 5EH         (in respect of right of access)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to install and maintain electricity cables)                                                                                                                                                                                                                     |

|                         | Awely                                                       | / Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                            |  |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2                                  |                                                                                                                                                                               |  |
|                         |                                                             | County of Denbighshire                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                         | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                |  |
| 489 cont'd              |                                                             | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and maintain gas pipelines) |  |
| 490                     | 1302 square metres of track (south of Glascoed Road, B5381) | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of shooting rights and restrictive covenants)                   |  |
|                         |                                                             | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)                       |  |
|                         |                                                             | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access to maintain water pipeline)     |  |
|                         |                                                             | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>(in respect of rights stated in Deed dated 6th November 2009)           |  |
|                         |                                                             | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed dated 6th November 2009)                               |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                      | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 491                     | 33 square metres of private road (south of Glascoed Road, B5381, and west of Lon<br>Coed Esgob) and footpath (DE 208 32) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)<br>ESP Electricity Limited<br>E S P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalls Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of access)<br>Gwynt Y Mor Offshore Wind Farm Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wittshire<br>SN5 6PB<br>(in respect of right to install and maintain cables)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N SEH<br>(in respect of right stated in Deed dated 6th November 2009) |  |
| 492                     | 1147 square metres of track (south of Glascoed Road, B5381)                                                              | (in respect of restrictive covenants)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |

|                         | Awel y Môr O                                                           | ffshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                    |  |  |
|-------------------------|------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | Awery More                                                             | BOOK OF REFERENCE - PART 2                                                                                                     |  |  |
|                         | County of Denbighshire                                                 |                                                                                                                                |  |  |
| Number                  |                                                                        |                                                                                                                                |  |  |
| Number on<br>Land Plans | Description of Land                                                    | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |  |
| Laliu Pialis            |                                                                        |                                                                                                                                |  |  |
| 492 cont'd              |                                                                        | ESP Electricity Limited                                                                                                        |  |  |
|                         |                                                                        | E S P Utilities Group Ltd                                                                                                      |  |  |
|                         |                                                                        | Bluebird House                                                                                                                 |  |  |
|                         |                                                                        | Unit 22 Mole Business Park                                                                                                     |  |  |
|                         |                                                                        | Randalls Road                                                                                                                  |  |  |
|                         |                                                                        | LEATHERHEAD                                                                                                                    |  |  |
|                         |                                                                        | Surrey                                                                                                                         |  |  |
|                         |                                                                        | KT22 7BA                                                                                                                       |  |  |
|                         |                                                                        | (in respect of right of access)                                                                                                |  |  |
|                         |                                                                        | Gwynt Y Mor Offshore Wind Farm Limited                                                                                         |  |  |
|                         |                                                                        | Windmill Hill Business Park                                                                                                    |  |  |
|                         |                                                                        | Whitehill Way                                                                                                                  |  |  |
|                         |                                                                        | SWINDON                                                                                                                        |  |  |
|                         |                                                                        | Wiltshire                                                                                                                      |  |  |
|                         |                                                                        | SN5 6PB                                                                                                                        |  |  |
|                         |                                                                        | (in respect of rights stated in Deed dated 6th November 2009)                                                                  |  |  |
|                         |                                                                        | Gwynt Y Mor OFTO PLC                                                                                                           |  |  |
|                         |                                                                        | Q14 Quorum Business Park                                                                                                       |  |  |
|                         |                                                                        | Benton Lane                                                                                                                    |  |  |
|                         |                                                                        | NEWCASTLE UPON TYNE                                                                                                            |  |  |
|                         |                                                                        | NE12 8BU                                                                                                                       |  |  |
|                         |                                                                        | National Grid Electricity Transmission PLC                                                                                     |  |  |
|                         |                                                                        | 1-3 Strand                                                                                                                     |  |  |
|                         |                                                                        | LONDON                                                                                                                         |  |  |
|                         |                                                                        | WC2N 5EH                                                                                                                       |  |  |
|                         |                                                                        | (in respect of rights stated in Deed dated 6th November 2009)                                                                  |  |  |
|                         |                                                                        |                                                                                                                                |  |  |
|                         |                                                                        | Unknown<br>(in respect of restrictive covenants)                                                                               |  |  |
|                         |                                                                        |                                                                                                                                |  |  |
| 493                     | 177 square metres of public road (Glascoed Road, B5381)                | Unknown                                                                                                                        |  |  |
| 494                     | 71 square metres of hedgerow (south of Glascoed Road, B5381)           | Unknown                                                                                                                        |  |  |
|                         |                                                                        | (in respect of rights contained in a Conveyance dated 20 March 1919)                                                           |  |  |
| 495                     | 15 square metres of track and hedgerow (south of Glascoed Road, B5381) | Diamond Transmission Partners BBE Limited                                                                                      |  |  |
|                         |                                                                        | Mid City Place                                                                                                                 |  |  |
|                         |                                                                        | 71 High Holborn                                                                                                                |  |  |
|                         |                                                                        | LONDON                                                                                                                         |  |  |
|                         |                                                                        | WC1V 6BA                                                                                                                       |  |  |
|                         |                                                                        | (in respect of right of access)                                                                                                |  |  |
| <u> </u>                |                                                                        |                                                                                                                                |  |  |

|                         | Awel y Môr Offs                                            | hore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans | Description of Land                                        | County of Denbighshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 495 cont'd              |                                                            | ESP Electricity Limited<br>E S P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalls Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of access)<br>Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>(in respect of rights stated in Deed dated 6th November 2009)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009) |
| 496                     | 60 square metres of public road (Glascoed Road, B5381)     | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 497                     | 194 square metres of track (south of Glascoed Road, B5381) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)<br>ESP Electricity Limited<br>E S P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalls Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of way)                                                                                                                                                                                                                                               |

|                         | Awel y Môr Of                                                             | fshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|---------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | ,                                                                         | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|                         | County of Denbighshire                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                                       | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 497 cont'd              |                                                                           | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>(in respect of rights stated in Deed dated 6th November 2009)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed dated 6th November 2009)                                                                                                                                                                                                                                                                                                                                    |  |
| 498                     | 13 square metres of hedgerow (south of Glascoed Road, B5381)              | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)<br>ESP Electricity Limited<br>E S P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalls Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of way)<br>Gwynt Y Mor Offshore Wind Farm Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SNS 6PB<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU |  |
| 400                     | 127 course matrice of using of public read (                              | (in respect of rights stated in Deed dated 6th November 2009)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 499                     | 137 square metres of verge of public road (south of Glascoed Road, B5381) | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |

|                         | Awel y Môr Offs                                                                        | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                       |  |  |
|-------------------------|----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | -,                                                                                     | BOOK OF REFERENCE - PART 2                                                                                                     |  |  |
|                         | County of Denbighshire                                                                 |                                                                                                                                |  |  |
| Number on<br>Land Plans | Description of Land                                                                    | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |  |
| N/A                     | Bryn Celyn Lodge, Glascoed, Abergele, Conwy, LL22 9DF                                  | Charlotte Anne Southern<br>Bryn Celyn Lodge<br>Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DF                                       |  |  |
| N/A                     | Land lying to the south side of St. Asaph Road, Bodelwyddan, Rhyl, Denbighshire, LL18  | Coed Du Hall Limited<br>6-8 Old Hall Road<br>Gatley<br>CHEADLE<br>Cheshire<br>SK8 4BE                                          |  |  |
| N/A                     | Waen Meredydd Farm House, Glascoed Road, St Asaph, LL17 0LG                            | Harry Watkin Williams-Wynn<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY                                      |  |  |
| N/A                     | Sun Valley Caravan Park, Marsh Road, Rhuddlan, Rhyl, L18 5UD                           | SF Sunvalley Limited<br>Banks House<br>Ty Isa Road<br>LLANDUDNO<br>Conwy<br>LL30 2PL                                           |  |  |
| N/A                     | Land adjoining Ty Ni, Nant Y Faenol Road, Bodelwyddan, Rhyl, Denbighshire, LL18<br>5UL | Andrew Anthony Gizzi<br>Ty Ni<br>Nant Y Faenol Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UL                                       |  |  |
| N/A                     | 49 Ffordd Ffynnon, Rhuddlan, Rhyl, LL18 2SP                                            | Reginald Graham Morgan<br>49 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP                                    |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                             |  |
| N/A                     | 43 Ffordd Ffynnon, Rhuddlan, Rhyl, LL18 2SP                                                                     | Patricia Hannah Roberts<br>43 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP<br>William David Roberts<br>43 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP |  |
| N/A                     | 41 Ffordd Ffynnon, Rhuddlan, Rhyl, LL18 2SP                                                                     | Jacqueline Carole Kay<br>41 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP                                                                                                 |  |
| N/A                     | 39 Ffordd Ffynnon, Ty Ddraig, Rhuddlan, Rhyl, LL18 2SP                                                          | Sian Elizabeth Griffiths<br>39 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP<br>Sion Haydn Griffiths<br>39 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP |  |
| N/A                     | 37 Ffordd Ffynnon, Rhuddlan, Rhyl, LL18 2SP                                                                     | Hannah Elizabeth Penlington<br>37 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                         |  |
| N/A cont'd              |                                                                                                                 | lan David Penlington<br>37 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP                                                                                              |  |
| N/A                     | 45 Ffordd Ffynnon, Rhuddlan, Rhyl, LL18 2SP                                                                     | Bronwen Tolley<br>45 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP<br>Steven Ralph Tolley<br>45 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP        |  |
| N/A                     | 35 Ffordd Ffynnon, Rhuddlan, LL18 2SP                                                                           | Catrin Wyn Williams<br>35 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP<br>Russell Owen Williams<br>35 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP |  |
| N/A                     | 31 Lon Cwybr, Rhuddlan, LL18 2SX                                                                                | Rowena Millward Roberts<br>31 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SX                                                                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                               |  |
| N/A cont'd              |                                                                                                                 | Steven John Roberts<br>31 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SX                                                                                          |  |
| N/A                     | 29 Lon Cwybr, Fairland Estate, Rhuddlan, LL18 2SX                                                               | Matthew John Bowyer<br>29 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SX                                                                                          |  |
| N/A                     | 27 Lon Cwybr, Rhuddlan, Rhyl, LL18 2SX                                                                          | Dean Joseph Martin<br>27 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SX<br>Donna Christine Martin<br>27 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SX |  |
| N/A                     | 25 Lon Cwybr, Rhuddlan, Rhyl, LL18 2SX                                                                          | Leigh Anne Carr<br>25 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SX<br>Martin William Carr<br>25 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SX       |  |

|                         | Α                                           | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                     |
|-------------------------|---------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                             | BOOK OF REFERENCE - PART 2                                                                                                                                                                                                 |
|                         |                                             | County of Denbighshire                                                                                                                                                                                                     |
| Number on<br>Land Plans | Description of Land                         | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                             |
| N/A                     | 33 Ffordd Ffynnon, Rhuddlan, Rhyl, LL18 2SP | Patricia Ann Hughes<br>33 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP<br>The Executor of the Estate of the Late Hugh Thomas Hughes<br>33 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SP |
| N/A                     | 24 Lon Cwybr, Rhuddlan, Rhyl, LL18 2SX      | Philip Twemlow<br>24 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SX<br>Phyllis Twemlow<br>24 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2SX                                                          |
| N/A                     | 31 Ffordd Ffynnon, Rhuddlan, LL18 2SP       | Susan Jane Williams<br>Glasfryn<br>Princes Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5RA<br>Trefor Rhydwen Williams<br>Glasfryn<br>Princes Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5RA                     |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                        |  |
| N/A                     | Bryn Cwybr Nursing Home, Cwybr, Rhuddlan, Rhyl, LL18 2YD                                                        | Bangor Centre For Developmental Disabilities Limited<br>54 Kinmel Street<br>RHYL<br>Denbighshire<br>LL18 1AR                                                                                          |  |
| N/A                     | 12 Bryn Hafod, Rhuddlan, Rhyl, LL18 2TB                                                                         | Gareth Lloyd Jones<br>12 Bryn Hafod<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2TB<br>Patricia Jean Ann Jones<br>12 Bryn Hafod<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2TB                       |  |
| N/A                     | Pant, 16 Highlands Close, Rhuddlan, Rhyl, LL18 2RU                                                              | Margaret Mary Jones<br>Pant<br>16 Highlands Close<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2RU<br>Richard Owen Jones<br>Pant<br>16 Highlands Close<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2RU |  |
| N/A                     | 6 Terfyn Pella Avenue, Rhyl, LL18 3YW                                                                           | Aileen Clayton<br>6 Terfyn Pella Avenue<br>RHYL<br>Denbighshire<br>LL18 3YW                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
| N/A cont'd              |                                                                                                                 | Robert Henry Francis Clayton<br>6 Terfyn Pella Avenue<br>RHYL<br>Denbighshire<br>LL18 3YW                                      |  |
| N/A                     | 8 Sherwood Avenue, Rhyl, LL18 3YN                                                                               | Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE         |  |
| N/A                     | 4 Terfyn Pella Avenue, Rhyl, LL18 3YW                                                                           | Adrienne Lesley Garry<br>424 Rhyl Coast Road<br>RHYL<br>Denbighshire<br>LL18 3YG                                               |  |
|                         |                                                                                                                 | Philip Anthony Garry<br>424 Rhyl Coast Road<br>RHYL<br>Denbighshire<br>LL18 3YG                                                |  |
| N/A                     | 423 Coast Road, Rhyl, LL18 3YE                                                                                  | Lyons Holiday Park Limited<br>Champion Allwoods Ltd<br>Refuge House<br>33-37 Watergate Row South<br>CHESTER<br>CH1 2LE         |  |
| N/A                     | 429 Rhyl Coast Road, Rhyl, LL18 3YE                                                                             | Anthony William Higgs<br>22 Maxstoke Close<br>REDDITCH<br>Worcestershire<br>B98 0EJ                                            |  |
| N/A                     | Terfyn Pella, Rhyl Coast Road, Rhyl, Denbighshire, LL18 3YH                                                     | Diane Elizabeth Inglis<br>24 Pennine Road<br>Woodley<br>STOCKPORT<br>Cheshire<br>SK6 1JS                                       |  |

|                         | A                               | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                                                                          |
|-------------------------|---------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| County of Denbighshire  |                                 |                                                                                                                                                                                               |
| Number on<br>Land Plans | Description of Land             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                |
| N/A cont'd              |                                 | Joyce Anne Signol<br>Terfyn Pella Camp<br>Rhyl Coast Road<br>RHYL<br>Denbighshire<br>LL18 3YH<br>Stewart Nicholas Signol<br>244 Colne Road<br>Sough<br>BARNOLDSWICK<br>Lancashire<br>BB18 6TD |
| N/A                     | 31 Garford Road, Rhyl, LL18 3UF | Charles Edward Moore<br>31 Garford Road<br>Y RHYL<br>LL18 3UF<br>Linda Ann Moore<br>31 Garford Road<br>Y RHYL<br>LL18 3UF                                                                     |
| N/A                     | 4 Garford Road, Rhyl, LL18 3UF  | Geraldine Anne Cyster<br>4 Garford Road<br>RHYL<br>LL18 3UF                                                                                                                                   |
| N/A                     | 9 Garford Road, Rhyl, LL18 3UF  | Angela Marie Rainbird<br>9 Garford Road<br>RHYL<br>LL18 3UF                                                                                                                                   |
| N/A                     | 2a Garford Road, Rhyl, LL18 3UF | Margaret Wardley<br>2A Garford Road<br>RHYL<br>LL18 3UF                                                                                                                                       |
| N/A                     | 2 Garford Road, Rhyl, LL18 3UF  | The Executor of the Estate of the Late Diane Curvis<br>2 Garford Road<br>RHYL<br>LL18 3UF                                                                                                     |

|                         | Av                                                   | vel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                         |  |
|-------------------------|------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                |  |
| Number on<br>Land Plans | Description of Land                                  | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
| N/A cont'd              |                                                      | William Curvis<br>2 Garford Road<br>Y RHYL<br>LL18 3UF                                                                         |  |
| N/A                     | 23 Garford Road, Rhyl, LL18 3UF                      | Paul David Smith<br>23 Garford Road<br>RHYL<br>LL18 3UF                                                                        |  |
|                         |                                                      | Sian Smith<br>23 Garford Road<br>RHYL<br>LL18 3UF                                                                              |  |
| N/A                     | 3 Brynhedydd Bay, Rhyl, LL18 3TP                     | Helen Christine Collard<br>3 Brynhedydd Bay<br>RHYL<br>LL18 3TP                                                                |  |
|                         |                                                      | John Paul Collard<br>3 Brynhedydd Bay<br>RHYL<br>LL18 3TP                                                                      |  |
| N/A                     | 29 Eaton Avenue, Rhyl, LL18 3UE                      | Tracy Lynn Bateman<br>29 Eaton Avenue<br>RHYL<br>LL18 3UE                                                                      |  |
| N/A                     | 5 Brynhedydd Bay, Rhyl, LL18 3TP                     | Connor Jack Cox<br>5 Brynhedydd Bay<br>RHYL<br>LL18 3TP                                                                        |  |
| N/A                     | 6 Garford Road, Rhyl, LL18 3UF                       | Bernard Jude Harrison<br>6 Garford Road<br>RHYL<br>LL18 3UF                                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
| N/A cont'd              |                                                                                                                 | Natalia Mikhaleva-Harrison<br>6 Garford Road<br>RHYL<br>LL18 3UF                                                               |  |
| N/A                     | 28 Eaton Avenue, Rhyl, LL18 3UE                                                                                 | Bernadette Conway<br>28 Eaton Avenue<br>RHYL<br>LL18 3UE<br>Neil Rawson<br>28 Eaton Avenue<br>RHYL                             |  |
| N/A                     | 12 Garford Road, Rhyl, LL18 3UF                                                                                 | LL18 3UE Debra Frances Moffat 23 Marine Drive Y RHYL LL18 3AY Russell Dean Moffat 23 Marine Drive Y RHYL LL18 3AY              |  |
| N/A                     | 29 Garford Road, Rhyl, LL18 3UF                                                                                 | Anthony Robert Hawthorn<br>29 Garford Road<br>Y RHYL<br>LL18 3UF                                                               |  |
| N/A                     | 2B Garford Road, Rhyl, LL18 3UF                                                                                 | Ralph Stuart Thomasson<br>2B Garford Road<br>Y RHYL<br>LL18 3UF<br>Susan Thomasson<br>2B Garford Road<br>Y RHYL<br>LL18 3UF    |  |
| N/A                     | 12 Brynhedydd Bay, Rhyl, LL18 3TP                                                                               | Frances Anne Kemp<br>12 Brynhedydd Bay<br>Y RHYL<br>LL18 3TP                                                                   |  |

|            | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire<br>Number on Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act |                                                                                                                                  |  |
|------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|--|
| Number on  |                                                                                                                                                                                                                                                        |                                                                                                                                  |  |
| Land Plans | Description of Land                                                                                                                                                                                                                                    | 2008                                                                                                                             |  |
| N/A cont'd |                                                                                                                                                                                                                                                        | Steven David Kemp<br>12 Brynhedydd Bay<br>Y RHYL<br>LL18 3TP                                                                     |  |
| N/A        | Land on the east side of Garford Road, Rhyl, LL18                                                                                                                                                                                                      | Firstage Limited<br>Leisure Buildings<br>Garford Road<br>RHYL<br>LL18 3UF                                                        |  |
| N/A        | 7 Garford Road, Rhyl, LL18 3UF                                                                                                                                                                                                                         | Sean John Long<br>7 Garford Road<br>Y RHYL<br>LL18 3UF<br>Valerie Long<br>7 Garford Road<br>Y RHYL                               |  |
| N/A        | 31 Ridgeway Avenue, Rhyl, LL18 3UG                                                                                                                                                                                                                     | LL18 3UF<br>Jacqueline Ellen Brown                                                                                               |  |
| 17.0       | ST Mugeway Avenue, Miyi, LEIS SOC                                                                                                                                                                                                                      | 31 Ridgeway Avenue<br>RHYL<br>LL18 3UG                                                                                           |  |
| N/A        | 27 Garford Road, Rhyl, LL18 3UF                                                                                                                                                                                                                        | John Kavanagh Jones<br>Lluest Pentre<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6HY                                            |  |
| N/A        | 21 Garford Road, Rhyl, LL18 3UF                                                                                                                                                                                                                        | Rita Carol Walsh<br>21 Garford Road<br>Y RHYL<br>LL18 3UF                                                                        |  |
| N/A        | 32 Ridgeway Avenue, Rhyl, LL18 3UG                                                                                                                                                                                                                     | Julie Brenda Griffiths<br>32 Ridgeway Avenue<br>Y RHYL<br>LL18 3UG<br>Neil Griffiths<br>32 Ridgeway Avenue<br>Y RHYL<br>LL18 3UG |  |

|                         | A                                | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                                                              |  |
|-------------------------|----------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire           |                                                                                                                                                                                   |  |
| Number on<br>Land Plans | Description of Land              | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                    |  |
| N/A                     | 25 Garford Road, Rhyl, LL18 3UF  | June Elizabeth Sadler<br>25 Garford Road<br>Y RHYL<br>LL18 3UF                                                                                                                    |  |
| N/A                     | 11a Garford Road, Rhyl, LL18 3UF | Florence Savory<br>11A Garford Road<br>RHYL<br>LL18 3UF<br>Ronald Arthur Savory<br>11A Garford Road<br>RHYL<br>LL18 3UF                                                           |  |
| N/A                     | 19 Garford Road, Rhyl, LL18 3UF  | Maureen Standish<br>19 Garford Road<br>RHYL<br>LL18 3UF<br>Nigel Peter Standish<br>19 Garford Road<br>RHYL<br>LL18 3UF<br>Rita Carol Walsh<br>19 Garford Road<br>RHYL<br>LL18 3UF |  |
| N/A                     | 16 Garford Road, Rhyl, LL18 3UF  | Raymond David Hinsley<br>16 Garford Road<br>Y RHYL<br>LL18 3UF<br>Sherrylynn Margaret Bacciochi<br>16 Garford Road<br>Y RHYL<br>LL18 3UF                                          |  |
| N/A                     | 11 Garford Road, Rhyl, LL18 3UF  | Marion Tiffney<br>11 Garford Road<br>RHYL<br>LL18 3UF                                                                                                                             |  |

|                         | A                                 | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2                                                                           |  |
|-------------------------|-----------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire            |                                                                                                                                                                |  |
| Number on<br>Land Plans | Description of Land               | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                 |  |
| N/A                     | 17 Garford Road, Rhyl, LL18 3UF   | lan Stringer<br>17 Garford Road<br>Y RHYL<br>LL18 3UF                                                                                                          |  |
| N/A                     | 3 Garford Road, Rhyl, LL18 3UF    | Alan Smith<br>3 Garford Road<br>RHYL<br>LL18 3UF<br>Margaret Smith<br>3 Garford Road<br>RHYL<br>LL18 3UF                                                       |  |
| N/A                     | 10 Brynhedydd Bay, Rhyl, LL18 3TP | Catherine Phyllis Shaw<br>10 Brynhedydd Bay<br>RHYL<br>LL18 3TP<br>Phillip Andrew Shaw<br>10 Brynhedydd Bay<br>RHYL<br>LL18 3TP                                |  |
| N/A                     | 14 Garford Road, Rhyl, LL18 3UF   | Anne Byrne<br>14 Garford Road<br>RHYL<br>LL18 3UF                                                                                                              |  |
| N/A                     | 1 Garford Road, Rhyl, LL18 3UF    | Sylvia Caroline Nicholson<br>1 Garford Road<br>RHYL<br>LL18 3UF                                                                                                |  |
| N/A                     | 10 Garford Road, Rhyl, LL18 3UF   | Maureen Gleave<br>10 Garford Road<br>RHYL<br>LL18 3UF<br>The Executor of the Estate of the Late Reginald Malcolm Gleave<br>10 Garford Road<br>RHYL<br>LL18 3UF |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
| N/A                     | 8 Garford Road, Rhyl, LL18 3UF                                                                                  | Maria Theresa Mardon<br>8 Garford Road<br>RHYL<br>LL18 3UF                                                                     |  |
| N/A                     | 5 Garford Road, Rhyl, LL18 3UF                                                                                  | Jodie Catrina Anne Hutton<br>5 Garford Road<br>RHYL<br>LL18 3UF<br>Lloyd George Edward Redgrave                                |  |
|                         |                                                                                                                 | 5 Garford Road<br>RHYL<br>LL18 3UF                                                                                             |  |
| N/A                     | 15 Garford Road, Rhyl, LL18 3UF                                                                                 | Karen Jayne Roberts<br>67 Maes Y Gog<br>Y RHYL<br>LL18 4QA                                                                     |  |
|                         |                                                                                                                 | Liam Brian Roberts<br>Flat 9<br>Newton Court<br>1 Axio Way<br>LONDON<br>E3 4QW                                                 |  |
| N/A                     | 22 Lon Cwybr, Rhuddlan, Rhyl, LL18 2SX                                                                          | Elliot Gough<br>22 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 2SX                                                          |  |
|                         |                                                                                                                 | Jeanette Gough<br>22 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 2SX                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                        |  |
| N/A                     | 20 Lon Cwybr, Rhuddlan, Rhyl, LL18 2SX                                                                          | Susan Parry Williams<br>20 Lon Cwybr<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 2SX                                                                         |  |
| N/A                     | 29 Ffordd Ffynnon, Rhuddlan, Rhyl, LL18 2SP                                                                     | Gwilym Owen<br>29 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 2SP<br>Pauline Jane Owen<br>29 Ffordd Ffynnon<br>Rhuddlan<br>Y RHYL<br>LL18 2SP |  |
| N/A                     | 27 Ffordd Ffynnon, Rhuddlan, Rhyl, LL18 2SP                                                                     | Enid Myfanwy Jones<br>27 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 2SP                                                                      |  |
| N/A                     | 25 Ffordd Ffynnon, Rhuddlan, Rhyl, LL18 2SP                                                                     | Ceridwen Jean Jones<br>25 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 2SP                                                                     |  |
| N/A                     | 30 Ffordd Ffynnon, Rhuddlan, LL18 2SR                                                                           | Helen Ann Brick<br>30 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 2SR                                                                         |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                           |  |
| N/A                     | 32 Ffordd Ffynnon, Fairlands Estate, Rhuddlan, LL18 2SR                                                         | David Littler<br>32 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 2SR<br>Loretta Littler<br>32 Ffordd Ffynnon<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 2SR                                                                                                                                                                                                                                             |  |
| N/A                     | 7 Llys Penyffordd, Pentre Lane, Rhuddlan, Rhyl, LL18 6HJ                                                        | Constance Audrey Webster<br>7 Liys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ<br>Francis Benson Webster<br>7 Liys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ<br>Karen Darrall<br>7 Liys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ<br>Karen Barrall<br>7 Liys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ |  |
| N/A                     | 6 Llys Penyffordd, Pentre Lane, Rhuddlan, Rhyl, LL18 6HJ                                                        | Arthur Edward Goddard<br>6 Llys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ                                                                                                                                                                                                                                                                                                       |  |

|                         | Awel y Môr                                                           | Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|----------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land                                                  | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                                                                                                                                                                                                                                                          |  |
| N/A cont'd              |                                                                      | David Joseph Goddard<br>6 Llys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ                                                                                                                                                                                                                                                                                                                                                       |  |
| N/A                     | 8 Llys Penyffordd, Pentre Lane, Rhuddlan, LL18 6HJ                   | Gordon Thomas Harrington         8 Llys Penyffordd         Pentre Lane         Rhuddlan         RHYL         Clwyd         LL18 6HJ         Marilyn Jean Harrington         8 Llys Penyffordd         Pentre Lane         Rhuddlan         RHYL         Clwyd         LL18 6HJ         Clwyd         LL18 6H         Marilyn Jean Harrington         8 Llys Penyffordd         Pentre Lane         Rhuddlan         RHYL         Clwyd         LL18 6HJ |  |
| N/A                     | Land adjoining 9 Llys Pen-Y-Ffordd, Pentre Lane, Rhuddlan, LL18 6HJ  | Hilary Jean Douglas<br>9 Llys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6HJ                                                                                                                                                                                                                                                                                                                                                               |  |
| N/A                     | 10 Llys Pen-y-Ffordd, Rhuddlan, LL18 6HJ                             | John White<br>10 Llys Pen-Y-Ffordd<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ                                                                                                                                                                                                                                                                                                                                                                             |  |
| N/A                     | Land adjoining 11 Llys Pen-Y-Ffordd, Pentre Lane, Rhuddlan, LL18 6HJ | Barbara Joyce Pierce<br>11 Llys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ                                                                                                                                                                                                                                                                                                                                                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                               |  |
| N/A cont'd              |                                                                                                                 | David Basil Pierce<br>11 Llys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ                                                                                             |  |
| N/A                     | Land adjoining Bryn Ebrill, Pentre Lane, Rhuddlan, Rhyl, LL18 6HY                                               | Emma Louise Williams<br>Bryn Ebrill<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HY<br>Stephen Matthew Haimes<br>Bryn Ebrill<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HY |  |
| N/A                     | 433 Rhyl Coast Road, Rhyl, LL18 3YE                                                                             | Amie Louise Pearson-Cook<br>433 Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3YE<br>Jonathan Kevin Bowron<br>433 Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3YE                                    |  |
| N/A                     | 435 Rhyl Coast Road, Rhyl, LL18 3YE                                                                             | Katie Marie Owens<br>435 Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3YE                                                                                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                 |  |
| N/A                     | 20 Griffiths Court, Rhyl, LL18 3YF                                                                              | Amie Jayne Owen<br>20 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                                                             |  |
|                         |                                                                                                                 | Connor David Gore<br>20 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                                                           |  |
| N/A                     | 19 Griffiths Court, Rhyl, LL18 3YF                                                                              | Amie Louise Bagnall         19 Griffiths Court         RHYL         Clwyd         LL18 3YF         Matthew Thomas Smith         19 Griffiths Court         RHYL         Clwyd         LL18 3YF |  |
| N/A                     | 18 Griffiths Court, Rhyl, LL18 3YF                                                                              | Majdi Jerbi<br>18 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                                                                 |  |
| N/A                     | 17 Griffiths Court, Rhyl, LL18 3YF                                                                              | Nichola Williams<br>17 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                                                            |  |
| N/A                     | 437 Rhyl Coast Road, Rhyl, LL18 3YE                                                                             | Chadleigh Steven Pritchard<br>437 Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3YE                                                                                                                 |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                       |  |
| N/A cont'd              |                                                                                                                 | Tasha Louise Metcalfe<br>437 Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3YE                                                                            |  |
| N/A                     | 15 Griffiths Court, Rhyl, LL18 3YF                                                                              | Amy Rebecca Cresswell<br>15 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                             |  |
| N/A                     | 16 Griffiths Court, Rhyl, LL18 3YF                                                                              | Rebecca Williams<br>16 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                  |  |
| N/A                     | 14 Griffiths Court, Rhyl, LL18 3YF                                                                              | Alison Evie Crompton<br>14 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                              |  |
| N/A                     | 13 Griffiths Court, Rhyl, LL18 3YF                                                                              | Jennifer Claire Sladen<br>13 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF<br>Michael James Sladen<br>13 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF |  |
| N/A                     | 12 Griffiths Court, Rhyl, LL18 3YF                                                                              | Charlotte Tonge<br>12 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                         |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                          |  |
| N/A                     | 439 Rhyl Coast Road, Rhyl, LL18 3YE                                                                             | Hannah Marie Hughes<br>439 Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3YE                                                                                 |  |
|                         |                                                                                                                 | Jay Daniel Thomas Semple<br>439 Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3YE                                                                            |  |
| N/A                     | 10 Cherry Close, Prestatyn, LL19 7DQ                                                                            | Graham David Evans<br>10 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ<br>Joanne Louise Inskip<br>10 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ    |  |
| N/A                     | 9 Cherry Close, Prestatyn, LL19 7DQ                                                                             | Anne Isobel Pennington<br>9 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ<br>David John Pennington<br>9 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ |  |
| N/A                     | 11 Griffiths Court, Rhyl, LL18 3YF                                                                              | Rachel May Cooper<br>11 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                    |  |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                     |                                                                                                                                |
|-----------------------------------------------------------------------------------------------------------------|-------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans                                                                                         | Description of Land                 | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |
| N/A                                                                                                             | 8 Cherry Close, Prestatyn, LL19 7DQ | Robert Howcroft<br>8 Cherry Close<br>PRESTATYN<br>LL19 7DQ                                                                     |
|                                                                                                                 |                                     | Amanda Jayne Howcroft<br>8 Cherry Close<br>PRESTATYN<br>LL19 7DQ                                                               |
| N/A                                                                                                             | 7 Cherry Close, Prestatyn, LL19 7DQ | Neil Wilkinson<br>55 Lysander Drive<br>Padgate<br>WARRINGTON<br>Cheshire<br>WA2 0GL                                            |
| N/A                                                                                                             | 10 Griffiths Court, Rhyl, LL18 3YF  | Oxford Critical Care Limited<br>264 Banbury Road<br>OXFORD<br>OX2 7DY                                                          |
| N/A                                                                                                             | 441 Rhyl Coast Road, Rhyl, LL18 3YE | David Joseph Weatherson<br>441 Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3YE                                                    |
| N/A                                                                                                             | 1 Griffiths Court, Rhyl, LL18 3YF   | Sarah Emily Jones<br>1 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                            |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                     |  |
| N/A                     | 11 Cherry Close, Dyserth Bay Estate, Prestatyn, LL19 7DQ                                                        | Janice Kathleen Grayston<br>11 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ<br>The Executor of the Estate of the Late Henry Edward Grayston<br>11 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ |  |
| N/A                     | 2 Griffiths Court, Rhyl, LL18 3YF                                                                               | Tina Michelle Jones<br>2 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                                                              |  |
| N/A                     | 6 Cherry Close, Prestatyn, LL19 7DQ                                                                             | Ruth McCarthy         6 Cherry Close         PRESTATYN         Clwyd         LL19 7DQ         Thomas McCarthy         6 Cherry Close         PRESTATYN         Clwyd         LL19 7DQ              |  |
| N/A                     | Griffiths Court, Rhyl Coast Road, Rhyl, LL18 3YE                                                                | Penrhyn Limited<br>10 Mostyn Street<br>LLANDUDNO<br>Gwynedd<br>LL30 2PS                                                                                                                            |  |
| N/A                     | 9 Griffiths Court, Rhyl, LL18 3YF                                                                               | Lucy Marie Tillotson<br>9 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                                                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
| N/A                     | 3 Griffiths Court, Rhyl, LL18 3YF                                                                               | Anna Louise Turton<br>3 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF<br>Joseph Frederick Gill                                  |  |
|                         |                                                                                                                 | 3 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                 |  |
| N/A                     | 4 Griffiths Court, Rhyl, LL18 3YF                                                                               | Paul Kenneth Howes<br>4 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                           |  |
| N/A                     | 5 Griffiths Court, Rhyl, LL18 3YF                                                                               | Alex John William Aspinwall<br>5 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                  |  |
|                         |                                                                                                                 | Holly May Spicer<br>5 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                             |  |
| N/A                     | 6 Griffiths Court, Rhyl, LL18 3YF                                                                               | Chelsey Newman<br>6 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                               |  |
| N/A                     | 12 Cherry Close, Prestatyn, LL19 7DQ                                                                            | Anne Shallcross<br>12 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                 |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                  |  |
| N/A                     | 8 Griffiths Court, Rhyl and parking space, LL18 3YF                                                             | Tracey Ann Rose<br>8 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                               |  |
| N/A                     | 7 Griffiths Court, Rhyl, LL18 3YF                                                                               | Elaine Hughes<br>7 Griffiths Court<br>RHYL<br>Clwyd<br>LL18 3YF                                                                                                 |  |
| N/A                     | 13 Cherry Close, Prestatyn, LL19 7DQ                                                                            | David Gwyn Bartley<br>69 Meliden Road<br>PRESTATYN<br>Clwyd<br>LL19 8RH                                                                                         |  |
| N/A                     | 105 Garnett Drive, Prestatyn, LL19 7DJ                                                                          | Ann Carolyn Minshull<br>105 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                                                     |  |
| N/A                     | 14 Cherry Close, Prestatyn, LL19 7DQ                                                                            | Agnes Margaret Wilkinson<br>14 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ<br>Malcolm Raymond Wilkinson<br>14 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ |  |
| N/A                     | 1 Cherry Close, Prestatyn, LL19 7DQ                                                                             | Norman Woodcock<br>1 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ                                                                                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
| N/A                     | 103 Garnett Drive, Prestatyn, LL19 7DJ                                                                          | Edna Marjorie Campbell<br>103 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ<br>Keith Campbell                                |  |
|                         |                                                                                                                 | 103 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                                            |  |
| N/A                     | 2 Cherry Close, Prestatyn, LL19 7DQ                                                                             | Lisa Marie Peard<br>2 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ                                                           |  |
| N/A                     | 101 Garnett Drive, Prestatyn, LL19 7DJ                                                                          | Edward Roy Thomas<br>101 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                       |  |
| N/A                     | 3 Cherry Close, Prestatyn, LL19 7DQ                                                                             | Jonathan Alan Massey<br>3 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ                                                       |  |
| N/A                     | 97 Garnett Drive, Prestatyn, LL19 7DJ                                                                           | Jane Holmes<br>97 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ<br>Robert Holmes<br>97 Garnett Drive                         |  |
|                         |                                                                                                                 | PRESTATYN<br>Clwyd<br>LL19 7DJ                                                                                                 |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                         |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                          |  |
| N/A                     | 99 Garnett Drive, Prestatyn, LL19 7DJ                                                                           |                                                                                                                                                         |  |
|                         |                                                                                                                 | Roy Finney<br>99 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                                                        |  |
| N/A                     | 95 Garnett Drive, Prestatyn, LL19 7DJ                                                                           | Joan Caroline Bowman<br>95 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ<br>John Richard Bowman<br>95 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ |  |
| N/A                     | 4 Cherry Close, Prestatyn, LL19 7DQ                                                                             | Beryl Ann Downie<br>4 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ<br>Thomas Downie<br>4 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ               |  |
| N/A                     | 5 Cherry Close, Prestatyn, LL19 7DQ                                                                             | Dorothy Joan Ferns<br>5 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ                                                                                  |  |

|                         | А                                     | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                         |
|-------------------------|---------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|
|                         |                                       | BOOK OF REFERENCE - PART 2                                                                                                     |
|                         |                                       | County of Denbighshire                                                                                                         |
| Number on<br>Land Plans | Description of Land                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |
| N/A                     | 93 Garnett Drive, Prestatyn, LL19 7DJ | Ray Woodcock<br>93 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ<br>Linne Woodcock                                           |
|                         |                                       | 93 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                                             |
| N/A                     | 91 Garnett Drive, Prestatyn, LL19 7DJ | Glynn Hughes<br>91 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                             |
|                         |                                       | Sandra Ann Hughes<br>91 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                        |
| N/A                     | 15 Cherry Close, Prestatyn, LL19 7DQ  | Allen Linforth<br>15 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ                                                            |
|                         |                                       | Susan Linforth<br>15 Cherry Close<br>PRESTATYN<br>Clwyd<br>LL19 7DQ                                                            |
| N/A                     | 87 Garnett Drive, Prestatyn, LL19 7DJ | Wayne David Harrison<br>87 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                     |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                           |  |
| N/A                     | 85 Garnett Drive, Prestatyn, LL19 7DJ                                                                           | Terence William Thomas<br>85 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ<br>Janet Thomas<br>85 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ       |  |
| N/A                     | 81 Garnett Drive, Dyserth Bay Estate, Prestatyn, LL19 7DJ                                                       | Edna Joy Stephens<br>81 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                                                  |  |
| N/A                     | 56 Garnett Drive, Prestatyn, LL19 7DN                                                                           | Glenys Elizabeth Downes<br>56 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN<br>Peter John Downes<br>56 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN |  |
| N/A                     | 54 Garnett Drive, Prestatyn, LL19 7DN                                                                           | Alan Frederick Taylor<br>54 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN<br>Elaine Taylor<br>54 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                       |  |
| N/A                     | 52 Garnett Drive, Prestatyn, LL19 7DN                                                                           | Arthur Roberts<br>52 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN                                                                                 |  |
| N/A                     | 79A Garnett Drive, Prestatyn, LL19 7DJ                                                                          | Sharon Denny<br>79A Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                                                  |  |
| N/A                     | 50 Garnett Drive, Prestatyn, LL19 7DN                                                                           | Judith Towler<br>50 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN                                                                                  |  |
| N/A                     | 48 Garnett Drive, Prestatyn, LL19 7DN                                                                           | Joan Baiton<br>48 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN<br>Lewis Baiton<br>48 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN              |  |
| N/A                     | 46 Garnett Drive, Prestatyn, LL19 7DN                                                                           | Richmond Westwell<br>46 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN<br>Carol Anne Westwell<br>46 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN |  |

|                         | A                                     | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                             |
|-------------------------|---------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                       | BOOK OF REFERENCE - PART 2<br>County of Denbighshire                                                                                                                                |
| Number on<br>Land Plans | Description of Land                   | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                      |
| N/A                     | 44 Garnett Drive, Prestatyn, LL19 7DN | Valerie Nadine Mitchell-Brown<br>6 Moreton Drive<br>Holmes Chapel<br>CREWE<br>Cheshire<br>CW4 7EL                                                                                   |
| N/A                     | 42 Garnett Drive, Prestatyn, LL19 7DN | John Anthony Swales<br>42 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN                                                                                                           |
| N/A                     | 79 Garnett Drive, Prestatyn, LL19 7DJ | Nancy Longley<br>79 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DJ                                                                                                                 |
| N/A                     | 40 Garnett Drive, Prestatyn, LL19 7DN | The Executor of the Estate of the Late Dorothy Ridgway<br>40 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN<br>Peter Ridgway<br>40 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN |
| N/A                     | 36 Garnett Drive, Prestatyn, LL19 7DN | Christine Alison Hindle<br>36 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN                                                                                                       |
| N/A cont'd              |                                       | Michael Hindle<br>36 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN                                                                                                                |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                      |  |
| N/A                     | 38 Garnett Drive, Prestatyn, LL19 7DN                                                                           | Alicia Bradley<br>38 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN                                                                                |  |
|                         |                                                                                                                 | John Lawrence Bradley<br>38 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN                                                                         |  |
| N/A                     | 34 Garnett Drive, Prestatyn, LL19 7DN                                                                           | Alan Cunningham<br>34 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN<br>Laura Ann Cunningham<br>34 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN |  |
| N/A                     | 8 Arfon Avenue, Prestatyn, LL19 7EN                                                                             | Sheila Irene Ford<br>8 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN<br>Stephen Michael Ford<br>8 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN   |  |
| N/A                     | 10 Arfon Avenue, Prestatyn, LL19 7EN                                                                            | Alan Colclough<br>c/o Debra Lloyd<br>10 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN                                                              |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |
| N/A                     | 12 Arfon Avenue, Prestatyn, LL19 7EN                                                                            | lan John Devenport<br>Woodlands<br>Bishopswood Road<br>PRESTATYN<br>Clwyd<br>LL19 9PL<br>Joanna Louise Devenport               |  |
|                         |                                                                                                                 | Woodlands<br>Bishopswood Road<br>PRESTATYN<br>Clwyd<br>LL19 9PL                                                                |  |
| N/A                     | 14 Arfon Avenue, Prestatyn, LL19 7EN                                                                            | John Holden<br>14 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN                                                               |  |
| N/A                     | 75 Garnett Drive, Prestatyn, LL19 7DL                                                                           | Beryl Newton<br>75 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                             |  |
|                         |                                                                                                                 | Jeffrey Newton<br>75 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                           |  |
| N/A                     | 32 Garnett Drive, Prestatyn, LL19 7DN                                                                           | John Dearden<br>32 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN                                                             |  |
| N/A                     | 16 Arfon Avenue, Prestatyn, LL19 7EN                                                                            | Judith Anne Burcham<br>16 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN                                                       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                       |  |
| N/A                     | 18 Arfon Avenue, Prestatyn, LL19 7EN                                                                            | Christine Michael<br>18 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN<br>Robert Michael<br>18 Arfon Avenue<br>PRESTATYN                             |  |
|                         |                                                                                                                 | Clwyd<br>LL19 7EN                                                                                                                                    |  |
| N/A                     | 20 Arfon Avenue, Prestatyn, LL19 7EN                                                                            | Edward George Beattie<br>20 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN<br>Wendy Ann Beattie<br>20 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN |  |
| N/A                     | 30 Garnett Drive, Prestatyn, LL19 7DN                                                                           | Stephanie Jane Powell<br>30 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DN                                                                          |  |
| N/A                     | 22 Arfon Avenue, Prestatyn, LL19 7EN                                                                            | Charles Barrie Yates<br>22 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN                                                                            |  |
| N/A                     | 73 Garnett Drive, Prestatyn, LL19 7DL                                                                           | Christine Moore<br>73 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                |  |

|                         | Awe                                                  | l y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                        |  |
|-------------------------|------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                             |  |
| Number on<br>Land Plans | Description of Land                                  | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                              |  |
|                         |                                                      | Keith George Moore<br>73 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                    |  |
| N/A                     | 71 Garnett Drive, Prestatyn, LL19 7DL                | Geoffrey Roden<br>71 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                        |  |
|                         |                                                      | Joyce Roden<br>71 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                           |  |
| N/A                     | 24 Arfon Avenue, Prestatyn, LL19 7EN                 | Michael William O'Neill<br>24 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN<br>Stephanie Emma O'Neill<br>24 Arfon Avenue<br>PRESTATYN<br>Clwyd<br>LL19 7EN |  |
| N/A                     | 69 Garnett Drive, Prestatyn, LL19 7DL                | Linda Corbett<br>69 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                         |  |
| N/A cont'd              |                                                      | Paul Corbett<br>69 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                                      |  |
| N/A                     | 67 Garnett Drive, Prestatyn, LL19 7DL                                                                           | Derrick Jeffrey<br>67 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                                                                               |  |
| N/A                     | 21 Arfon Avenue, Prestatyn, LL19 7EN                                                                            | Anne Margaret Pridding<br>4 Pendre Avenue<br>PRESTATYN<br>Clwyd<br>LL19 9SL<br>Ian Charles Pridding<br>4 Pendre Avenue<br>PRESTATYN<br>Clwyd<br>LL19 9SL                                                            |  |
| N/A                     | 65 Garnett Drive, Prestatyn, LL19 7DL                                                                           | Herbert William Winstanley         65 Garnett Drive         PRESTATYN         Clwyd         LL19 7DL         Lilian Antoinette Winstanley         65 Garnett Drive         PRESTATYN         Clwyd         LL19 7DL |  |
| N/A                     | 63 Garnett Drive, Prestatyn, LL19 7DL                                                                           | Adrian Davenport<br>63 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                                                                              |  |
| N/A cont'd              |                                                                                                                 | Beverley Davenport<br>63 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                                                                            |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                                           |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                            |  |
| N/A                     | 61 Garnett Drive, Prestatyn, LL19 7DL                                                                           | Julia Margaret Pierce<br>61 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                                               |  |
| N/A                     | 59 Garnett Drive, Prestatyn, LL19 7DL                                                                           | Alice Doreen Monaghan<br>59 Garnett Drive<br>PRESTATYN<br>Clwyd<br>LL19 7DL                                                                                                               |  |
| N/A                     | 89 Garnett Drive, Prestatyn, LL19 7DJ                                                                           | Charles Lair<br>89 Garnett Drive<br>PRESTATYN<br>LL19 7DJ<br>(as reputed owner)<br>Unknown                                                                                                |  |
| N/A                     | 83 Garnett Drive, Prestatyn, LL19 7DJ                                                                           | Jennifer Roberts<br>83 Garnett Drive<br>PRESTATYN<br>LL19 7DJ<br>(as reputed owner)<br>Stephen Alan Roberts<br>83 Garnett Drive<br>PRESTATYN<br>LL19 7DJ<br>(as reputed owner)<br>Unknown |  |
| N/A                     | 77 Garnett Drive, Prestatyn, LL19 7DJ                                                                           | Grace Alker<br>77 Garnett Drive<br>PRESTATYN<br>LL19 7DJ<br>(as reputed owner)                                                                                                            |  |
| N/A cont'd              |                                                                                                                 | Unknown                                                                                                                                                                                   |  |

|                         | Awe                                                       | l y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                   |
|-------------------------|-----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                           | BOOK OF REFERENCE - PART 2                                                                                                                                                                             |
|                         |                                                           | County of Denbighshire                                                                                                                                                                                 |
| Number on<br>Land Plans | Description of Land                                       | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                                                         |
| N/A                     | 9 Llys Penyffordd, Pentre Lane, Rhuddlan, Rhyl, LL18 6HJ  | Hilary Jean Douglas<br>9 Llys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6HJ<br>(as reputed owner)<br>Unknown                                                                             |
| N/A                     | 11 Llys Penyffordd, Pentre Lane, Rhuddlan, Rhyl, LL18 6HJ | Barbara Joyce Pierce<br>11 Llys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ<br>David Basil Pierce<br>11 Llys Penyffordd<br>Pentre Lane<br>Rhuddlan<br>RHYL<br>Clwyd<br>LL18 6HJ |
| N/A                     | Land adjoining 15 Ferguson Avenue, Prestatyn              | Julian Calland<br>15 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                                                                                          |
| N/A                     | 15 Ferguson Avenue, Prestatyn                             | Julian Calland<br>15 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                                                                                          |
| N/A                     | 13 Ferguson Avenue, Prestatyn, LL19 7YA                   | June Phyllis Naden<br>13 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                                                                                      |
| N/A                     | 11 Ferguson Avenue, Prestatyn, LL19 7YA                   | Esther Mary Davordjie<br>11 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                                                                                   |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                |  |
|                         |                                                                                                                 | Jerry Rankin Davordjie<br>11 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                         |  |
| N/A                     | 9 Ferguson Avenue, Prestatyn, LL19 7YA                                                                          | Susan Constance Marley<br>9 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA<br>Thomas Patrick Marley<br>9 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA   |  |
| N/A                     | Daysprings, Ferguson Avenue                                                                                     | Matthew David Humphreys<br>7 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                         |  |
| N/A                     | 5 Ferguson Avenue, Prestatyn, LL19 7YA                                                                          | Christopher Alexander Sheen<br>5 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA<br>Rachel Jayne Sheen<br>5 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA |  |
| N/A                     | 3 Ferguson Avenue, Prestatyn, LL19 7YA                                                                          | Lisa Clare Roberts<br>3 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA<br>Paul Antony Roberts<br>3 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA         |  |
| N/A                     | 1 Ferguson Avenue, Prestatyn, LL19 7YA                                                                          | Agnieszka Murnane<br>1 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                               |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                                             |  |
|                         |                                                                                                                 | Andrew Shaun Murnane<br>1 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                                                         |  |
| N/A                     | 260 Victoria Road, Prestatyn, LL19 7UU                                                                          | Pauline Valerie Taylor-Clague<br>260 Victoria Road<br>PRESTATYN<br>LL19 7UU                                                                                                |  |
| N/A                     | 262 Victoria Road, Prestatyn, LL19 7UU                                                                          | Gillian Lawrence<br>262 Victoria Road<br>PRESTATYN<br>LL19 7UU<br>Raymond Anthony Lawrence<br>262 Victoria Road<br>PRESTATYN<br>LL19 7UU                                   |  |
| N/A                     | 258 Victoria Road, Prestatyn, LL19 7UU                                                                          | lan Stacey Sherwood<br>258 Victoria Road<br>PRESTATYN<br>LL19 7UU<br>Janice Mary Sherwood<br>258 Victoria Road<br>PRESTATYN<br>LL19 7UU                                    |  |
| N/A                     | 2 Ferguson Avenue, Prestatyn                                                                                    | Adrian Garth Townsend<br>1 Coed Mor Drive<br>PRESTATYN<br>Denbighshire<br>LL19 9RA<br>Anne Christine Townsend<br>1 Coed Mor Drive<br>PRESTATYN<br>Denbighshire<br>LL19 9RA |  |
| N/A                     | 6 Ferguson Avenue, Prestatyn, LL19 7YA                                                                          | Alan Harry Groom<br>6 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                                                             |  |

|                         | A                                                    | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                         |  |  |
|-------------------------|------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                                |  |  |
| Number on<br>Land Plans | Description of Land                                  | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008                 |  |  |
|                         |                                                      | Hilda Irene Groom<br>6 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                                |  |  |
| N/A                     | 4 Ferguson Avenue                                    | Cynthia Cheshire<br>4 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA<br>Peter Bramwell Cheshire<br>4 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA        |  |  |
| N/A                     | 8 Ferguson Avenue, Prestatyn, LL19 7YA               | Jason Lewis Metcalfe<br>8 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                             |  |  |
| N/A                     | 10 Ferguson Avenue, Prestatyn, LL19 7YA              | Robert Jason Hayward<br>10 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA<br>Veronica Margaret Parker<br>10 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA |  |  |
| N/A                     | Ashbourne, 12 Ferguson Avenue, Prestatyn, LL19 7YA   | Doreen Hall<br>12 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                                     |  |  |
| N/A                     | 14 Ferguson Avenue, Prestatyn, LL19 7YA              | Brenda Ann Killip<br>14 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                               |  |  |
| N/A cont'd              |                                                      | George Robert Killip<br>14 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                                            |  |  |

| <u> </u>                | Awel v Môr                                                          | Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                   |  |  |
|-------------------------|---------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         |                                                                     | BOOK OF REFERENCE - PART 2                                                                                                     |  |  |
|                         | County of Denbighshire                                              |                                                                                                                                |  |  |
| Number on<br>Land Plans | Description of Land                                                 | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008 |  |  |
| N/A                     | Land adjoining 14 Ferguson Avenue, Prestatyn, LL19 7YA              | Brenda Ann Killip<br>14 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                               |  |  |
|                         |                                                                     | George Robert Killip<br>14 Ferguson Avenue<br>PRESTATYN<br>LL19 7YA                                                            |  |  |
| N/A                     | 164 Ffordd Idwal, Prestatyn, LL19 7US                               | George Alway Gibbs<br>5 Pearse Grove<br>Walton Park<br>MILTON KEYNES<br>MK7 7HD                                                |  |  |
| N/A                     | 162 Ffordd Idwal, Prestatyn, LL19 7US                               | Stephen Astle<br>2 Glyn Avenue<br>PRESTATYN<br>LL19 9NN                                                                        |  |  |
| N/A                     | 160 Ffordd Idwal, Prestatyn, LL19 7US                               | Stephen Astle<br>2 Glyn Avenue<br>PRESTATYN<br>LL19 9NN                                                                        |  |  |
| N/A                     | 158 Ffordd Idwal, Prestatyn, LL19 7US                               | Stephen Astle<br>2 Glyn Avenue<br>PRESTATYN<br>LL19 9NN                                                                        |  |  |
| N/A                     | Rhodfa Peris, Rhodfa Wyn, Rhodfa Padarn and Ffordd Idwal, Prestatyn | Wallace Estates Limited<br>Botanic House<br>100 Hills Road<br>CAMBRIDGE<br>CB2 1PH                                             |  |  |
| N/A                     | 156 Ffordd Idwal, Prestatyn, LL19 7US                               | Hollie Jane Hughes<br>156 Ffordd Idwal<br>PRESTATYN<br>LL19 7US                                                                |  |  |
| N/A cont'd              |                                                                     | Thomas Michael Bishop<br>156 Ffordd Idwal<br>PRESTATYN<br>LL19 7US                                                             |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 2<br>County of Denbighshire |                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008           |  |
| N/A                     | 154 Ffordd Idwal, Prestatyn, LL19 7US                                                                           | Terri Leanne Spargo-Thompson<br>154 Fford Idwal<br>PRESTATYN<br>LL19 7US                                                                 |  |
| N/A                     | 152 Ffordd Idwal, Prestatyn, LL19 7US                                                                           | Christine Ann Beaumont<br>152 Ffordd Idwal<br>PRESTATYN<br>LL19 7US<br>Simon Colin Beaumont<br>152 Ffordd Idwal<br>PRESTATYN<br>LL19 7US |  |
| N/A                     | 47 Rhodfa Wyn, Prestatyn, LL19 7UN                                                                              | Della Shoel<br>47 Rhodfa Wyn<br>PRESTATYN<br>LL19 7UN                                                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                 |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 1                       | 217151 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)    | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                                                                                 |  |
| 2                       | 111434 square metres of foreshore land seaward of the Mean High Water lying to<br>the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                |  |
| 4                       | 155 square metres of foreshore land seaward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown)            | Unknown                                                                                                                                                                                                                                                                                                                                                |  |
| 7                       | 34 square metres of foreshore land seaward of the Mean High Water at Rhyl<br>(excluding all interests of the Crown)                             | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                     |  |
| 8                       | 4615 square metres of footway (north of Rhyl Coast Road, A548) and footpath<br>(DE 207 11)                                                      | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)                                                                                                                                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                          |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                       |                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                                                                           |                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                                              | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 8 cont'd                |                                                                                                                                  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)<br>Unknown                                                   |  |
|                         |                                                                                                                                  | (in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                                                                                                                                                                                                             |  |
| 9                       | 2 square metres of footway (north of Rhyl Coast Road, A548)                                                                      | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                               |  |
| 10                      | 113 square metres of foreshore landward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 11                      | 12 square metres of slipway seaward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown)     | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 12                      | 86 square metres of slipway landward of Mean High Water north of Rhyl (excluding all interests of the Crown)                     | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 13                      | 345 square metres of footway and grassed area (north of Garford Road)                                                            | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                                                                                                                                                   |  |

| <b></b>                 | Awel v Môr Offe                                                                                           | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|                         | County of Denbighshire                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                                                                       | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                            |  |
| 13 cont'd               |                                                                                                           | The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 OTP<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                    |  |
| 14                      | 347 square metres of footway, hardstanding and grassed area (north of Garford Road)                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access) |  |
| 15                      | 3125 square metres of hardstanding and grassed area (north of Brynhedydd Bay)<br>and footpath (DE 207 11) | Rhyl Golf Club Limited         Rhyl Coast Road         RHYL         Clwyd         LL18 3RE         (in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of electricity apparatus)                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
|-------------------------|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | County of Denbighshire                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
| Number on<br>Land Plans | Description of Land                                                                       | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |
| 15 cont'd               |                                                                                           | Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
| 16                      | 2728 square metres of private road, buildings and hardstanding at Rhyl Golf Club,<br>Rhyl | AP Wireless II (UK) Limited<br>A P Wireless Ltd<br>16-18 Conduit Street<br>LICHFIELD<br>WS13 6JR<br>(in respect of rights stated in Lease dated 28th September 2012)<br>Cornerstone Telecommunications Infrastructure Limited<br>Hive 2<br>1330 Arlington Business Park<br>Theale<br>READING<br>Berkshire<br>RG7 45A<br>(in respect of rights granted by Lease dated 17th November 2017)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>EE Limited<br>Trident Place<br>Mosquito Way<br>HATFIELD<br>Hertfordshire<br>AL10 98W<br>(in respect of rights stated in Lease dated 31st March 2004)<br>Hutchison 3G UK Limited<br>450 Longwater Avenue<br>READING<br>RE2 6GF<br>(in respect of rights stated in Lease dated 31st March 2004) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER  |                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|----------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|                         |                                                          | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land                                      | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                               |  |
| 16 cont'd               |                                                          | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications apparatus)                 |  |
| 17                      | 12 square metres of hardstanding at Rhyl Golf Club, Rhyl | EE Limited<br>Trident Place<br>Mosquito Way<br>HATFIELD<br>Hertfordshire<br>AL10 9BW<br>(in respect of rights stated in Lease dated 31st March 2004)<br>Hutchison 3G UK Limited<br>450 Longwater Avenue<br>READING<br>RG2 6GF<br>(in respect of rights stated in Lease dated 31st March 2004)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 18                      | 206 square metres of grassed area at Rhyl Golf Club, Rhyl                                                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of right of access)                                                                                                                                                                                                                    |  |
| 19                      | 15 square metres of grassed area at Rhyl Golf Club, Rhyl                                                        | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of right of access)                                                                                                                                                                                                                    |  |
| 20                      | 123 square metres of private road at Rhyl Golf Club, Rhyl                                                       | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                   |  |
| 21                      | 483 square metres of footway of public road (north of Rhyl Coast Road, A548)                                    | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                 |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 21 cont'd               |                                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way                                                                                                                                                             |  |
|                         |                                                                                                                                 | North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                                                                                                                                                                                                     |  |
|                         |                                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                            |  |
| 22                      | 152117 square metres of grassed area and hardstanding (Rhyl Golf Club, north of Rhyl Coast Road, A548) and footpath (DE 207 11) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                               |  |
|                         |                                                                                                                                 | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)                                                                                                                                                            |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                          |                                                                                                                                                                                                                                                                                                                                                                                           |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                           |  |
|                         | County of Denbighshire                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                                                                                              | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                |  |
| 22 cont'd               |                                                                                                                                  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |  |
| 23                      | 34 square metres of foreshore landward of the Mean High Water lying to the north of Rhyl, (excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 24                      | 4159 square metres of foreshore landward of the Mean High Water and footway<br>(north of Rhyl Coast Road, A548)                  | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                                                        |  |
| 25                      | 15 square metres of hardstanding (north of Rhyl Coast Road, A548)                                                                | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)                                                                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                               |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                            |                                                                                                                                                                                                                                                                                                                                                        |  |
|                         | County of Denbighshire                                                                                                                |                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 25 cont'd               |                                                                                                                                       | Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                                                                                                                                                                                                        |  |
| 26                      | Plot No Longer Included In DCO                                                                                                        | Plot No Longer Included In DCO                                                                                                                                                                                                                                                                                                                         |  |
| 27                      | 1209 square metres of foreshore land seaward of the Mean High Water at Rhyl<br>(excluding all interests of the Crown)                 | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                     |  |
| 28                      | 16 square metres of foreshore land seaward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown)   | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                     |  |
| 29                      | 466 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                                                                                |  |
| 30                      | 8349 square metres of foreshore landward of the Mean High Water and footway<br>lying to the north of Rhyl                             | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                               |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                                  |                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 31                      | 218 square metres of foreshore land seaward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown)  | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                         |  |
| 32                      | 6074 square metres of foreshore land seaward of the Mean High Water lying to the north of Rhyl (excluding all interests of the Crown) | Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter as contained in a Lease dated 8th July 2009)<br>Unknown<br>(in respect of right of access and maintenance as contained in an Indenture dated 20th October 1883)                                         |  |
| 33                      | 63 square metres of footway (north of Rhyl Coast Road, A548)                                                                          | Unknown                                                                                                                                                                                                                                                                                                                                    |  |
| 34                      | 3534 square metres of foreshore landward of the Mean High Water and footway<br>(north of Rhyl Coast Road, A548)                       | Unknown<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)                                                                                                                                                                                   |  |
| 35                      | 9198 square metres of foreshore land seaward of the Mean High Water and footway at Prestatyn (excluding all interests of the Crown)   | Unknown                                                                                                                                                                                                                                                                                                                                    |  |
| 36                      | 160 square metres of footway (north of Rhyl Coast Road, A548)                                                                         | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Unknown                                                                                                                                                                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                             |  |
| 37                      | 275 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn                                                | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Unknown                                                                                                                                                                                                                                                                                                                     |  |
| 38                      | 379 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)     | Unknown<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                     |  |
| 39                      | 3649 square metres of foreshore land seaward of the Mean High Water lying to the<br>north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of access) |  |

|                         | Awel y Môr Offsl                                                                                                                                        | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|                         | County of Denbighshire                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                     | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                       |  |
| 40                      | 1228 square metres of foreshore landward of the Mean High Water and<br>hardstanding lying to the north of Prestatyn and footpath (DE 205 48, DE 205 49) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of access) |  |
| 41                      | 1013 square metres of footway (north of Rhyl Coast Road, A548) and footpath<br>(DE 205 48)                                                              | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Rhyl Golf Club Limited<br>Rhyl Coast Road<br>RHYL<br>Clwyd<br>LL18 3RE<br>(in respect of right to service media and right to enter)                                               |  |

|                         | Awel v Môr Offs                                                                                                                                            | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 41 cont'd               |                                                                                                                                                            | Unknown<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                       |  |
| 42                      | 35 square metres of footway (north of Rhyl Coast Road, A548)                                                                                               | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 43                      | 67 square metres of footway and foreshore landward of the Mean High Water<br>(north of Rhyl Coast Road, A548) and footpath (DE 205 49)                     | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 44                      | 47 square metres of hardstanding and foreshore land seaward of the Mean High Water (north of Rhyl Coast Road, A548) (excluding all interests of the Crown) | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 45                      | 183 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)                  | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                                                                             |  |
|                         |                                                                                                                                                            | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                      |  |
| 46                      | 28705 square metres of foreshore land seaward of the Mean High Water lying to<br>the north of Prestatyn, (excluding all interests of the Crown)            | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Diamond Transmission Partners BBE Limited                                                                                                                                                       |  |
|                         |                                                                                                                                                            | Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                                                                                                                        |  |
| 47                      | 73 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)                       | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                                                                           |  |
|                         |                                                                                                                                                            | Unknown                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offst                                                                                                                              | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                        |  |
|                         | County of Denbighshire                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 48                      | 4312 square metres of foreshore land seaward of the Mean High Water lying to the<br>north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)                                                                |  |
| 49                      | 7 square metres of footway seaward of the Mean High Water (north of Victoria<br>Road, A548) (excluding all interests of the Crown)            | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                    |  |
| 50                      | 17734 square metres of footway landward of the Mean High Water (north of Victoria Road, A548) and footpath (DE 205 49)                        | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                    |  |
| 51                      | 5853 square metres of footway (north of Victoria Road, A548) and footpath<br>(DE 205 49)                                                      | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                                                                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                            |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 51 cont'd               |                                                                                                                                            | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                |  |
| 52                      | 3329 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)     | Unknown                                                                                                                                                                                                                                                                                                                                    |  |
| 53                      | 56 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)       | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Unknown                                                                                                                                                                                          |  |
| 54                      | 1968 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                                                                     |  |
| 55                      | 1175 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                                                                     |  |
| 56                      | 167 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)      | Unknown                                                                                                                                                                                                                                                                                                                                    |  |
| 57                      | 2173 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                                                                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                    |                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                                                        | County of Denbighshire<br>Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 58                      | 1839 square metres of foreshore land seaward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                                                                                               |  |
| 59                      | 174 square metres of foreshore land seaward of the Mean High Water lying north of Prestatyn (excluding all interests of the Crown)         | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                                                                                               |  |
| 60                      | 26 square metres of foreshore landward of the Mean High Water lying north of Prestatyn (excluding all interests of the Crown)              | Unknown                                                                                                                                                                                                                                                                                                                                                              |  |
| 61                      | 15 square metres of foreshore landward of the Mean High Water lying to the north of Prestatyn (excluding all interests of the Crown)       | Unknown                                                                                                                                                                                                                                                                                                                                                              |  |
| 62                      | 795 square metres of footway and foreshore landward of the Mean High Water<br>(north of Victoria Road, A548)                               | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of easement dated 7th March 2005)                                     |  |
| 63                      | 155 square metres of slipway at Ffrith Beach landward of Mean High Water (north of Victoria Road, A548)                                    | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                          |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                             |                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                              | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 63 cont'd               |                                                                                                                  | Unknown<br>(in respect of rights granted by Deed of easement dated 7th March 2005)                                                                                                                                                                                                                                                               |  |
| 64                      | 94 square metres of hardstanding and foreshore landward of the Mean High Water<br>(north of Victoria Road, A548) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of easement dated 7th March 2005)                 |  |
| 65                      | 25 square metres of hardstanding and foreshore landward of the Mean High Water<br>(north of Victoria Road, A548) | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 66                      | 17 square metres of footway (north of Victoria Road, A548)                                                       | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 67                      | 53 square metres of grassed area (north of Victoria Road, A548)                                                  | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 68                      | 3178 square metres of grassed area (north of Victoria Road, A548)                                                | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed of easement dated 7th March 2005)<br>Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of easement dated 7th March 2005)                 |  |
| 69                      | 3442 square metres of grassed area (north of Victoria Road, A548)                                                | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)                                                                                                                                                                                              |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                        |  |  |
|-------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | County of Denbighshire                                                                |                                                                                                                                                                                                                                                                                                                                                        |  |  |
| Number on<br>Land Plans | Description of Land                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |  |
| 69 cont'd               |                                                                                       | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                    |  |  |
| 69a                     | Plot No Longer Included In DCO                                                        | Plot No Longer Included In DCO                                                                                                                                                                                                                                                                                                                         |  |  |
| 70                      | 45 square metres of footway (north of Victoria Road, A548)                            | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                    |  |  |
| 71                      | 55 square metres of grassed area (north of Victoria Road, A548)                       | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)                                                                                                                          |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                     |  |
| 71 cont'd               |                                                                                                     | Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                                                                                                                                                                                                                                         |  |
| 72                      | 891 square metres of grassed area and track (north of Victoria Road, A548)                          | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951) |  |
| 73                      | 57 square metres of grassed area at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                        |  |
| 74                      | 66 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548)      | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)                                                                                                                                                                                                                                                              |  |

|                         | Awel y Môr Offsl                                                              | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|-------------------------|-------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                               | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Number on<br>Land Plans | Description of Land                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                     |
| 74 cont'd               |                                                                               | Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 75                      | 2624 square metres of grassed area (north of Victoria Road, A548)             | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951) |
| 76                      | 469 square metres of private road and hardstanding (north of Ferguson Avenue) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                       |  |
| 76 cont'd               |                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951) |  |
| 77                      | 22 square metres of private road and verge (north of Ferguson Avenue)                                           | Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 78                      | 10 square metres of public road and verge (Ferguson Avenue)                                                     | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                           |  |
| 79                      | 52 square metres of public road (Ferguson Avenue)                                                               | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                                               |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER     |                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|                         | County of Denbighshire                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                         | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                |  |
| 79 cont'd               |                                                             | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                   |  |
|                         |                                                             | Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                                                                                                |  |
| 80                      | 24 square metres of private road (north of Ferguson Avenue) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| 81                      |                                                                                                                 | Denbighshire Leisure Limited<br>8-11 Trem Y Dyffryn<br>Colomendy Industrial Estate<br>DENBIGH<br>LL16 5TX<br>(in respect of rights stated in Lease)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Dwr Cymru Cyfwgedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Melions<br>CARDIFF<br>CF3 0LT<br>(in respect of vater apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>S PManweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 82                      | 225 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548)                 | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                             |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                    |  |
|                         |                                                                                                                 | Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)                                                                                                                                                                                                                                                            |  |
|                         |                                                                                                                 | Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                                                                                               |  |
| 83                      | 1746 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548)                | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)                                                                                                                                                                                                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                        |                                                                                                                                                                                                                                                                                                                                                        |  |  |
|-------------------------|------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                                                     |                                                                                                                                                                                                                                                                                                                                                        |  |  |
|                         | County of Denbighshire                                                                         |                                                                                                                                                                                                                                                                                                                                                        |  |  |
| Number on<br>Land Plans | Description of Land                                                                            | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |  |
| 83 cont'd               |                                                                                                | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                               |  |  |
|                         |                                                                                                | (in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                                                                                                                                                                       |  |  |
| 84                      | 27 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                    |  |  |
| 85                      | 19 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)                    |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 86                      | 16 square metres of footway at Ffrith Beach Festival Gardens (north of Victoria<br>Road, A548)                  | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of way)<br>Unknown<br>(in respect of restrictive covenants in Conveyance dated 14 February 1934)<br>Unknown<br>(in respect of restrictive covenants as contained in a Transfer dated 15th August 1951)  |  |
| 87                      | 30836 square metres of grassed area (Rhyl Golf Club, north of Rhyl Coast Road, A548)                            | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)                                                                                                                                                                                              |  |
| 88                      | 14607 square metres of grassed area and footway (Rhyl Golf Club, north of Rhyl<br>Coast Road, A548)             | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                              |  |
| 89                      | 361 square metres of public road (Rhyl Coast Road, A548)                                                        | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                                                                                                                                       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 89 cont'd               |                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way                                                                                                                                                             |  |
|                         |                                                                                                                 | North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                |  |
|                         |                                                                                                                 | Unknown<br>Wales and West Utilities Limited                                                                                                                                                                                                                                                                                                |  |
|                         |                                                                                                                 | Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                                |  |
| 90                      | 5 square metres of hardstanding (north of Rhyl Coast Road, A548)                                                | Unknown                                                                                                                                                                                                                                                                                                                                    |  |
| 91                      | 1363 square metres of public road (Rhyl Coast Road, A548)                                                       | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)                                                                                                                                                                                                     |  |
|                         |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 91 cont'd               |                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                     |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                    |  |
|                         |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                  |  |
| 92                      | 232 square metres of footway of public road (south of Rhyl Coast Road, A548)                                    | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to enter)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|                         |                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                             |  |
| 92 cont'd               |                                                                              | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWWDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |
| 93                      | 116 square metres of footway of public road (south of Rhyl Coast Road, A548) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                             |  |
| 93 cont'd               |                                                                                                            | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                                                                                                                                                                      |  |
| 94                      | 11 square metres of electricity substation (west of Sherwood Avenue and south of<br>Rhyl Coast Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for maintenance)<br>lain Alexander Douglas Blair Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as contained in a Conveyance dated 30th November 1906)<br>Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(in respect of restrictive covenants as contained in a Conveyance dated 16th October 1933)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
|-------------------------|------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| Number on<br>Land Plans | Description of Land                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                        |  |  |
| 94 cont'd               |                                                                              | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| 95                      | 808 square metres of footway of public road (south of Rhyl Coast Road, A548) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WCLH 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|                         | County of Denbighshire                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 96                      | 18635 square metres of buildings, caravans, hardstanding and grassed area (Robin<br>Hood Holiday Camp, Rhyl) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for maintenance)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Iain Alexander Douglas Blair Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as contained in a Conveyance dated 30th November 1906)<br>Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(in respect of restrictive covenants as contained in a Conveyance dated 16th October 1933)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of lecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights stated in Lease dated 14 May 1959) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                 |  |
| 96 cont'd               |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of way as contained in a Conveyance dated 10th December 1935)                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 97                      | 22566 square metres of buildings, caravans, hardstanding and grassed area (Robin<br>Hood Holiday Camp, Rhyl)    | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V GBA<br>(in respect of right of access for maintenance)<br>lain Alexander Douglas Blair Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as contained in a Conveyance dated 30th November 1906)<br>Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(in respect of restrictive covenants as contained in a Conveyance dated 16th October 1933)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire           |                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                       | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                           |  |
| 97 cont'd               |                                                                                                                           | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of way as contained in a Conveyance dated 10th December 1935)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                    |  |
| 98                      | 7994 square metres of private road, buildings, caravans, hardstanding and grassed<br>area (Robin Hood Holiday Camp, Rhyl) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access for maintenance)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 98 cont'd               |                                                                                                                 | lain Alexander Douglas Blair Cochrane<br>Lochnell Castle<br>Benderloch<br>OBAN<br>PA37 1QT<br>(in respect of right of drainage as contained in a Conveyance dated 30th November 1906)                                                                                                                                                |  |
|                         |                                                                                                                 | Network Rail Infrastructure Limited<br>1 Eversholt Street<br>LONDON<br>NW1 2DN<br>(in respect of restrictive covenants as contained in a Conveyance dated 16th October 1933)                                                                                                                                                         |  |
|                         |                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                         |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights stated in Lease dated 14 May 1959)                                                                                                                                                                     |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                        |  |
|                         |                                                                                                                 | Unknown<br>(in respect of right of way as contained in a Conveyance dated 10th December 1935)                                                                                                                                                                                                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                          |  |
| 98 cont'd               |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                     |  |
| 99                      | 2590 square metres of railway (Prestatyn and Rhyl) and hedgerow                                                 | Unknown                                                                                                                                                                                                                                                                                                                                                             |  |
| 100                     | 27103 square metres of agricultural land (north of The Cut and south of Rhyl Coast<br>Road, A548)               | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |  |

|                         | Awel v Môr Offs                                                                                  | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|--------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                       |                                                                                                                                                                                                                                                                                                                                                                                         |  |
|                         | County of Denbighshire                                                                           |                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land                                                                              | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                  |  |
| 101                     | 2759 square metres of agricultural land (north of The Cut and south of Rhyl Coast<br>Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |  |
| 102                     | 15 square metres of agricultural land (north of The Cut and south of Rhyl Coast<br>Road, A548)   | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |  |
| 103                     | 323 square metres of agricultural land (north of The Cut and south of Rhyl Coast<br>Road, A548)  | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964)                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                     |                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land                                                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                              |  |
| 104                     | 417 square metres of hedgerow, track and watercourse (The Cut) (north of Dyserth<br>Road, B5119 and south of Rhyl Coast Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |  |
| 105                     | 8131 square metres of agricultural land and watercourse (The Cut) (north of Dyserth Road, B5119 and south of Rhyl Coast Road, A548) | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959)                                                                                                                                                                |  |
| 106                     | 259 square metres of agricultural land (north of Dyserth Road, B5119 and south of The Cut)                                          | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959)                                                                                                                                                                |  |
| 107                     | 17 square metres of watercourse (The Cut) and banks (north of Dyserth Road, B5119)                                                  | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959)                                                                                                                                                                |  |

| l                       | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                            |  |
|                         |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 108                     | 1 square metre of agricultural land (north of Dyserth Road, B5119 and south of The<br>Cut)                      | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959)                                                                                                                   |  |
| 109                     | 1489 square metres of agricultural land (north of Dyserth Road, B5119 and south of The Cut)                     | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959)                                                                                                                   |  |
| 110                     | 470 square metres of hedgerow (north of Dyserth Road, B5119 and south of The Cut)                               | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Conveyance dated 1st May 1959)                                                                                                                   |  |
| 111                     | 2039 square metres of track (north of Dyserth Road, B5119)                                                      | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                              |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                               |  |
| 112                     | 6479 square metres of agricultural land and track (north of Dyserth Road, B511 and<br>south of Rhyl Coast Road, A548) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964) |  |
| 113                     | 2 square metres of agricultural land (north of Dyserth Road, B5119)                                                   | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)                                                                                                                                                                                                                                                                                                                                                                                  |  |

|                         | Awel v Môr                                                            | Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                     |  |  |
|-------------------------|-----------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                            |                                                                                                                                                                                                                                                                                                                                                  |  |  |
|                         | County of Denbighshire                                                |                                                                                                                                                                                                                                                                                                                                                  |  |  |
| Number on<br>Land Plans | Description of Land                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |  |
| 113 cont'd              |                                                                       | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                  |  |  |
| 114                     | 681 square metres of agricultural land (north of Dyserth Road, B5119) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                    |  |  |
| 115                     | 449 square metres of agricultural land (north of Dyserth Road, B5119) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                                                                     |  |  |
| 116                     | 3362 square metres of track (north of Dyserth Road, B5119)            | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                     |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|--------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|                         | County of Denbighshire                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                  | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                     |  |
| 116 cont'd              |                                                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access for maintenance as contained in an Agreement dated 14th February 1964)                                                                                   |  |
| 117                     | 2 square metres of wooded area (north of Dyserth Road, B5119)                        | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |
| 118                     | 21327 square metres of agricultural land and hedgerow (north of Dyserth Road, B5119) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                                                                                                                                                                               |  |

| <b></b>                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                       |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                    |                                                                                                                                                                                                                                                                                                                                                        |  |
|                         |                                                                                               | County of Denbighshire                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 118 cont'd              |                                                                                               | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                          |  |
| 119                     | 974 square metres of agricultural land (north of Dyserth Road, B5119)                         | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                                                                           |  |
| 120                     | 9265 square metres of agricultural land (north of Dyserth Road, B5119)                        | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)                                                                                                                                                                                           |  |
| 121                     | 2650 square metres of agricultural land, hedgerow and track (north of Dyserth<br>Road, B5119) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                           |  |
| 122                     | 2 square metres of wooded area (north of Dyserth Road, B5119)                                 | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)                                                                                                                                                                                                |  |

|                         | Awel y Môr (                                                          | Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|                         | County of Denbighshire                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                           |  |
| 122 cont'd              |                                                                       | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                                                                                                            |  |
| 123                     | 843 square metres of agricultural land (south of Dyserth Road, B5119) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights to lay and maintain services)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                               |  |
| 124                     | 113 square metres of wooded area (north of Dyserth Road, B5119)       | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Wales and West Utilities Limited<br>Wales a West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 125                     | 56 square metres of verge of public road (north of Dyserth Road, B5119)                                         | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                                                                                                                                                                          |  |
| 126                     | 21 square metres of track (north of Dyserth Road, B5119)                                                        | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                     |  |
| 127                     | 106 square metres of hedgerow (north of Dyserth Road, B5119)                                                    | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|---------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                   | County of Denbighshire Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 128                     | 305 square metres of agricultural land (north of Dyserth Road, B5119)                 | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                            |  |
| 129                     | 593 square metres of agricultural land (north of Dyserth Road, B5119)                 | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of easement granted dated 29 March 2017)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                            |  |
| 130                     | 150 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119)    | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights to lay and maintain apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                            |  |

|                         | Awel y Môr Offs                                                                               | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|                         | County of Denbighshire                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                 |  |
| 131                     | 444 square metres of public road (Dyserth Road, B5119) (excluding all interests of the Crown) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| 133                     | 677 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119)            | George Berwyn Lloyd Kerfoot<br>Henliys<br>Towyn Road<br>Towyn Road<br>UL22 9HF<br>(in respect of right to take water and use water pipe)<br>Iola Morris Kerfoot<br>Henliys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Comwy<br>LL22 9HF<br>(in respect of right to take water and use water pipe)<br>Jason Wayne Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to septic tank)<br>I eremy Charles Salisbury<br>Salisbury & Co Business Solutions Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 0RN<br>(in respect of right to take water and use water pipe) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                        |  |
| 133 cont'd              |                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Pamela Elaine Dakin<br>Greenacres<br>Dyserth Road<br>RHYL                                                                                                                                                                                                                               |  |
|                         |                                                                                                                 | LL18 4DY<br>(in respect of right of drainage to septic tank)<br>Unknown                                                                                                                                                                                                                                                                                                                                                                 |  |
| 134                     | 5 square metres of hedgerow (south of Dyserth Road, B5119)                                                      | George Berwyn Lloyd Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of right to take water and use water pipe)<br>Iola Morris Kerfoot<br>Henllys<br>Towyn Road<br>Towyn<br>ABERGELE<br>Conwy<br>LL22 9HF<br>(in respect of right to take water and use water pipe)<br>Jason Wayne Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to septic tank) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                             |  |
| 134 cont'd              |                                                                                                                 | Jeremy Charles Salisbury<br>Sailsbury & Co Business Solutions Ltd<br>Irish Square<br>Upper Denbigh Road<br>ST. ASAPH<br>Denbighshire<br>LL17 ORN<br>(in respect of right to take water and use water pipe)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Pamela Elaine Dakin<br>Greenacres<br>Dyserth Road<br>RHYL<br>LL18 4DY<br>(in respect of right of drainage to septic tank)<br>Unknown |  |
| 135                     | 19 square metres of verge of public road (south of Dyserth Road, B5119) and footpath (DE 206 5)                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                                                                                                                                                                                                                                                                                                                            |  |
| 136                     | 16 square metres of verge of public road (south of Dyserth Road, B5119) and footpath (DE 206 5)                 | Richard Gratton<br>8 Eldon Drive<br>ABERGELE<br>LL22 7BY<br>(in respect of right to passage of service media)                                                                                                                                                                                                                                                                                                                                                                                      |  |

|                         | Awel y Môr Offsh                                                                   | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                               |  |  |
|-------------------------|------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                                         |                                                                                                                                                                                                                                                                                                                                                        |  |  |
|                         | County of Denbighshire                                                             |                                                                                                                                                                                                                                                                                                                                                        |  |  |
| Number on<br>Land Plans | Description of Land                                                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |  |
| 136 cont'd              |                                                                                    | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline)                                                                                                                                                                               |  |  |
| 137                     | 50 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119)  | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                        |  |  |
| 138                     | 54 square metres of agricultural land (south of Dyserth Road, B5119)               | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                        |  |  |
| 139                     | 158 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights to lay and maintain services)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                            |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                                    | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 140                     | 32097 square metres of agricultural land, hedgerow and track (south of Dyserth<br>Road, B5119) and footpath (DE 206 5) | Centrica PLC<br>Milistream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right of access for maintenance)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WCLH 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline stated in deed dated 19 May 1955) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                       |  |
| 141                     | 2528 square metres of agricultural land (south of Dyserth Road, B5119)                                          | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                       |  |
| 142                     | 3651 square metres of agricultural land and track (south of Dyserth Road, B5119)<br>and footpath (DE 206 5)     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right of access for maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of ga apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                    | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                              |  |
| 142 cont'd              |                                                                                                                        | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline stated in deed dated 19 May 1955)                                                                                                                                                                                                                                                                                           |  |
| 143                     | 14073 square metres of agricultural land, hedgerow and track (south of Dyserth<br>Road, B5119) and footpath (DE 206 5) | Centrica PLC<br>Milistream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right of access for maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales 3 & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline stated in deed dated 19 May 1955) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                       |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                          |                                                                                                                                                                                                                                                                                                                                                  |  |
|                         |                                                                                               |                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 144                     | 6563 square metres of agricultural land and hedgerow (south of Dyserth Road, B5119)           | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right of access for maintenance)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline stated in deed dated 19 May 1955) |  |
| 145                     | 206 square metres of agricultural land (south of Dyserth Road, B5119)                         | Wales and West Utilities LimitedWales & West HouseSpooner CloseCoedkernewCASNEWYDDNP10 8FZ(in respect of gas apparatus)Wales and West Utilities LimitedWales & West HouseSpooner CloseCoedkernewCASNEWYDDNP10 8FZ(in respect of gas apparatus)                                                                                                   |  |
| 146                     | 4674 square metres of agricultural land, hedgerow and ditch (south of Dyserth<br>Road, B5119) | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                  |  |
| 147                     | 149 square metres of hedgerow (south of Dyserth Road, B5119)                                  | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 148                     | 21 square metres of hedgerow (south of Dyserth Road, B5119)                                   | Unknown                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                              |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                           |                                                                                                                                                                                                                                                                                                                                            |  |
|                         | County of Denbighshire                                                                                               |                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                                  | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 149                     | 55 square metres of hedgerow (south of Dyserth Road, B5119)                                                          | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                            |  |
| 150                     | 35 square metres of agricultural land (south of Dyserth Road, B5119)                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                            |  |
| 151                     | 123 square metres of agricultural land (south of Dyserth Road, B5119) and footpath<br>(DE 206 3)                     | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                            |  |
| 152                     | 10 square metres of agricultural land (south of Dyserth Road, B5119)                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                            |  |
| 153                     | 1494 square metres of agricultural land (south of Dyserth Road, B5119) and footpaths (DE 206 4, DE 206 6, DE 206 11) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                              |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                         | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                   |  |
| 153 cont'd              |                                                                                                                                                             | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 154                     | 3934 square metres of agricultural land, hedgerow, track and private road (south of<br>Dyserth Road, B5119) and footpaths (DE 206 12, DE 206 13, DE 206 15) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)                                                                                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                 |  |
| 154 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 155                     | 1166 square metres of private road (south of Dyserth Road, B5119) and footpaths<br>(DE 204 7, DE 206 15a)       | Gwilym Howatson<br>Aberkinsey Farm<br>Dyserth<br>Y RHYL<br>LL18 6BL<br>(in respect of right of access)<br>The Bodrhyddan Farming Company Limited<br>Bodrhyddan<br>Rhuddlan<br>Clwyd<br>LL18 5SB<br>(in respect of right of access)<br>Unknown                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                        |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                     |                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                                                         |                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                            | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 155 cont'd              |                                                                                                                | Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications apparatus)                                                                                                                                                                                                                       |  |
| 156                     | 3178 square metres of agricultural land and hedgerow (south of Dyserth Road,<br>B5119) and footpath (DE 206 3) | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)               |  |
| 157                     | 5902 square metres of agricultural land and hedgerow (south of Dyserth Road,<br>B5119) and footpath (DE 206 3) | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)               |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                        |                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                          | County of Denbighshire Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 158                     | 2128 square metres of agricultural land and shrubbery (south of Dyserth Road, B5119) and footpath (DE 206 3) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                                           |  |
|                         |                                                                                                              | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                                                                      |  |
|                         |                                                                                                              | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                         |  |
|                         |                                                                                                              | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                         |  |
| 159                     | 1535 square metres of agricultural land (south of Dyserth Road, B5119)                                       | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                         |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|--------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|                         | County of Denbighshire                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                  | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                             |  |
| 159 cont'd              |                                                                                      | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                                                                                                                        |  |
| 160                     | 84 square metres of hedgerow (south of Dyserth Road, B5119)                          | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 161                     | 1552 square metres of agricultural land and shrubbery (south of Dyserth Road, B5119) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                      |                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                  | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009            |  |
| 161 cont'd              |                                                                                                                                                      | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 162                     | 48411 square metres of agricultural land, hedgerow and shrubbery (north of Pentre<br>Lane and south of Dyserth Road, B5119) and footpath (DE 206 44) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                         |  |
|                         |                                                                                                                                                      | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Wales and West Utilities Limited                                                                                                                                                |  |
|                         |                                                                                                                                                      | Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                                           |  |

| <b> </b>                | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
|                         | County of Denbighshire                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
| Number on<br>Land Plans | Description of Land                                                                                              | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                 |  |  |
| 162 cont'd              |                                                                                                                  | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                                                                                                                        |  |  |
| 163                     | 12537 square metres of agricultural land and hedgerow (north of Pentre Lane and<br>south of Dyserth Road, B5119) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 SGD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CHS 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |  |
| 164                     | 254 square metres of agricultural land (north of Pentre Lane)                                                    | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                                                                                                                                                                          |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 164 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                                   |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                      |  |
| 165                     | 1876 square metres of agricultural land and hedgerow (north of Pentre Lane)                                     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                        |  |
|                         |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                                   |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                               |                                                                                                                                                                                                                                                                                                                                                        |  |  |
|-------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                                            |                                                                                                                                                                                                                                                                                                                                                        |  |  |
|                         | County of Denbighshire                                                                |                                                                                                                                                                                                                                                                                                                                                        |  |  |
| Number on<br>Land Plans | Description of Land                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |  |
| 166                     | 11721 square metres of agricultural land, hedgerow and pond (north of Pentre<br>Lane) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                          |  |  |
|                         |                                                                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT                                                                      |  |  |
| 167                     | 358 square metres of agricultural land (north of Pentre Lane)                         | Centrica PLC         Millstream         Maidenhead Road         WINDSOR         Berkshire         SL4 5GD         (in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                            |  |  |
|                         |                                                                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                                                     |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
|-------------------------|------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
|                         | County of Denbighshire                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
| Number on<br>Land Plans | Description of Land                                                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                 |  |  |
| 167 cont'd              |                                                                                    | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                                                                                                                        |  |  |
| 168                     | 245 square metres of agricultural land (north of Pentre Lane)                      | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CHS 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |  |
| 169                     | 4031 square metres of agricultural land, track and hedgerow (north of Pentre Lane) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                                                                                                                                                                          |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 169 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                                                     |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                        |  |
| 170                     | 776 square metres of agricultural land and hedgerow (north of Pentre Lane and south of Bryn Cwnin Road)         | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                          |  |
|                         |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                                                     |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                           |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                        |                                                                                                                                                                                                                                                                                                                                            |  |
|                         | County of Denbighshire                                                                                                            |                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                                               | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 171                     | 573 square metres of agricultural land (north of Pentre Lane and south of Bryn<br>Cwnin Road)                                     | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                              |  |
|                         |                                                                                                                                   | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House                                                                                                                    |  |
|                         |                                                                                                                                   | Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                 |  |
| 172                     | 4649 square metres of agricultural land and hedgerow (north of Pentre Lane and south of Bryn Cwnin Road) and footpath (DE 206 44) | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan                                                                                                                                             |  |
|                         |                                                                                                                                   | Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                                                                                                                                                                                                                                                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                          |  |
| 172 cont'd              |                                                                                                                 | Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                                                                                                                                                                                                                            |  |
| 173                     | 156 square metres of agricultural land and hedgerow (south of Bryn Cwnin Road)                                  | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)<br>Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|                         | County of Denbighshire                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                                                       | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| 174                     | 1251 square metres of agricultural land, track and woodland (south of Bryn Cwnin<br>Road) | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RTUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access)<br>Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)<br>Openreach Limited<br>Keivin House<br>123 Judd Street<br>LONDON<br>WCIH 9NP<br>(in respect of telecommunications apparatus) |  |

|                         | Awel v Môr Offst                                                                              | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                    |                                                                                                                                                                                                                                                                                                                                                       |  |
|                         | County of Denbighshire                                                                        |                                                                                                                                                                                                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009      |  |
| 175                     | 227 square metres of agricultural land (north of Pentre Lane and south of Bryn<br>Cwnin Road) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                         |  |
|                         |                                                                                               | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |
| 176                     | 43 square metres of agricultural land (north of Pentre Lane and south of Bryn Cwnin<br>Road)  | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | County of Denbighshire                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
| Number on<br>Land Plans | Description of Land                                                                                            | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                   |  |  |
| 176 cont'd              |                                                                                                                | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                                                                                                                                                        |  |  |
| 177                     | 5302 square metres of agricultural land and wooded area (south of Bryn Cwnin<br>Road) and footpath (DE 206 44) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021) |  |  |
| 178                     | 10 square metres of public road (south of Bryn Cwnin Road)                                                     | Unknown<br>Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications apparatus)                                                                                                                                                                                                                                                                                                                                                                  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|---------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|                         | County of Denbighshire                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| Number on<br>Land Plans | Description of Land                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                            |  |
| 179                     | 5 square metres of footway of public road (south of Bryn Cwnin Road)            | Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| 180                     | 2 square metres of footway of public road (south of Bryn Cwnin Road)            | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 181                     | 1137 square metres of agricultural land and hedgerow (south of Bryn Cwnin Road) | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access)<br>Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access) |  |
| 182                     | 1694 square metres of agricultural land (south of Bryn Cwnin Road)              | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 182 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                                               |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                  |  |
| 183                     | 447 square metres of track (south of Bryn Cwnin Road)                                                           | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                    |  |
|                         |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                                               |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                  |  |

| <u> </u>                | Awel v N                                                     | Λôr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|--------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|                         |                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| Number on<br>Land Plans | Description of Land                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                  |  |
| 184                     | 7765 square metres of wooded area (south of Bryn Cwnin Road) | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>L118 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>L118 6LA<br>(in respect of right of access)<br>Unknown<br>(in respect of right of access)<br>Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced)<br>Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>L118 6LA<br>(in respect of right of access) |  |
| 185                     | 77 square metres of wooded area (south of Bryn Cwnin Road)   | Caroline Morris<br>Office Tan Y Bryn<br>Pentre Lane<br>RHUDDLAN<br>Y Rhyl<br>LL18 6LA<br>(in respect of right of access)<br>Christopher Jenner<br>1 Cwrt Bryn Cwnin<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 185 cont'd              |                                                                                                                 | Unknown<br>(in respect to such covenants and easements as may have been imposed thereon before 5 July 2006 and are still subsisting and<br>capable of being enforced)                                                                                                                                                                                  |  |
|                         |                                                                                                                 | Wynne Randles<br>Tan Y Bryn<br>Pentre Lane<br>Rhuddlan<br>Y RHYL<br>LL18 6LA<br>(in respect of right of access)                                                                                                                                                                                                                                        |  |
| 186                     | 5551 square metres of agricultural land and wooded area (south of Bryn Cwnin<br>Road) and footpath (DE 206 17)  | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                          |  |
|                         |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                                                                                                                                                                     |  |
|                         |                                                                                                                 | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                        |  |

|                         | Awel v Môr Offs                                                                                                         | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                              |                                                                                                                                                                                                                                                                                                                                                        |  |
|                         | County of Denbighshire                                                                                                  |                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                                                     | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 187                     | 8968 square metres of agricultural land and hedgerow (south of Bryn Cwnin Road)<br>and footpaths (DE 206 17, DE 206 18) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)                                                                                                                                                                                                          |  |
|                         |                                                                                                                         | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)  |  |
| 188                     | 517 square metres of agricultural land (south of Bryn Cwnin Road, north of Pentre<br>and east of A525)                  | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire     |                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009          |  |
| 188 cont'd              |                                                                                                                     | Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                                                                                                                                                 |  |
| 189                     | 338 square metres of agricultural land and hedgerow (south of Bryn Cwnin Road,<br>north of Pentre and east of A525) | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage) |  |
| 192                     | 477 square metres of agricultural land (east of A525)                                                               | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                 |  |
| 192 cont'd              |                                                                                                                 | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                                                                                                                          |  |
| 193                     | 704 square metres of hedgerow (east of A525)                                                                    | Paul Raymond Cook         Cwybr Bach         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2VD         (in respect of right to service media and right to enter)         Sandra Louise Cook         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2VD         (in respect of right to service media and right to enter)         Sandra Louise Cook         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2YD         (in respect of right to service media and right to enter)         Unknown         (in respect of right of drainage) |  |

|                         | Awel v Môr                                                           | Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|----------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                      | BOOK OF REFERENCE - PART 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| Number on<br>Land Plans | Description of Land                                                  | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                            |  |
| 194                     | 15513 square metres of agricultural land and hedgerow (east of A525) | Openreach Limited<br>British TelecomKelvin House123 Judd StreetLONDONWCLH 9NP(in respect of telecommunication apparatus)Paul Raymond CookCwybr BachCwybrRhuddlanRHVLDenbighshireL18 ZYD(in respect of right to service media and right to enter)Sandra Louise CookCwybrRhuddlanRHYLDenbighshireL18 ZYD(in respect of right to service media and right to enter)Sandra Louise CookCwybrRhuddlanRHYLDenbighshireL18 ZYD(in respect of right to service media and right to enter)SP Manweb PLC3 Prenton WayNorth Cheshire Trading EstatePRENTONMerseysideCH43 3ET(in respect of electricity apparatus)Unknown(in respect of right to drainage) |  |
| 195                     | 2316 square metres of agricultural land (east of A547 and A525)      | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                      |                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| County of Denbighshire  |                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                |  |
| 195 cont'd              |                                                                                                            | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                          |  |
| 196                     | 612 square metres of agricultural land (east of A547 and north of Pentre Lane) and<br>footpath (DE 206 46) | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage) |  |
| 197                     | 21252 square metres of agricultural land and hedgerow (east of A547) footpaths<br>(DE 206 20, DE 206 46)   | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                   |                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                                                     | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009          |  |
| 197 cont'd              |                                                                                                                         | Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of drainage)         |  |
| 198                     | 373 square metres of agricultural land and hedgerow (north of Pentre Lane and east<br>of A547) and footpath (DE 206 20) | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage) |  |

| <u> </u>                | Awel v Môr Offs                                                                                                       | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                             |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                       | BOOK OF REFERENCE - PART 3                                                                                                                                                                                                                                                                                                           |  |
|                         | County of Denbighshire                                                                                                |                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 200                     | 2829 square metres of agricultural land and hedgerow (north of Pentre Lane and east of A547) and footpath (DE 206 20) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 17th May 1991)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St Mellons                                                                                                |  |
|                         |                                                                                                                       | St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain apparatus)<br>Anwyl Construction Company Limited<br>Anwyl House<br>Clos Dewi Sant<br>Ewloe<br>DEESIDE<br>Conwy<br>CH5 3DT<br>(in respect of rights stated in Transfer dated 01 December 2021)                                                          |  |
| 201                     | 97 square metres of agricultural land and hedgerow (north of Pentre Lane and east<br>of A547)                         | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                      |  |

|                         | Awel y Môr Offsh                                                                               | ore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                                                                                                                                                                                                                                                                           |  |
|-------------------------|------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                         |                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                            | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                      |  |
| 201 cont'd              |                                                                                                | Unknown<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                                                    |  |
| 202                     | 122 square metres of agricultural land and hedgerow (north of Pentre Lane and east<br>of A547) | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Unknown<br>(in respect of right of drainage) |  |
| 203                     | 8 square metres of hedgerow (west of Pentre Lane, Rhyl)                                        | Unknown                                                                                                                                                                                                                                                                                                                                                         |  |
| 208                     | 4 square metres of hedgerow (west of Pentre Lane and east of A547)                             | Unknown                                                                                                                                                                                                                                                                                                                                                         |  |
| 209                     | 3106 square metres of agricultural land (east of A547 and north of Afon Ffyddion)              | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 209 cont'd              |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                    |  |
| 210                     | 39 square metres of verge of public road (east of A547)                                                         | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Unknown                                                                                                                                                                                        |  |
| 211                     | 19 square metres of verge of public road (east of A547)                                                         | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                   |  |
| 212                     | 751 square metres of grassed area public road (east of A547)                                                    | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                   |  |

| <u> </u>                | ۵wel v Môr Off                                                           | shore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|--------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
|                         | County of Denbighshire                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                                      | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                          |  |
| 212 cont'd              |                                                                          | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications apparatus) |  |
| 213                     | 10 square metres of grassed area (east of A547) and footpath (DE 206 46) | Paul Raymond Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)<br>Sandra Louise Cook<br>Cwybr Bach<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right to service media and right to enter)                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER    |                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|-------------------------|------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         | BOOK OF REFERENCE - PART 3                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|                         | County of Denbighshire                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
| Number on<br>Land Plans | Description of Land                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                |  |  |  |
| 213 cont'd              |                                                            | Unknown<br>(in respect of right of drainage)                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |  |
| 214                     | 17 square metres of public road (east of A547)             | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                    |  |  |  |
| 216                     | 1449 square metres of public road and verge (east of A525) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER   |                                                                                                                                                                                                                                                                                                                                            |  |  |
|-------------------------|-----------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                |                                                                                                                                                                                                                                                                                                                                            |  |  |
|                         | County of Denbighshire                                    |                                                                                                                                                                                                                                                                                                                                            |  |  |
| Number on<br>Land Plans | Description of Land                                       | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |  |
| 216 cont'd              |                                                           | Unknown<br>Wales and West Utilities Limited                                                                                                                                                                                                                                                                                                |  |  |
|                         |                                                           | Wales and West Countes Enhited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                              |  |  |
| 217                     | 2828 square metres of grassed area and public road (A525) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                               |  |  |
| 218                     | 7 square metres of grassed and wooded area (west of A525) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                                     |  |  |
| 219                     | 528 square metres of footway and verge (west of A525)     | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                              |  |  |

|                         | Awel y                                                          | Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                       |  |  |  |
|-------------------------|-----------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         | BOOK OF REFERENCE - PART 3                                      |                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
|                         | County of Denbighshire                                          |                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| Number on<br>Land Plans | Description of Land                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |  |  |
| 220                     | 317 square metres of agricultural land and track (west of A525) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                    |  |  |  |
|                         |                                                                 | Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                       |  |  |  |
| 221                     | 245 square metres of agricultural land (west of A525)           | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                                                 |  |  |  |
| 222                     | 1 square metre of track (west of A525)                          | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                          |  |  |  |
| 223                     | 3 square metres of footway and track (west of A525)             | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                           |  |  |  |

|                         | Awel                                                        | y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                             |  |  |  |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         | BOOK OF REFERENCE - PART 3                                  |                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
|                         | County of Denbighshire                                      |                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| Number on<br>Land Plans | Description of Land                                         | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009         |  |  |  |
| 224                     | 1 square metre of footway (west of A525)                    | Unknown                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 225                     | 3 square metres of footway (west of A525)                   | Unknown                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 226                     | 25 square metres of footway and track (west of A525)        | Openreach Limited         British Telecom         Kelvin House         123 Judd Street         LONDON         WC1H 9NP         (in respect of telecommunication apparatus)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of electricity apparatus) |  |  |  |
| 227                     | 4 square metres of footway (west of A525)                   | Unknown                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| 228                     | 16 square metres of footway and grassed area (west of A525) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                   |  |  |  |
| 229                     | 10 square metres of grassed area (west of A525)             | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                   |  |  |  |
| 231                     | 1 square metre of grassed area (west of A525)               | Unknown                                                                                                                                                                                                                                                                                                                                                        |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                |                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 232                     | 58 square metres of verge and public road (A525)                                      | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                        |  |
| 233                     | 105 square metres of public road (A525) and footpath (DE 206 42)                      | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                        |  |
| 234                     | 97 square metres of public road (A525)                                                | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                   |  |

| <u> </u>                | Awel y Môr Offs                                                            | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|----------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|                         |                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                |  |
| 234 cont'd              |                                                                            | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                                                                                                                                            |  |
| 235                     | 1159 square metres of private road (west of A525) and footpath (DE 206 42) | Aaron Haldane         The Maisonette         Cwybr House         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Anthony Ashworth         14 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Anthony Ashworth         14 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Anthony Roberts         Flat 1         Cwybr House         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 235 cont'd              |                                                                                                                 | Christopher George Roberts<br>1 Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Helen McCahill<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>James Keith Nixon<br>Fiat 3<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |

|                         |                            | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|----------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|                         |                            | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| 235 cont'd              |                            | lames Massey<br>Fit 2<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 27D<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Nuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Kristy Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>House<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Kristy Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr<br>House<br>Cwybr |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                   |  |
| 235 cont'd              |                                                                                                                 | Martin Nixon<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHV<br>LL18 ZYD<br>(in respect of right of access)<br>Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 ZYD<br>(in respect of right of access)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WCL19 Syn<br>(in respect of telecommunication apparatus)<br>Peter Baddeley<br>2 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 ZYD<br>(in respect of right of access)<br>Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 ZYD<br>(in respect of right of access) |  |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|-----------------------------------------------------------------------------------------------------------------|---------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans                                                                                         | Description of Land | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 235 cont'd                                                                                                      |                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Susan Davies<br>S Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MX7 GIP<br>(in respect of right of access) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009        |  |
| 236                     | 14 square metres of private road (west of A525)                                                                 | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 SGD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown<br>(in respect of restrictive covenants)  |  |
| 237                     | 8537 square metres of agricultural land (west of A525)                                                          | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|--------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                      | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 238                     | 18 square metres of private road (west of A525) and footpath (DE 206 42) | Aaron Haldane         The Maisonette         Cwybr         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Anthony Ashworth         14 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Anthony Roberts         Flat 1         Cwybr         Cwybr House         Cwybr         Cwybr House         Cwybr         Ital 3 YD         (in respect of right of access)         Anthony Roberts         Flat 1         Cwybr House         Cwybr         Cwybr House         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Christopher George Roberts         1 Cwybr Uchaf Farm         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Helen McCahill         13 Cwybr Cottages |  |

|                         | Awel y Môr O        | ffshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|-------------------------|---------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                     | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| Number on<br>Land Plans | Description of Land | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                 |
| 238 cont'd              |                     | In Towers11 Cwybr CottagesCwybrRhuddlanY RHYLLL18 2VDJack Christopher Mousley9 Cwybr CottagesCwybrRhuddlanY RHYLLL18 2VD(in respect of right of access)James Keith NixonFlat 3CwybrRhuddlanY RHYLLL18 2VD(in respect of right of access)James Keith NixonFlat 3Cwybr HouseCwybrRhuddlanY RHYLLL18 2VD(in respect of right of access)James Keith Shaw1 Cwybr VillaCwybrRhuddlanY RHYLLL18 2VD(in respect of right of access)James Keith Shaw1 Cwybr VillaCwybrRhuddlanY RHYLLL18 2VD(in respect of right of access)James MasseyFlat 2Cwybr HouseCwybrRhuddlanY RHYLLL18 2VD(in respect of right of access)James MasseyFlat 2Cwybr HouseCwybrRhuddlanY RHYLLL18 2VD(in respect of right of access) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| 238 cont'd              |                                                                                                                 | John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L118 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L118 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L118 2YD<br>(in respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L118 2YD<br>(in respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L118 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L118 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L118 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L118 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L118 2YD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 238 cont'd              |                                                                                                                 | Martin Nixon<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                        |  |
|                         |                                                                                                                 | Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                           |  |
|                         |                                                                                                                 | Peter Baddeley<br>2 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                              |  |
|                         |                                                                                                                 | Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                           |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of rights granted by Deed of grant dated 20 December 2017)                                                                                                                                                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| 238 cont'd              |                                                                                                                 | Stepher Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2VD<br>(in respect of right of access)<br>Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2VD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2VD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2VD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 GiP<br>(in respect of right of access)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
| 239                     | 349 square metres of private road (west of A525 and north-east of Cwybr-uchaf)<br>and footpath (DE 206 42)      | Aaron Haldane         The Maisonette         Cwybr         Rhuddlan         Y RHYL         L18 2YD         (in respect of right of access)         Anthony Ashworth         14 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         L18 2YD         (in respect of right of access)         Anthony Roberts         Flat 1         Cwybr         Rhuddlan         Y RHYL         L18 2YD         (in respect of right of access)         Anthony Roberts         Flat 1         Cwybr         Rhuddlan         Y RHYL         L18 2YD         (in respect of right of access)         Christopher George Roberts         1 Cwybr Uchaf Farm         Cwybr         Rhuddlan         Y RHYL         L18 2YD         (in respect of right of access)         Helen McCahill         13 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         L18 2YD         (in respect of right of access)         Helen McCahill </td |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                       | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| 239 cont'd              |                                                                                       | Ian Towers         11 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         L13 27D         (in respect of right of access)         Jack Christopher Mousley         9 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         L13 27D         (in respect of right of access)         James Keith Nixon         Flat 3         Cwybr         Rhuddlan         Y RHYL         L13 27D         (in respect of right of access)         James Keith Nixon         Flat 3         Cwybr         Rhuddlan         Y RHYL         L13 27D         (in respect of right of access)         James Keith Shaw         1 Cwybr Villa         Cwybr         Rhuddlan         Y RHYL         L13 27D         (in respect of right of access)         James Keith Shaw         1 Cwybr Villa         Cwybr         Rhuddlan         Y RHYL         L13 27D         (in respect of right of access)         James Massey |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 239 cont'd              |                                                                                                                 | John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL13 2YD<br>(in respect of right of access)<br>Keith Jones<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL13 2YD<br>(in respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL13 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL13 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL13 2YD<br>(in respect of right of access)<br>Markt Nicholls<br>6 Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL13 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL13 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL13 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL13 2YD<br>(in respect of right of access) |  |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|-----------------------------------------------------------------------------------------------------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans                                                                                         | Description of Land | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                         |
| 239 cont'd                                                                                                      |                     | Matt Hession2 Gwybr Uchaf FarmCwybrRhuddanY RHYLL13 2 YD(in respect of right of access)Peter Baddeley2 Cwybr VilaCwybrCwybrRhuddanY RHYLL13 2 YD(in respect of right of access)Shaun Williams6 Cwybr CottagesC CwybrRhuddanY RHYLL13 2 YD(in respect of right of access)Shaun Williams6 Cwybr CottagesC CwybrRhuddanY RHYLL13 2 YD(in respect of right of access)Stephen Allen NicksonFlat 3C Cwybr HuoseC Wybr HuoseC Wybr PhoseC Wybr CottagesC CwybrSusan Davies5 Cwybr CottagesC CwybrNaddanY RHYLL13 2 YD(in respect of right of access)Susan Davies5 Cwybr CottagesCwybrCwybrRhuddanY RHYLL13 2 YD(in respect of right of access)Susan Davies5 Cwybr CottagesCwybrCwybrRhuddanY RHYLL13 2 YD(in respect of right of access)Uthrown |

|                         | Awel y Môr Offs                                                                                                                         | shore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                              |  |
|                         | County of Denbighshire                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                              |  |
| Number on<br>Land Plans | Description of Land                                                                                                                     | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                       |  |
| 239 cont'd              |                                                                                                                                         | Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 6JP<br>(in respect of right of access)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |  |
| 240                     | 933 square metres of private road (west of A525 and north of Cwybr-uchaf) and<br>footpaths (DE 206 24, DE 206 25, DE 206 31, DE 206 42) | Aaron Haldane<br>The Maisonette<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Ashworth<br>14 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                             |  |
| 240 cont'd              |                                                                                                                 | Anthony Roberts<br>Fiat 1<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Helen McCahill<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access) |  |

|                         | A                   | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|-------------------------|---------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                     | BOOK OF REFERENCE - PART 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                         |                     | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| Number on<br>Land Plans | Description of Land | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 240 cont'd              |                     | James Ketth Nixon<br>Fila 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L138 2YD<br>(in respect of right of access)<br>James Ketth Shaw<br>1 Cwybr Villa<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L138 2YD<br>(in respect of right of access)<br>James Massey<br>Filat 2<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L138 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L138 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L138 2YD<br>(in respect of right of access)<br>Keith Jones<br>Filat 4<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L138 2YD<br>(in respect of right of access)<br>Keith Jones<br>Filat 4<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L138 2YD<br>(in respect of right of access)<br>Keith Jones<br>Filat 4<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>L138 2YD<br>(in respect of right of access) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| 240 cont'd              |                                                                                                                 | Kirsty Massey         Flat 2         Cwybr         Rhuddlan         Y RHYL         LL18 27D         (in respect of right of access)         Mark Nicholls         6 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 27D         (in respect of right of access)         Mark Nicholls         6 Cwybr Cottages         Cwybr         Rhuddlan         Y RHYL         LL18 27D         (in respect of right of access)         Martin Nixon         Flat 4         Cwybr House         Cwybr         Cwybr House         Cwybr         Cwybr House         Cwybr Woldan         Y RHYL         LL18 27D         (in respect of right of access)         Matt Hession         2 Cwybr Uchaf Farm         Cwybr         Rhuddlan         Y RHYL         LL18 27D         (in respect of right of access)         Openreach Limited         British Telecom         Keivin House         123 Ludd Street |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                      |  |
| 240 cont'd              |                                                                                                                 | Peter Baddeley<br>2 Cwybr Villa<br>Cwybr Rhuddlan<br>Y RHVL<br>LL13 2YD<br>(in respect of right of access)<br>Shaun Williams<br>6 Cwybr Cottages<br>Cwybr Cottages<br>Cwybr Rhuddlan<br>Y RHVL<br>LL18 2YD<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right granted by Deed of grant dated 20 December 2017)<br>Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL13 2YD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                            |                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                       |  |
| 240 cont'd              |                                                                                                                                            | Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 6JP<br>(in respect of right of access) |  |
| 241                     | 1126 square metres of private road, hardstanding and agricultural land (east of<br>Cwybr-uchaf and north of A525) and footpath (DE 206 24) | Aaron HaldaneThe MaisonetteCwybr HouseCwybrRhuddlanY RHYLLL18 2YD(in respect of right of access)Anthony Ashworth14 Cwybr CottagesCwybrRhuddlanY RHYLLL18 2YD(in respect of right of access)                                                                                                                                                                                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| 241 cont'd              |                                                                                                                 | Anthony Roberts<br>Flat 1<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Helen McCahill<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                        |  |
| 241 cont'd              |                                                                                                                 | Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2VD<br>(in respect of right of access)<br>James Keith Nixon<br>Flot 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2VD<br>(in respect of right of access)<br>James Massey<br>Flot 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2VD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2VD<br>(in respect of right of access)<br>Keith Jones<br>Flot 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2VD<br>(in respect of right of access)<br>Keith Jones<br>Flot 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2VD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                               |  |
| 241 cont'd              |                                                                                                                 | Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 27D<br>(in respect of right of access)<br>Kirsty Massey<br>Flat 2<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2YD<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2YD<br>(in respect of right of access)<br>Mart Nicholls<br>118 2YD<br>(in respect of right of access)<br>Mart Nichols<br>Flat 4<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2YD<br>(in respect of right of access)<br>Mart Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2YD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 241 cont'd              |                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                           |  |
|                         |                                                                                                                 | Shaun Williams<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                                       |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                          |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain stated in deed dated 20 December 2017)                                                                                                                                                                |  |
|                         |                                                                                                                 | Stephen Allen Nickson<br>Flat 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                                                                                                                                                                                                                           |  |

| <b></b>                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
|                         | County of Denbighshire                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                                                                                    | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                      |  |
| 241 cont'd              |                                                                                                                        | Susan Davies<br>5 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Unknown<br>(in respect of restrictive covenants)<br>Unknown<br>(in respect of right of access)<br>Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES |  |
|                         |                                                                                                                        | MK7 6JP<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                                                |  |
| 242                     | 43453 square metres of agricultural land (north of A525 and south of Cwybr-uchaf) and footpaths (DE 206 23, DE 206 24) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)                                                                                                                                                                                                                                                                        |  |
|                         |                                                                                                                        | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                            |  |

| <b></b>                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER     |                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                  |                                                                                                                                                                                                                                                                                                                                                                                              |  |
|                         | County of Denbighshire                                      |                                                                                                                                                                                                                                                                                                                                                                                              |  |
| Number on<br>Land Plans | Description of Land                                         | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                         |  |
| 242 cont'd              |                                                             | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain stated in deed dated 20 December 2017)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants) |  |
| 243                     | 869 square metres of private road and track (north of A525) | Aaron Haldane<br>The Maisonette<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Ashworth<br>14 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>Anthony Roberts<br>Flat 1<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| 243 cont'd              |                                                                                                                 | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 SGD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Christopher George Roberts<br>1 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Helen McCahill<br>13 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Ian Towers<br>11 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2VD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>N RhyL<br>LL18 2VD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>RhyL<br>LL18 2VD<br>(in respect of right of access)<br>Jack Christopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>RhyL<br>LL18 2VD<br>(in respect of right of access)<br>Jack Thristopher Mousley<br>9 Cwybr Cottages<br>Cwybr<br>RhyL<br>LL18 2VD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| 243 cont'd              |                                                                                                                 | James Keith Nixon<br>Field 3<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL18 2YD<br>(in respect of right of access)<br>James Massey<br>Field 2<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL3 2YD<br>(in respect of right of access)<br>John Appleton<br>12 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL3 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL3 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL3 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL3 2YD<br>(in respect of right of access)<br>Keith Jones<br>10 Cwybr Cottages<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL3 2YD<br>(in respect of right of access)<br>Keith Jones<br>Field 4<br>Cwybr House<br>Cwybr<br>Rhuddlan<br>Y RHVL<br>LL3 2YD<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 243 cont'd              |                                                                                                                 | Kirsty Massey<br>Flat 2<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 27D<br>(in respect of right of access)<br>Mark Nicholls<br>6 Cwybr Cottages<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 27D<br>(in respect of right of access)<br>Martin Nixon<br>Flat 4<br>Cwybr House<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 27D<br>(in respect of right of access)<br>Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 27D<br>(in respect of right of access)<br>Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 27D<br>(in respect of right of access)<br>Matt Hession<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddian<br>Y RHYL<br>LL18 27D<br>(in respect of right of access)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WCL1H 9NP<br>(in respect of telecommunication apparatus) |  |

| Awel y Môr                               | Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BOOK OF REFERENCE - PART 3               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                                          | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Number on Description of Land Land Plans | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 243 cont'd                               | Shaun Williams         6 Cwybr Cottages         Cwybr         Rhuddlan         Y RHV         LL18 2YD         (in respect of right of access)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to lay and maintain stated in deed dated 20 December 2017)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to lay and maintain stated in deed dated 20 December 2017)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of electricity apparatus)         Stephen Allen Nickson         Filat 3         Cwybr House         Cwybr         Rhuddlan         Y RHYL         LL18 2YD         (in respect of right of access)         Unknown         (in respect of right of access)         Unknown      < |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 243 cont'd              |                                                                                                                 | Unknown<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                             |  |
|                         |                                                                                                                 | Victoria Jones<br>2 Cwybr Uchaf Farm<br>Cwybr<br>Rhuddlan<br>Y RHYL<br>LL18 2YD<br>(in respect of right of access)<br>William Richard Basketfield<br>4 Copthorne Place<br>Kents Hill<br>MILTON KEYNES<br>MK7 GJP<br>(in respect of right of access)                                                                                                    |  |
| 244                     | 731 square metres of track(west of Cwybr-uchaf and north-west of A525) and footpath (DE 206 31)                 | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of rights stated in Deed dated 26th February 1992)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                                                 |  |
| 245                     | 5 square metres of agricultural land (south-west of Cwybr-uchaf and north-west of A525)                         | Unknown                                                                                                                                                                                                                                                                                                                                                |  |
| 246                     | 535 square metres of agricultural land (south-west of Cwybr-uchaf and north-west of A525)                       | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of drainage)                                                                                                                                                                                                                                  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                       | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Number on<br>Land Plans | Description of Land                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 246 cont'd              |                                                                                       | Hugh Henry Wynne Davies         Morfa Cwybr         Norfa Cyybr         RIYL         Denbighshire         L18 2YB         (In respect of right of drainage)         Mary Williams         Cefn Cestyll         Rhydanfair         BETWS-Y-COED         L124 0LY         (In respect of right of drainage)         Michael Williams         Penlee         Flordd Fynnon         Dyserth         RHYL         L18 6HL         (In respect of right of drainage)         Robert Haydn Wynne Davies         Glan Yr Arfon Nurseries         RHYL         L18 2YB         (In respect of right of drainage)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (In respect of electricity apparatus)         Wales & West Utilities Limited         Wales and West Utilities Limited         Wales & West Utilities Limited         Wales & West House         Spooner Close         Coedkernew         CASNEWYDD         NPID 8FZ |

| [                       | Awel y Môr Offsh                                                                                                   | ore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|                         |                                                                                                                    | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| Number on<br>Land Plans | Description of Land                                                                                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 247                     | 14555 square metres of agricultural land and watercourse (Afon Ffyddion, north of<br>River Clwyd and west of A525) | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of drainage)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>RHYL<br>Denbighshire<br>LL18 2YB<br>(in respect of right of drainage)<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of drainage)<br>Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 GHL<br>(in respect of right of drainage)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 ZPB<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENDN<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017) |  |

|                         | Awel v Môr                                                       | Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|                         |                                                                  | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| Number on<br>Land Plans | Description of Land                                              | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| 247 cont'd              |                                                                  | SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of electricity apparatus)         Wales and West Utilities Limited         Wales & West House         Spooner Close         Coedkernew         CASNEWYDD         NP10 8FZ         (in respect of gas apparatus)         Wales and West Utilities Limited         Wales & West House         Spooner Close         Coedkernew         CASNEWYDD         NP10 8FZ         (in respect of right of access for maintenance as contained in an Agreement dated 19th June 1952) |  |
| 248                     | 7057 square metres of agricultural land (north of Afon Ffyddion) | Arthur Horron Wynne Davies         Trescawen Farm         LLANGEFNI         Gwynedd         LL77 7UR         (in respect of right of drainage)         Hugh Henry Wynne Davies         Morfa Cwybr         Morfa Cwybr         RHYL         Denbighshire         LL18 2YB         (in respect of right of drainage)         Mary Williams         Cefn Cestyll         Rhydlanfair         BETWS-Y-COED         LL24 0LY         (in respect of right of drainage)                                                                                                                                                                                                                                                                                                                              |  |

| Summer on Land Plan         Names of all those entilies to enjoy essements or other private rights over value (including private rights of navigation over value) (Application: Prescribed forms and Procedures) Regulations 2009           248 cont d         Michael Williams         Peniee           Floride Figure 1         Floride Figure 2         Floride Figure 2           11 Unservice 1         Michael Williams         Peniee           Floride Figure 2         Floride Figure 2         Floride Figure 2           11 Unservice 1         Floride Figure 2         Floride Figure 2           11 Unservice 1         Floride Figure 2         Floride Figure 2           12 Private 2         Floride Figure 2         Floride Figure 2           13 Private 2         Floride Figure 2         Floride Figure 2           14 Unservice 2         Floride Figure 2         Floride Figure 2           14 Private 2         Floride Figure 2         Floride Figure 2           14 Private 2         Floride Figure 2         Floride Figure 2           14 Private 2         Floride Figure 2         Floride Figure 2           14 Private 2         Floride Figure 2         Floride Figure 2           14 Private 2         Floride Figure 2         Floride Figure 2           14 Private 2         Floride Figure 2         Floride Figure 2           14                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Land Plans         Description of Land         which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)SO of the infrastructure Planning.<br>(Applications: Prescribed Forms and Procedures) Regulations 2009           245 cont'd         Michael Williams<br>Perifies         Perifies           Fordid Fyrnon<br>Dysertin         Perifies         Perifies           Find of Fyrnon<br>Dysertin         RNN.         LL3 6HL<br>(In respect of right of drainage)           Robert Haydn Wyme Davies<br>Glam Y Arton Nurseries         RNN.         LL3 2HB<br>(In respect of right of drainage)           SP Manweb PLC<br>3 Prenton Way<br>North Cheshite Trading Estate         PERIfies as contained in a Deed dated 20th December 2017)           SP Manweb PLC<br>3 Prenton Way<br>North Cheshite Trading Estate         PRENTON<br>Merscycide<br>CH43 3ET<br>(In respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017)           SP Manweb PLC<br>3 Prenton Way<br>North Cheshite Trading Estate         PERITON<br>Merscycide<br>CH43 3ET<br>(In respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017)           SP Manweb PLC<br>3 Prenton Way<br>North Cheshite Trading Estate         PERITON<br>Merscycide<br>CH43 3ET<br>(In respect of electricity apparatus)           Wales and West Utilities Limited         Wales and West Utilities Limited                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Penlee<br>Ford frynon<br>Dyserth<br>HYU<br>LLB 6HL<br>LLB 6HL<br>LLB 6HL<br>LLB 2H<br>Robert Haydn Wyne Davies<br>Glan Yr Arfon Nurseries<br>RHL<br>LLB 2Y8<br>I LLB 2Y8<br>I LLB 2Y8<br>I LLB 2Y8<br>I Comparison<br>SP Manweb PLC<br>SP enton Way<br>North Cheshiter Tading Estate<br>PEDETNON<br>Merseyside<br>CH3 32 TT<br>(in respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017)<br>SP Manweb PLC<br>CH3 32 TT<br>(in respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017)<br>SP Manweb PLC<br>SP Annoweb PLC<br>SP Annoweb PLC<br>SP Annoweb PLC<br>SP Annoweb PLC<br>SP Annoweb PLC<br>CH3 32 TT<br>(in respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017)<br>SP Manweb PLC<br>CH3 32 TT<br>(in respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017)<br>SP Manweb PLC<br>SP Annoweb PLC<br>SP Ann | Description of Land                                                                   | which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Spooner Close<br>Coedkernew<br>CASNEWYDD                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                       | Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHVL<br>LL13 6HL<br>(in respect of right of drainage)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHVL<br>LL13 2YB<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain electricity cables as contained in a Deed dated 20th December 2017)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASINEWYDD<br>NP10 8FZ |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                       | (in respect of right of access for maintenance as contained in an Agreement dated 19th June 1952)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                   | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| Number on<br>Land Plans | Description of Land                                                                                               | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| 249                     | 5304 square metres of agricultural land and watercourse (Afon Ffyddion, north of<br>River Clwyd and west of A525) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 SUH<br>(in respect of right of drainage)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr Fawr<br>LL18 2VD<br>(in respect of right of drainage)<br>Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)<br>Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2VD<br>(in respect of right of drainage)<br>Sandra Archdale<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2VD<br>(in respect of right of drainage)<br>S Andra Archdale<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2VD<br>(in respect of right of drainage)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| 249 cont'd              |                                                                                                                                                            | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain pipeline)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 250                     | 21814 square metres of watercourse (Afon Ffyddion), hedgerow, agricultural land,<br>footway and river bank (north of River Clwyd) and footpath (DE 206 29) | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddian<br>RHVL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr I fawr<br>Chyfer Fawr<br>Cwybr Fawr<br>Chyfer Fawr<br>Cwybr Cwr<br>Cwybr Cwr<br>Cwybr Fawr<br>Cwybr Fawr<br>Cwybr Cwr<br>Cwybr Cwr<br>Cwybr Cwr<br>Cwybr Cwr<br>Cwybr Cwr<br>Cwybr Cwr<br>Cwybr Cwr<br>Cwybr Cwr<br>Cwr<br>Cwr<br>Cwr<br>Cwr<br>Cwr<br>Cwr<br>Cwr |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|--------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                                            | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                            |  |
| 250 cont'd              |                                                                                | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain pipeline)                                                                                                                                                                                 |  |
| 251                     | 264 square metres of agricultural land (north of River Clwyd and west of A525) | Helen Owen Proffitt         Fferm         Bodelwyddan Road         Rhuddlan         RHYL         Denbighshire         LL18 5UH         (in respect of right of drainage)         Janet Johnson         Cwybr Fawr         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2VD         (in respect of right of drainage)         Rachel Georgina Hughes         Brook House Farm         DENBIGH         Denbighshire         LL16 4RD         (in respect of right of drainage) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                 |                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|-------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                            |  |
| 251 cont'd              |                                                                                                       | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain pipeline)                                                                                                   |  |
| 252                     | 726 square metres of agricultural land, hedgerow and track (north of River Clwyd<br>and west of A525) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of way)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RhVL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>RHYL<br>Denbighshire<br>LL18 ZYD<br>(in respect of right of drainage) |  |

|                         |                                                    | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|-------------------------|----------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans | Description of Land                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                  |
| 252 cont'd              |                                                    | Rachel Georgina Hughes         Brook House Farm         DENBIGH         Denbighshire         LL16 4RD         (in respect of right of drainage)         Sandra Archdale         Cwybr Fawr         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2YD         (in respect of right of drainage)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of electricity apparatus)         Wales and West Utilities Limited         Wales and West House         Spooner Close         Coedkernew         CASNEWYDD         NP10 BFZ         (in respect of right to lay and maintain pipeline) |
| 253                     | 1675 square metres of track (north of River Clwyd) | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of way)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |

|                         | Awel y Môr          | Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|-------------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Number on<br>Land Plans | Description of Land | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                            |
| 253 cont'd              |                     | Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Morfa Cwybr<br>RHVL<br>Denbighshire<br>LL18 278<br>(in respect of right of way)<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of way)<br>Michael Williams<br>Penlee<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 6HL<br>(in respect of right of way)<br>Robert Haydn Wynne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 278<br>(in respect of right of way)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 254                     | 77 square metres of track (north of A525 and River Clwyd)                                                       | Dwr Cymru Cyfyngedig         Dwr Cymru Welsh Water         Linea         Fortran Road         St. Mellons         CARDIFF         CF3 U.T         (in respect of right of way)         Dwr Cymru Cyfyngedig         Dwr Cymru Welsh Water         Linea         Fortran Road         St. Mellons         CARDIFF         CF3 0.LT         (in respect of water apparatus)         Helen Owen Proffitt         Fferm         Bodelwyddan Road         Rhuddlan         RHYL         Denbighshire         LLI3 SUH         (in respect of right of drainage)         Janet Johnson         Cwybr         RhVL         Denbighshire         LLI3 2VU         In respect of right of drainage)         Janet Johnson         Cwybr         RhVL         Denbighshire         LLI3 2VD         (in respect of right of drainage)         Rachel Georgina Hughes         Brook House Farm         DENISGH         Denbighshire         LLIG 4RD         Unsphyline |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                           |                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | County of Denbighshire                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                           |  |  |
| 254 cont'd              |                                                                                                                                 | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain pipeline)                                                                                                                  |  |  |
| 255                     | 52 square metres of track and grassed area (north of River Clwyd and south-east of A525) (excluding all interests of the Crown) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHVL<br>Denbighshire<br>LL18 SUH<br>(in respect of right of drainage)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 ZYD<br>(in respect of right of drainage) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 255 cont'd              |                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                               |  |
|                         |                                                                                                                 | Rachel Georgina Hughes<br>Brook House Farm<br>DENBIGH<br>Denbighshire<br>LL16 4RD<br>(in respect of right of drainage)                                                                                                                                                                                                                     |  |
|                         |                                                                                                                 | Sandra Archdale<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of drainage)                                                                                                                                                                                                                |  |
|                         |                                                                                                                 | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(in respect of annual Crown rent)                                                                                                                                                                                                 |  |
|                         |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain pipeline)                                                                                                                                                                       |  |

|                         | Awel y Môr (                                                             | Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
|-------------------------|--------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
|                         | County of Denbighshire                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
| Number on<br>Land Plans | Description of Land                                                      | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| 256                     | 290 square metres of track (north of River Clwyd and south-east of A525) | Arthur Horron Wynne Davies<br>Trescawen Farm<br>LLANGEFNI<br>Gwynedd<br>LL77 7UR<br>(in respect of right of access)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Melions<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Hugh Henry Wynne Davies<br>Morfa Cwybr<br>Mirfa Cwybr<br>Mirfa Cwybr<br>Mirfa Cwybr<br>Mirfa Cwybr<br>Mirfa Cwybr<br>Mirfa Caybr<br>Mary Williams<br>Cefn Cestyll<br>Rhydlanfair<br>BETWS-Y-COED<br>LL24 0LY<br>(in respect of right of access)<br>Michael Williams<br>Penlea<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 27B<br>(in respect of right of access)<br>Michael Williams<br>Penlea<br>Ffordd Ffynnon<br>Dyserth<br>RHYL<br>LL18 27B<br>(in respect of right of access)<br>Robert Haydn Wyne Davies<br>Glan Yr Arfon Nurseries<br>RHYL<br>LL18 27B<br>(in respect of right of access) |  |  |

| <b></b>                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                            |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                         |                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                                                                             |                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 256 cont'd              |                                                                                                                                    | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications apparatus)                                                           |  |
| 257                     | 4323 square metres of watercourse and river (River Clwyd), foreshore, bed and banks thereof (excluding all interests of the Crown) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights stated in Lease dated 2nd June 2015)<br>Rhyl & Conwy Wildfowlers Club<br>Ty Gwyn<br>Cwm<br>Rhuddlan<br>Y RHYL<br>LL18 5RY<br>(in respect of sporting rights)                                                       |  |
| 258                     | 340 square metres of grassed area and river banks (south of River Clwyd) (excluding all interests of the Crown)                    | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 259                     | 4924 square metres of grassed area and river banks (south of River Clwyd)<br>(excluding all interests of the Crown)                | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 260                     | 1516 square metres of grassed area and track (south of River Clwyd) and footpath<br>(DE 201 12)                                    | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 260 cont'd              |                                                                                                                 | The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 0TP<br>(in respect of right of access)<br>Unknown                                                                                                                                                                                                          |  |
| 261                     | 13531 square metres of agricultural land and hedgerow (south of River Clwyd)                                    | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of access for maintenance)                                                                                                                                                                            |  |
| 262                     | 52 square metres of track and grassed area (south of River Clwyd) and footpath<br>(DE 201 12)                   | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights granted by Deed dated 20 June 1979)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                               |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 262 cont'd              |                                                                                                                 | The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 OTP<br>(in respect of right of access)<br>Unknown<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                           |  |
| 263                     | 537 square metres of track (south of River Clwyd) and footpath (DE 201 12)                                      | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire    |                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                      |  |
| 263 cont'd              | 158 square metres of agricultural land (south of River Clwyd)                                                      | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Natural Resources Body For Wales<br>Cambria House<br>29 Newport Road<br>CARDIFF<br>CF24 OTP<br>(in respect of right of access)<br>Unknown<br>Dwr Cymru Cyfyngedig                                          |  |
| 264                     | 158 square metres of agricultural land (south of River Clwyd)                                                      | Dwr Cymru Cymru Cymru Cymru<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)                                                                                                                                                                                                           |  |
| 265                     | 729 square metres of agricultural land (Hafod-Ilwyn Farm north of Abergele Road,<br>A457 and south of River Clwyd) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of access for maintenance stated in Deed dated 20 June 1979)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus) |  |

|                         | Awel y Môr Offs                                                                                                               | shore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                          |                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 266                     | 40 square metres of agricultural land and hedgerow (Hafod-Ilwyn Farm north of Abergele Road, A457 and east of Marsh Road)     | Unknown                                                                                                                                                                                                                                                                                                                                    |  |
| 267                     | 72 square metres of hedgerow (Hafod-Ilwyn Farm north of Abergele Road, A457<br>and east of Marsh Road)                        | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Frances Marion Davies<br>Bryn Y Mor<br>Dimlands Road<br>LLANTWIT MAJOR<br>South Glamorgan<br>CF61 1YX<br>(in respect of right of access)                                                 |  |
| 268                     | 378 square metres of agricultural land and hedgerow (Hafod-Ilwyn Farm north of<br>Abergele Road, A457 and east of Marsh Road) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of access for maintenance)                                                                                                                                                                |  |
| 269                     | 681 square metres of agricultural land and hedgerow (north of Abergele Road, A457)                                            | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Frances Marion Davies<br>Bryn Y Mor<br>Dimlands Road<br>LLANTWIT MAJOR<br>South Glamorgan<br>CF61 1YX<br>(in respect of right of access)                                                 |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                |  |
| 270                     | 18 square metres of hedgerow (Hafod-Ilwyn Farm north of Abergele Road, A457)                                    | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access)<br>Unknown<br>(in respect of rights stated in Conveyance dated 14th July 1924)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road                                                                |  |
|                         |                                                                                                                 | St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of rights stated in Deed dated 20th June 1979)                                                                                                                                                                                                                                                                 |  |
| 271                     | 483 square metres of agricultural land (Hafod-Ilwyn Farm north of Abergele Road,<br>A457)                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of easement and right of access for maintenance stated in Deed dated 20 June 1979)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                      |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                   |                                                                                                                                                                                                                                                                                                                                                        |  |
|                         |                                                                                                              | County of Denbighshire                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 271 cont'd              |                                                                                                              | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                          |  |
| 272                     | 10631 square metres of agricultural land (north of Abergele Road, A457 and east of<br>Gipsy Lane)            | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of easement and right of access for maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                           |  |
| 273                     | 35 square metres of agricultural land (north of Abergele Road, A457 and east of Gipsy Lane)                  | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of easement and right of access for maintenance)                                                                                                                                                                            |  |
| 274                     | 64 square metres of agricultural land and hedgerows (north of Abergele Road, A457<br>and east of Gipsy Lane) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right of access)                                                                                                                                                                                                         |  |

|                         | Awel v Môr Offe                                                                               | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                    |                                                                                                                                                                                                                                                                                                                                                        |  |
|                         |                                                                                               | County of Denbighshire                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 274 cont'd              |                                                                                               | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only)                                                                                                                                                            |  |
| 275                     | 5910 square metres of agricultural land (north of Abergele Road, A457 and east of Gipsy Lane) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only)          |  |
| 276                     | 2256 square metres of agricultural land (north of Abergele Road, A457 and east of Gipsy Lane) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right of access)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only)          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                      |                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                         |                                                                                                                                                                                                                                                                                                                                               |  |
| Number on<br>Land Plans | Description of Land                                                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009    |  |
| 277                     | 358 square metres of agricultural land and hedgerow (north of Abergele Road, A457<br>and east of Gipsy Lane) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only) |  |
| 278                     | 620 square metres of agricultural land (north of Abergele Road, A457 and east of Gipsy Lane)                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 SUH<br>(in respect of right of drainage and right to pass for agricultural purposes only) |  |
| 279                     | 27 square metres of hedgerow (east of Gipsy Lane and north of Abergele Road, A45)                            | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access)                                                                                                                                                                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|                         |                                                                             | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land                                                         | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                         |  |
| 279 cont'd              |                                                                             | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of drainage and right to pass for agricultural purposes only)                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| 280                     | 562 square metres of private road (Gipsy Lane north of Abergele Road, A457) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 ODH<br>(in respect of right of access)<br>Police & Crime Commissioner for North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWWN BAY<br>Convy<br>LL29 BAW<br>(in respect of right of access) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 280 cont'd              |                                                                                                                 | R G B Webster & Sons Limited<br>Banks House<br>1 Paradise Street<br>RHYL<br>Denbighshire<br>LL18 3LW<br>(in respect of right of access)<br>Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of access)<br>Unknown                                                         |  |
| 281                     | 51 square metres of public road (Gipsy Lane north of Abergele Road, A457)                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Unknown                                                                                                                                                                                              |  |
| 282                     | 19597 square metres of agricultural land (north of Abergele Road, A457 and east of<br>Gipsy Lane)               | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH<br>(in respect of right to pass and right to lay and maintain a water pipeline)                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                        |                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                     |                                                                                                                                                                                                                                                                                                                                      |  |
|                         | County of Denbighshire                                                                                         |                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                            | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 283                     | 302 square metres of verge of public road (south of Abergele Road, A547)                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                            |  |
| 284                     | 400 square metres of verge of public road (north of Abergele Road, A547)                                       | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                            |  |
| 285                     | 7031 square metres of agricultural land (Hafod-Ilwyn Farm north of Abergele Road, A457 and east of Gipsy Lane) | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 0DH<br>(in respect of right to pass and right to lay and maintain a water pipeline)                                                                                                                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire            |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 286                     | 644 square metres of agricultural land and hedgerow (Hafod-Ilwyn Farm north of Abergele Road, A457 and east of Gipsy Lane) | Philip Edward Beech<br>Ty Fferm<br>Pentre Uchaf Farm<br>Ffordd Y Bryn<br>LLANELWY<br>LL17 ODH<br>(in respect of right to pass and right to lay and maintain a water pipeline)                                                                                                                                                                          |  |
| 287                     | 120 square metres of track (Hafod-Ilwyn Farm north of Abergele Road, A457)                                                 | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of access)<br>The Executor of the Estate of the Late Richard Dod<br>Bryn Carrog Farm<br>Bodelwyddan Road<br>RHUDDLAN<br>Denbighshire<br>LL18 5UH<br>(in respect of right of access)<br>Unknown                           |  |
| 288                     | 6 square metres of public road (Hafod-Ilwyn Farm north of Abergele Road, A457)                                             | Unknown                                                                                                                                                                                                                                                                                                                                                |  |
| 289                     | 208 square metres of verge of public road (north of Abergele Road, A457)                                                   | Unknown                                                                                                                                                                                                                                                                                                                                                |  |
| 290                     | 172 square metres of verge of public road (south of Abergele Road, A457)                                                   | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                                                                                                                                                         |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|                         | County of Denbighshire                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                          |  |
| 290 cont'd              |                                                                                                               | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                                                                                                                                                                                                                                                         |  |
| 291                     | 904 square metres of public road and verges (Abergele Road, A457)                                             | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                                                                                                       |  |
| 293                     | 19992 square metres of agricultural land and hedgerow (south of Abergele Road, A457) and footpath (DE 206 40) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain electric lines and right to fell trees that would obstruct the electric lines as contained in a<br>Deed dated 21st July 1994) |  |

|                         | Awel y Môr Offsh                                                                                                      | ore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                  |                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 296                     | 2073 square metres of agricultural land (north of Bodelwyddan Road) and footpaths<br>(DE 201 8, DE 206 38, DE 206 41) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                            |  |
| 298                     | 12 square metres of verge of public road (north of Bodelwyddan Road)                                                  | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Unknown                                                                                                                                                                                  |  |
| 300                     | 18 square metres of verge of public road (north of Bodelwyddan Road)                                                  | Unknown                                                                                                                                                                                                                                                                                                                                    |  |
| 302                     | 42 square metres of agricultural land and hedgerow (north of Bodelwyddan Road)<br>and footpath (DE 206 4)             | Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right to lay and maintain services)                                                                                                                                                                          |  |
| 303                     | 174 square metres of agricultural land and hedgerows (north of Bodelwyddan Road)                                      | Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right to lay and maintain services)                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                        |  |
| 304                     | 231 square metres of agricultural land and hedgerow (north of Bodelwyddan Road)<br>and footpath (DE 206 4)      | Rhyl Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right of access and right to lay and maintain services)                                                                                                                                                                                                                                                                                                                             |  |
| 306                     | 472 square metres of private road (north of Bodelwyddan Road)                                                   | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Rhyl Town Council<br>Town Council<br>Town Clerk<br>Civic Offices<br>Wellington Road<br>RHYL<br>LL18 1LE<br>(in respect of right to enter to lay and maintain water pipes)<br>The Executor of the Estate of the Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of right of way) |  |
| 308                     | 6339 square metres of agricultural land (north of Bodelwyddan Road)                                             | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                                 |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                     |  |
| 309                     | 9496 square metres of agricultural land and hedgerow (north of Bodelwyddan<br>Road) and footpath (DE 206 4)     | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                                                                                                                                                                       |  |
|                         |                                                                                                                 | Rhyl Town Council         Town Clerk         Civic Offices         Wellington Road         RHYL         LL18 1LE         (in respect of right of access and right to lay and maintain services)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside                                                 |  |
|                         |                                                                                                                 | CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                    |  |
| 310                     | 862 square metres of verge of public road (south of Bodelwyddan Road)                                           | Dwr Cymru Cyfyngedig         Dwr Cymru Welsh Water         Linea         Fortran Road         St. Mellons         CARDIFF         CF3 0LT         (in respect of water apparatus)         Openreach Limited         British Telecom         Kelvin House         123 Judd Street         LONDON         WC1H 9NP         (in respect of telecommunication apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                   |  |
| 311                     | 17 square metres of public road (south of Bodelwyddan Road)                                                     | Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                              |  |
| 312                     | 252 square metres of verge of public road (north of Bodelwyddan Road)                                           | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to enter, right to construct, place, lay, inspect, maintain, clean, repair, renew, enlarge, replace, conduct, use<br>and manage water pipes and right to pass)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus) |  |
| 313                     | 102 square metres of hedgerow and verge of public road (north of Bodelwyddan Road)                              | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 314                     | 203 square metres of agricultural land and hedgerow (south of Bodelwyddan Road)                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                                                                                                       |  |
| 316                     | 14 square metres of verge of public road (north of Bodelwyddan Road)                                            | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|--------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                   |                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                    | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 317                     | 523 square metres of agricultural land, hedgerows, private road and verges (north of Bodelwyddan Road) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Executor of the Estate of the Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of right of way)                            |  |
| 318                     | 5 square metres of private road and verges (north of Bodelwyddan Road)                                 | The Executor of the Estate of the Late George Edward Brookes<br>Ystrad Fechan<br>Ystrad<br>DENBIGH<br>Denbighshire<br>LL16 4RL<br>(in respect of right of access)<br>Unknown                                                                                                                                                               |  |
| 319                     | 9 square metres of private road and verges (north of Bodelwyddan Road)                                 | Helen Owen Proffitt<br>Fferm<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of access)<br>Janet Johnson<br>Cwybr Fawr<br>Cwybr<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 2YD<br>(in respect of right of access)                                                                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                     |  |
| 319 cont'd              |                                                                                                                 | Rachel Georgina Hughes         Brook House Farm         DENBIGH         Denbighshire         LL16 4RD         (in respect of right of access)         Sandra Archdale         Cwybr Fawr         Cwybr         Rhuddlan         RHYL         Denbighshire         LL18 2YD         (in respect of right of access)         The Executor of the Estate of the Late George Edward Brookes         Ystrad         PENBIGH         Denbighshire         LL16 4RL         (in respect of right of access) |  |
| 320                     | 272 square metres of verge of public road (north of Bodelwyddan Road)                                           | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                               |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|--------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                      | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                     |  |
| 321                     | 167 square metres of verge of public road (south of Bodelwyddan Road)    | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                              |  |
| 322                     | 521 square metres of public road and hedgerow (part of Bodelwyddan Road) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown |  |
| 323                     | 67 square metres of public road (part of Bodelwyddan Road)               | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 323 cont'd              |                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                                                                                                                                                                    |  |
| 325                     | 124 square metres of verge of public road (north of Bodelwyddan Road)                                           | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of way)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                             |  |
| 326                     | 4 square metres of public road (part of Bodelwyddan Road)                                                       | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of way)                                                                                                                                                                                                      |  |
| 327                     | 207 square metres of verge of public road (north of Bodelwyddan Road)                                           | Unknown                                                                                                                                                                                                                                                                                                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                   |                                                                                                                                                                                                                                                                                                                                                  |  |
|                         |                                                                                        |                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                    | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 328                     | 1728 square metres of agricultural land and hedgerow (north of Bodelwyddan<br>Road)    | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)<br>Gwynt Y Mor Offshore Wind Farm Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Wiltshire<br>SNS 6PB                                             |  |
|                         |                                                                                        | (in respect of right to an easement as contained in a Deed of Grant dated 8th September 2010)<br>Unknown<br>(in respect of right of drainage)                                                                                                                                                                                                    |  |
| 329                     | 272 square metres of verge of public road (north of Bodelwyddan Road)                  | Raymond Ivor Beech<br>Bryn Carrog<br>Bodelwyddan Road<br>Rhuddlan<br>RHYL<br>Denbighshire<br>LL18 5UH<br>(in respect of right of way)                                                                                                                                                                                                            |  |
| 330                     | 4 square metres of public road (part of Nant-y-Faenol Road, south of Bodelwyddan Road) | Unknown                                                                                                                                                                                                                                                                                                                                          |  |
| 331                     | 1305 square metres of agricultural land and hedgerows (west of Nant-γ-Faenol<br>Road)  | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Unknown<br>(in respect of right of way)                                                                                                                                                        |  |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER |                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|---------------------------------------------------------|------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                                                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                                                         |                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans                                 | Description of Land                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                           |  |
| 332                                                     | 86 square metres of hedgerow and track (west of Nant-y-Faenol Road)                                        | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 333                                                     | 154 square metres of hedgerow (north and east of Nant-y-Faenol Road)                                       | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 335                                                     | 11216 square metres of agricultural land (north and east of Nant-y-Faenol Road, south of Bodelwyddan Road) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| 337                                                     | 19 square metres of verge of public road (north of Nant-y-Faenol Road)                                     | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 338                                                     | 37 square metres of agricultural land and hedgerow (north of Nant-y-Faenol Road)                           | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 340                                                     | 149 square metres of agricultural land and hedgerow (south of Nant-y-Faenol Road)                          | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 341                                                     | 1557 square metres of agricultural land and hedgerow (south of Nant-y-Faenol<br>Road)                      | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes) |  |
| 342                                                     | 420 square metres of public road and verges (Nant-y-Faenol Road)                                           | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         |                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                           |  |
| 343                     | 2 square metres of verge of public road (west of Nant-y-Faenol Road)                                | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 344                     | 23 square metres of hedgerow and verge of public road (south of Nant-y-Faenol Road)                 | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 345                     | 17 square metres of verge of public road (south of Nant-y-Faenol Road)                              | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 346                     | 103 square metres of agricultural land (south of Nant-y-Faenol Road)                                | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes) |  |
| 347                     | 15477 square metres of agricultural land (west of Little Pengwern, south of Nant-y-<br>Faenol Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)                                                                                                                                                                                                              |  |

| <u> </u>                | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                                      | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                           |  |
| 347 cont'd              |                                                                                                                          | The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes)                                                                                                                                                                                                                                                                                                                                        |  |
| 348                     | 2398 square metres of agricultural land and hedgerows (west of Little Pengwern,<br>south and west of Nant-y-Faenol Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes) |  |
| 349                     | 2525 square metres of woodland (west of Little Pengwern and south-west of Nant-<br>y-Faenol Road)                        | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 350                     | 286 square metres of woodland (west of Little Pengwern, south-west of Nant-y-<br>Faenol Road)                            | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 351                     | 11635 square metres of agricultural land and hedgerows (west of Little Pengwern<br>and south-west of Nant-y-Faenol Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)                                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 351 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                         |  |
|                         |                                                                                                                 | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)                                                                                                                             |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                          |  |
|                         |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)                                                                                                                                 |  |
| 352                     | 5480 square metres of agricultural land (west of Little Pengwern and Nant-y-Faenol Road)                        | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)                                                                                                                                                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                 |  |
| 352 cont'd              |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 5UL<br>(in respect of right of access for maintenance of pipes)                                                                                                                                                                            |  |
| 353                     | 22471 square metres of agricultural land (west of Little Pengwern and Nant-y-Faenol Road)                       | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 SUL<br>(in respect of right of access for maintenance of pipes) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 354                     | 2603 square metres of agricultural land (south of Little Pengwern and west of Nant-<br>y-Faenol Road)           | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LIONDON<br>WC1V 6BA<br>(in respect of rights granted in Deed dated 12th December 2016)<br>Openreach Limited<br>British Telecom<br>Keivin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Executor of the Estate of the Late Edwin Richard Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Clwyd<br>LL18 SUL<br>(in respect of right of access for maintenance of pipes) |  |
| 355                     | 11 square metres of public road (Nant-y-Faenol Road)                                                            | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                              |  |
| 356                     | 26 square metres of agricultural land (west of Nant-y-Faenol Road)                                                           | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 357                     | 236 square metres of agricultural land and hedgerow (north of North Wales<br>Expressway, A55 and west of Nant-y-Faenol Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Cyfngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>51. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>51. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                   |  |
| 357 cont'd              |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 358                     | 1504 square metres of track (north of Tyddyn Isaf Farm and North Wales<br>Expressway, A55)                      | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>51. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>51. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 358 cont'd              |                                                                                                                                                                              | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 359                     | 35572 square metres of agricultural land, woodland and hedgerows (north-east of<br>Tyddyn Isaf Farm, north of North Wales Expressway, A55 and west of Nant-y-Faenol<br>Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline as contained in a Deed dated 5th September 1967)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUY<br>(in respect of right to free passage of service media as contained in a Conveyance dated 18th November 1958) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                       | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                               |  |
| 359 cont'd              |                                                                                                                                           | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)                                                                                                                                                                              |  |
| 360                     | 19135 square metres of agricultural land (north of North Wales Expressway, A55<br>and west of Nant-y-Faenol Road) and footpath (DE 201 7) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain water pipeline as contained in a Deed dated 5th September 1967) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                          | County of Denbighshire Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                           |  |
| 360 cont'd              |                                                                                                                              | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957) |  |
| 361                     | 828 square metres of agricultural land and hedgerow (north of North Wales<br>Expressway, A55 and west of Nant-y-Faenol Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
|                         |                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                  |  |  |
| 361 cont'd              |                                                                                                                              | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)                                                                                                    |  |  |
| 362                     | 559 square metres of agricultural land and hedgerow (north of North Wales<br>Expressway, A55 and west of Nant-y-Faenol Road) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | County of Denbighshire                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| 362 cont'd              |                                                                                                                                              | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |
| 363                     | 77 square metres of hedgerow (north of North Wales Expressway, A55 and west of<br>Nant-y-Faenol Road) (excluding all interests of the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 DDX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Secretary Of State For Wales<br>William Morgan House<br>G-7 Central Square<br>CARDIFF<br>CF10 IEP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                  | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 364                     | 151 square metres of hedgerow (north of North Wales Expressway, A55 and west of<br>Nant-y-Faenol Road) (excluding all interests of the Crown)        | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 0DX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 365                     | 2597 square metres of agricultural land (north of North Wales Expressway, A55 and west of Nant-y-Faenol Road) (excluding all interests of the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 365 cont'd              |                                                                                                                 | The Executor of the Estate of the Late Elizabeth Alice Jones Pengwern Farm Nant Y Faenol Road Bodelwyddan RHYL Denbighshire LL18 SUL (in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th November 1958) Jane Elizabeth Grindal Ty Mawr Farm St. Asaph Road Bodelwyddan RHYL Denbigshire LL18 SUY (in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003) The Secretary Of State For Wales William Morgan House 6-7 Central Square CARDIFF CH10 IEP (in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 366                     | 16 square metres of agricultural land (north of Tyddyn Isaf Farm and North Wales Expressway, A55)               | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                 |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 366 cont'd              |                                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                             |  |
|                         |                                                                                                                                 | Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)                                                                                                                 |  |
|                         |                                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to lay and maintain gas pipeline as contained in a Deed dated 26th August 1957)                                                                                                                     |  |
| 367                     | 2 square metres of agricultural land and hedgerows (north-east of Tyddyn Isaf Farm<br>and north of North Wales Expressway, A55) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                                                                                                                                                                   |  |
| 368                     | 393 square metres of agricultural land (north-east of Tyddyn Isaf Farm and north of North Wales Expressway, A55)                | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to install and maintain electricity lines)                                                                                                                                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| 368 cont'd              |                                                                                                                 | Owr Cymru Welsh Water         Linea         Fortran Road         St. Melions         CARDIF         CROIF         CROIF         CROIF         CF3 0LT         (in respect of water apparatus)         Dwr Cymru Cyfyngedig         Dwr Cymru Cyfyngedig         Dwr Cymru Welsh Water         Linea         Fortran Road         St. Mellons         CARDIF         CF3 0LT         (in respect of right to lay and maintain water pipeline as contained in a Deed dated 5th September 1967)         Jane Elizabeth Grindal         Ty Mawr Fam         St. Asaph Road         Bodelwyddan         RHYL         Denbigshire         LL18 SUY         (in respect of right of free passage of service media as contained in a Conveyance dated 18th November 1958)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH33 3ET         (in respect of electricity apparatus)         Wales and West House         Sponer Close         Coekernew         CANUVDD         NP10 872 |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 369                     | 10 square metres of agricultural land (east of Tyddyn Isaf Farm, north of North<br>Wales Expressway, A55 and west of Nant-y-Faenol Road) (excluding all interests of<br>the Crown)  | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Secretary Of State For Wales<br>William Morgan House<br>G-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 370                     | 27217 square metres of agricultural land (north of North Wales Expressway, A55<br>and west of Nant-y-Faenol Road) and footpath (DE 201 7) (excluding all interests of<br>the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |

| Number on       Description of Land       Names of all those entitled to enjoy easements or other private rights over land (including private rights of na which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infra                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |  |  |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|--|--|
| Land Plans (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                 |  |  |
| 370 cont'd       The Executor of the Estate of the Late Elizabeth Alice Jones         Pengwern Farm       Next Y Faenol Road         Bodelwyddan       HYL         HYL       Deblighshire         L18 SUV       (in respect of right of access for maintenance and right to free passage of service media as contained in a Con         November 1558)       Jane Elizabeth Alice Jones         Jane Elizabeth Orindal       Ty Maw Farm         SL Asaph Road       Bodelwyddan         RHYL       Derbligshire         L13 SUV       (in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)         SP Manweb PLC       3 Prenton Way         North Cheshire Trading Estate       PENTON         Merseyide       CH3 3ET         CH3 3ET       (in respect of right of access for maintenance electricity cables as contained in a Deed dated 17th Octob         SP Manweb PLC       3 Prenton Way         North Cheshire Trading Estate       PENTON         Merseyide       CH3 3ET         CH3 3ET       (in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th Octot         SP Manweb PLC       3 Prenton Way         North Cheshire Trading Estate       PENTON         Merseyide       CH3 3ET         (In respect o | er 2003)                                                                                                        |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                                                                         | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| 371                     | 830 square metres of agricultural land (east of Tyddyn Isaf Farm, north of North<br>Wales Expressway, A55 and west of Nant-y-Faenol Road) and footpath (DE 201 7)<br>(excluding all interests of the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>57. ASAPH<br>Clwyd<br>LL17 0DX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 SUL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>TY Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th<br>November 1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                                    | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| 371 cont'd              |                                                                                                                                                                        | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| 372                     | 1429 square metres of agricultural land, tracks and hardstanding (at Tyddyn Isaf<br>Farm, north of North Wales Expressway, A55) (excluding all interests of the Crown) | Catherine Nerys Thomas 1 Oak Close ST. ASAPH Clwyd LL17 ODX (in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958) Dwr Cymru Cyfyngedig Dwr Cymru Welsh Water Linea Fortran Road St. Mellons CARDIFF CF3 OLT (in respect of water apparatus) The Executor of the Estate of the Late Elizabeth Alice Jones Pengwern Farm Nant Y Faenol Road Bodelwyddan RHYL Denbigshire LL18 5UU (in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th November 1958) Jane Elizabeth Grindal Ty Mawr Farm St. Asaph Road Bodelwyddan RHYL Denbigshire LL18 5UV (in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                           |  |
| 372 cont'd              |                                                                                                                    | SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of electricity apparatus)         The Secretary Of State For Wales         William Morgan House         6-7 Central Square         CARDIFF         CF10 1EP         (in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 373                     | 786 square metres of private road (north of North Wales Expressway, A55)<br>(excluding all interests of the Crown) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables and right of access)                                                                                                                                                                                                                                                                                                                                                                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                                               | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| 373 cont'd              |                                                                                                                                                                                   | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| 374                     | 10879 square metres of agricultural land (north of North Wales Expressway, A55<br>and east of Tyddyn Isaf Farm) and footpath (DE 201 7) (excluding all interests of<br>the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 SUL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUV<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th<br>November 1958)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | County of Denbighshire                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                   |  |  |
| 374 cont'd              |                                                                                                                                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991)                                                                                                                                                                                                                                                                                                                                            |  |  |
| 375                     | 56 square metres of agricultural land (north of North Wales Expressway, A55 and<br>east of Tyddyn Isaf Farm) (excluding all interests of the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbighshire<br>LL18 5UL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                    |  |
| 375 cont'd              |                                                                                                                                                                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991)                                                                                                                                                                                                                                                        |  |
| 376                     | 16642 square metres of agricultural land (north of North Wales Expressway, A55<br>and west of Nant-y-Faenol Road) and footpath (DE 201 7) (excluding all interests of<br>the Crown) | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)<br>The Executor of the Estate of the Late Elizabeth Alice Jones<br>Pengwern Farm<br>Nant Y Faenol Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUL<br>(in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th<br>November 1958)<br>Jane Elizabeth Grindal<br>Ty Mawr Farm<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUY<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                        |  |
| 376 cont'd              |                                                                                                                            | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain overhead electricity cables as contained in a Deed dated 17th October 2003)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 377                     | 122 square metres of private road and verges (north of North Wales Expressway, A55) (excluding all interests of the Crown) | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of restrictive covenants)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                     | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                |  |
| 377 cont'd              |                                                                                                                                                         | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for maintenance)                                                                                                                                                                                                                                                                                                                                                                                               |  |
| 378                     | 2220 square metres of agricultural land (north of North Wales Expressway, A55 and<br>west of Nant-y-Faenol Road) (excluding all interests of the Crown) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to lay and maintain cables and right of access stated in Deed dated 17 October 2003)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991) |  |
| 379                     | 122 square metres of agricultural land (north of North Wales Expressway, A55 and west of Nant-y-Faenol Road) (excluding all interests of the Crown)     | Catherine Nerys Thomas<br>1 Oak Close<br>ST. ASAPH<br>Clwyd<br>LL17 ODX<br>(in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958)                                                                                                                                                                                                                                                                                                                                                                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|                         |                                                                                                                                                     | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                             |  |
| 379 cont'd              |                                                                                                                                                     | The Executor of the Estate of the Late Elizabeth Alice Jones Pengwern Farm Nant Y Faenol Road Bodelwyddan RHYL Denbighshire LL18 5UL (in respect of right of access for maintenance and right to free passage of service media as contained in a Conveyance dated 18th November 1958) Jane Elizabeth Grindal Ty Mawr Farm St. Asaph Road Bodelwyddan RHYL Denbigshire LL18 5UV (in respect of right of access for maintenance as contained in a Conveyance dated 18th November 1958) The Secretary Of State For Wales William Morgan House 6-7 Central Square CARDIFF CF10 1EP (in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 380                     | 1383 square metres of private road and verges (north of North Wales Expressway, A55) and footpath (DE 201 7) (excluding all interests of the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| 381                     | 945 square metres of private road and verges (north of North Wales Expressway, A55) and footpath (DE 201 7) (excluding all interests of the Crown)  | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                         |  |
| 382                     | 249 square metres of verge (North Wales Expressway, A55) (excluding all interests<br>of the Crown)              | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of restrictive covenants)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of restrictive covenants)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for maintenance) |  |
| 384                     | 1481 square metres of public road (North Wales Expressway, A55) (excluding all interests of the Crown)          | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right to enter and right to service media)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of right to enter and right to service media)                                                                                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire     |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 385                     | 194 square metres of verge (North Wales Expressway, A55) (excluding all interests of the Crown)                     | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 5UY<br>(in respect of right to enter and right to service media)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 5UY<br>(in respect of right to enter and right to service media)                        |  |
| 386                     | 359 square metres of verge (North Wales Expressway, A55) (excluding all interests of the Crown)                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                          |  |
| 387                     | 782 square metres of public road (North Wales Expressway, A55) (excluding all interests of the Crown)               | Unknown                                                                                                                                                                                                                                                                                                                                                |  |
| 389                     | 145 square metres of public road and verges (North Wales Expressway, A55)<br>(excluding all interests of the Crown) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                           |  |
| 390                     | 5 square metres of public road and verges (North Wales Expressway, A55)<br>(excluding all interests of the Crown)   | Unknown                                                                                                                                                                                                                                                                                                                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                           |  |
| 391                     | 4 square metres of verges (North Wales Expressway, A55) (excluding all interests of the Crown)                  | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| 392                     | 562 square metres of verge(North Wales Expressway, A55)(excluding all interests of the Crown)                   | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunications apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                              |  |
| 393                     | 21 square metres of public road and verges (North Wales Expressway, A55)<br>(excluding all interests of the Crown) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                                                                                                                                                                                                                                                                                                             |  |
| 394                     | 14 square metres of public road (North Wales Expressway, A55) (excluding all interests of the Crown)               | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect to rights granted by Deed dated 6 June 1958 which also contains restrictive covenants)                                                                                                                                                                                                                                                                          |  |
| 395                     | 175 square metres of verges (North Wales Expressway, A55) (excluding all interests of the Crown)                   | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect to rights granted by Deed dated 6 June 1958 which also contains restrictive covenants)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunications apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                 |  |
| 396                     | 104 square metres of public road and verge (North Wales Expressway, A55)<br>(excluding all interests of the Crown) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| 397                     | 67 square metres of verges (North Wales Expressway, A55) (excluding all interests<br>of the Crown)                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WCIH 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunications apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                |  |
| 398                     | 418 square metres of public road (North Wales Expressway, A55) (excluding all interests of the Crown)           | Elizabeth Eirlys Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>RHYL<br>Denbigshire<br>LL18 SUY<br>(in respect of restrictive covenants)<br>Islwyn Williams<br>Tyddyn Isaf<br>St. Asaph Road<br>Bodelwyddan<br>Y RHYL<br>LL18 SUY<br>(in respect of restrictive covenants)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access for maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 398 cont'd              |                                                                                                                            | Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunications apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 399                     | 86 square metres of public road and verge (south of North Wales Expressway, A55)<br>(excluding all interests of the Crown) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WiNDSOR<br>Berkshire<br>SL4 SGD<br>(in respect of right to lay and maintain gas pipeline)<br>Kent Periscopes Limited<br>Dowlish Ford<br>ILMINSTER<br>TA19 0PF<br>(in respect of right to service media and right to enter)<br>North Wales Fire Authority<br>Site G<br>Ffordd Salesbury<br>Sit. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0J<br>(in respect of rights stated in Lease dated 26th April 2005)<br>NW (BBP) Limited<br>3 More London Riverside<br>LONDON<br>SE1 2AQ<br>(in respect of rights stated in Lease dated 9th October 2002)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H SNP<br>(in respect of telecommunication apparatus) |  |

| Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on Description of Land                                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 399 cont'd                                                                                                      | Police & Crime Commissioner for North Wales         North Wales Police         Headquarters         Abergele Road         COLWYN BAY         Conwy         L129 BAW         (in respect of rights stated in Lease dated 1st March 2000)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right of access along roadways and right to lay and maintain gas pipeline)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right of access along roadways and right to lay and maintain gas pipeline)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of electricity apparatus)         TRB United         1TRB Drive         S1. Asaph Business Park         ST. ASAPH         Denbighshire         L17 0B         (in respect of right of access along roadways and right to free passage of service media) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|                         | County of Denbighshire                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                                               | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                             |  |
| 399 cont'd              |                                                                                                                   | WGIF (Jersey) Trustee I Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)<br>WGIF (Jersey) Trustee II Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)<br>Wrexham Glyndwr University<br>Plas Coch<br>Mold Road<br>WREXHAM<br>LL11 2AW<br>(in respect of rights stated in Lease dated 9th July 2013) |  |
| 400                     | 622 square metres of public road (south of North Wales Expressway, A55)<br>(excluding all interests of the Crown) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and maintain gas pipeline)<br>Kent Periscopes Limited<br>Dowlish Ford<br>ILMINSTER<br>TA19 OPF<br>(in respect of right to service media and right to enter)<br>North Wales Fire Authority<br>Site G<br>Ffordd Salesbury<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0J<br>(in respect of rights stated in Lease dated 26th April 2005)                                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 400 cont'd              |                                                                                                                 | NW (BBP) Limited<br>3 More London Riverside<br>LONDON<br>SE1 2AQ<br>(in respect of rights stated in Lease dated 9th October 2002)<br>Openreach Limited                                                                                                                                                                                           |  |
|                         |                                                                                                                 | British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                          |  |
|                         |                                                                                                                 | Police & Crime Commissioner for North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWYN BAY<br>Conwy<br>LL29 8AW<br>(in respect of rights stated in Lease dated 1st March 2000)                                                                                                                                             |  |
|                         |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right of access along roadways and right to lay and maintain gas pipeline)                                                                                                                                                |  |
|                         |                                                                                                                 | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access along roadways and right to free passage of service media)                                                                                                                                                      |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                                     | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 400 cont'd              |                                                                                                                                                         | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of access along roadways)<br>WGIF (Jersey) Trustee I Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)<br>WGIF (Jersey) Trustee II Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)<br>WGIF (Jersey) Trustee II Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)<br>Wrexham Glyndwr University<br>Plas Coch<br>Mold Road<br>WREXHAM<br>LL11 2AW<br>(in respect of rights stated in Lease dated 9th July 2013) |  |
| 401                     | 583 square metres of agricultural land, hedgerow and verge of public road (south of North Wales Expressway, A55) (excluding all interests of the Crown) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and maintain gas pipeline)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                     | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                       |  |
| 401 cont'd              |                                                                                                                                         | Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of bridleway 201 10)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                                                                                         |  |
| 402                     | 10 square metres of public road (Old A55 Faenol Bropor, south of North Wales<br>Expressway, A55) (excluding all interests of the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>G-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support)<br>Unknown<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media) |  |

| <b></b>                 | Awel v Môr Offst                                                                                                                              | nore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
|                         | County of Denbighshire                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                            |  |  |
| 402 cont'd              |                                                                                                                                               | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                                                                                                                                                                                                                                                                                               |  |  |
| 403                     | 905 square metres of agricultural land, hedgerow and verge (south of North Wales<br>Expressway, A55) (excluding all interests of the Crown)   | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and maintain gas pipeline)                                                                                                                                                                                                                                                                                                                                  |  |  |
| 404                     | 1479 square metres of agricultural land, verges and hedgerow (south of North<br>Wales Expressway, A55) (excluding all interests of the Crown) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to lay and maintain gas pipeline)<br>Denbighshire County Council<br>County Hall<br>Wynnstay Road<br>RUTHIN<br>Clwyd<br>LL15 1YN<br>(in respect of bridleway 201 10)<br>North Wales Fire Authority<br>Site G<br>Ffordd Salesbury<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0J<br>(in respect of rights stated in Lease dated 26th April 2005) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | ,<br>BOOK OF REFERENCE - PART 3                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|                         | County of Denbighshire                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 404 cont'd              |                                                                                                                                     | NW (BPP) Limited<br>3 More London Riverside<br>LONDON<br>SE1 2AQ<br>(in respect of rights stated in Lease dated 9th October 2002)<br>Police & Crime Commissioner for North Wales<br>North Wales Police<br>Headquarters<br>Abergele Road<br>COLWYN BAY<br>CONWY<br>LL29 BAW<br>(in respect of rights stated in Lease dated 1st March 2000)<br>WGIF (Jersey) Trustee I Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)<br>WGIF (Jersey) Trustee II Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)<br>WGIF (Jersey) Trustee II Limited<br>Third Floor Liberation House<br>Castle Street<br>St Helier<br>Jersey<br>JE1 2LH<br>(in respect of rights stated in Lease dated 26th April 2005)<br>Wrexham Glyndwr University<br>Plas Coch<br>Mold Road<br>WREXHAM<br>LL11 2AW<br>(in respect of rights stated in Lease dated 9th July 2013) |  |
| 405                     | 77 square metres of agricultural land and hedgerow (south of North Wales<br>Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|                         | County of Denbighshire                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| Number on<br>Land Plans | Description of Land                                                                                                     | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                            |  |
| 405 cont'd              |                                                                                                                         | The Secretary Of State For Wales William Morgan House G-7 Central Square CARDIFF CF10 1EP (in respect of right of access for maintenance of fence and drain) TRB Limited 1 TRB Drive St. Asaph Business Park ST. ASAPH Denbighshire LL17 0JB (in respect of right of access to maintain drainage and right of support) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of right of support and right to free passage of service media) Wild Ground 3-4 Plas Power Road Tanyfron WREXHAM LL11 5SZ (in respect of right of access to maintain drainage) |  |
| 406                     | 509 square metres of agricultural land (south of North Wales Expressway, A55)<br>(excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                                                                                                                                                                                                                                                                  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|                         | County of Denbighshire                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                      |  |  |
| 406 cont'd              |                                                                                                                                     | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OJB<br>(in respect of right of access to maintain drainage and right of support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                   |  |  |
| 407                     | 54 square metres of track and verge (Faenol-Bropor, south of North Wales<br>Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|                         | County of Denbighshire                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                          |  |
| 407 cont'd              |                                                                                                                                                            | Wales and West Utilities Limited         Wales & West House         Spooner Close         Coedkernew         CASNEWYDD         NP10 8FZ         (in respect of right of support and right to free passage of service media as contained in a Conveyance dated 11th March 1991)         Wales and West Utilities Limited         Wales & West House         Spooner Close         Coedkernew         CASNEWYDD         NP10 8FZ         (in respect of gas apparatus)         Wild Ground         3-4 Plas Power Road         Tanyfron         WREXHAM         LL11 5SZ         (in respect of right of access to maintain drainage) |  |
| 408                     | 499 square metres of agricultural land, hedgerow and verge (Faenol-Bropor, south<br>of North Wales Expressway, A55) (excluding all interests of the Crown) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                                                                                                                                    |  |

| <b></b>                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|                         | County of Denbighshire                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                                                                                                  | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                          |  |
| 408 cont'd              |                                                                                                                                      | TRB Limited1 TRB DriveSt. Asaph Business ParkST. ASAPHDenbighshireLL17 OJB(in respect of right of access to maintain drainage and right of support)Wales and West Utilities LimitedWales & West HouseSpooner CloseCoedkernewCASNEWYDDNP10 8FZ(in respect of right of support and right to free passage of service media)Wild Ground3-4 Plas Power RoadTanyfronWREXHAMLL11 5SZ(in respect of right of access to maintain drainage)                                                                                               |  |
| 409                     | 120 square metres of track and verge (Faenol-Bropor, south of North Wales<br>Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|                         | County of Denbighshire                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                   |  |
| 409 cont'd              |                                                                                                                                                            | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                                                                                                                                                                                                                                                                                                     |  |
| 410                     | 386 square metres of agricultural land, hedgerow and verge (Faenol-Bropor, south<br>of North Wales Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|----------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                       |  |
| 410 cont'd              |                                                                                                                                              | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media as contained in a Conveyance dated 11th March 1991)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                                                                                                                                                                                                                                                                                                |  |
| 411                     | 1198 square metres of agricultural land (Faenol-bropor farm, south of North Wales<br>Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0IB<br>(in respect of right of access to maintain drainage and right of support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         | County of Denbighshire                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                               | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 411 cont'd              |                                                                                                                                                                   | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| 412                     | 312 square metres of agricultural land and private road (east of Faenol-bropor farm, south of North Wales Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>L17 0IB<br>(in respect of right of access to maintain drainage and right of support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 SSZ<br>(in respect of right of access to maintain drainage) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                                                      |                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 413                     | 26731 square metres of agricultural land, track, hedgerows (Faenol-Bropor, south of<br>North Wales Expressway, A55) (excluding all interests of the Crown) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>The National Assembly for Wales<br>Crown Building<br>Cathays Park                                                                                                                          |  |
|                         |                                                                                                                                                            | CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)                                                                                                                                                                                                                                               |  |
|                         |                                                                                                                                                            | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                                                                                                                                                                |  |
|                         |                                                                                                                                                            | TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support)                                                                                                                                                                |  |
|                         |                                                                                                                                                            | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right of support and right to free passage of service media as contained in a Conveyance dated 11th March 1991)                                                                                           |  |
|                         |                                                                                                                                                            | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                            |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|                         |                                                                                                                                                                     | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| 413 cont'd              |                                                                                                                                                                     | Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5SZ<br>(in respect of right of access to maintain drainage)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| 414                     | 2872 square metres of agricultural land, ditch and hedgerow (east of Faenol-Bropor,<br>south of North Wales Expressway, A55) (excluding all interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 0JB<br>(in respect of right of access to maintain drainage and right of support)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8F2<br>(in respect of right of support and right to free passage of service media)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 5S2<br>(in respect of right of access to maintain drainage) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|                         |                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                    | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|                         |                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 415                     | 11946 square metres of agricultural land and hedgerows (Faenol-Bropor, south of<br>North Wales Expressway, A55) (excluding all interests of the Crown) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>(F10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>(CF10 12P<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LUT7 01B<br>(in respect of right of access to maintain drainage and right of support)<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 BFZ<br>(in respect of right of support and right to free passage of service media)<br>Will Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXHAM<br>LL11 552<br>(in respect of right of access to maintain drainage) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                                                                                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 416                     | 10443 square metres of woodland, agricultural land and hedgerow (south of North<br>Wales Expressway, A55 and west of William Morgan Road) and footpath (DE   201 9)<br>(excluding all interests of the Crown) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 DLT<br>(in respect of water apparatus)<br>The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 SNQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 SIP<br>(in respect of right of access for maintenance of fence and drain)<br>TRB Limited<br>TRB buintes<br>St. Asaph Business Park<br>St. Asaph Bu |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                      | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| 417                     | 325521 square metres of agricultural land and private roads (north of Glascoed<br>Road, B5381, and south of North Wales Expressway, A55) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WCLH 9NP<br>(in respect of telecommunication apparatus)<br>SP Marweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>TRB Limited<br>1 TRB Drive<br>St. Asaph Business Park<br>ST. ASAPH<br>Denbighshire<br>LL17 OIB<br>(in respect of right of access to maintain drainage and right to support as contained in a Conveyance dated 11th March 1991)<br>Unknown<br>(in respect of right of drainage as contained in a Conveyance dated 11th March 1991)<br>Wild Ground<br>3-4 Plas Power Road<br>Tanyfron<br>WREXEMAM<br>LL11 SSZ<br>(in respect of right of access to maintain drainage as contained in a Conveyance dated 11th March 1991) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009       |  |
| 417 cont'd              |                                                                                                                 | Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                                    |  |
| 418                     | 50 square metres of verge of public road (Glascoed Road, B5381)                                                 | Openreach Limited         British Telecom         Kelvin House         123 Judd Street         LONDON         WC1H 9NP         (in respect of telecommunication apparatus)         Unknown         Dwr Cymru Cyfyngedig         Linea         Fortran Road         St. Mellons         CARDIFF         CF3 0LT         (in respect of water apparatus) |  |
| 419                     | 201 square metres of verge of public road (Glascoed Road, B5381)                                                | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                              |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 419 cont'd              |                                                                                                                 | Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 420                     | 74 square metres of pavement (Glascoed Road, B5381)                                                             | Alun Lloyd Davies         6 Ffordd Glascoed         ABERGELE         L122 9DW         (in respect of right of access)         Amy Louise Evans         5 Ffordd Glascoed         ABERGELE         Conwy         L122 9DW         (in respect of right of access)         Martin Griffiths         5 Ffordd Glascoed         ABERGELE         Conwy         L122 9DW         (in respect of right of access)         Martin Griffiths         5 Ffordd Glascoed         ABERGELE         Conwy         L122 9DW         (in respect of right of access)         Carrie Hargraves         8 Ffordd Glascoed         ABERGELE         L122 9DW         (in respect of right of access)         Carrie Hargraves         8 Ffordd Glascoed         ABERGELE         L122 9DW         (in respect of right of access)         Darrie Carrie Hargraves         8 Ffordd Glascoed         ABERGELE         L122 9DW         (in respect of right of access)         Dwr Cymru Cyfyngedig         Dwr Cymru Welsh Water |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|-------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                                                                                       | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Number on<br>Land Plans | Description of Land                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 420 cont'd              |                                                                                       | (Applications: Prescribed Forms and Procedures) Regulations 2009 Ethan James Homer 4 Ford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Hayley Elizabeth Doroshenko-Nuttall 4 Ford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Lynn Alison Hughes 3 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right to use common sewers and right to enter to maintain sewers and drains) Meirick Lloyd Davies 1 Ford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy L122 9DW (in respect of right of access) Nesta Davies 1 Fford Glascoed ABERGELE Conwy (in respect of right of access) |
|                         |                                                                                       | (in respect of telecommunication apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                |  |
| 420 cont'd              |                                                                                                                 | The Executor of the Estate of the Late Ian Charles Hughes<br>3 Ffordd Glascoed<br>CONWY<br>Conwy<br>LL22 9DW<br>(in respect of right to use common sewers and right to enter to maintain sewers and drains)<br>William Owen Morris<br>7 Ffordd Glascoed<br>ABERGELE<br>LL22 9DW<br>(in respect of right of access)                                                                                                                                                                                                                                                                                              |  |
| 421                     | 48 square metres of pavement (south of Glascoed Road, B5381)                                                    | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Meirick Lloyd Davies<br>1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)<br>Nesta Davies<br>1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)<br>Nesta Davies<br>1 Ffordd Glascoed<br>ABERGELE<br>Conwy<br>LL22 9DW<br>(in respect of right of access)<br>Openreach Limited<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunications apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                        |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 422                     | 1 square metre of verge of public road (south of Glascoed Road, B5381)                                          | Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                  |  |
| 423                     | 101 square metres of verge of public road (south of Glascoed Road, B5381)                                       | Unknown<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                       |  |
| 424                     | 33 square metres of verge of public road (Glascoed Road, B5381)                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                                                                                                                                                                                |  |
| 425                     | 509 square metres of public road and verges (Glascoed Road, B5381)                                              | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown<br>Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunication apparatus)                                                                         |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                           |  |
| 425 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| 426                     | 278 square metres of verge of public road (south of Glascoed Road, B5381)                                       | Openreach Limited         British Telecom         Kelvin House         123 Judd Street         LONDON         WC1H 9NP         (in respect of telecommunication apparatus)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of electricity apparatus)         Unknown         Dwr Cymru Cyfyngedig         Linea         Fortran Road         St. Mellons         CARDIFF         CF3 0LT         (in respect of water apparatus) |  |
| 427                     | 490 square metres of agricultural land (south of Glascoed Road, B5381)                                          | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                                                                                                                                                                                               |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| 428                     | 258 square metres of agricultural land (south of Glascoed Road, B5381)                                          | Centrica PLC<br>Milistream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 SGD<br>(in respect of right to inspect, maintain, protect, use, replace, remove or render unusable a gas pipeline and right to pass)<br>Duncan Robertson<br>Llantysilio Hall<br>LlaNGOLLEN<br>LL20 8BT<br>(in respect of right to enter and right to all game)<br>David Watkin Williams-Wynn Bt<br>3 Upper Brook Street<br>OSWESTRY<br>SY11 2TB<br>(In respect of right to enter and right to all game)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WCZN SEH<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |
| 429                     | 3 square metres of public road (Glascoed Road, B5381)                                                           | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 430                     | 7 square metres of public road (Glascoed Road, B5381)                                                           | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                          |  |
| 430 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                           |  |
| 431                     | 104 square metres of public road and verge (Glascoed Road, B5381)                                               | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                                                                                                                                                                                                                                                        |  |
| 432                     | 71 square metres of verge of public road (north of Glascoed Road, B5381)                                        | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>Zayo Group UK Limited<br>100 New Bridge Street<br>LONDON<br>EC4V 6JA<br>(in respect of telecommunication apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                 |                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009     |  |
| 432 cont'd              |                                                                                                                                 | Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                      |  |
| 435                     | 7765 square metres of agricultural land and hedgerow (south of Glascoed Road,<br>B5381)                                         | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                              |  |
| 436                     | 1226 square metres of agricultural land and hedgerows (west of Memorail Park and<br>Crematorium, south of Glascoed Road, B5381) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 5GD<br>(in respect of right to inspect, maintain, protect, use, replace, remove or render unusable a gas pipeline and right to pass)<br>Duncan Robertson<br>Llantysilio Hall<br>Llantysilio<br>LLANGOLLEN<br>LL20 8BT<br>(in respect of right to enter and right to all game) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 436 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig           Dwr Cymru Weish Water           Linea           Fortran Road           St. Melions           CARDIFF           CF3 0LT           (in respect of right to lay a water main and right to enter as contained in a Deed dated 12th January 1984)           Dwr Cymru Cyfyngedig           Dwr Cymru Weish Water           Linea           Fortran Road           St. Mellons           CARDIFF           CF3 0LT           (In respect of right Valver           Linea           Fortran Road           St. Mellons           CARDIFF           CF3 0LT           (In respect of water apparatus)           David Watkin Williams-Wynn Bt           3 Upper Brook Street           OSWESTRY           SY11 2TB           (In respect of right to enter and right to all game)           National Grid Electricity Transmission PLC           1-3 Strand           LONDON           WC2N SEH           (In respect of electricity apparatus)           SP Manweb PLC           3 Prenton Way           North Cheshire Trading Estate           PRENTON           Meressylde |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|                         | County of Denbighshire                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                                                                              | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 437                     | 9776 square metres of agricultural land and hedgerow (north of Pentre-Mawr and<br>south of Glascoed Road, B5381) | Centrica PLC<br>Millstream<br>Maidenhead Road<br>WINDSOR<br>Berkshire<br>SL4 SGD<br>(in respect of right to inspect, maintain, protect, use, replace, remove or render unusable a gas pipeline and right to pass)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay a water main and right to enter as contained in a Deed dated 12th January 1984)<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N SEH<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|                         |                                                                                                                  | County of Denbighshire Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)                                                                                                                                                                                                                                                                                                                                                                                |  |
| Number on<br>Land Plans | Description of Land                                                                                              | which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                         |  |
| 437 cont'd              |                                                                                                                  | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                                                                                                                                                                                            |  |
| 438                     | 5595 square metres of agricultural land and hedgerow (north of Pentre-Mawr and<br>south of Glascoed Road, B5381) | Memoria - Denbighshire Memorial Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support, and maintenance)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of rights granted by deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 439                     | 156 square metres of agricultural land and hedgerow (north of Pentre-Mawr and south of Glascoed Road, B5381)     | Memoria - Denbighshire Memorial Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support, and maintenance)<br>Unknown<br>(in respect of rights granted by deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966)                                                                                                                                                  |  |
| 440                     | 8 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)                 | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                      |                                                                                                                                                                                                                                                                                                                                                                                           |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                           |  |
|                         |                                                                                                              | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                    |  |
| 441                     | 3 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)             | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 442                     | 63 square metres of agricultural land and hedgerow (north of Pentre-Mawr and south of Glascoed Road, B5381)  | Memoria - Denbighshire Memorial Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support, and maintenance)<br>Unknown<br>(in respect of rights granted by deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 443                     | 40 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)            | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 444                     | 258 square metres of public road (road from Glascoed Road past Waen Meredydd)                                | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 445                     | 635 square metres of agricultural land and hedgerow (north of Pentre-Mawr and south of Glascoed Road, B5381) | Memoria - Denbighshire Memorial Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support, and maintenance)<br>Unknown<br>(in respect of rights granted by deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 446                     | 103 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)           | Unknown                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 447                     | 99 square metres of agricultural land and hedgerow (north of Pentre-Mawr and south of Glascoed Road, B5381)  | Memoria - Denbighshire Memorial Park Ltd<br>The Pool House<br>Bicester Road<br>Stratton Audley<br>BICESTER<br>OX27 9BS<br>(in respect of right of access, support, and maintenance)                                                                                                                                                                                                       |  |

|            | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on  | County of Denbighshire Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| Land Plans | Description of Land                                                                                                                                        | which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
| 447 cont'd |                                                                                                                                                            | Unknown<br>(in respect of rights granted by deed of grant dated 19th September 1979, and of restrictive covenants)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|            |                                                                                                                                                            | Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |  |
| 448        | 2 square metres of agricultural land (north of Pentre-Mawr and south of Glascoed Road, B5381)                                                              | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 449        | 16 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)                                                          | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 450        | 143 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)                                                         | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 451        | 188 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)                                                         | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| 452        | 3016 square metres of agricultural land (north of Pentre-Mawr and south of<br>Glascoed Road, B5381)                                                        | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

| <u> </u>                | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|                         |                                                                                                     | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| Number on<br>Land Plans | Description of Land                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                               |  |
| 453                     | 1990 square metres of agricultural land (north of Pentre-Mawr and south of<br>Glascoed Road, B5381) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |
| 454                     | 2903 square metres of agricultural land (north of Pentre-Mawr and south of Glascoed Road, B5381)    | SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of electricity apparatus)         Robert Bryn Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         LL22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                          |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| 454 cont'd              |                                                                                                                 | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
| 455                     | 245 square metres of verge of public road (north of Pentre-Mawr and south of Glascoed Road, B5381)              | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| 456                     | 514 square metres of agricultural land and hedge (south of Waen Meredydd)                                       | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                     | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                     |  |
| 457                     | 290 square metres of agricultural land (north of Pentre-Mawr and south of Glascoed Road, B5381) and footpath (DE 105 7) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>Groesffordd Marli<br>ARERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |
| 458                     | 137 square metres of road from Glascoed road past Waen Meredydd                                                         | Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| 459                     | 158 square metres of agricultural land (south of Waen Meredydd)                                                         | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                                                                                                                                                                                                                                                                                            |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                           |  |
| 459 cont'd              |                                                                 | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                    |  |
| 460                     | 239 square metres of agricultural land (south of Waen Meredydd) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                     |  |
| 460 cont'd              |                                                                                                                 | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
| 461                     | 524 square metres of agricultural land (south of Waen Meredydd)                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                           |  |
| 462 cont'd              |                                                                                                                                               | Robert Bryn Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         LL22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)         Mona Elizabeth Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         LL22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |
| 462                     | 24500 square metres of agricultural land and private road (north of Pentre-Mawr<br>and south of Glascoed Road, B5381) and footpath (DE 105 7) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| 462 cont'd              |                                                                                                                 | SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of right to install and maintain electricity cables)         Wales and West Utilities Limited         Wales & West House         Spooner Close         Codekernew         CASNEWYDD         NP10 8FZ         (in respect of gas apparatus)         Robert Bryn Davies         Pentre Mawr Farm         Groesffordd Marii         ABERGELE         Conwy         L122 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)         Mona Elizabeth Davies         Pentre Mawr Farm         Groesffordd Marii         ABERGELE         Conwy         L122 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |
| 463                     | 13597 square metres of agricultural land and hedge (south of Waen Meredydd)                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| Number on<br>Land Plans | Description of Land                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                   |  |
| 463 cont'd              |                                                                                       | Robert Bryn Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         LL22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)         Mona Elizabeth Davies         Pentre Mawr Farm         Groesffordd Marli         ABERGELE         Conwy         LL22 9DR         (in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                               |  |
| 464                     | 17129 square metres of agricultural land and hedge (south of Waen Meredydd)           | National Grid Electricity Transmission PLC         1-3 Strand         LONDON         WC2N 5EH         (in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)         Unknown         (in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)         Unknown         (in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)         Unknown         (in respect of rights granted by conveyance dated 14th June 1966) |  |
| 465                     | 11550 square metres of agricultural land (south of Waen Meredydd)                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                                                                             |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|----------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|                         | County of Denbighshire                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                                                    | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                             |  |
| 465 cont'd              |                                                                                        | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                                                      |  |
| 466                     | 10905 square metres of agricultural land and hedge (south of Ffordd Richard<br>Davies) | National Grid Electricity Transmission PLC         1-3 Strand         LONDON         WC2N 5EH         (in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)         Unknown         (in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)         Unknown         (in respect of rights granted by conveyance dated 14th June 1966)         Wales and West Utilities Limited         Wales & West House         Spooner Close         Coedkernew         CASNEWYDD         NP10 8FZ         (in respect of gas apparatus) |  |
| 467                     | 40 square metres of agricultural land and hedge (south Waen Meredydd)                  | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                    | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                     |  |
| 467 cont'd              |                                                                                                                        | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |
| 468                     | 48 square metres of public road (road from Glascoed Road past Waen Meredydd)<br>(excluding all interests of the Crown) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|----------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|                         |                                                                                                    | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                                                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                   |  |
| 468 cont'd              |                                                                                                    | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| 469                     | 304 square metres of track (south of Ffordd Richard Davies) (excluding all interests of the Crown) | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of easements granted by Deed of grant dated 15 July 2013)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                     |  |
| 470                     | 46 square metres of agricultural land and track (south of Ffordd Richard Davies)                   | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants) |  |

|                         | Awel y Môr Offst                                                                                                              | hore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|-------------------------|-------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                             |  |
| 471                     | 1394 square metres of agricultural land and hedge (south of Ffordd Richard Davies)                                            | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 472                     | 3007 square metres of agricultural land (south of Ffordd Richard Davies)                                                      | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966) |  |
| 473                     | 10383 square metres of agricultural land and hedgerow (west of Electricity Distribution Site, south of Ffordd Richard Davies) | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)                                                                                                                                                                                                                                                                                                                                                         |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                  |  |
| 473 cont'd              |                                                                                                                       | Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)                                                                                                                                                                        |  |
| 474                     | 2441 square metres of agricultural land and hedgerow (west of Electricity<br>Distribution Site, north of Hendy Gorse) | Intrespect of electricity apparatosy         National Grid Electricity Transmission PLC         1-3 Strand         LONDON         WC2N 5EH         (in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)         Unknown         (in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)         Unknown         (in respect of rights granted by conveyance dated 14th June 1966)         Vodafone Limited         Vodafone House         The Connection         NEWBURY         Berkshire         RG14 2FN         (in respect of telecommunication apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                                                                                    | BOOK OF REFERENCE - PART 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
|                         | County of Denbighshire                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                                | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                    |  |  |  |
| 474 cont'd              |                                                                                                                                    | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| 475                     | 15344 square metres of grassland area and shrubbery and plyon (south of<br>Bodelwyddan Substation, south of Ffordd Richard Davies) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain water pipeline)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N SEH<br>(in respect of electricity apparatus)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables) |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 475 cont'd              |                                                                                                                 | Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                                                                                                                                                                           |  |
|                         |                                                                                                                 | Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                          |  |
|                         |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and maintain gas pipelines)                                                                                                                                                                    |  |
|                         |                                                                                                                 | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                              |  |
|                         |                                                                                                                 | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| 476                     | 204 square metres of agricultural land (north of Hendy Gorse)                                                   | Dwr Cymru Cyfyngedig         Dwr Cymru Welsh Water         Linea         Fortran Road         St. Melions         CARDIFF         CF3 0LT         (In respect of right to lay and maintain water pipeline)         Richard Thomas Owen Williams         Cefn Estate Office         ST. ASAPH         Denbighshire         L17 OEY         (In respect of right of access)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (In respect of right to install and maintain electricity cables)         Timothy Michael Bell         Cefn Estate Office         ST. ASAPH         Denbighshire         L17 0EY         (In respect of right to install and maintain electricity cables)         Timothy Michael Bell         Cefn Estate Office         ST. ASAPH         Denbighshire         L17 0EY         (In respect of right of access)         Vodafone Limited         Vodafone Limited         Vodafone House         The Connection         NEWBURY         Ber |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                         |  |
| 476 cont'd              |                                                                                                                 | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N SEH<br>(in respect of electricity apparatus) |  |
| 477                     | 2889 square metres of agricultural land and hedge (north of Hendy Gorse)                                        | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                                                                                                                                                                                |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 477 cont'd              |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                                                                                                                                                                         |  |
|                         |                                                                                                                 | Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                                                                                                                                                                           |  |
|                         |                                                                                                                 | Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                          |  |
|                         |                                                                                                                 | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                              |  |
|                         |                                                                                                                 | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | BOOK OF REFERENCE - PART 3                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|                         | County of Denbighshire                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
| Number on<br>Land Plans | Description of Land                                                                                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                      |  |  |
| 477 cont'd              |                                                                                                                              | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| 478                     | 168 square metres of agricultural land (west of Electricity Distribution Site, south of Ffordd Richard Davies)               | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                                                                                                                                                                                                                                                                                                                                                        |  |  |
| 479                     | 2209 square metres of agricultural land and hedge (west of Electricity Distribution<br>Site, south of Ffordd Richard Davies) | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966)<br>Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication apparatus) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                       |  |
| 479 cont'd              |                                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                       |  |
| 480                     | 3332 square metres of agricultural land and hedgerow (west of Electricity<br>Distribution Site, south of Ffordd Richard Davies) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables) |  |

|                         | A                                                    | wel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| Number on<br>Land Plans | Description of Land                                  | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| 480 cont'd              |                                                      | Trinothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 OEY<br>(In respect of right of access)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(In respect of gas apparatus)<br>Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(In respect of right to install and maintain gas pipelines)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marii<br>ABERGELE<br>Comwy<br>LL22 9DR<br>(In respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marii<br>ABERGELE<br>Comwy<br>LL22 9DR<br>(In respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marii<br>ABERGELE<br>Comwy<br>LL22 9DR<br>(In respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                         | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                     |  |
| 481                     | 174 square metres of agricultural land and hedge (west of Electricity Distribution<br>Site, south of Ffordd Richard Davies) | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights granted by Deed dated 6th November 2009, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by Deed of grant dated 19th September 1979, and of restrictive covenants)<br>Unknown<br>(in respect of rights granted by conveyance dated 14th June 1966)<br>Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication apparatus)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N SEH<br>(in respect of electricity apparatus) |  |
| 482                     | 90 square metres of agricultural land (west of Electricity Distribution Site, south of Ffordd Richard Davies)               | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                                                                                                                                                                                                                                                                                                  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 482 cont'd              |                                                                                                                 | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                                                                                                                                                                         |  |
|                         |                                                                                                                 | Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                                                                                                                                                                           |  |
|                         |                                                                                                                 | Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                          |  |
|                         |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and maintain gas pipelines)                                                                                                                                                                    |  |
|                         |                                                                                                                 | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                              |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                    |  |
| 482 cont'd              |                                                                                                                 | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)                                                                                         |  |
| 483                     | 4800 square metres of agricultural land (north of Electricity Distribution Site)                                | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of electricity apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                         | County of Denbighshire                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| Number on<br>Land Plans | Description of Land                                                                                 | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                    |  |  |
| 483 cont'd              |                                                                                                     | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication apparatus)                                                                                                                                                                     |  |  |
| 484                     | 9467 square metres of substation (Electricity Distribution Site, south of Ffordd<br>Richard Davies) | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Vodafone Limited<br>Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication apparatus) |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
| Number on<br>Land Plans | Description of Land                                                                           | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                       |  |
| 484 cont'd              |                                                                                               | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and maintain gas pipelines)<br>Dwr Cymru Cyfyngedig<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                           |  |
| 485                     | 2378 square metres of agricultural land and hedgerow (south of Electricity Distribution Site) | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N SEH<br>(in respect of electricity apparatus)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 |  |
| 485 cont'd              |                                                                                                                 | Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)                                                                                                                                                                                                                           |  |
|                         |                                                                                                                 | Vodafone Limited<br>Vodafone House<br>The Connection<br>NEWBURY<br>Berkshire<br>RG14 2FN<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                          |  |
|                         |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)                                                                                                                                                                                                  |  |
|                         |                                                                                                                 | Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                              |  |
|                         |                                                                                                                 | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                           |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |
| 486                     | 189 square metres of ditch (south of Electricity Distribution Site)                                             | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right to lay and maintain water pipeline)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Martii<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|-------------------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         |                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |
|                         |                                                                                       | County of Denbighshire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| Number on<br>Land Plans | Description of Land                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                 |  |
| 486 cont'd              |                                                                                       | Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |
| 487                     | 232 square metres of hardstanding (Electricity Distribution Site)                     | Dwr Cymru Qyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of right to lay and maintain water pipeline)<br>Richard Thomas Owen Williams<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)<br>Timothy Michael Bell<br>Cefn Estate Office<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of right to install and maintain electricity cables) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
| 487 cont'd              |                                                                                                                 | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of gas apparatus)<br>Robert Bryn Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Mona Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>Nota Elizabeth Davies<br>Pentre Mawr Farm<br>Groesffordd Marli<br>ABERGELE<br>Conwy<br>LL22 9DR<br>(in respect of right of drainage and access for maintenance stated in Deed dated 26th January 1996)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N SEH<br>(in respect of electricity apparatus) |  |
| 488                     | 29 square metres of private road (south of Electricity Distribution Site)                                       | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of right of access)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of right to install and maintain electricity cables)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |

|                         | Awel v N                                                             | 1ôr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                       |  |  |  |
|-------------------------|----------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                         |                                                                      | BOOK OF REFERENCE - PART 3                                                                                                                                                                                                                                                                                                                                                                             |  |  |  |
|                         | County of Denbighshire                                               |                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
| Number on<br>Land Plans | Description of Land                                                  | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                       |  |  |  |
| 488 cont'd              |                                                                      | Wales and West Utilities Limited<br>Wales & West House<br>Spooner Close<br>Coedkernew<br>CASNEWYDD<br>NP10 8FZ<br>(in respect of right to install and maintain gas pipelines)                                                                                                                                                                                                                          |  |  |  |
| 489                     | 1012 square metres of track (south of Electricity Distribution Site) | National Grid Electricity Transmission PLC1-3 StrandLONDONWC2N 5EH(in respect of right of access)SP Manweb PLC3 Prenton WayNorth Cheshire Trading EstatePRENTONMerseysideCH43 3ET(in respect of right to install and maintain electricity cables)Wales and West Utilities LimitedWales & West HouseSpooner CloseCoedkernewCASNEWYDDNP10 8FZ(in respect of right to install and maintain gas pipelines) |  |  |  |
| 490                     | 1302 square metres of track (south of Glascoed Road, B5381)          | David Watkin Williams-Wynn Bt<br>Plas Yn Cefn<br>Cefn<br>ST. ASAPH<br>Denbighshire<br>LL17 0EY<br>(in respect of shooting rights and restrictive covenants)<br>Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right to lay and maintain cable)                                                                                 |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                   |  |
| 490 cont'd              |                                                                                                                 | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of right of access to maintain water pipeline)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>(in respect of rights stated in Deed dated 6th November 2009) |  |
|                         |                                                                                                                 | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                    |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Description of Land                                                                                                      | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                            |  |
| 490 cont'd              |                                                                                                                          | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |
| 491                     | 33 square metres of private road (south of Glascoed Road, B5381, and west of Lon<br>Coed Esgob) and footpath (DE 208 32) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)<br>ESP Electricity Limited<br>ES P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalis Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of access)<br>Gwynt Y Mor Offshore Wind Farm Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Witshire<br>SNS 6PB<br>(in respect of right to install and maintain cables)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N SEH<br>(in respect of right stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of restrictive covenants) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|-------------------------|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
|                         | County of Denbighshire                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| Number on<br>Land Plans | Description of Land                                         | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 492                     | 1147 square metres of track (south of Glascoed Road, B5381) | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WCL1V 66A<br>(in respect of right of access)<br>ESP Electricity Limited<br>E S P Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalis Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of access)<br>Gwynt Y Mor Offshore Wind Farm Limited<br>Windmill Hill Business Park<br>Whitehill Way<br>SWINDON<br>Willshire<br>SNS 6PB<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WCZN 5EH<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WCCIM SHP |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER      |                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|-------------------------|--------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
|                         | County of Denbighshire                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |
| Number on<br>Land Plans | Description of Land                                          | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                |  |
| 492 cont'd              |                                                              | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown<br>(in respect of restrictive covenants)                                                                                                                                                                                                                                               |  |
| 493                     | 177 square metres of public road (Glascoed Road, B5381)      | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 OLT<br>(in respect of water apparatus)<br>Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus) |  |
| 494                     | 71 square metres of hedgerow (south of Glascoed Road, B5381) | SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                   |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | County of Denbighshire                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| Number on<br>Land Plans | Description of Land                                                                   | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| 494 cont'd              |                                                                                       | Unknown<br>(in respect of rights contained in a Conveyance dated 20 March 1919)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
| 495                     | 15 square metres of track and hedgerow (south of Glascoed Road, B5381)                | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>7.1 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of right of access)<br>ESP Electricity Limited<br>E SP Utilities Group Ltd<br>Bluebird House<br>Unit 22 Mole Business Park<br>Randalis Road<br>LEATHERHEAD<br>Surrey<br>KT22 7BA<br>(in respect of right of access)<br>Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NEI2 BBU<br>(in respect of rights stated in Deed dated 6th November 2009)<br>National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Unknown<br>(in respect of rights stated in Deed dated 6th November 2009)<br>Merseyside |  |
|                         |                                                                                       | CH43 3ET<br>(in respect of electricity apparatus)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |

|                         | Awel y M                                                   | lôr Offshore Wind Farm DEVELOPMENT CONSENT ORDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
|                         | BOOK OF REFERENCE - PART 3<br>County of Denbighshire       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |
| Number on<br>Land Plans | Description of Land                                        | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)© of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                            |  |
| 496                     | 60 square metres of public road (Glascoed Road, B5381)     | Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)<br>Openreach Limited                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |
|                         |                                                            | British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>Unknown                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |
| 497                     | 194 square metres of track (south of Glascoed Road, B5381) | Diamond Transmission Partners BBE Limited         Mid City Place         71 High Holborn         LONDON         WC1V 6BA         (in respect of right to lay and maintain cable)         Dwr Cymru Cyfyngedig         Dwr Cymru Welsh Water         Linea         Fortran Road         St. Mellons         CARDIFF         CF3 0LT         (in respect of water apparatus)         ESP Electricity Limited         E S P Utilities Group Ltd         Bluebird House         Unit 22 Mole Business Park         Randalls Road         LEATHERHEAD         Surrey         KT22 7BA         (in respect of right of way) |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                             |  |  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                            |  |  |  |  |
| 497 cont'd              |                                                                                                                 | Gwynt Y Mor OFTO PLC<br>Q14 Quorum Business Park<br>Benton Lane<br>NEWCASTLE UPON TYNE<br>NE12 8BU<br>(in respect of rights stated in Deed dated 6th November 2009)                                                                                                                                                                                                                                                         |  |  |  |  |
|                         |                                                                                                                 | National Grid Electricity Transmission PLC<br>1-3 Strand<br>LONDON<br>WC2N 5EH<br>(in respect of rights stated in Deed dated 6th November 2009)                                                                                                                                                                                                                                                                             |  |  |  |  |
|                         |                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)                                                                                                                                                                                                                                                                                |  |  |  |  |
| 498                     | 13 square metres of hedgerow (south of Glascoed Road, B5381)                                                    | Diamond Transmission Partners BBE Limited         Mid City Place         71 High Holborn         LONDON         WC1V 6BA         (in respect of right to lay and maintain cable)         ESP Electricity Limited         E S P Utilities Group Ltd         Bluebird House         Unit 22 Mole Business Park         Randalls Road         LEATHERHEAD         Surrey         KT22 7BA         (in respect of right of way) |  |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009                                                                                                                                                                                                                                                                  |  |  |  |  |
| 498 cont'd              |                                                                                                                 | Gwynt Y Mor Offshore Wind Farm Limited         Windmill Hill Business Park         Whitehill Way         SWINDON         Wiltshire         SN5 6PB         (in respect of rights stated in Deed dated 6th November 2009)         Gwynt Y Mor OFTO PLC         Q14 Quorum Business Park         Benton Lane         NEWCASTLE UPON TYNE         NE12 8BU         (in respect of rights stated in Deed dated 6th November 2009)         SP Manweb PLC         3 Prenton Way         North Cheshire Trading Estate         PRENTON         Merseyside         CH43 3ET         (in respect of electricity apparatus) |  |  |  |  |
| 499                     | 137 square metres of verge of public road (south of Glascoed Road, B5381)                                       | Diamond Transmission Partners BBE Limited<br>Mid City Place<br>71 High Holborn<br>LONDON<br>WC1V 6BA<br>(in respect of electricity apparatus)<br>Dwr Cymru Cyfyngedig<br>Dwr Cymru Welsh Water<br>Linea<br>Fortran Road<br>St. Mellons<br>CARDIFF<br>CF3 0LT<br>(in respect of water apparatus)                                                                                                                                                                                                                                                                                                                   |  |  |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 3<br>County of Denbighshire |                                                                                                                                                                                                                                                                                                                                                        |  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| Number on<br>Land Plans | Description of Land                                                                                             | Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water)<br>which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1) <sup>©</sup> of the Infrastructure Planning<br>(Applications: Prescribed Forms and Procedures) Regulations 2009 |  |  |  |
| 499 cont'd              |                                                                                                                 | Openreach Limited<br>British Telecom<br>Kelvin House<br>123 Judd Street<br>LONDON<br>WC1H 9NP<br>(in respect of telecommunication apparatus)<br>SP Manweb PLC<br>3 Prenton Way<br>North Cheshire Trading Estate<br>PRENTON<br>Merseyside<br>CH43 3ET<br>(in respect of electricity apparatus)<br>Unknown                                               |  |  |  |

| Land wł                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                             |                                                                                                                                            |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                         | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made |  |
| 1                       | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 217151 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |
| 2                       | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 111434 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |
| 3                       | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 8428 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)        | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |
| 4                       | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 155 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)         | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 5                       | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 33060 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)       | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |
| 6                       | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 3496 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)        | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                    |                                                                                                                                            |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made |  |
| 7                       | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 34 square metres of<br>foreshore land seaward of<br>the Mean High Water at<br>Rhyl (excluding all<br>interests of the Crown)                       | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 10                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 113 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Rhyl<br>(excluding all interests of<br>the Crown)    | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 11                      | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 12 square metres of<br>slipway seaward of the<br>Mean High Water lying to<br>the north of Rhyl<br>(excluding all interests of<br>the Crown)        | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 12                      | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 86 square metres of<br>slipway landward of<br>Mean High Water north<br>of Rhyl (excluding all<br>interests of the Crown)                           | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 23                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 34 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Rhyl,<br>(excluding all interests of<br>the Crown)    | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 27                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 1209 square metres of<br>foreshore land seaward of<br>the Mean High Water at<br>Rhyl (excluding all<br>interests of the Crown)                     | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 28                      | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 16 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown) | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                          |                                                                                                                                            |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                      | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made |  |
| 29                      | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 466 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown)     | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 31                      | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 218 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)      | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 32                      | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 6074 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)     | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 35                      | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 9198 square metres of<br>foreshore land seaward of<br>the Mean High Water and<br>footway at Prestatyn<br>(excluding all interests of<br>the Crown)       | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 38                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 379 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                              |                                                                                                                                            |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                          | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made |  |
| 39                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 3649 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown)                    | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 44                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 47 square metres of<br>hardstanding and<br>foreshore land seaward of<br>the Mean High Water<br>(north of Rhyl Coast Road,<br>A548) (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 45                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 183 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown)                     | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 46                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 28705 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn, (excluding all<br>interests of the Crown)                  | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |
| 47                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 73 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown)                          | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 48                      | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 4312 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown)                    | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |

| Land wł                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                           |                                                                                                                                            |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                       | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made |  |
| 49                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 7 square metres of<br>footway seaward of the<br>Mean High Water (north<br>of Victoria Road, A548)<br>(excluding all interests of<br>the Crown)            | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 52                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 3329 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown)     | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 53                      | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 56 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown)       | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 54                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 1968 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |
| 55                      | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 1175 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |
| 56                      | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 167 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown)      | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                           |                                                                                                                                            |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                       | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made |  |
| 57                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 2173 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |
| 58                      | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 1839 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |
| 59                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 174 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying north of Prestatyn<br>(excluding all interests of<br>the Crown)         | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                      |  |
| 60                      | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 26 square metres of<br>foreshore landward of the<br>Mean High Water lying<br>north of Prestatyn<br>(excluding all interests of<br>the Crown)              | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 61                      | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 15 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown)       | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                |  |
| 255                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 52 square metres of track<br>and grassed area (north<br>of River Clwyd and south-<br>east of A525) (excluding<br>all interests of the Crown)              | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(in respect of annual Crown rent) |  |

| Land wl                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                      |                                                                                                                                                                                                                              |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                  | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                   |  |
| 257                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 4323 square metres of<br>watercourse and river<br>(River Clwyd), foreshore,<br>bed and banks thereof<br>(excluding all interests of<br>the Crown)                    | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH                                                                                                                        |  |
| 258                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 340 square metres of<br>grassed area and river<br>banks (south of River<br>Clwyd) (excluding all<br>interests of the Crown)                                          | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                                                                                                  |  |
| 259                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 4924 square metres of<br>grassed area and river<br>banks (south of River<br>Clwyd) (excluding all<br>interests of the Crown)                                         | The King's Most Excellent Majesty In Right Of His Crown<br>1 St. James's Market<br>LONDON<br>SW1Y 4AH<br>(as reputed owner)                                                                                                  |  |
| 363                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 77 square metres of<br>hedgerow (north of North<br>Wales Expressway, A55<br>and west of Nant-y-<br>Faenol Road) (excluding<br>all interests of the Crown)            | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 364                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 151 square metres of<br>hedgerow (north of North<br>Wales Expressway, A55<br>and west of Nant-y-<br>Faenol Road) (excluding<br>all interests of the Crown)           | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 365                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 2597 square metres of<br>agricultural land (north of<br>North Wales Expressway,<br>A55 and west of Nant-y-<br>Faenol Road) (excluding<br>all interests of the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                                                                               |                                                                                                                                                                                                                              |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                                                                           | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                   |  |
| 369                     | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 10 square metres of<br>agricultural land (east of<br>Tyddyn Isaf Farm, north<br>of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) (excluding all<br>interests of the Crown)                             | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 370                     | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 27217 square metres of<br>agricultural land (north of<br>North Wales Expressway,<br>A55 and west of Nant-y-<br>Faenol Road) and<br>footpath (DE 201 7)<br>(excluding all interests of<br>the Crown)                           | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 371                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 830 square metres of<br>agricultural land (east of<br>Tyddyn Isaf Farm, north<br>of North Wales<br>Expressway, A55 and<br>west of Nant-y-Faenol<br>Road) and footpath<br>(DE 201 7) (excluding all<br>interests of the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 372                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 1429 square metres of<br>agricultural land, tracks<br>and hardstanding (at<br>Tyddyn Isaf Farm, north<br>of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                      | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 373                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 786 square metres of<br>private road (north of<br>North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown)                                                                                                   | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991)                                              |  |

| Land wł                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                                                     |                                                                                                                                                                                                                              |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                                                 | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                   |  |
| 374                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 10879 square metres of<br>agricultural land (north of<br>North Wales Expressway,<br>A55 and east of Tyddyn<br>Isaf Farm) and footpath<br>(DE 201 7) (excluding all<br>interests of the Crown)       | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 375                     | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 56 square metres of<br>agricultural land (north of<br>North Wales Expressway,<br>A55 and east of Tyddyn<br>Isaf Farm) (excluding all<br>interests of the Crown)                                     | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 376                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 16642 square metres of<br>agricultural land (north of<br>North Wales Expressway,<br>A55 and west of Nant-y-<br>Faenol Road) and<br>footpath (DE 201 7)<br>(excluding all interests of<br>the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |
| 377                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 122 square metres of<br>private road and verges<br>(north of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                                           | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP                                                                                                                        |  |
| 378                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 2220 square metres of<br>agricultural land (north of<br>North Wales Expressway,<br>A55 and west of Nant-y-<br>Faenol Road) (excluding<br>all interests of the Crown)                                | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991)                                              |  |
| 379                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 122 square metres of<br>agricultural land (north of<br>North Wales Expressway,<br>A55 and west of Nant-y-<br>Faenol Road) (excluding<br>all interests of the Crown)                                 | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain as contained in a Transfer dated 11th March 1991) |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                       |                                                                                                                                                                                 |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                   | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                      |  |
| 380                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 1383 square metres of<br>private road and verges<br>(north of North Wales<br>Expressway, A55) and<br>footpath (DE 201 7)<br>(excluding all interests of<br>the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991) |  |
| 381                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 945 square metres of<br>private road and verges<br>(north of North Wales<br>Expressway, A55) and<br>footpath (DE 201 7)<br>(excluding all interests of<br>the Crown)  | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access stated in Transfer dated 11 March 1991) |  |
| 382                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 249 square metres of<br>verge (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                                           | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP                                                                           |  |
| 383                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 147 square metres of<br>verge (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                                           | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                    |  |
| 384                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 1481 square metres of<br>public road (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                                    | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                        |  |
| 385                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 194 square metres of<br>verge (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                                           | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                        |  |
| 386                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 359 square metres of<br>verge (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                                           | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                    |  |

| Land wł                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                              |                                                                                                                                                                                                                                                                                                                                   |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                          | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                                                                                                                        |  |
| 387                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 782 square metres of<br>public road (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)            | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath public highway)                                                                                                                                                                                                    |  |
| 388                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 39 square metres of verge<br>(North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown)                      | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                                                                                                                                                                      |  |
| 389                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 145 square metres of<br>public road and verges<br>(North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown) | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                                                                                                                                                                      |  |
| 390                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 5 square metres of public<br>road and verges (North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown)   | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath half width of public highway)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of subsoil beneath half width of public highway) |  |
| 391                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 4 square metres of verges<br>(North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown)                      | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as reputed owner)                                                                                                                                                                                                                    |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                             |                                                                                                                                                                                                                                                                                                                       |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                         | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                                                                                                            |  |
| 391 cont'd              |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                             | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(as reputed owner)                                                                                                                                                                                           |  |
| 392                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 562 square metres of<br>verge(North Wales<br>Expressway,<br>A55)(excluding all<br>interests of the Crown)                   | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP                                                                                                                                                                                                                 |  |
| 393                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 21 square metres of<br>public road and verges<br>(North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown) | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of subsoil beneath half width of public highway)<br>The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath half width of public highway) |  |
| 394                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 14 square metres of<br>public road (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)            | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                                                                                                                                                          |  |
| 395                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 175 square metres of<br>verges (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                                                                                                                                                          |  |
| 396                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 104 square metres of<br>public road and verge<br>(North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown) | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath half width of public highway)                                                                                                                                                                          |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                           |                                                                                                                                                          |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                       | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made               |  |
| 397                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 67 square metres of<br>verges (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                                               | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP                                                    |  |
| 398                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 418 square metres of<br>public road (North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                                         | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP                                                    |  |
| 399                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 86 square metres of<br>public road and verge<br>(south of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown)                                    | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                 |  |
| 400                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 622 square metres of<br>public road (south of<br>North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown)                                                | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                 |  |
| 401                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 583 square metres of<br>agricultural land,<br>hedgerow and verge of<br>public road (south of<br>North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                 |  |
| 402                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 10 square metres of<br>public road (Old A55<br>Faenol Bropor, south of<br>North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown)                       | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath half width of public highway) |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                         |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                        | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                                                                                                                                              |  |
| 402 cont'd              |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                            | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998) |  |
| 403                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 905 square metres of<br>agricultural land,<br>hedgerow and verge<br>(south of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                                                                                                                                                                                |  |
| 404                     | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 1479 square metres of<br>agricultural land, verges<br>and hedgerow (south of<br>North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown)  | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ                                                                                                                                                                                                                                                                |  |
| 405                     | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 77 square metres of<br>agricultural land and<br>hedgerow (south of North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown)            | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)        |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                             | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                                                                                                                                                                                                                                                                                                                   |  |  |
| 405 cont'd              |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                 | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998)                                                                                                                                                                                                                                                                                                                                                     |  |  |
| 406                     | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 509 square metres of<br>agricultural land (south of<br>North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown)                | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998) |  |  |
| 407                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 54 square metres of track<br>and verge (Faenol-Bropor,<br>south of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                                                                                                                                                                             |  |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                       | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                                                                                                                                                                                                                                                                                                                   |  |  |
| 407 cont'd              |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                           | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998)                                                                                                                                                                                                                                                                                                                                                     |  |  |
| 408                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 499 square metres of<br>agricultural land,<br>hedgerow and verge<br>(Faenol-Bropor, south of<br>North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998) |  |  |
| 409                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 120 square metres of<br>track and verge (Faenol-<br>Bropor, south of North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown)                         | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                                                                                                                                                                             |  |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                       | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                                                                                                                                                                                                                                                                                                                   |  |  |
| 409 cont'd              |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                           | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998)                                                                                                                                                                                                                                                                                                                                                     |  |  |
| 410                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 386 square metres of<br>agricultural land,<br>hedgerow and verge<br>(Faenol-Bropor, south of<br>North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998) |  |  |
| 411                     | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 1198 square metres of<br>agricultural land (Faenol-<br>bropor farm, south of<br>North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown)                 | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                                                                                                                                                                             |  |  |

| Land wi                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                                    | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 411 cont'd              |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                        | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998)                                                                                                                                                                                                                                                                                                                                                     |  |
| 412                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 312 square metres of<br>agricultural land and<br>private road (east of<br>Faenol-bropor farm,<br>south of North Wales<br>Expressway, A55)<br>(excluding all interests of<br>the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998) |  |
| 413                     | Acquisition of new rights and<br>imposition of restrictions                                                                                                                                                                                                                                                                                    | 26731 square metres of<br>agricultural land, track,<br>hedgerows (Faenol-<br>Bropor, south of North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown)             | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)                                                                                                                                                                             |  |

| Land wh                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                                | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                                                                                                                                                                                                                                                                                                                                                                   |  |
| 413 cont'd              |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                    | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998)                                                                                                                                                                                                                                                                                                                                                     |  |
| 414                     | Temporary Possession                                                                                                                                                                                                                                                                                                                           | 2872 square metres of<br>agricultural land, ditch<br>and hedgerow (east of<br>Faenol-Bropor, south of<br>North Wales Expressway,<br>A55) (excluding all<br>interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain)<br>The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998) |  |
| 415                     | Acquisition of new rights and imposition of restrictions                                                                                                                                                                                                                                                                                       | 11946 square metres of<br>agricultural land and<br>hedgerows (Faenol-<br>Bropor, south of North<br>Wales Expressway, A55)<br>(excluding all interests of<br>the Crown)             | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)<br>The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 IEP<br>(in respect of right of access for maintenance of fence and drain)                                                                                                                                                                             |  |

| Land wl                 | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 4<br>Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009<br>County of Denbighshire |                                                                                                                                                                                                                               |                                                                                                                                                                             |  |
|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                                                                                                                                                                                                                                                   | Description of land                                                                                                                                                                                                           | The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made                                  |  |
| 415 cont'd              |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                               | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998)    |  |
| 416                     | Acquisition of Freehold                                                                                                                                                                                                                                                                                                                        | 10443 square metres of<br>woodland, agricultural<br>land and hedgerow (south<br>of North Wales<br>Expressway, A55 and<br>west of William Morgan<br>Road) and footpath<br>(DE 201 9) (excluding all<br>interests of the Crown) | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of right of maintenance of drainage and right of support)           |  |
|                         |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                               | The Secretary Of State For Wales<br>William Morgan House<br>6-7 Central Square<br>CARDIFF<br>CF10 1EP<br>(in respect of right of access for maintenance of fence and drain) |  |
|                         |                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                               | The Welsh Ministers<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as beneficiary of option to purchase contained in option agreement dated 31 March 1998)    |  |
| 468                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 48 square metres of<br>public road (road from<br>Glascoed Road past Waen<br>Meredydd) (excluding all<br>interests of the Crown)                                                                                               | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(in respect of subsoil beneath half width of public highway)                    |  |
| 469                     | Acquisition of new rights                                                                                                                                                                                                                                                                                                                      | 304 square metres of<br>track (south of Ffordd<br>Richard Davies) (excluding<br>all interests of the Crown)                                                                                                                   | The National Assembly for Wales<br>Crown Building<br>Cathays Park<br>CARDIFF<br>CF10 3NQ<br>(as reputed owner)                                                              |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 5<br>County of Denbighshire |                                                                                                                                                             |                  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                    | Description of land                                                                                                                                         | Category of Land |  |
| 1                       | Acquisition of new rights and imposition of restrictions                                                        | 217151 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Open Space       |  |
| 2                       | Temporary Possession                                                                                            | 111434 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Open Space       |  |
| 3                       | Temporary Possession                                                                                            | 8428 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)        | Open Space       |  |
| 4                       | Temporary Possession                                                                                            | 155 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)         | Open Space       |  |
| 5                       | Temporary Possession                                                                                            | 33060 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)       | Open Space       |  |
| 6                       | Acquisition of new rights                                                                                       | 3496 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)        | Open Space       |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 5<br>County of Denbighshire |                                                                                                                                                 |                  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------|------------------|--|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                    | Description of land                                                                                                                             | Category of Land |  |  |
| 7                       | Temporary Possession                                                                                            | 34 square metres of<br>foreshore land seaward of<br>the Mean High Water at<br>Rhyl (excluding all<br>interests of the Crown)                    | Open Space       |  |  |
| 8                       | Acquisition of new rights                                                                                       | 4615 square metres of<br>footway (north of Rhyl<br>Coast Road, A548) and<br>footpath (DE 207 11)                                                | Open Space       |  |  |
| 9                       | Temporary Possession                                                                                            | 2 square metres of<br>footway (north of Rhyl<br>Coast Road, A548)                                                                               | Open Space       |  |  |
| 10                      | Temporary Possession                                                                                            | 113 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Rhyl<br>(excluding all interests of<br>the Crown) | Open Space       |  |  |
| 11                      | Acquisition of new rights                                                                                       | 12 square metres of<br>slipway seaward of the<br>Mean High Water lying to<br>the north of Rhyl<br>(excluding all interests of<br>the Crown)     | Open Space       |  |  |
| 12                      | Acquisition of new rights                                                                                       | 86 square metres of<br>slipway landward of<br>Mean High Water north<br>of Rhyl (excluding all<br>interests of the Crown)                        | Open Space       |  |  |
| 13                      | Temporary Possession                                                                                            | 345 square metres of<br>footway and grassed area<br>(north of Garford Road)                                                                     | Open Space       |  |  |
| 14                      | Acquisition of new rights                                                                                       | 347 square metres of<br>footway, hardstanding<br>and grassed area (north<br>of Garford Road)                                                    | Open Space       |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 5<br>County of Denbighshire |                                                                                                                                                      |                  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------|------------------|--|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                    | Description of land                                                                                                                                  | Category of Land |  |  |
| 23                      | Temporary Possession                                                                                            | 34 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Rhyl,<br>(excluding all interests of<br>the Crown)      | Open Space       |  |  |
| 24                      | Temporary Possession                                                                                            | 4159 square metres of<br>foreshore landward of the<br>Mean High Water and<br>footway (north of Rhyl<br>Coast Road, A548)                             | Open Space       |  |  |
| 27                      | Temporary Possession                                                                                            | 1209 square metres of<br>foreshore land seaward of<br>the Mean High Water at<br>Rhyl (excluding all<br>interests of the Crown)                       | Open Space       |  |  |
| 28                      | Acquisition of new rights and imposition of restrictions                                                        | 16 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)   | Open Space       |  |  |
| 29                      | Acquisition of new rights and imposition of restrictions                                                        | 466 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown) | Open Space       |  |  |
| 30                      | Acquisition of new rights and imposition of restrictions                                                        | 8349 square metres of<br>foreshore landward of the<br>Mean High Water and<br>footway lying to the north<br>of Rhyl                                   | Open Space       |  |  |
| 31                      | Acquisition of new rights and imposition of restrictions                                                        | 218 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)  | Open Space       |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 5 |                                                                                                                                                           |                        |  |  |
|-------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------|--|--|
|                         |                                                                                       |                                                                                                                                                           | County of Denbighshire |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                          | Description of land                                                                                                                                       | Category of Land       |  |  |
| 32                      | Acquisition of new rights and<br>imposition of restrictions                           | 6074 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of Rhyl<br>(excluding all interests of<br>the Crown)      | Open Space             |  |  |
| 33                      | Acquisition of new rights and<br>imposition of restrictions                           | 63 square metres of<br>footway (north of Rhyl<br>Coast Road, A548)                                                                                        | Open Space             |  |  |
| 34                      | Acquisition of new rights and<br>imposition of restrictions                           | 3534 square metres of<br>foreshore landward of the<br>Mean High Water and<br>footway (north of Rhyl<br>Coast Road, A548)                                  | Open Space             |  |  |
| 35                      | Acquisition of new rights and<br>imposition of restrictions                           | 9198 square metres of<br>foreshore land seaward of<br>the Mean High Water and<br>footway at Prestatyn<br>(excluding all interests of<br>the Crown)        | Open Space             |  |  |
| 36                      | Acquisition of new rights                                                             | 160 square metres of<br>footway (north of Rhyl<br>Coast Road, A548)                                                                                       | Open Space             |  |  |
| 37                      | Temporary Possession                                                                  | 275 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn                                                   | Open Space             |  |  |
| 38                      | Temporary Possession                                                                  | 379 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown)  | Open Space             |  |  |
| 39                      | Temporary Possession                                                                  | 3649 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Open Space             |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 5 |                                                                                                                                                                              |                  |  |  |
|-------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------|--|--|
|                         | County of Denbighshire                                                                |                                                                                                                                                                              |                  |  |  |
| Number on<br>Land Plans | Extent of acquisition or use                                                          | Description of land                                                                                                                                                          | Category of Land |  |  |
| 40                      | Temporary Possession                                                                  | 1228 square metres of<br>foreshore landward of the<br>Mean High Water and<br>hardstanding lying to the<br>north of Prestatyn and<br>footpath (DE 205 48,<br>DE 205 49)       | Open Space       |  |  |
| 41                      | Acquisition of new rights                                                             | 1013 square metres of<br>footway (north of Rhyl<br>Coast Road, A548) and<br>footpath (DE 205 48)                                                                             | Open Space       |  |  |
| 42                      | Acquisition of new rights                                                             | 35 square metres of<br>footway (north of Rhyl<br>Coast Road, A548)                                                                                                           | Open Space       |  |  |
| 43                      | Temporary Possession                                                                  | 67 square metres of<br>footway and foreshore<br>landward of the Mean<br>High Water (north of Rhyl<br>Coast Road, A548) and<br>footpath (DE 205 49)                           | Open Space       |  |  |
| 44                      | Temporary Possession                                                                  | 47 square metres of<br>hardstanding and<br>foreshore land seaward of<br>the Mean High Water<br>(north of Rhyl Coast Road,<br>A548) (excluding all<br>interests of the Crown) | Open Space       |  |  |
| 45                      | Temporary Possession                                                                  | 183 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown)                     | Open Space       |  |  |
| 46                      | Temporary Possession                                                                  | 28705 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn, (excluding all<br>interests of the Crown)                  | Open Space       |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 5<br>County of Denbighshire |                                                                                                                                                           |                  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|------------------|--|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                    | Description of land                                                                                                                                       | Category of Land |  |  |
| 47                      | Temporary Possession                                                                                            | 73 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown)       | Open Space       |  |  |
| 48                      | Acquisition of new rights                                                                                       | 4312 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Open Space       |  |  |
| 49                      | Temporary Possession                                                                                            | 7 square metres of<br>footway seaward of the<br>Mean High Water (north<br>of Victoria Road, A548)<br>(excluding all interests of<br>the Crown)            | Open Space       |  |  |
| 50                      | Temporary Possession                                                                                            | 17734 square metres of<br>footway landward of the<br>Mean High Water (north<br>of Victoria Road, A548)<br>and footpath<br>(DE   205   49)                 | Open Space       |  |  |
| 51                      | Acquisition of new rights                                                                                       | 5853 square metres of<br>footway (north of Victoria<br>Road, A548) and footpath<br>(DE 205 49)                                                            | Open Space       |  |  |
| 52                      | Temporary Possession                                                                                            | 3329 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown)     | Open Space       |  |  |
| 53                      | Acquisition of new rights                                                                                       | 56 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown)       | Open Space       |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 5<br>County of Denbighshire |                                                                                                                                                           |                  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|------------------|--|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                    | Description of land                                                                                                                                       | Category of Land |  |  |
| 54                      | Temporary Possession                                                                                            | 1968 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Open Space       |  |  |
| 55                      | Acquisition of new rights                                                                                       | 1175 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Open Space       |  |  |
| 56                      | Acquisition of new rights                                                                                       | 167 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown)      | Open Space       |  |  |
| 57                      | Temporary Possession                                                                                            | 2173 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Open Space       |  |  |
| 58                      | Acquisition of new rights                                                                                       | 1839 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying to the north of<br>Prestatyn (excluding all<br>interests of the Crown) | Open Space       |  |  |
| 59                      | Temporary Possession                                                                                            | 174 square metres of<br>foreshore land seaward of<br>the Mean High Water<br>lying north of Prestatyn<br>(excluding all interests of<br>the Crown)         | Open Space       |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 5<br>County of Denbighshire |                                                                                                                                                     |                  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------|------------------|--|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                    | Description of land                                                                                                                                 | Category of Land |  |  |
| 60                      | Temporary Possession                                                                                            | 26 square metres of<br>foreshore landward of the<br>Mean High Water lying<br>north of Prestatyn<br>(excluding all interests of<br>the Crown)        | Open Space       |  |  |
| 61                      | Acquisition of new rights                                                                                       | 15 square metres of<br>foreshore landward of the<br>Mean High Water lying to<br>the north of Prestatyn<br>(excluding all interests of<br>the Crown) | Open Space       |  |  |
| 62                      | Temporary Possession                                                                                            | 795 square metres of<br>footway and foreshore<br>landward of the Mean<br>High Water (north of<br>Victoria Road, A548)                               | Open Space       |  |  |
| 63                      | Acquisition of new rights                                                                                       | 155 square metres of<br>slipway at Ffrith Beach<br>landward of Mean High<br>Water (north of Victoria<br>Road, A548)                                 | Open Space       |  |  |
| 64                      | Temporary Possession                                                                                            | 94 square metres of<br>hardstanding and<br>foreshore landward of the<br>Mean High Water (north<br>of Victoria Road, A548)                           | Open Space       |  |  |
| 65                      | Temporary Possession                                                                                            | 25 square metres of<br>hardstanding and<br>foreshore landward of the<br>Mean High Water (north<br>of Victoria Road, A548)                           | Open Space       |  |  |
| 66                      | Acquisition of new rights                                                                                       | 17 square metres of<br>footway (north of Victoria<br>Road, A548)                                                                                    | Open Space       |  |  |
| 67                      | Temporary Possession                                                                                            | 53 square metres of<br>grassed area (north of<br>Victoria Road, A548)                                                                               | Open Space       |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 5<br>County of Denbighshire |                                                                                                              |                                |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|--------------------------------|--|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                    | Description of land                                                                                          | Category of Land               |  |  |
| 68                      | Temporary Possession                                                                                            | 3178 square metres of<br>grassed area (north of<br>Victoria Road, A548)                                      | Open Space                     |  |  |
| 69                      | Temporary Possession                                                                                            | 3442 square metres of<br>grassed area (north of<br>Victoria Road, A548)                                      | Open Space                     |  |  |
| 69a                     | Plot No Longer Included In<br>DCO                                                                               | Plot No Longer Included<br>In DCO                                                                            | Plot No Longer Included In DCO |  |  |
| 70                      | Temporary Possession                                                                                            | 45 square metres of<br>footway (north of Victoria<br>Road, A548)                                             | Open Space                     |  |  |
| 71                      | Temporary Possession                                                                                            | 55 square metres of<br>grassed area (north of<br>Victoria Road, A548)                                        | Open Space                     |  |  |
| 72                      | Temporary Possession                                                                                            | 891 square metres of<br>grassed area and track<br>(north of Victoria Road,<br>A548)                          | Open Space                     |  |  |
| 73                      | Temporary Possession                                                                                            | 57 square metres of<br>grassed area at Ffrith<br>Beach Festival Gardens<br>(north of Victoria Road,<br>A548) | Open Space                     |  |  |
| 74                      | Temporary Possession                                                                                            | 66 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north of<br>Victoria Road, A548)         | Open Space                     |  |  |
| 75                      | Temporary Possession                                                                                            | 2624 square metres of<br>grassed area (north of<br>Victoria Road, A548)                                      | Open Space                     |  |  |
| 76                      | Acquisition of new rights                                                                                       | 469 square metres of<br>private road and<br>hardstanding (north of<br>Ferguson Avenue)                       | Open Space                     |  |  |

|                         | Awel y Môr Offshore Wind Farm DEVELOPMENT CONSENT ORDER<br>BOOK OF REFERENCE - PART 5<br>County of Denbighshire |                                                                                                                       |                  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------|------------------|--|--|
| Number on<br>Land Plans | Extent of acquisition or use                                                                                    | Description of land                                                                                                   | Category of Land |  |  |
| 77                      | Acquisition of new rights                                                                                       | 22 square metres of<br>private road and verge<br>(north of Ferguson<br>Avenue)                                        | Open Space       |  |  |
| 80                      | Temporary Possession                                                                                            | 24 square metres of<br>private road (north of<br>Ferguson Avenue)                                                     | Open Space       |  |  |
| 81                      | Temporary Possession                                                                                            | 1729 square metres of<br>grassed area and footway<br>(north of Victoria Road,<br>A548 and west of<br>Ferguson Avenue) | Open Space       |  |  |
| 82                      | Acquisition of new rights                                                                                       | 225 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north of<br>Victoria Road, A548)                 | Open Space       |  |  |
| 83                      | Acquisition of new rights                                                                                       | 1746 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north of<br>Victoria Road, A548)                | Open Space       |  |  |
| 84                      | Temporary Possession                                                                                            | 27 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north of<br>Victoria Road, A548)                  | Open Space       |  |  |
| 85                      | Temporary Possession                                                                                            | 19 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north of<br>Victoria Road, A548)                  | Open Space       |  |  |
| 86                      | Temporary Possession                                                                                            | 16 square metres of<br>footway at Ffrith Beach<br>Festival Gardens (north of<br>Victoria Road, A548)                  | Open Space       |  |  |



## RWE Renewables UK Swindon Limited

Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB T +44 (0)8456 720 090

Registered office: RWE Renewables UK Swindon Limited Windmill Hill Business Park Whitehill Way Swindon